

Policy Capturing with Local Models: The Application of the ALD Technique in Modeling Judgment

ŝ

ð

L. L. Gooch, Major, USAF



Approved for public rel 1807 Distribution Unit of d

ii

SECURITY CLASSIFICATION OF THIS PAGE(Wien Data Entered)

Block #.0

3

1

ې تې چېنې

9 IV

 The past decade has witnessed increased research into the use of linear statistical models to describe subjective human judgment processes. Past studies have focused on, 1) the superior predictive capability of statistical models, vis-avis, man himself, 2) the detection and modeling of configural judgment process, and, 3) the use of judgment models as catalysts to conflict resolution. This research has generally been conducted in the context of clinical experiments and the results have not been extensively applied to practical decision situations. This lack of diffusion is primarily the result of failure of researchers to present their models in a format that the non-technical decision maker can understant and appreciate.

This dissertation helps to bridge the gap between the laboratory and operational environments by applying Folicy Capturing to the decision process of an installment loan officer who approves or denies credit on the basis of a written application. The Automatic Interaction Detection (AID) algorithm was used to present the "structural image" of the policy in a format that is readily comprehensible by the decision maker. A unique conceptualization of configural decision processes is proposed and demonstrated. Experimental results given.

UNCLASSIFICATION OF THIS PAGE(When Date Ent ed)

The dissertation is dedicated to

• • •

., , , 1

·7.

.

ці, в

*

្តំគ្ន

(* TP.

jł.

्रहेड

Ì

. •

Penny, Jennifer, Winnie, and Laura,

the four individuals who gave more to this effort than I did.

Acknowledgments

John Donne's words, "No man is an island, entire of itself", were never more appropriate than when used to acknowledge the many people whose help and support made this research possible. Outstanding among these supporters are those who sponsored, supervised, and participated in this research effort.

I would like to first express my appreciation to the Air Force Institute of Technology for having made my entire graduate education possible. Next, I would like to recognize the considerable contribution of the members of the Operations Research and Applied Statistics Group at The University of Texas at Austin. It was their guidance, instruction, and metivation that allowed me to exploit the opportunity, offered by AFIT, to undertake this effort.

Throughout the research itself, the unwavering support and guidance of my supervising professor, Dr. G. R. Wagner, has been a continuing source of inspiration and will be forever appreciated. The additional encouragement and timely advice offered by the other members of my committee, Dr. F. Brons, Dr. E. Jennings, Dr. F. May, and Dr. J. H. Ward, Jr. undoubtedly made this research a more meaningful experience. It is indeed an honor to have been supervised by such an august group of educators.

ž

My special appreciation must be expressed to Mr. Dave Statham, and his staff at Lamar Savings and Loan Association for their enthusiastic participation in this project. Their patient response to my continual requests for "more data" will hopefully reap tangible dividends in the near future.

The considerable assistance of the Personnel Research Division, Air Force Human Resources Laboratory, Lackland, AFB, Texas must be acknowledged. Their own pioneering efforts in the field

iv

of Policy Capturing and their generous support of my efforts to advance the frontier of knowledge in the field were of inestimable value in the completion of this research.

Finally, the individual efforts of the many other people who contributed both directly and indirectly will long be remembered as I reflect upon this experience. To them, I simply say, THANKS!

v

L.L.G.

The University of Texas at Austin December, 1972

. . .

5

,

ų

.

= ,

, तुः च

ABSTRACT

The use of linear statistical models to describe subjective judgment processes has received considerable attention in the fields of clinical psychology and psychometrics over the last decade. The majority of this research has been conducted within the basic conceptual framework of the Brunswikian Lens. The studies have focused on three areas. They are: 1) the superior predictive capability of the statistical models, vis-a-vis, man himself, 2) the detection and modeling of configural judgment processes, and, 3) the use of judgment models as catalysts to conflict resolution. This research has generally been conducted in the context of clinical experiments with only limited implementation of the results in the business, academic, and administrative environments. The lack of diffusion of this innovative concept is primarily attributable to the failure of the researchers to present their models in a format that the non-technical decision maker can understand and appreciate.

ţ

This dissertation attempts to help bridge the gap between the laboratory and operational environments by applying the techniques of Policy Capturing to a practical decision process. The judgment process used is that of the installment loan officer who approves or denies credit solely on the basis of information contained on a written credit application and the past credit rating of the applicant. The Automatic Interaction Detection (AID) algorithm is used as the primary means of describing the loan officers' policies. It presents the "structural image" of the policy in a graphical format that is readily comprehensible by the decision maker. The analytical data provided by the technique is used to hypothesize the appropriate mathematical model.

vi

ŝ

A unique conceptualization of configural decision processes as a series of local models over subspaces of the predictor set is proposed and demonstrated. A heuristic procedure for defining local models is presented and used to develop policy models for the loan officers. These models achieve cross-validation hit rates in excess of 91 percent in predicting actual decisions on new loan applications. Models for identifying the top 30 percent of the applications, in the absence of credit rating data, are developed and discussed as a means of improving the level of service that the lending institution can provide during non-banking hours.

TABLE OF CONTENTS

5

3

2. i . . .

ù., 1

đ

1

CHAPTER		Page
	DEDICATION	<u>i</u> ii
	ACKNOWLEDGEMENTS	iv
	ABSTRACI	vi
	INTRODUCTION	1
Ι.	CONCEPT OVERVIEW	
	The Judgment Modeling Concept	5
	The Judgment Modeling Process	6
	The Brunswikian Lens	8
	Correlational vs. Functional Measurement	
	Paradigm	11
	Definition of Policy Capturing	12
	Definitions	12
II.	REVIEW AND ANALYSIS OF PAST RESEARCH	
	Overview	15
	The Early Regressionists	16
	The Configurality Debate	23
	Duststrapping	28
	The Nature of the Decision Process	33
	Summary	35
III.	FOCUS OF THIS RESEARCH	
	Applications of Policy Capturing in Proctorial	
	Environments	41
	The Transition from Hypothetical to	
	Methodological	48
	Criteria for Attaining Effective Models	49
	The Research Approach	51
	An Overview of the Research Tasks	52
	Summary	57

Circles 1

No. of Concepts of

CHAPTER

58

Conceptualization of the Configural Judgment	
Model as a Series of Subspace Hyperplanes	. 58
The Automatic Interaction Detection Algorithm	. 62
Problems in the Application of AID for Hypo-	
thesizing Policy Models	. 70
The Development of the AID4UT/AIDTRE	
Computer Program	. 72
Application of AUD4UT/AIDTRE in Defining	
Subspaces	. 73
The Criteria for Defining Subspaces	. 74
Building Local Models Within the Subspaces	. 80
The Determination and Evaluation of Model	
Coefficients	. 81
Comparative Results of Various Modeling	
Approaches	. 83
Summary of Guidelines for Applying AID4UT/	
AIDTRE in Policy Capturing	. 90

37	
- ¥	2

÷.

BACKGROUND, ENVIRONMENT, AND CHRONOLOGY OF CASE STUDY

Introduction	96
A Brief Resume of Numerical Credit Scoring	96
The Decision Process Considered in this Study .	98
The Lending Institution	101
The Decision-Makers	102
Two Functional Objectives of the Policy Models;	
Tentative Approvals and Surrogate Judgments	103
Chronology of the Project	105
Summary	113

ix

CHA^{L,}TER

5

D

.

-

. 6

ţ

ĉ

n, v

v

i

Page

VI. ANALYSIS, RESULTS, AND IMPLICATIONS OF LOAN OFFICER POLICY MODELS

Introduction.	114
Section I: Results and Decisions of Initial	
Analysis	116
Section II: Analysis of Decisions for Codified	
Predictor Sets	126
Section III: Presentation of Individual Policy	
Models	134
Section IV: Comparison of Models and Judges	155
Section V: Discussion of Voting and Composite	
Models	163
Section VI: Results of Cross-Validation Field-	
Test	167
Section VII: Implications for Implementation	180
Section VIII: Summary	189

VII. CONCLUSIONS, COMPARISONS, AND RECOM-MEDATIONS

Viability of Applying Policy Capturing to	
Operational Decisions	191
Modeling Configural Judgment Processes	
with Local Models	192
The Character of the Model Seeking Technique,	
AID	192
Comparison of Results of this Research with	
Those of Past Research	193
Recommendations for Further Research	195
Final Comments	201
APPENDIX A: Definition and Comparison of	
Mathematical Models	204
APPENDIX B: Exhibits 1-19	237

CHAPTER

3

1

APPENDIX C:Users Description and Listing
for AID4UT/AIDTRE and
Ancillary Programs296

BIELIOGRAPHY 368

VITA

Page

LIST OF TABLES

Table		Page
2-1	Rater Consistency	20
2-2	Policy Comparison	21
3-1	Model Characteristics Versus Functional Applications	51
4-1	BSS/TSST Profile	77
4-2	BSS/TSS(i) Profile	78
4-3	Comparison of Predictive Efficiency for Alternative Coefficient Determination Procedures	83
	Comparison of Cross-Validity on Split Sample of 80/20	84
4-5	Comparison of Ranked Importance of Predictors	89
4-6	Guidelines for Employing AID4UT/AIDTRE in Building Policy Models	91
· - 1	By-Category Success Rates Used for Ranking Predictor Categories	116
6-2	Tentative Approval Models	129
6-3	Coded vs. Actual Decision Statistics	130
6-4	Ranked Importance of Predictors	133
6-5	First-Order Model Coefficients	136
6-6	Second-Order Model Coefficients	137

xii

Table

6

0

herrori

7

6-7	Local Model Coefficients: Judge 1	146
6-8	Local Model Coefficients: Judge 2	148
6-9	Local Model Coefficients: Judge 3	150
6-10	Local Model Coefficients: Judge 4	152
6-11	Local Model Coefficients: Judge 5	154
6-12	Interjudge Similarity	161
6-13	Coefficients for 4/5 Vote and Composite Models	164
6-14	Approval Rate vs. Credit Rating Category	167
6- 15	Maximum Hit Rates	169
6-16	Expert vs. Clerical Measurement Differences on Application Variables	173
6-17	Comparison of 4/5 Vote and Decisions of Judges 6 and 7	187

44.1464

CONTRACTOR OF STREET, OR OTHER

LIST OF FIGURES

, .

. .

ł

A

.

<u>,</u>6

,

> د م ق

к •

ي. ا

\$

à

257.

/

Figure		Page
1-1	Functional Elements of the Judgment Modeling Process	7
1-2	Brunswikian Lens Model	9
3-1	Characterization of Decision Tasks	43
3-2	A Hypothetical Configural Effects Judgment Process	53
4-1	Logic of AID Algorithm	64
4-2	Example of an AID-Tree	65
4-3	Annotated AID-Tree for Example Problem	76
4-4	Comparative Profiles of Hit Rates	85
5-1	Dealer Paper Flow Diagram	100
6-1	Predictability of Scaled and Binary Criterion Measures	120
6-2	Success Rates vs. Marital Status	123
6-3	AID-Tree for Actual Decisions with Credit Rating Included	1 32
6-4	AID Split DiagramJudge 1	139
6- 5	AID Split DiagramJudge 2	140
6-6	AID Split DiagramJudge 3	141

xiv

Ĵ

é

Figure

6-7	AID Split DiagramJudge 4
6-8	AID Split DiagramJudge 5
6-9	Subspaces: Judge 1
6-10	Subspaces: Judge 2
6-11	Subspaces: Judges 3
6-12	Subspaces: Judge 4
6-13	Subspaces: Judge 5
6-14	Error Curves for Judge 1
6-15	Error Curves for Judge 2
6-16	Error Curves for Judge 3 158
6-17	Error Curves for Judge 4
6-18	Error Curves for Judge 5
6-19	Merge Diagram from Clustering Program
6-20	Error Curves: "4/5" Voting Strategy 166
6-21	Cross-Validation Hit Rates for Various First-Order Models Using Credit Rating
6-22	Cross-Validation Hit Rates vs. Measurement Type
6-23	Risk Curves for "4/5" Policy for Tentative Approval (Expert Measurement)

Page

Figure		Page
6-24	Risk Curves for "4/5" Policy for Tentative Approval (Clerical Measurement)	177
6-25	Risk Curves for "4/5" Policy for Tentative Approval (with Dings/Expert Measurement)	178
6-26	Hit Rate vs. Score for Various Model/Criterion Aliernatives	179
6-27	Risk Curves for "4/5" Policy (First-Order) with Credit Rating	183
6-28	Risk Curves for "4/5" Policy (Local Model) with Credit Rating	183
6-29	Distribution of Times Applications Spent in "System" and "Credit System"	185

្រី រូវ

•

.

....

۰. ۱۰.

> .1 8

3

INTRODUCTION

The many definitions of Operations Research all involve a common conceptual thread; that is, "the application of the scientific approach and mathematical techniques to describe, control, and improve operations of organizational systems". This dissertation represents an effort to adapt, expand, and introduce yet another technique into the inventory of tools available to the practitioner of Operations Research.

The voluminous literature in the field of Operations Research is replete with decision models and solution algorithms for problems in which both the objective functions and the relevant parameters are thought to be well known. The general statement of such problems is:

OPTIMIZE:	$F(\widetilde{x})$	χεV
Subject to:		
equality constraints:	$h_i(\widetilde{x}) =$	$0 i = 1 \dots m$
inequality constraints:	g _i (x) ≤	$0 i = m + 1 \dots p$

In all cases the coefficients of the objective function and constraints are either deterministic or, at least, estimable by some probability distribution. In those cases where the objective function, $F(\tilde{x})$, is not known, it is either assumed, or the problem is left unsolved.

The lack of objective functions has spurred research in such fields as "Utility Theory" (Fishburn, 1968) and "Measurement Theory" (Roberts, 1970), and has brought about such techniques as Delphi (Helmer, 1967) (Dalkey, 1971) and the Echo Method (Milburn, 1968). However, these techniques have not really been refined to the stage that they represent readily usable tools for the practitioner of Operations Research.

The technique proposed here represents a more direct attack on solving the problem of determining what is important to the decision maker and provides a readily-asable methodology for analyzing, describing, simulating, and possibly improving the decision-making function. This approach involves the inference of the values of the decision-maker through the analysis of his decisions over a sufficiently large number of sample decision situations. Development of this type of procedure was advocated by Churchman (1961) in his philosophical discussion of optimal decision processes.

The research chronicled by this dissertation includes the exploration and adaptation of a concept known as "Policy Capturing" for use in ecologically valid environments. The basic concept of "Policy Capturing" has been an area of continued research in the field of clinical psychology during the past decade. This research has been pursued and results reported under such titles as "Judgment Modeling", "Decision Modeling", and "Analysis of Cognitive Processes". Although there has been much work done and many articles written in the various journals of clinical psychology, the transition from a clinical experiment to an applications methodology has only been attempted in a limited number of instances. The lack of diffusion of this innovative concept is mainly due to the approach taken by many researchers of working with contrived and synthetic situations, a practice which has not enhanced the adaptability or implementability of the scientific findings of the psychologist to the extra-laboratory or practical decision situation.

It is the purpose of this research to broaden the applicability of the Policy Capturing concept by bridging the gap between the

findings of the clinical psychologists and the realities of typical operational decision situations facing the Operations Research practitioner. This bridge is afforded by the adaptation of recently developed satistical techniques for the purpose of defining policy models that reflect the underlying judgment process of the decision-maker and retain a high degree of predictive stability.

This dissertation is divided into seven chapters and three appendices. Chapter 1 presents the basic framework and the experimental paradigm within which the Policy Capturing concept is formulated and the current research was performed. Chapter 2 provides a review and analysis of the past research that is relevant to the current efforts and Chapter 3 provides the focus and motivations for the current research. Chapter 4 describes the development, modification, and interpretation of the statistical techniques used in this research. This chapter also presents a methodology for developing "structural image" models that provide the primary bridge between the concept of Policy Capturing and the practical methodology for its application. Chapter 5 presents the background and case history of the practical environment within which this research was conducted and the viability of the methodology is demonstrated. Chapter 6 presents results, conclusions and implications generated by this research for the particular practical environment in which it was used. Chapter 7 relates the current research to the findings and hypotheses generated by various previous researchers and discusses the implications of applying the basic methodology to other environments.

Finally, three appendices are provided. Appendix A reflects a comparison of the various statistical tools available and their relationship with the Policy Capturing methodology. Appendix B

provides the forms, data, and detailed structural-image models for the case study. Appendix C provides a self-contained description for using the computer programs and the analysis scheme, developed in this research, in future Policy Capturing efforts.

Guide to the Reader:

Whereas this dissertation attempts to straddle the gap between the technical intricacies of statistical modeling and the operational concerns of manager, it necessarily addresses itself to two audiences. A guide to those sections of interest to each seems in order. The reader who is interested in the details of <u>HOW</u> the analysis was performed and the models were built should concern himself with Chapters 1-4, 7, and Appendices A and C. The reader who is interested mainly in the results of this application of Policy Capturing to modeling the policies of loan officers will be most interested in Chapters 1, 5-7, and Appendix B.

CHAFTER I: CONCEPT OVERVIEW

The Judgment Modeling Concept:

ð

The judgment process is defined as "the process of forming an opinion or evaluation by discerning and comparing". Implicit in this definition is the existence of one or more attributes or characteristics of the situation that need to be compared or evaluated, and the identification of these attributes. If one makes many judgments of the same nature, the logical premise of consistency would dictate that the same set of evaluations and comparisons should be carried out in each decision situation. Such a set of comparisons could be classified as a model or <u>policy</u> for making all judgments of a particular nature. If the attributes can be quantified in some mathematical form, the comparisons and evaluations can be carried out mathematically with the <u>policy</u> being embodied in the various mathematical operations to be performed.

The fundamental premise of the Judgment Modeling Concept is that it is possible to represent subjective human judgment with objective mathematical models. This premise, in itself, has been a source of considerable debate relative to the superiority of clinical vs. actuarial decisions. This debate was initiated when Meehl (1954) proposed that psychologists could make better diagnoses, on the average, if they were to actuarily combine the same data that they normally used in their clinical diagnosis. The ensuing research has resulted in over 600 studies in the rather narrow field of human information processing in the last decade. (Slovic and Lictenstein, 1970)

The Judgment Modeling Process:

 γ_{ij}

10 1

The definition of judgment modeling itself implies a rather straight-forward, logical sequence of events that has been the basis for most of the research in the field. The functional elements of this process are shown in Figure 1-1. As might be expected, each element has been the focal point for much detailed research, with particular attention being paid to the formulation of the mathematical models in step number three and to the comparison of judgment models in step number four. Very little research has been accomplished on the process as a whole and virtually none has been attempted on the process in practical, extra-laboratory environments.

In an excellent review of past research, Slovic and Lictenstein (1970), have reviewed and compared the Regression and the Bayesian Approaches of judgment modeling. The basic differences between these two approaches being that in the former, the judgment process is formulated in the context of the general linear hypothesis while in the latter, the judgment process is formulated in the context of conditional probabilities and Bayes' Theorem.

The work pursued in the research discussed herein has all been within the Regression Approach and, therefore, the Bayesian Approach will not be addressed. Further, within the Regression Approach, the main thrust of this research has been within the "correlational paradigm" with some efforts encroaching on the area traditionally known as the "functional measurement paradigm". The difference between these paradigms rests essentially in the elements of the judgment process considered, the statistical techniques used, and the goals pursued within a particular research effort.

FUNCTIONAL ELEMENTS OF THE JUDGMENT MODELING PROCESS

Ť

ß

<u>____</u>

772.5 1. 7**3出**





The Brunswikian Lens:

The conceptual model for the Regression Approach of judgment modeling was first proposed by Egon Brunswik in 1952. The model as shown in Figure 1-2, or minor modifications thereof, has served as the conceptual framework for both the correlational and functional measurement paradigms.

The Lens model graphically depicts the underlying hypothesis that each attribute (cue) of the decision situation is probabilistically related to an ecological criterion of a "true state-of-nature". This relationship can be characterized by a general linear model of the form:

$$\mathbf{Y}_{\mathbf{e}} = \mathbf{\hat{Y}}_{\mathbf{e}} + \mathbf{\hat{c}}_{\mathbf{e}} = \mathbf{\hat{\Sigma}} \mathbf{B}_{\mathbf{i}} \mathbf{X}_{\mathbf{i}} + \mathbf{\hat{c}}_{\mathbf{e}}$$

where:

,÷

$$Y_e =$$
 the actual criterion value
 $\hat{Y}_e =$ the predicted criterion value
 $\varepsilon_e =$ the error in the prediction
 $B_{i,e} =$ the coefficient of the ith variable
 $X_i =$ the value of the ith variable
 $e =$ the subscript representing the ecological
side of the Lens

On the judgmental side of the Lens, a similar model is hypothesized which relates the judged value of the criterion to the

and a sublemp of the second second





FIGURE 1-2

8

8

Ô

·

Second States

stimulus cue values. This equation is:

$$\mathbf{Y}_{j} = \mathbf{Y}_{j} + \mathbf{\varepsilon}_{j} = \sum_{i} \mathbf{b}_{i,j} \mathbf{X}_{i} + \mathbf{\varepsilon}_{j}$$

where:

ŧ

÷

<u>(</u>1

Ř

а,

ç

283

224

٩Ô

1

Y _j	=	the judged value of the criterion
Y,	Ξ	the predicted value of the criterion per the
ŗ		judgment model
€ _i	Ξ	the error in the prediction
b ij	=	the coefficient of the i th variable
x	Ξ	the value of the i variable
j	=	the subscript representing the judgmental side
		of the Lens

Further statistics that are relevant to using and discussing the model are:

R _{i.e}	=	ecological validity: correlation of the i th cue to
" , -		the true criterion value (Y)
r i. i	=	utilization coefficient: correlation of the i th
1 ,)		cue to the judged criterion value (Y _i)
r e	Ξ	$r_{Y} $; multiple correlation between the actual e e
		and predicted criterion value in nature
r j	=	$r_{Y,Y}$: response consistency: multiple correlation j_j

between decision makers judgments and the value of those judgments predicted by the model

In essence, the left-hand side of the lens represents a modeling of reality and the right-hand side of the lens represents a modeling of the judge's perception and interpretation of reality. On the ecological side of the lens, the error, e_e , is made up of random error and possibly any lack-of-fit of the hypothesized model to reality, whereas, on the judgmental side, e_i , is composed of lack-of-fit of judge's model and his own inconsistency in following his model. It has been pointed out by Ward (1970) that the lack-cf-fit portion of the error in the judgmental equation can be attributed to two main causes, 1) missing stimulus variables in the hypothesized variables in the model.

Correlational versus Functional Measurement Paradigm:

The distinction between the correlational and the functional measurement paradigms can be clarified in terms of the lens model. In the correlational paradigm, the emphasis is on the judgmental side of the lens with the structure of the judgmental model, the cue coefficients $(b_{i,j})$, and the response consistency (r_i) being of most interest. Elements 3, 4, 5, and 8 of the judgment modeling process have been the main areas of concern for researchers within this paradigm. In the functional measurement paradigm, interest is still evidenced in the judgmental side of the lens, but particular emphasis is put on the modification of the judgmental equation as a result of feedback from

× , × . .

success and failures of applying the model. Statistics of particular interest to workers within this paradigm are the achievement index (r_a) and the matching index (r_m) , with most of the experimental work occurring within elements 1, 2, 3, 6, and 7 of the judgment modeling process.

Definition of Policy Capturing:

It should be noted here that in past research, the term "Policy Capturing" has been generally used to describe the analysis of the judgmental side of the Brunswikian Lens. However, for want of more descriptive terminology, the connotation of "Policy Capturing" has been expanded within this research to be as follows:

IDENTIFICATION AND QUANTIFICATION OF THE AT-TRIBUTES THAT ARE PERTINENT TO A DECISION AND THE SUB-SEQUENT MATHEMATICAL DESCRIPTION OF THE DECISION POLICY FOR THE EVALUATION OF THESE ATTRIBUTES is defined as POLICY CAPTURING.

Definitions:

ı,

At this point, it is beneficial to explicitly define various terminology that will be used throughout the description of past and current research in order to avoid any semantic difficulties that might arise relative to various types of mathematical models.

Linear Models: A linear model is of the form

 $Y = \hat{Y} + E = \sum_{i} b_i X_i + E$ $i = 1 \dots n$

where: \hat{Y} : $\sum_{i} b_{i} X_{i}$ = predicted value of the criterion i i: are parameters to be estimated X_{i} : are variables that are known for any given data entity or that can be compounded from other basic variables such as $X_{i} = Z_{i} * W_{i}$ or $X_{i} - Z_{i}^{3}$ E: the error

13

The definition "linear" refers to the fact that the parameters " b_i " are constant over the entire predictor space (V^n) and <u>does not</u> relate to the order or power of the variables " X_i ". All models considered as being linear models.

<u>Main Effects Models</u>: A linear model in which the highest power of any variable is unity.

<u>Curvilinear Models</u>: A linear model in which the power of some variable is greater than unity, i.e., X_i^3 .

<u>Configural Effects Models</u>: A linear model in which crossproduct terms of the form $b_{ij} X_i X_j$ where X_i and X_j are different variables, are included. This is also referred to as an "interaction" model.

Local Models: This is a linear model that applies over some subspace of the predictors, $S^m \in V^n$, and can be considered as a special case of the configural model. Whereas in configural models, the terms $b_{ij} X_i X_j$ are generally considered to apply throughout the entire predictor space, V^n , in local models, such terms are applied only over the local subspace S^m .

Judges: This is the common term for the individual rater or decision-maker who is making the judgment. Many of the past works use the terminology "subjects". <u>Cues</u>: Variables, attributes, predictors, or other characteristics of the decision situation which are perceived by the judge and are hypothesized as the stimuli for his decisions.

ñ

. . . .

• •

ò

e'i

• A

à,

3.52

Historical Data: Cue and decision data taken from recorded files, that is. the cues were considered and the decisions made prior to the introduction of the concept of Policy Capturing to the subjects.

CHAPTER II: REVIEW AND ANALYSIS OF PAST RESEARCH

Overview:

ş

. 14

÷.,

٠.

 \cdot

The bulk of past research in the area of judgment modeling has been accomplished at three institutions, each with a somewhat separate focal point. At the Oregon Research Institute (ORI) the interest of such researchers as Hoffman, Slovic, Goldberg, and Dawes has been on elements 3, 4, and 5 of the judgment modeling process as shown in Figure 1-1. Their main thrust has been in evaluating how well a modelof-man can do versus man himself and in investigating the best way to formulate the judgment models.

At Colorado's Institute for Behavioral Science, the focus of researchers Hammond, Hursch, Todd, Earle, and Brehmer has been on elements 4 and 7 of the process. Their main efforts have been in investigation of policy conflicts, interpersonal learning, and the training of subjects to use various types of cues.

The only real experience in using judgment models in a practical environment to date has been at the Air Force Human Resources Laboratory (AFHRL) where researchers Ward, Christal, Koplyay, Madden, and Mead have focused mainly on elements 4, 5, 6 and 7 of the process. They have specialized in capturing and clustering the policies that individual board members use in making USAF personnel decisions.

Although these institutions have taken a major lead in the past research in Policy Capturing, independent researchers including Einhorn (elements 3 and 8), Green (element 3), and Kleinmuntz (element 1) have made substantial contributions to the current body of knowledge in the field.

In 1960, Hoffman reported that one could "paramorphically" represent the policy of a decision maker very well with first-order linear regression models. He applied the term "paramorphic" to denote the fact that, even though the individual terms in the policy model did not necessarily represent meaningful physical constructs relative to the judge's logical processes, these terms could be combined so as to predict a set of judgments (usually rankings) that had a high correlation with the actual judgments. Hoffman contended that even though these policy models were paramorphic and might not portray the decision process accurately from a structural standpoint, they could account for a significant amount of the variability in individual judgments. He felt that useful information could be gained about the judge's dependence on each cue by examination of the Relative Weight for each cue. He defined the Relative Weight for cue "i" in the policy of judge "j" as:

$$RW_{i,j} = \frac{\beta_{i,j}r_{i,j}}{r_{i}^{2}}$$

where:

r_i

- $\beta_{i,j} = \text{standardized regression weight of the ith cue for the jth judge's policy.$
- r_{i,j} = correlation coefficient of the ith cue with the criterion for the jth judge. (Hoffman calls this the utilization coefficient.)

= response consistency of the jth judge.

This contention was quickly refuted by Ward (1962) in a demonstration that the Relative Weight was meaningless except in the very special case of orthogonality among the predictors.

Consequently, many researchers resorted to near total reliance on factorially designed experiments in order to have meaningful measures of comparison among predictor variables. To facilitate the comparison of policies on an inter-judge basis, a trend toward rather strict control of the data sets used in the analysis also developed. Hoffman (1960) justified such control by stating:

> ... restricting the situation (by controlling the stimuli) assures that each person is evaluated with respect to the same information. Ambiguous and equivocal cues are removed, and all judges are thereby certain to have at their disposal the same information and no more. The inferences made beyond this point are certain to have their origins in the data provided, (p. 118).

With his original work, Hoffman sparked two areas of controversy that are germain to the problem of applying Policy Capturing to an actual decision making environment. The first area is the use of first-order linear models as an adequate representation of the judgment process regardless of the underlying logical procedure. This approach can present a nearly insurmountable obstacle to practical implementation in convincing the decision maker, who may be unsophisticated mathematically, that an unexplainable equation can do as well as he can with his explainable judgment procedure. The second area is the use of highly structured and controlled data sets for capturing the policy of the decision maker. This technique often leads to a lack of credibility since the data used to capture the policy may not conform to the "real world" data that the decision maker must assess in his normal judgment process.

The adequacy of a paramorphic first-order model must really be evaluated relative to its intended use. If prediction of the final outcome is in the major purpose and there is no "selling to the management" problem involved, then a highly efficient first-order paramorphic is probably better than a more complex configural model.

....

This is exactly the situation confronting the research personnel at the USAF Human Resources Laboratory, Lackland AFB, Texas. These researchers were particularly concerned with providing decision aids to the numerous personnel boards which make such decisions as "Who should be promoted?" or "In which occupational specialty should a recruit with a given set of background qualifications be trained?" The goal in these instances is adequate prediction of the composite decisions of many judges for many cases and not the detailed analysis of how the individual judges make up their minds. The "selling job" is also minimized in that they are operating in an environment where they do not have to convince a given judge that the policy model is congruent with "the way he thinks he makes decisions", rather, the paramorphic policy model sells itself if it is predictively efficient in matching the composite decisions of the board members and can result in increased operational efficiency.

The AFHRL has published numerous studies in the development of policy models for specific areas. A few of these areas include assignment of personnel to jobs (Ward, 1962), merited pay and merited grade for jobs (Madden, 1963), promotion of enlisted men (Koplyay, 1969), and evaluation of jeb difficulty (Mead, 1970). In most of these studies, the key points of interest has been the comparison of the individual policies of the board members and the attainment of a composite policy model. Thus, they have emphasized the clustering and comparison of the policy equations and have developed and applied the JAN

i.8

Technique. With JAN, the captured policies of the individual judges are hierarchically clustered in a manner that minimizes the loss in predictive efficiency of the composite policies at each step. (Ward and Hook, 1961)(Bottenberg and Christal, 1961)(Christal, 1963)

An example of the information to be gained about the consistency of a given rater in applying "his policy" and about the homogeneity of "policies" within a group of raters can be illustrated by considering the results published by Madden. (1963) The data on Table shows the distribution of "rater efficiencies" (the multiple r_j^2 for the rater's paramorphic policy model) for each of 38 raters performing the ratings on each of 50 separate job-descriptor profiles. The "rater efficiency" reflects the ratio of the variation in ratings explained by the raters' policy model to the total variation in the ratings. It has a theoretical upper limit of unity. This ratio is an indication of the quality of the "fit" of the policy model to the policy used by the rater in making the ratings. Note that an efficiency of .99 was attained in modeling the policy of one individual relative to merited pay and this efficiency was attained for five individuals relative to merited grade.

·**.

с <u>Г</u>

ŝ

∂•

12

Table 1. shows the results of hierarchial clustering of the policy models of the raters. At each stage of the hierarchial clustering procedure, the number of groups or clusters is reduced by one by combining that set of policy models that reduces the composite r_j^2 the least amount. A sudden drop in r_j^2 at a given stage would indicate that two non-homogeneous clusters had been grouped at that stage. Note that in the last step of the procedure for the merited pay policies, a group of 37 raters was combined with a lone rater and the resultant r_j^2 dropped from .87 to .78. By using clustering techniques, one can determine if significant differences in policy exist among individuals or groups of individuals on a board. Then action can be taken to eliminate
RATER CONSISTENCY

Distribution of r²'s Achieved in Predicting Rank Order j of Merited Pay and Merited Grade from Factor (predictor) Values

2 r	Merited	Mer ited				
Ĵ	Pay	Grade				
. 99	1	5				
. 98	2	7				
. 97	6	2				
. 96	6	2				
. 95	6	2				
. 94	3	2				
. 93	1	0				
. 92	4	3				
. 91	1	0				
. 90	3	2				
. 89	1	5				
. 88	1	1				
. 87	1	2				
. 86	1	1				
. 85		1				
. 74	1					
. 72		1				
. 31		1				
. 15		1				
N=	38					

From: (Madden, 1963)

.

5,

že s

:8

9.

- سی .

್ರ**ಿ** ನ್ರ

 a^{j}

POLICY COMPARISON

Effects on r_j^2 of Number of Groups

Basis of Ranking	Number of Groups	Number of Judges Each Group	Comp r ²
Pav	38	1	. 93
- /	3	33, 4, 1	. 88
	2	37, 1	. 87
	1	38	. 78
Grade	34	1	. 92
	3	16, 16, 2	. 88
	2	32, 2	.86
	1	34	. 84
Pay & Grade	72	1	. 93
•	3	64, 5, 3	. 86
	2	67, 5	. 83
	1	72	. 81

50 Subjects: 10 Predictors: No Interactions: 10 Levels per predictor

*Note that only 34 raters were grouped on the Merited Grade equation due to obvious inconsistency in 4 of the raters.

From: (Madden, 1963)

÷

2

\$.b

=3

ji

members with deviate policies or to resolve conflicts and differences of opinion among the raters.

In a AFHRL sponsored study of the effectiveness of identifying and resolving policy differences by first capturing the policies and then using the JAN Technique, Mullins and Usdin (1970) compared this procedure with one in which each judge estimated the relative importance of each variable and then the judges mediated their weights for each of the variables and came up with a composite equation. They found that by using the JAN Technique, the quality of the group judgment always improved but that by using the latter technique, the mediated weight for a variable was often only a reflection of the opinion of the most powerful, or the most vocal, judge. Further, they found the quality of the group estimate was worse as often as it was better.

The use of policy aggregation as a catalyst to conflict resolution was further pursued and reported on by Stephenson and Ward (1970) in an experiment wherein they captured the policy model of each of several members of a promotion board and clustered their individual policy models. This resulted in a focusing of attention on the previously non-isolateable reasons for differences between board members in the ratings of potential promotees. With areas of difference brought into focus by means of different weights on the relevant variables, some limited progress was reported in getting the board members to come to a common agreement on the relative importance of the various factors. They reported that even though compromise on a policy did not result to the desired degree, the judges in the study reported that the technique had served a very valuable purpose in clarifying the positions and policies of each of the board members.

Both of these studies would suggest that policy capturing could provide a solution to the problem in face-to-face communications described by Robinson (1971) as "noise"; that is, discussion of various interests and not problem solving.

0 7. p

.....

y.**

_U

In most of the work carried on or sponsored by the AFHRL, a difference in policy between several judges, or groups of judges, has not been decomposed further than necessary to effect a resolution to a common policy or to identify and eliminate judges with greatly deviant policies. More recently, Brehmer (1970) has pointed out that apparent policy disagreements should really be decomposed into spearate components, one part due to individual policy inconsistency, and the other part due to inter-judge policy conflicts. His experimental results indicate that as judges with policy disagreements work to reduce the conflict that exists on an inter-judge basis, there is a counter-balancing effect with reduction in policy conflicts being accompanied by increases in individual policy inconsistency, resulting in very little net change in apparent differences. This would indicate that once people recognize the innate causes of policy conflicts, they may be willing to change their weightings of the cues, but they have a difficult time effecting the change because they cannot apply the new or "compromise" policy consistently. Brehmer also found that policy differences were reduced faster by free communication and free communication plus objective cue weights than they were by just communication of differences in judgments, a result that would support the utility of Policy Capturing as a catalyst for reducing policy differences.

The Configurality Debate:

A somewhat different set of interests has motivated the researchers for whom the first-order paramorphic model was untenable. They have persisted in what Goldberg (1968) has referred to as the "search for the configural judge". This continued effort to model the

policy of the judge with terms that are more reflective of the actual logic process of the judge is stimulated by: 1) the judges' introspective beliefs that they process cues in complex and configural ways, 2) the possibility that truly configural judgment tasks have not yet been studied, and, 3) the possibility that the experimental designs and statistical techniques available to the researchers were best suited for discovery of first-order effects and not particularly suited for unmasking configural relationships. (Slovic and Lictenstein, 1970) Many publications have appeared on both sides of the "first-order" versus "higherorder and configural" issue; the principle participants being Wiggins and Hoffman (1968), Goldberg (1968), Hoffman, Slovic and Rorer (1968), Einhorn (1970, 1971) and Valenzi (1970).

Evidence for the tremendous power of the first-order model was presented by Yntema and Torgerson (1961) when they demonstrated that 94% of the variance within a completely configural design could be accounted for by a first-order model.

Their analysis was based on synthetically generated data from a purely configural 3^7 design of the form:

W = aXZ + bXY + cYZ

with their paramorphic surrogate taking the form:

W = aX + bY + cZ

In reviewing the many studies using first-order model up to 1970, Slovic and Lictenstein (1970) state:

In all of these situations the linear 1 model has done a fairly good job of predicting judgments, as indicated by r^{2} values in the .80's and .90's for the artificial j tasks and the .79's for the more complex real-world situations. (p. 36)

On the pro-configural side of the issue, continued efforts have been made to find situations where the higher-order and configural models will consistently out-perform the approximation afforded by the linear surrogate, superior performance being indicated by the amount of variance explainable by the policy model. In one study, Wiggins and Hoffman (1968) compared the performance of a quadratic model and a configural "sign model" to the first-order models for 29 psychologists in making a diagnosis from mental symptoms of 861 cases of MMPI (Minnesota Multi-Phasic Inventory) data.² The results showed that for 16 of the 29 judges, configural models were marginally better, but not to a statistically significant extent. In general, other such efforts have not been impressively successful in increasing the amount of explained variance and have caused some frustration to the researchers. This frustration has led the researchers to search for different statistical methods and devise new tactics in applying old tools. Among the methods attempted have been the application of Analysis-of-Variance procedures (Hoffman, Slovic and Rorer, 1968) (Slovic, 1969) and the "a priori partialling out" of variance due to the hypothesized configural cifects (Valenzi, 1970). Even with these techniques, statistically significant evidence of configurality could not be established by the researchers,

å

Much of the current literature is rife with the misuse of the term "linear" to denote "first-order" models as opposed to including all models with linear parameters.

^c This set of data has become known as the "Meehl data" and has been used extensively by various researchers.

lending credence to Green's (1968) critique of all such efforts:

Another difficulty with standard configural techniques analysis-of-variance, and the new index³ is that they essentially are fishing expeditions. The experimenter will covet any configural effect, any interaction term he can find. He cannot begin to examine all the possible non-linear effects, and is very likely to miss those that are present, unless he knows where to fish. (p. 94)

Undaunted, other researchers have taken the approach of formulating completely new types of models and working with different decision situations in order to demonstrate configurality. The most prominent of these efforts has been the work by Einhorn (1970, 1971) in the formulation of the "conjunctive" and "disjunctive" models. Einhorn, hypothesized that the appropriate decision model is really a function of the type of decision task and other factors of the decision situation including: inter-judge differences, distribution properties of the cue values, number of stimulus objects, the correlation among the cues, the number of stimulus or cue dimensions to be evaluated for each object, and the context within which the decision is made. He stated that for certain types of situations, such as those in which the strongest attribute or the weakest attribute of the stimulus object was important, non-compensatory models of either the conjunctive or disjunctive type would apply respectively. The proposed conjunctive model is of the form:

 $Y = \prod_{i=1}^{N} x_{i}^{a_{i}} \quad \text{or} \quad \log Y = \sum_{i=1}^{N} a_{i} \log x_{i}$

⁵ The index referred to here is an index of cue consistency proposed by Hoffman (1968) as the "believability index" and relates to the degree to which separate cues match the preconceived ideas of how the sep. rate cues should be correlated with the criterion.

and the disjunctive model is of the form:

÷.,

et :

ź

4

. <u>'</u>

ы Ш

$$Y = \prod_{i=1}^{N} \left(\frac{1}{a_i - x_i}\right)^{b_i} \text{ or } \log Y = \sum_{i=1}^{N} b_i \log(a_i - x_i)$$

In his experiments, Einhorn pitted these models against the first-order models for four judges in the task of ranking graduate school applicants based on Graduate Record Exam scores. He found his models gave significantly better results for three of the four judges and the shrinkage of his models upon cross-validation was much less than that achieved with first-order models for all four of the judges. This last finding is of special interest, in that Ward (1954) found that configural models generally did not cross validate as well as first-order models due to "over-fitting". Hence, high cross validities for a configural model would appear to enhance its claim to authenticity.

Einhorn's work has offered new evidence for the long fostered opinion that configurality existed, if it could just be uncovered. It is important to note that Einhorn himself recognized that his models would probably not perform as well in different decision situations. This admission did not placate Goldberg, however, who was apparently disgruntled by Einhorn's refutation of his previous assertion that firstorder models fit best because they are mathematically simple and man thinks in simple ways. (Goldberg, 1968) Instead, Goldberg (1971) immediately set about the task of disproving Einhorn's findings on the much analyzed "Meehl data". The results of this vendetta showed that, as was nearly pre-ordained, the linear model did a better job than either the conjunctive or disjunctive models in depicting the judgment of clinical psychologist in making diagnoses. This merely tends to confirm Einhorn's hypothesis that the best model is a function of the decision situation. More importantly, in justifying his results, Goldberg gave the first indication that he and other researchers, within the heretofore "pure policy capturing" camp, were awakening to the necessity for more emphasis on "representative" research designs as espoused by Brunswik (1956) would be necessary.

Bootstrapping:

Ĵ,

8

ð j

<u>д</u> Э

u G

а. 0

90

نيونې مېر ک

8

., .

,

j. F

-Ø

111<u>1</u>

à

 v^{\prime}

Independent of the researchers belief or disbelief in <u>HOW</u>? the policy of a decision maker can best be modeled, there is always the problem of justifying <u>WHY</u>? they should attempt to model the decision maker at all. This debate has continued since Meehl's original proposal of the "actuarial combination of cues", with some clinicians, apparently justifying their jobs, claiming that the actual combination of the cues is only a small part of the overall judgment process. (Holt, 1970)

Most of the arguments for using policy models of the judge, as opposed to the judge himself, center around the consistency of a known and fixed computational formula as compared to the inherent variability of the human judgment process. One of the main proponents of using Policy Capturing has been Goldberg (1970) who described the human reliability problem by saying:

> He 'has his days': Boredom, fatigue, illness, situational and interpersonal distractions all plague him, with the result that his repeated judgments of the exact same stimulus configuration are not identical. He is subject to all these human frailties which lower the reliability of his judgments below unity. And, if the judge's reliability is less than unity, there must be error in his judgments-error which can serve no other purpose than to attenuate his accuracy. If we could...eliminate the random error in his judgments, we should thereby increase the validity of the resulting predictions. (p. 423)

In an analysis of the performance of first-order policy models versus the actual judgments on the "Meehl data", Goldberg (1970) found that the models were marginally better for all 29 judges than the judges themselves.

....

ः 4

т. 1

٩Ņ

Dawes (1971) also demonstrated the superiority of policy models over actual judgments in predicting the rank of graduate admissions. Dawes and his fellow researchers at ORI have termed the phenomena of the policy model's superiority to the human in predicting judgments as "bootstrapping." The essence of the justification for "bootstrapping" is that mathematical models, by their very nature, are abstractions of the processes that they model. The efore, if the decision maker's behavior involves following valid principles, but he is unable to follow them reliably, these valid principles will be abstracted by the model, and once abstracted, the superior reliability of the machine can be enlisted to follow them exactly, resulting in better quality judgments. It is recognized by this argument that this reduction in variability will occur if there is no systematic deviation from the principles based on the variables themselves, i.e., if the structural form of the model is correct. On this point, Dawes asserts that bootstrapping can occur even if there are structural defects of the model, such as in cases where a first-order model is used to represent a truly configural decision situation, if the cues themselves are monotonic and the data is fallible.

Dawes and Diller (1970) derived, and experimentally verified, a mathematical criterion based on the lens model parameters for determining when bootstrapping could be expected to occur for the case of first-order models of very good (near-optimal) judges. This condition for bootstrapping is:

$$r_{j}\sqrt{\frac{1-r_{j}}{1+r_{j}}} > \sqrt{r_{N_{j1}N_{j2}}} \sqrt{r_{Y_{j1}Y_{j2}} - \frac{r_{j}^{2}}{r_{\hat{Y}_{j1}\hat{Y}_{j2}}}}$$

where:

 \mathbf{r}_{j}

 ${\rm d}^{\rm d}$

 \mathcal{L}^{*}

)

ŝ

41

$$r_{j}$$
 = the predictability of the judge $(r_{Y_{j}}, \hat{Y}_{j})$
 $r_{N_{j1}N_{j2}}$ = reliability of the residuals (Y_{j}, \hat{Y}_{j}) , i.e.
the correlation of residuals in cross-
validation samples
 $r_{Y_{j1}Y_{j2}}$ = reliability of criterion values in cross-
validation samples

validation samples

reliability of predicted correlation values 2 ${}^{r}\hat{Y}_{j1}\hat{Y}_{j2}$ (\hat{Y}_{j}) in cross-validation samples.

In this same study, they proposed amalgamating the firstorder prediction of the judge from the policy model with the judge's actual prediction to get a hybrid prediction. This amalgamation is based on weighting the current judgment and the policy-model predicted judgment in the form:

$$\mathbf{A} = \hat{\mathbf{r}}_{\mathrm{m}} \mathbf{z} \hat{\mathbf{Y}}_{\mathrm{j}} + \hat{\mathbf{r}}_{\mathrm{n}} \mathbf{z}_{\mathrm{N}_{\mathrm{j}}}$$

where:

^zŶj the standardized "z-score" form of the predicted value \hat{Y}_{j}

$$z_{N_{j}} = \text{the standardized "z-score" form of the residual}} (Y_{j}-\hat{Y}_{j})$$

$$\hat{r}_{m} = \text{bounded validity} = r_{j}/\sqrt{r_{Y_{j1}Y_{j2}}} coefficient for the model bounded validity} = \sqrt{r_{N_{j1}N_{j2}}} \sqrt{1 - \frac{r_{j}^{2}}{r_{Y_{j1}Y_{j2}}}} \frac{r_{Y_{j1}Y_{j2}}}{r_{Y_{j1}Y_{j2}}} \frac{r_{Y_{j1$$

Their experimental work on this procedure indicated that the amalgamation of predicted and judge criterion values was superior to the judge alone or the prediction alone, especially in the case where non-linear⁴ cue use was considerable.

р

ن ن

; [0

..

,

. 8

IJ,

Δ.,

Bowman (1963) and Kunreuther (1969), have also proposed and studied the use of policy models of expert managers to reduce variance in management decisions relative to production management and inventory control. They have suggested that such a technique would be an excellent surrogate for the more traditional analysis, much of which is dependent on relationships of decision and responses that are greatly separated in time, or which is based on cost parameters that are intangible and must be estimated or assumed. This suggestion is tantamount to stating that it is more efficient to model the expert's opinion of the proper policy to follow, than to model the actual response as a function of the decision policy. This rather intriguing thought of using experts to estimate the cue-response relationships when the actual relationships were unattainable was also studied by Mullins and Usdin (1970). In that study, the use of judges to estimate the relation-

⁴Here "non-linear" refers to "higher order and configural" per the definition in Chapter I.

ships between the actual criterion values and the cues produced estimated criterion values that were virtually no different than the actual criterion values.

Sawyer (1966) analyzed the results of some 45 studies accomplished prior to 1966 and proposed that the judgment task be divided into two subtasks, measurement and prediction. He concluded that the perceptive capabilities of the human were best put to use in the measurement task and that the inherent reliability of the machine made the policy models superior in the prediction task. Sawyer's lead in decomposing the global judgment task into these two subtasks has more recently been followed by Einhorn (1972). Einhorn has agreed with Sawyer and proposes that not only should the measurement task be relegated to the expert judge, but that the whole process might be improved even further by using multiple judges in the measurement task with each judge being an expert in the measurement of some particular attribute. Such a technique would then use the expert measurements as an input to an actuarial combination of the cues in attaining a global judgment.

Apart from the justification of the relative accuracy and efficiency of using policy models, the <u>WHY</u>? of Policy Capturing can be answered relative to management's effective use of computers in the current man-machine era. Many authors such as Yntema and Torgerson (1961) have discussed management's need to "teach the computer how they want decisions to be made", and others state that in the operation of large organizations in the future, "specialist's knowledge will be stored in computer banks". (Vandell, 1970) Since computers are best at manipulating mathematical expressions, the use of captured policy rodels to instill man's value judgments in the computer appears to be the only logical approach to the task.

The Nature of the Decision Process:

93 B

In addition to studying the more mundane aspects of policy modeling such as the practicality of using these models as decision aids, a very considerable effort has been undertaken by psychologists in using Folicy Capturing as a descriptive tool for understanding the fundamental cognitive processes of the decision maker. Some of these results have been the direct goal of specific research, while others have come about as by-products of the more applied research efforts. In any event, it is well to be cognizant of a number of these findings in any project in which a viable set of policy models is to be sought for use in a practical environment.

One of the original postulates in the area of cognitive processes was the concept of "cognitive strain" advanced by Bruner, Goodnow, and Austin. (1956) They hypothesized:

> Where the nature of a task imposes a high degree of strain on the memory and inference, the strategy used for coping with the task will tend to be less conducive to cognitive strain. To put it in terms of an analogy, if someone has to move a heavy weight, there is...likelihood that the mover will have recourse to strain reducing techniques for carrying out his task. (p.112)

Another of the fundamental postulates resulted from the work of Miller (1956) who found that there are rather low limits on the number of items of data that the human can evaluate and integrate in a consistent manner.

In line with these postulates, Einhorn (1971) found that the judge made more consistent judgment when fewer cues were available. He proposed that man uses excess data in some hybrid fashion instead of discarding it as might be expected in any effort to reduce cognitive strain. Similarly, Slovic and Lictenstein (1970) concluded that the results of past research showed that more information gave the judge a false confidence without improving his prediction consistency.

J

de

័ររីទ័

8.1

Slovic (1969) investigated the relative uses of first-order and higher-order cues in individual and group judgments. He found some evidence that when multiple cues were present, the tendency to use curvilinear models was less in the case of group judgments, while it was more in the case of individual judgments. This suggests that the averaging effect of group judgments may tend to obscure configurality in individual policies, especially if the configurality is different for different judges.

Earle (1970) studied the learning and interpersonal learning characteristics of judges trained on first-order and higher-order cues. He found that those judges trained initially to use a higher order cue could readily adapt to using a first-order cue without the aid of a teacher, but that those judges who had been trained on a first-order cue had to receive help in adapting to the use of a higher-order cue.

Another point of investigation has been in the area of sequential decision processes. Although most of this work has been carried out within the previously mentioned Bayesian Approach, Slovic (1966) and Hoffman (1968) have proposed that the judge first assesses the "believeability" of the cues and then proceeds to adjust the cue weights of his first-order policy model accordingly. In a more exhaustive work, Bettman (1969) essentially pursues this same notion except that he hypothesizes that the decision maker first assesses the consistency of the cues and given they are consistent, uses a simple or first-order model; however, if they are inconsistent, the decision maker uses a configural model in what Bettman calls a "problem

solving" approach. These findings are suggestive of the "lexicographic" model mentioned by Einhorn (1970) in which it is hypothesized that a decision is made on the most important variable first and then remaining variables are considered in the order of their importance. Unfortunately, Einhorn and others defer from pursuing this model due to their inability to mathematically model the process.

Summary:

.

" . și

Probably the strongest general impression to be gained from a review of the past research efforts in Policy Capturing is the artificial nature of the environments in which the research has generally been conducted. This is especially paradoxical in that one of the basic tenents of the Brunswikian Lens conceptualization is the use of "representative" designs. Only recently, researchers such as Goldberg (1970), Einhorn (1972), and Dawes (1971) have begun to voice concern over this departure from reality, with Dawes expressing his reservations of past research results in the statement:

> Most of these studies have been conducted in analogue experimental situations and it is reasonable to ask whether the situations were constructed in such a way that the linear model works, but are not representative of the real world decision making situations. (p. 181)

Most of the past analyses have been performed on decision tasks where only a limited number of well-defined cues have been used. As a typical example, Valenzi (1970) arbitrarily reduced the number of cues in his study from 20 to 5 based on a preliminary interview with one subject in the study. Such practices would seem to be risky in the face of data indicating that even experienced judges have a difficult time accurately describing the relative weights of the variables in their policies (Slovic and Lictenstein, 1970). Valenzi's own results indicated that only one of the four judges in his study was able to correctly estimate the relative importance on the variables in their policies.

2

. ອີງ ແມ່ນ ເບ In most studies even the procedure of asking the subject which variables they "felt" were important, before deciding on a predictor set, has been omitted with the subjects being given a set of predefined cues. This practice, which is equivalent to ignoring steps 1 and 2 of the Policy Capturing process as depicted in Figure 1-1, page 7. was first challenged by Garner (1970) when he noted that the psychologists modeling clinical judgments had paid virtually no attention to the problems of identifying. scaling, and coding appropriate stimuli. Einhorn (1971) later asserted that:

The practice of presenting cues to the judge in a decomposed form not only imposes the experimenter's own judgment as to what the relevant cues actually are but more importantly it does a considerable part of the cognitive work for the judge. (p. 8)

A related problem with the artificial environment is the extent to which the experimenters have relied upon the statistical and computational niceties of factorial designs in choosing their stimulus data sets. In an early study of the possible effects of using sample data sets that were statistically different than the underlying population, Dudycha and Naylor (1966) perturbed the predictor intercorrelation matrix from the true values and studied the resultant effects on the policies derived for the judges. They found that perturbation of the interrelationships between predictor variables to minimize the correlations resulted in lower rater consistency and better discrimination of the effects of the individual variables, while increasing these correlations resulted in higher rater consister :y and worse discrimination at the individual variable level. This would indicate that artificial orthogonalization of the predictor variables by use of factorial designs with equal-cell frequencies, would tend to produce policies that were less predictive than they should be. Hence, computational efficiency has been purchased with decreases in accuracy in such experiments.

સ્ત્

Å

а г

ð

In addition to these rather arbitrary decisions relative to which variables and which data units to use in the experiment, the criterion for evaluation of the various policy models has usually been restricted to using the percentage of explained variance, as expressed by the squared multiple correlation coefficient (r_j^2) . Slovic and Lictenstein (1970) found that most of the models that they reviewed were <u>NOT</u> cross-validated. Further review by this investigator, indicates that only in the case of those models developed or sponsored by the researchers at the AFHRL, and in a few other cases such as Bettman (1969), has cross-validation of the derived policy equations been accomplished on a systemmatic basis. Admittedly, explained variance is a commonly used "measure of goodness", but with small data samples models can achieve inflated values of r_j^2 solely due to the idiosyncratic nature of the sample data.

A second general impression of the past work is the rather restricted range of modeling possibilities that have been entertained. Within the correlational paradigm and the Brunswikian lens framework, almost all of the work has been confined to linear first-order, higherorder, and configural models of the type:

 $Y = \sum_{ij} b_i x_i + \dots + b_j x_j + b_{ij} x_i x_j + E$

where the values of h_i , b_j , and b_{ij} are considered to apply over the total predictor space, V^n . Hoffman (1968) characterized such models as being n-dimensional hyperplanes and states:

The concept of a best-fitting hyperplane implies sort of an averaging of the effects of all points in the multiple dimensioned space, and, since linearity is assumed, the regression coefficients are invariant over the entire stimulus space. (p. 59)

Even Einhorn's (1970) departure from the more standard main-effects or main-effects-plus-configural-effects modeling approaches by using the conjunctive and disjunctive models still represents the fitting the entire predictor space by a single, continuous hyperplane.

*

. . The only works that reflect a departure from this concept are the efforts of Slovic (1968), Hoffman (1968) and especially Bettman (1969). In the first two of these studies, it is proposed that the judge adjusts the coefficients in his first-order model based on an initial assessment of cue-consistency or "believability", while in the latter study, the whole structure of the policy model is dependent upon this initial assessment. These procedures effectively divide the predictor space into a set of discontinuous hyperplanes, each hyperplane representing a local policy or a local model. Bettman (1969) appears to be the only one of these researchers to recognize this implication with his conclusion:

> The major and most useful conclusion to draw from the above ... (research)... is that a linear or less configural model can be viewed as a complex model conditioned by a given cue configuration. (p. 134)

The real potential of the concept of modeling the entire predictor space, V^n , with a series of disjoint hyperplanes over subspaces, S^m , of the predictors has not been further addressed. However, this technique presents a possible solution to the problem of obtaining models that retain the desirable characteristics of first-order models while accounting for the configural logic processes that many judges feel they use. One possible reason that this area has remained unexplored is the problem of determining the basis upon which to deaggregate the total predictor space into appropriate subspaces. Such a division would surely have to be based on the inherent characteristics of the policy and the predictors for it to be meaningful. Any arbitrary approach would be analogous to Green's (1968) "fishing expeditions" and would be doomed to frustration. Blaylock (1969) voices a similar opinion in his discussion of building theories with multivariate models by stating:

> One cannot simply build theories involving interaction effects without some rationale to suggest the conditions under which non-additive relationships might be expected. (p. 156)

Thus, a viable methodology for identifying appropriate subspaces over which local models could be fit would appear to be a necessary precursor to the exploration of the approach of using multiple hyperplanes, as opposed to using a single hyperplane, in the modeling of policies.

зŧў

46.

As a final comment in this review, the observation of Slovic and Lictenstein (1970) that resulted from their own exhaustive review of the previous research is germain. They concluded that for as much work as had been accomplished in many different areas of



judgment modeling, surprisingly little cross-communication among the efforts had transpired. They compared this situation to a speeding vehicle without side-windows, the only visibility being directly down the path of travel of a given paradigm, or modeling strategy. With this comment they have noted the requirement for a side-window; the review of the recent work of others such as Hoffman, Bettman, and Einhorn suggests a faint outline as to where a side-window might be carved; it is the purpose of the work which follows to open this sidewindow so that future researchers can look out and the practitioners of Operations Research can look in.

CHAPTER III: FOCUS OF THIS RESEARCH

Application of Policy Capturing in Practical Environments:

۰.

ą

45

The introduction of the Policy Capturing Concept and a review of the past work can provide the genesis for many tantalizing ideas of the potential benefits that such a technique could offer to Managers, Educators and other Administrators in extra-laboratory environments. However, as with any scientific concept, a problem of the conversion from a fairly general collection of thoughts to a specific methodology exists in the adaptation of the Policy Capturing Concept.

The review of past research pointed out that such a conversion has only been previously attempted in a few instances, specifically, those studies undertaken by the researchers associated with the Air Force Human Resources Laboratory (AFHRL). In these cases the effective implementation of the concept has been materially aided by the particular decision tasks and the environments involved; namely, decision tasks in which the predictors were such quantities as test scores, aptitude indices, and performance ratings; data sets that consisted of large quantities of data; underlying judgment processes that were very well represented by first-order policy models; and, minimal emphasis upon correlation of the paramorphic models with the actual logical procedures of particular judges.¹ Thus, even the AFHRL work has been

¹In the latter aspect, the difference between a policy model that is visible and can be explained and a policy model that is reflective of a particular judge's self-perceived logical process is important. The only requirement for the former is that the variables and structural relationships in the model be relatable in some rational manner to the decision process; the requirement in the latter is that those relationships must not only be rational, but also, they must be in harmony with the way the individual judge "believes" he makes decisions. in environments with attributes that make them pseudo-clinical in nature since they are closer to ideal experimental situations than can be generally expected in the business, academic, and governmental or other administrative types of environments. In the extra-laboratory environments, categorical predictors, limited data sets, complex and configural logic processes, and skepticism for "black box" answers are probably more characteristic of the decision situations.

The focus of this dissertation is directed at exploring the hypothesis that Policy Capturing can be applied effectively in the extralaboratory environment. The fundamental assumption underlying this hypothesis is that the successful application of Policy Capturing to practical decision tasks is dependent upon providing the decision modeler with an efficient methodology for deriving explainable policy models that reflect the mental procedures that the decision maker feels he goes through in making judgments. This assumption is the basis for the first part of the effort of the research, i.e., the analysis of various modeling alternatives and the development of a methodology that will lead to models that are both accurate and descriptive of the underlying logical processes. The stated hypothesis is the basis for the remainder of this research, i.e., the application of Policy Capturing in an extra-laboratory environment and the analysis and demonstration of the benefits attainable by applying the Policy Capturing technique in this environment.

At this point, a further characterization of which types of decision tasks would probably be most amenable to the use of Policy Capturing is warranted. In making such a characterization, one might conceptualize all decision tasks as lying in a two-dimensional envelope such as is shown in Figure 3-1. One axis of this envelope corresponds to the inherent subjectivity of the decision situation and the other axis

CHARACTERIZATION OF DECISION TASKS

Α.

ം. ട

Υ.

2 A 48

. • •

Ł



EXAMPLE JUDGMENTS

- A, B: Computation of Perimeter Length; only judgment involved is in reading measurement scale, A--perimeter of a circle; B-perimeter of a rectangle
- C: Acceptability for Graduate School (Dawes, 1971); two judgments involve weights for variables and evaluation of the quality of the undergraduate school (QS).
- D: Probability of Being Hired as Secretary (Valenzi, 1970); up to 20 different predictors could have been used.
- E: Evaluation of Job Difficulty (Mead, 1970); both subjective and quantitative variables used to predict job difficulty.

FIGURE 3-1

corresponds to the dimensionality of the predictor space. Any decision situation in which the relationship between the predictors and the criterion cannot be fully explained by some immutable natural law would fall somewhere within this envelope. Any decision situation within the envelope has a requirement for some degree of human judgment in making the final decision. At one end of the subjectivity axis, the selection of which equation or physical law should be applied is the only part of the total decision left to the judge, while on the other end of this axis, the judge is totally responsible for the subjective identification of the relevant variables, the assessment of their individual relationships with the criterion, and the combination of their contributions into a global judgment.

ց.

Policy Capturing would be particularly applicable in the region of this envelope where considerable fuzziness exists in the relationship between predictors and the criterion, and in that region where the dimensionality of the problem is high.

The more mechanical aspects of decision scenarios that solicit the application of the technique are:

- The decision task is routine and repeated on a frequent basis with the relative importance of the various predictors remaining stable in the short term.
- There are multiple decision makers for whom a composite decision is desired or whose individual decisions should be consistent.
- 3) The quantity of the data for each decision task might lead the judge to simplify the decision process and focus on different subsets of the cues from case to case in an effort to reduce cognitive strain.

- 4) The decision process is possibly subject to biasedness and the detection and removal of this biasedness would be beneficial.
- 5) The actual predictor/criterion relationship can only be determined after a considerable time lapse and such a time interval can render any model of this relationship inappropriate and obsolete due to environmental changes.
- 6) The relative importance of the parameters of the decision process are not known and are not estimable from explicit physical relationships.

Given these basic characteristics of "fuzzy" decision processes and the mechanical aspects of target applications, a more detailed discussion of several functions that Policy Capturing serve is possible. Five generic roles in which Policy Capturing would be beneficial are:

۲. ۵

30

نە :

3

A. Factoring Expert Judgment into Automated Decision Making

As noted in the Introduction, the Operations Research practitioner has long been concerned with optimization of value based on models of various operational systems, but these models often include imprecise "guess-timates" of Management's utilities for risk and profit in the form of objective function coefficients. The more precise approach of implicitly determining a manager's expert opinion of the relevant variables and their weights via Policy Capturing would assure that the essence of the manager's expertise is factored into the decision model and not just those aspects of his experience that are easy to quantify or verbalize. Whereas the objective functions of many organizations are really a reflection of the composite judgment of managers at many levels, the use of explicit policy models for the contributors would provide an improved basis for discussion and integration of policy opinions of each manager in a manner commensurate with his proper role in the company. With less explicit procedures, a person's impact upon organizational policies are often based solely upon his debating ability, dominant personality or title, and not upon the quality of his opinions.

B. Standardization of Policy

The use of Policy Capturing techniques to assess intra-judge consistency, inter-judge consistency, and even inter-organizational consistency would appear to have merit. On an intra-judge basis, the inability of a given judge to make consistent judgments over a set of similar judgmental situations could indicate a fundamental uncertainty in his perception of the task or possibly a higher susceptibility to random external pressures. On an inter-judge basis, the assessment of the consistency of two or more persons performing identical decision tasks would help to make the performance of this task invariant to which judge makes the decision.

The use of Policy Capturing to detect systematic differences on an inter-organizational basis is particularly intriguing from the standpoint of administering a large de-centralized organization. In cases where a policy is set at some central headquarters, but administered at various subordinates levels such as divisions of large corporations or regional offices of government agencies, Policy Capturing could provide a vehicle for identifying communications and interpretation problems between the policy makers and the policy administrators.

C. Training

The use of Management's policy models would also be of value in the training of new personnel. This is especially true where the judgment process is highly subjective and Management is continually faced with the dilemma of whether to use the experts to "do-the-job" or to "train-the-new-people". Codifying the expertise of the seasoned employee in the form of policy models and using these models as training aids would enhance the transfer of knowledge and experience without requiring continual participation of the expert judge. Such models could also be used as a baseline or standard with which experienced employees, who may have lost contact with the lower level decision making process, could "recalibrate" themselves.

D. Policy Simulation

Once a policy has been captured, the modification of the policy model to simulate changes in the policy could provide a valuable tool for Management in performing a "sensitivity analysis" and assessing impacts of such changes on an <u>a priori</u> basis. This application was proposed by Ward and Davis (1963) as an efficient method for analyzing many possible policy alternatives before institution of any changes, especially in cases where the effect of a change might take a considerable time to become evident.

E. Mechanization of Routine Judgments

The phenomena of "bootstrapping" and the inherent reliability of machines presents possibly the greatest opportunity for benefit to be gained from Policy Capturing. Not only does the reliability of the machine, in evaluating and integrating numerous stimulus variables, offer advantages, but further, the potential increases in speed and efficiency to be gained in performing the function with minimal human intervention are substantial. The use of the machine allows for the analysis of more data than can be used by the human on a continual basis. For example, in the only previously published estimate of savings to be gained by using policy models, Dawes (1971) estimates that approximates \$18 million could be saved annually, on a national basis, by mechanizing the evaluation of graduate student applications using captured policies of admission boards. Substantial savings can be imagined even if the computer does not make the final decision, but only pre-processes the data and provides its evaluation to the judge in such a form that allows the judge to make his decision more efficiently.

The Transition from Hypothetical to Methodological:

Speculating and hypothesizing as to the potential of a concept can generate considerable enthusiasm, but such activities do not constitute verification that the concept is anything other than a dream. If a concept is purely theoretical, the researcher can set about defining ppstulates, lemmas, and theorems which will result in necessary and sufficient logical proofs of the concept's validity. However, if one is dealing with such "fuzzy" entities as man and the human judgmental process, he is confined to the world of empiricism wherein the only available approach to verification is demonstration. Given numerous repeated demonstrations over a broad range of conditions, these empirical demonstrations can result in acceptance and tacit verification of the concept. However, such verifications only hold within the bounds of the underlying assumptions and population characteristics for which the demonstrations were performed, a point often overlooked when assumptions become so commonly used that they cease to be emphasized in the analysis and description of the results.

Many researchers who find themselves operating in the world of empiricism try to regain some measure of "theoretical legitimacy", or at least try to simplify the verification procedure, by relying upon the theoretical developments of mathematical statistics and demonstrating "statistical significance". This too, requires the making of assumptions which often go unsubstantiated, being justified only by the claim that they are commonly used. Among the most commonly used assumptions are those of randomness, univariate or multivariate normality, independence, and homoscedacity. The carte blanche use of these assumptions is one of the major problem areas in translating many of the findings from the clinical environment to operational environments. If one is to effectively apply his results to the real-world he must derive these results under real-world experimental conditions or demonstrate that any underlying assumptions do not vitiate his results in the real world.

Criteria for Attaining Effective Models:

5

٠.

υā

Marrie I.

Consistent with the philosophy of performing Policy Capturing with ecologically valid data and being fully cognizant of any assumptions, the criterion for the measurement of the quality of a model must also be congruent with the requirements in the environment being modeled. Establishment of such requirements represents a further delineation of the previously stated fundamental assumption of this research. The basic characteristics that models should have to be viable in the business, academic and administrative fields are:

A) Accuracy: The models must possess at least that degree of accuracy necessary to produce a "bootstrapping" effect and to provide an improvement in operational efficiency. 49

Mill. 1999 - 1999

B) Reflectivity and Interpretability: The models should be reflective of the logical processes of the judge in order that he may identify with the model and have confidence in it.

\$

C) Communicability: The model should be readily communicable if it is to be used as a training device. Simplicity in a model would add to its communicability to and by non-scientifically oriented judges.

D) Data Amenability: The model must be able to accept and use data that is defined and categorized similarly to the manner in which the judge perceives and uses it in his normal judgment procedure.

E) Data Conservatism: The modeling procedure should be operable with data quantities that are not prohibitively large in terms of data collection and should not require major disruption of the ongoing decision process for data collection.

The above delineation of the characteristics of effective models holps to clarify the criteria for measuring model goodness, but the very nature of these characteristics does not readily admit them to quantitative measurement. The researcher is still left with subjectivity in the evaluation of various models; however, the dimensions of this subjective judgment have been better identified.

These criteria are not only qualitative, but their applicability is also dependent upon the functional application that is intended for the model. Table 3-1 reflects an attempt by this researcher to list the three most important model characteristics for each of the functions that were identified previously.

TABLE 3-1

FUNCTION

MODEL CHARACTERISTICS

Factoring "Expert Judgment"	A, B, C
Standardization of Policy	A, B, C
Training	B, C, D
Policy Simulation	A, D, E
Policy Mechanization	A, B, D

The Research Approach:

1

The foregoing definition of the experimental arena and the research objectives is a necessary prelude to the definition of the specific research approach and research tasks to be accomplished. The two major objectives of the effort are:

- the analysis and development of a modeling methodology that will produce policy models with the desired characteristics, and,
- 2) the demonstration of the potential viability and utility of policy models of an actual decision process in an extra-laboratory environment.

This research effort is unique with respect to past judgment analysis studies since it includes the pursuit of both of these objectives simultaneously. The effort to adapt and apply a new methodology for defining policy models is actually embedded within the effort to obtain and validate policy models in an operational environment. In light of the current state-of-the-art in judgment modeling, the success of the methodology development effort is considered a pre-requisite for the accomplishment of the second of these objectives. In terms of the functional steps of the traditional Policy Capturing process as depicted in Figure 1-1, page 7, the total research effort represents the tailoring of all of the functions to accommodate the conditions in an extra-laboratory decision situation. More importantly, the methodology effort involves the development of a completely new procedure for step #3 of that process.

In this respect, two separate contributions are expected from this research. The first is a set of procedures or guidelines for constructing policy model in extra-laboratory decision situations. These guidelines and the discussion of the model building methodology are addressed in detail in Chapter IV. The second contribution consists of the facts and implications that are generated in the process of conducting the Policy Capturing exercise in the particular decision situation. The background of the decision environment and the results of the Policy Capturing exercise are discussed in the format of a case study in Chapter V and Chapter VI.

An Overview of the Research Tasks:

霻

The fundamental task of developing a model building technique is accomplished through the definition of "structural image" models with the Automatic Interaction Detection (AID) procedure. AID was recently developed by a group of sociologists at the University of Michigan for the purpose of discovering structural relationships between variables in large quantities of sociological data.

The motivation for pursuing this approach is the similarity between the conceptual diagram of a configural-effects judgment process, as shown in Figure 3-2, and the binary tree that results from the application of the AID technique. Figure 3-2 schematically diagrams the configural effects judgment process in which the relevant stimulus

.'S	6	< 35	<u>Age</u> > 35	6 7	Eager	Aggressiveness Reluctant	4			7	Dependable	<u>Erratic</u>	4			
A HYPOTHETICAL "CONFIGURAL-EFFECT JUDGMENTAL PROCESS		80	Over Two Years 6	College Job Experience	Two Years or Less	2		Educational Level		-	6	Skilled	High School	4 Mechanical Skills Level	Unskilled	۲
							Estimated	Employee	Value to	Organization //_///						

FIGURE 3-2

Ø

1

3**88**5

at io 17

Q

variables and their relative importance is obviously dependent upon the value of the previously considered stimulus variables. In a conceptually similar manner, AID performs a "local analysis" of the data and builds a binary tree that reflects the "structural image" of the data. In this tree each node represents a subgroup of the total data set. Each subgroup is uniquely defined by a common set of predictors and, each split or branching reflects the most important variable relative to reducing the unexplained variability in the data at that node. Hence, the splitting process reflects the importance of the predictor variables relative to those predictor variables defining the subgroup or node. If interaction and configural relationships exist among the predictors, it can be detected via analysis of the predictors involved in the splits and the relative importance of each predictor after each split.

54

In this research, a conceptual link between the "structural image" model of AID and configural judgments processes is made in terms of modeling judgment processes with a series of local hyperplanes. A local hyperplane is a hyperplane defined over a subspace of the predictor set.

The major subtasks within this effort that relate specifically to the building and application of policy models include:

- Identification of an extra-laboratory decision situation in which most of the potential benefits from policy capturing could be derived.
- Identification of <u>all</u> potentially pertinent variables
 based on discussion with actual decision makers.
- Collection of data representative of the population of decision cases.
- Coding, analysis, and refinement of data categorizations and predictor definitions.

- 5) Collection of decisions by each participating judge on codified predictors over the data set.
- 6) Investigation of alternative modeling approaches.
- 7) Development of models for each of the participants.
- Verification and field-testing of models in an actual extra-laboratory environment.
- Analysis of benefits and problems in actual application of the policy models.

The decision process chosen in subtask 1 as the basic vehicle for this research is that of the installment credit officer in granting or denying installment credit on durable appliances. This decision is based on credit applications forwarded to the credit institution by the appliance dealer and does not involve direct personal contact between the decision maker and the applicant. Several of the more pertinent reasons for using this situation as the basis for the research are:

ï., 1

- The decision process is felt to be configural in nature.
- The data and decisions are real and typical of the types of data and decisions made throughout business, academic, and governmental agencies.
- The decision task is a routine and frequent part of the judges daily duties.
- 4) The judge has an economic motive for making the proper decision in the most efficient and accurate manner and is willing to use any decision aids that are understandable and effective.
- 5) Most of the hypothesized benefits of Policy Capturing could accrue in this environment.

55
6) Attainment of meaningful results in this application
 should constitute a demonstration of the general
 viability and utility of the Policy Capturing methodology
 in the extra-laboratory environment.

Subtasks 2, 3, 4, and 5 are accomplished with the objective of minimizing the influence that the procedures and decisions of the data definition, collection, or coding phases have upon the derived policy models. Particularly important aspects of these subtasks include:

> Initial identification and coding of the predictors is accomplished through extensive discussions with the loan officers involved with refinement and recoding of the predictors variables resulting from the initial analytical efforts.

and,

R

2) The decision data consists of a random sample of approximately half of the actual decision cases that occurred during a five-month period from May to September, 1971. Actual decisions made prior to the start of the project are used in one part of the analysis. Decisions subsequently collected for each of five loan officers on the codified version of the predictor variables are also used.

This emphasis in collecting data and decisions "au naturel" represents a departure from the usual procedures of policy modeling studies which work with predefined predictors and fixed categorizations. This approach complicates the whole Policy Capturing procedure, but hopefully justifies itself by resulting in models that are more realistic and easier to implement in actual environments. Subtasks 6 and 7 involve the development of first-order and second-order policy models for the judges and the comparison of these models with the "local models" attained through the application of the techniques resulting from the methodological development effort.

Finally, the collection and use of cross-validation data by the field test approach in subtask 8 is unique to this study. This subtask not only provided a measure of cross-validation of the derived models, but also, identified potential implementation problems and served as a demonstration of the viability of the concept to the participants.

Summary:

The focus of this research project is unique in that it not only involves the entire Policy Capturing process, and the application of the entire process in "real-world" environments, but it also involves a major methodology adaptation as an integral part of research.

The primary goal of the research is obtaining policy models that are "implementable". The criteria for "implementable" models are subjective at best, but a set of desirable model characteristics has been proposed in this chapter. The absence of these characteristics may not dictate inviability for a model, however, the fundamental assumption of this research is that possession of these attributes will enhance the viability of a model. As noted by Rosenthal (1966), such an assumption reflects a bias in the attitude of the investigator and could be a significant determinant in the results. Therefore, the research approach has been formulated so as to enhance the credibility of the results as much as possible by modeling actual decision makers who were making actual decisions in their natural environment, and then subjecting the resulting models to the most rigorous of all validation tests -- analysis of their performance under actual environmental conditions.

CHAPTER IV: THE DEVELOPMENT AND USE OF STRUCTURAL IMAGE MODELS IN POLICY CAPTURING

Introduction:

·· 1.

S. M

...

This chapter presents the conceptualization, an example, and guidelines for the modeling of configural judgment processes as a series of local models which define hyperplanes over subspaces of the predictor set. The adaptation and use of the AID algorithm as a tool in portraying the structural image of the judgment process and in defining appropriate subspaces and hypothesizing local models is discussed.

The technique is applied to define local models for the judgment data published by Valenzi (1970) as well as to define the policies of installment loan officers. The results of the former analysis are compared with the results published by Valenzi for his second-order regression and ANOVA models. The modeling effort for the loan officers comprises a major effort for this research and is discussed in detail in Chapters V and VI.

Conceptualization of the Configural Judgment Model as a Series of Subspace Hyperplanes:

Past modeling of judgment processes as either first-order, higher-order, or configural linear statistical models has generally been in the context of "continuous" models. In this discussion, the descriptor "continuous" has the following connotation: 1. If a model is of first-order, the coefficients of each of the predictors applies to all values which that predictor can assume.

ŝ

, e.,

antitez.

2. If the model has interaction terms, the configural relationship between the interacting variables is assumed to be the same over all possible values which the individual variables can assume and the coefficient for the configural term is applicable to all possible combinations of these values. Hence the adjective "continuous" is relating to the domain over which the terms of the model are defined and does not relate solely to the "linearity" of the model coefficients.

Continuity has been inherently imposed through the formulation of the judgment models as linear regression models of the form:

 $Y = b_0 + b_1 x_1 + b_2 x_2 + \dots + b_{12} x_1 x_2 + \epsilon$

where the coefficients, "b,", apply over the entire predictors space, v^n .

As discussed in Chapter II, Hoffman (1968) and others have described such models as representing the "best fitting hyperplane" over the predictor set with the coefficients reflecting the "average" effects over the entire stimulus space.

The near-universal representation of judgment processes as "continuous" models appears to largely have been a function of the availability and the understanding of the standard regression computational routines and the types of predictors that were considered in the models. The alternative of formulating judgment models in terms of discrete or binary regression models has been mysteriously avoided. This avoidance may be due to the fact that the predictors considered in past studies have generally been interval variables. In those cases where there were categorical predictors, the approach usually has been to attempt various multidimensional scaling techniques to transform the variables into interval form.

ST.

5

ų

• ₩ *1 }

• 47 0 There are certain advantages to "continuous formulations," two in particular being the economy of degrees of freedom required by the predictor vectors, and the parsimony in describing models that have fewer predictors.

On the other hand, binary or categorical regression formulations, in which each possible value of each predictor represents a separate category, consumes one degree of freedom per category and results in one predictor vector and one coefficient for each category. Such models do not represent hyperplanes, but rather, a series of unique points in the stimulus space. The advantage of binary models is that they inherently avoid the assumption of continuity and each of the coefficients reflects the exact effect for a single point in the stimulus space. Thus, there is no "averaging" of the effects represented by a particular coefficient.

In the past research, the use of ANOVA models by the researchers at the Oregon Research Institute and by Valenzi (1970) are the only serious ventures into the use of categorical models.¹

Appendix A discusses equivalency of ANOVA and categorical/binary regression models.

Clearly, a compromise between the de facto assumption of continuity and the alternative of considering each point in the predictor space is desirable to reach an optimum tradeoff between parsimony and exactness.

A viable approach to such a compromise appears to lie in the definition of a mixed model in which both the parsimony of the continuous model and the exactness of the discrete model are utilized. In terms of hyperplanes and subspaces, this approach envisions defining a number of discrete subspaces, each of which can be modeled adequately with its own hyperplane.

In reference to judgment processes this corresponds to defining subprocesses over various patterns, or subsets, of the cue variables. Past researchers, Slovic (1969), Hoffman (1968), and Bettman (1969), actually provide the genesis for this type of conceptualization with their suggestion that judges may actually think in a sequential fashion. Their contention that man first surveys the predictor values, assesses their consistency or believability, and then selects and applies the appropriate decision logic is conceptually analogous to the initial definition of a particular subspace, followed by the application of a model within that subspace.

In terms of the "structural image" of a configural effects judgment process as shown in Figure 3-2, page 53, the subspaces could be represented by the branches of the tree and the hyperplanes are defined by those variables considered in any branch.

At the left of the tree, one subspace is defined as consisting of all possible categories of the individual predictors and all possible combinations thereof. After the first branching, two subspaces are defined on the basis of the predictor used in that stage. In terms of

Hoffman's (1968) work, that predictor may be some index of "believability"; in terms of Bettman's (1969) work, it could be an index of cue "consistency". Each sequential branching results in a further de-aggregation of the predictor space, and in the limit, a completely categorical model would be defined in which each subspace consisted of a single pattern of predictor values.

If one pursues this analogy between a sequential decision process and the division of the entire predictor space into subspaces, the major problem in modeling the process becomes one of defining the appropriate subspaces. This problem is combinatorial in nature. Considering a decision process with five trichotomous predictors, there are $3^{5}(243)$ individual predictor combinations, or single-point subspaces.² Furthermore, there are 2^{243} -1 (a very large number!) possible combinations of these points into unique patterns of subspaces. Obviously, a trial and error search would be prohibitive and an efficient method of defining the subspaces is required.

The Automatic Interaction Detection Algorithm:

ł.,

đ.

6

.4

••,

n

đ

. . .

ų,

The Automatic Interaction Detection (AID) algorithm was originally developed by J. A. Sonquist and J. N. Morgan of the Institute for Social Research, Ann Arbor, Michigan. In a series of monographs and presentations, they proposed, explained, tested, and expanded the computer code for the algorithm. (Sonquist and Morgan,

² This would be the case if enough data points existed such that each possible combination of the predictors existed in the data set. Otherwise, the maximum possible number of combinations is the upper limit on the number of single point subspaces.

1964)(Sonquist, 1967) (Sonquist, 1970)(Sonquist, Baker, and Morgan, 1971)

The algorithm accomplishes the sequential division of the data set into subsets based on that split which causes the greatest reduction in the unexplained variability. The procedure involves the iterative application of one-way analysis of variance over every possible split in the predictor set. The major logical steps of the algorithm are depicted in Figure 4-1; a detailed mathematical description is given in Appendix C.

The output from this computer program consists of a series of splits and a set of terminal subgroups which best explain the variability in the criterion variable. A model of the form:

$$Y = \sum_{i} b_{i} x_{i} + \epsilon$$

where:

ь

- the mean response of those data units falling
 in subgroup "i".
- a binary group membership flag for those data
 units falling in subgroup "i".

can be written for these subgroups.³ A major benefit of the program is the graphical display of the splitting process in the form of an AID-Tree as shown in Figure 4-2.

This graphic display of the splits was one of the primary motivations for developing the program since it allows visualization

^D The relationship between the AID model and regression models are discussed in Appendix A.



۲۳,

ŝ.

. .

Į,

. .

5

1

9-94 - 115



. ₩	<pre>************************************</pre>	@ /*	EAN 275.00 RSC = .9469 .0.= 30.00 PRCBE .042 .2												
	6484444444446 6600P 74F1MAL 7 Nm 2 PXEUICTOR 2	. COUES 1 	GROUP 64FINAL	COUES 0											
	6 0 0 0 9 6 7 7 8 8	8410 4410 4410		****	****	.84] .	•••			960		•••			
*		346.00 RSD = 73.99 PH08=				165,00 FSA # 40,00 PKOB								646444 1,000 1,000 1,000 1,00 1,00 1,00 1,0	AID-TREE
	* * * * *	8 8 4 0 8 9 8 9 8 9 8 8 9 8 8 8 8 8 8 8 8 8 8 8	****			10 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -				# • 1 • • • • •				н Не ин 5. о н Х2	F AN
		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	, ,		******	FINAL H 3				*******	N X	****		€ € € € € € € € € € € € € € € € € € €	LE O
	** 1 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	• 680UP • 88 • 87 • 87 • 87 • 8 • 8 • 8 • 8 • 8 • 8 • 8 • 9 • 9 • 9 • 9 • 9 • 9 • 9 • 9		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		•680UP +	• COUES G			*6800P	+ NE + PREUICTO				EXAMP
				9 9 9 7	¥ \$ 8										4-2,
			마마마 아마마 아마마 아마마 아마마 아마마 아마마 아마마 아마마 아마마	281,67 R50 = 104.83 PHOB	************				2.50 193•71 193•72					# # # # # # # # # # # # # # # # # # #	FIGURE
				2 2 2 2 3 3 4 3 4 3 4 3 4 3 4 3 4 3 4 3	******	* • •	₽ I 6	****	SeDen S'Den		• • • •				
5				568009 2 N= 5 PREDICTOR 1 CCOES 0		1 2 3 4 1		****	GRCUP 1 N= 32 LEVEL	*********			CODES 1 CODES 1	# # # # # # # # # # # # # # # # # # #	

Ś

\$

65_.

,

of the inherent structure in the data. A second motivation was the elimination of the requirement for assuming that linearity and additivity were properties possessed by the proper model. In essence, AID is a heuristic approach to searching the raw data for structure and is purely a mechanization of the procedure a researcher might go through manually in hypothesizing a model.

Songuist and Morgan (1964) hypothesized that the structure of the proper model would be indicated by the variables used in the splitting process and by the shape of the AID-Tree. They further hypothesized that symmetrical trees would be indicative of additive terms in the model which could be properly modeled by first-order predictors. More importantly, they felt that asymmetry, especially relative to the predictors involved in the splitting process, would be indicative of configural, or interactive, relationships among the variables. In this regard, they categorized tree structures as being either of the "trunktwig" or of the "trunk-branch" type. The trunk-twig structure could be either of the top termination or of the bottom termination types. In the top termination type, small homogeneous subgroups (twigs) would be split off from the tree with each subgroup possessing some particular advantage. In the bottom terminating tree, the twig subgroups would contain data units which possessed alternative disadvantages. In the trunk-branch structure, the tree would be reasonably symmetric and would be reflective of cumulative edvar ages and disadvantages.

Sonquist (1970) presented the results of his continuing evaluation of the AID computer program as his doctoral dissertation. In this evaluation he was concerned with the analysis of synthetic data from three contrived models and the performance of the AID algorithm, vis-a-vis, the binary regression approach as embodied in the Multiple

5

٧.,

Classification Analysis (MCA) program.⁴ The data set consisted of seven variables divided into 49 categories, on each of 2036 data cases. The models used to generate the data were:

- 1) An additive model (trunk-branch)
- A configural model in which advantages substitute for one-another but disadvantages are cumulative (upper terminating trunk-twig)
- A configural model in which disadvantages substitute for one-another but advantages are cumulative (lower terminating trunk-twig)

Runs without noise, low noise ( $\approx$  10 percent), and high noise ( $\approx$  20 percent) were made and analyzed to see if they would result in the same models as were input.

The major conclusions of the investigation as presented by Sonquist (1970) were:

- An additive model would produce a symmetric tree (Sonquist recognized that additivity was a sufficient condition for a symmetric tree but did not address the problem of necessity).
- Small but universal effects may not be discovered in the splitting process.
- Irrelevant variables which are correlated with powerful predictors may be used spuriously in the splitting process.
- The presence of noise (at the levels considered) did not materially affect the resulting AID-Trees.

5

.....

*;*;

÷

⁴MCA is described in Appendix A.

- 5) One evidence of interaction is a fairly large increase in the predictive power of one of the variables after the effects of some of the other predictors had been removed by the splitting process.
- 6) When a nonsymmetric structure appears, and the groups isolated first have approximately the same, low or high, means: a clear departure from additivity has occur.ed.

1.4

5

4

- The AID-Tree was effective in depicting departures from additivity for the models considered.
- 8) The order in which the variables are used in the splitting process is not necessarily reflective of their overall importance in describing the process, rather the total variation explained by a variable may be more indicative of importance.
- 9) The B statistics calculated and output by the tested version of the AID program were virtually meaningless in depicting the relative importance of the individual variables.

Two important aspects of this investigation deserve special mention. The first is that the underlying models and the quality of the data set were known on an <u>a priori</u> basis. The second is that a data set of approximately 40 points per predictor category was available and used in the analysis.

It is entirely different problem to see a known model in the AID-Tree than it is to infer an unknown model from the AID-Tree. Further, the stability of the category means offered by a high data/ predictor ratio helps significantly in discriminating between actual

69

effects that are due to the predictors and the spurious effects that are due to the idiosyncracies of the data sample.

Sonquist recognized these deficiencies in the generality of his results and suggested the application of the following heuristic procedures when analyzing AID-Trees.

- The detailed profiles of each variable used in a split (and its competitors for the split) should be examined within the major subgroups throughout the tree.
- 2) All variables that are competitors in the splitting process should be examined closely for dependence. If two variables are measures of the same aspect of the situation, the variable that is least supportable purely on logical or theoretical grounds should be discarded.
- 3) The residual profiles of the final groups should be examined to see if effects from less powerful predictors may still be present and unaccounted for by the splitting process.

In regard to the size of the data set required for AID analysis, Sonquist (1970) stated:

Data sets with a thousand or more cases are necessary, otherwise the power of the search process must be restricted unastically or those processes will carry one into a never-never land of idiosyncratic results. (p. 1)

In addition to generally confirming his original hypotheses relative to the utility of AID in detecting interaction effects, Sonquist's (1970) investigation did provide some insight into the problems to be

•

encountered in running AID. For these problems he offered the following general guidelines:

ς.

. . .

ņî.

8.

- Do not categorize the variables with excessive detail;
   five to seven categories per predictor should be the maximum segmentation of the predictors.
- Categorize the variables so as to minimize the problems created by skewness in the predictors: if a predictor is skewed, put those values in the tail of the distribution in separate categories.
- Beware of skewness in the criterion; discard outlier cases or perform appropriate transformation of scale on them.
- Perform appropriate initial runs with the split termination limits disabled in order to detect problems such as measurement or coding errors.

Finally, Sonquist (1967, 1970) proposed a procedure for using AID to hypothesize the terms of the model and then using MCA to determine the proper weights of the individual predictors. The additional analysis with MCA was espoused since the AID program does not produce statistics that accurately reflect the relative weights of the individual predictors.

Problems in the Application of AID for Hypothesizing Policy Models:

The application of this "structure searching" technique to the hypothesis of policy models would be straightforward except for two reservations--the amount of data required to obtain stable means for the subgroups, and the lack of rules for inferring an "unknown" model from the data set.

Given the objective of "data conservatism", as presented in Chapter III, and the difficulties attendant in obtaining large quantities of decision data in operational environments, the freedom of the AID search would probably have to be somewhat restricted. Initial efforts by this investigator in analyzing the loan application data confirmed this fact. Splits resulting from groups of less than 10 cases were generally found to be meaningless and uninterpretable. In runs consisting of between 224 to 400 data units, with approximately 50 predictor categories, AID displayed a tendency to make spurious splits that did not hold up under cross-validation. When the splitting processes were restricted to avoid this overfitting, the explained variation in the data set was disappointingly low (< 50 percent).

The hypothesis of configural judgment models based solely on the splitting process was found to be less than successful, as measured by the lack of improvement in the predictive capability of the resulting configural models. The transformation from various AID-Tree structures into appropriate terms was not apparent. "Continuous" interaction terms of the form,  $b_{ij} x_i x_j$ , were found to be of little value in improving the fit of policy models. It was demonstrated that the number of categories and the definition of these categories could materially affect the appearance of the AID-Tree. Experimentation with various configural models revealed that perfectly symmetrical AID-Trees could be generated with data from highly configural models. Hence, additivity was found to be a sufficient condition for symmetry, but not a necessary condition.

()

1

÷

۰.,

¢

Although the AID-Tree itself proved less interpretable than anticipated in inferring configural models for the decision process of the installment loan officers, it did retain the advantage of displaying the variables and their relationships in a manner that was readily comprehensible and believable to the judges involved. In all cases it identified those predictors that were most important to the judges. The predictors used in the early splits of the tree appeared to be pivotal elements in the judges' own descriptions of their thought processes.

In order to better define the proper configural relationships, it was concluded that the predictive efficiency and the correlation of the criterion with each of the predictors, within each subgroup of the AID-Tree, would be required in addition to the AID-Tree itself.

This same conclusion was independently reached by B. M. Finifter, of Michigan State University, who published the description of a proposed addition to the AID program in September, 1971. This addition consists of a routine to collect and display profiles of the Between Sums of Squares/Total Sum of Squares (BSS/TSS) from the AID splitting process. A normalized parameter reflective of the predictive ability of each predictor over the entire AID-Tree is calculated and designated as the Potential Explanatory Power (PEP) coefficient (Finifter, 1971, 1972).

The Development of the AID4UT/AIDTRE Computer Program:

The modification and extension proposed by Finifter was incorporated into an improved version of the AID4 computer program that was acquired from the Air Force Human Resources Laboratory.

The new program was implemented on the U.T. CDC-6600 computer system. It has the capacity to consider unlimited quantities of data with up to 80 predictors, divided into up to 700 predictor categories. The program can build, cross-validate, and double cross-validate AID-Trees, and it can produce profiles of the BSS/TSS for each predictor. A complete description of the modified program with Users' instructions is provided in Appendix C.

Application of AID4UT/AIDTRE in Defining Subspaces:

٩.

.71

Although the addition of the Finister modification does provide more useful statistics for discerning unknown structure in the data, the problem of overfitting can only be avoided by limiting the number of splits that are allowed. In general, it was found that the limiting subgroups to contain no less than 5 percent of the cases would provide splits that were relatively stable. This early termination of the splitting process leaves considerable residual variability within the final subgroups. Review of these final subgroups revealed that more of the residual variability could be explained by various predictors, but the importance of any particular predictor was generally not as dominant as was the case in the earlier splits. This condition suggested that the residual variability in the subgroups might better be explained by a combination of the predictors instead of a binary split on a single predictor. In repeated applications on the 224 data units for the decisions of the installment loan officers, the criterion of retaining at least 5 percent of the total in each of the final subgroups resulted in termination of the splitting process within 10 to 15 splits.⁵ It was discovered that the most important predictors, in terms of their configural relationships with the other variables, tended to be involved in the early splits. Assessment of the subgroups resulting from these early splits indicated that they were often homogeneous. Within the individual subgroups, different combinations of the predictors appeared to explain the variation.

By relating this phenomena to the judges' own verbalizations of their policies, and reflecting upon the findings of Hoffman (1968), Bettman (1969), and Sonquist (1970), the concept of using AID-Trees solely for the purpose of defining a series of subspaces, and then modeling each subspace with a local model evolved. This technique was perceived as a means of taking advantage of the configural effects of the predominant variables without falling into the trap of overfitting the idiosyncracies of small subsets of data.

The Criteria for Defining Subspaces:

.7

12

The decision of how many, and which subspaces to divide the predictor space into is based on heuristic rules developed during this approach. Experience with the loan officer data indicates the best criteria appear to be homogeneity of the subgroup and the predictive capability (both absolute and relative) of the predictors within the subgroups.

74

Allowing the tree to split into smaller groups does not affect the first few splits since the tree can be manually truncated; this was the case with the loan application data.

The logic for defining appropriate subspaces is best explained by an example. The example that will be used is the definition of subspaces from the AID-Tree for one of the five installment loan officers whose policies were captured as a part of this research. An annotated AID-Tree for this policy model is shown in Figure 4-3; profiles for the BSS/TSS values for all 16 variables are shown in Tables 4-1 and 4-2. The data shown in Table 4-1 reflects the maximum between sum of squares for each predictor, in a given subgroup, normalized over the Total Sum of Squares for the entire data set. It will be designated as BSS/TSST. Table 4-2 reflects the profile in which the maximum between sum of squares for each predictor has been normalized over the Total Sum of Squares within the subgroup under consideration at a given step or trial. It will be designated as BSS/TSS(i). These two profiles reflect the "overall" and "local" predictive capability of each predictor. Comparison of these profiles are very helpful in revealing configural and nonlinear effects between the individual predictors even when one of the predictors involved is not actually used in the splitting process.

In the step-by-step consideration of the AID-Tree that follows, the superscripts refer to the corresponding entries on Tables 4-1 and 4-2. More detailed output for this analysis is shown in Exhibit 8 of Appendix B.

Step 1. Credit Rating has the largest value for BSS/TSST. Since there is only one group at this point, entries of  $.38^{(a, al)}$  appear in both Tables 4-1 and 4-2. (The next largest value is for Financial Reference which has  $.17^{b}$ ) Group 1 splits into groups 2 and 3. Group 2 is a homogeneous "twig" with a mean criterion value of .01. The input parameters used for this particular run of AID4UT⁶ allow no

⁶ The setting of the parameters for AID4UT is discussed in Appendix C.





1

٩

`'

in in in ΰ

715 716 716

¥

* 精麗 

76

į

()

	-	ZWARITL	JLCCAL	4 PHONE	5 JABTIM	58EST1M	
0.1122     0.0000     0.0000     0.0000     0.0000       0.1122     0.00000     0.0000     0.0000     0.0000       0.1122     0.00000     0.0000     0.00000     0.0000       0.1125     0.00000     0.00000     0.00000     0.00000       0.1125     0.00000     0.00000     0.00000     0.00000       0.1125     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.1100     0.11100     0.11100     0.00000       0.11100     0.11100     0.11100     0.00000     0.00000       0.00000     0.00000     0.00000     0.00000     0.00000       0.00000     0.11100     0.11100     0.000000       0.000000     0.0		N+1500 •		•055844	192/00 ·	12 12 14 ·	
01318     01075     01075     01075     01149       07909     020000     001908     001906     00190     001906       07909     020000     001905     001905     001906     001906       01318     001902     01305     001905     001906     001906       01318     001902     01305     011905     001905     001906     001906       019050     001902     011905     011905     011905     001905     001906       019051     019051     011905     011905     011905     011906     010905       0190510     011915     011915     011915     011916     011916     011905       0190510     011915     011915     011916     011916     011916     011916       0111108     011915     019191     019289     01919     019289     01919       0111108     011915     019289     011916     011916     011916     011916       0111108     011915     019289     011916     019289     01019       0111108     011916     011916     019289     019289     01939       0111108     011916     019289     019289     019289     019289       0111108     011915<		22297			047718	~026762	
-011307     001518     -001518     -001518     -001518     -001518       -01316     -01326     -01315     -001518     -0015       -01316     -01326     -01315     -01555     -0156     -015       -01316     -01326     -01317     -01555     -01575     -015       -01316     -01326     -01326     -01317     -02556     -013       -01316     -01326     -01326     -0157     -02566     -013       -01316     -01326     -01326     -01347     -01527     -012       -01316     -01327     -01327     -01347     -012546     -013       -01316     -01327     -01327     -01347     -01527     -013       -01316     -011108     -11615     -12549     -01357     -01357       -011108     -11615     -25545     -117364     -013       -011108     -11615     -25545     -117364     -013       -011108     -11615     -01367     -01367     -01357       -011108     -11615     -003217     -117364     -013       -011108     -11615     -01367     -110365     -01357       -011108     -11615     -01367     -010367     -01367       -01145		Aditio.	5/2600 ·	.015649	.010720	.019124	
.07301     .030000     .07544     .00146     .001       .0730     .03340     .0355     .0355     .00146     .070       .07909     .03340     .03340     .0355     .0355     .00156     .013       .07909     .03340     .03340     .0356     .0015     .01346     .013       .07909     .03340     .01376     .0136     .01356     .01356       .07909     .03340     .01376     .117150     .121044       .00141     .001791     .002594     .00299     .0019       .00141     .011106     .11615     .35545     .013     .013       .011106     .11615     .35545     .117456     .013       .011108     .11615     .35545     .013567     .013       .011108     .11615     .35545     .013567     .013       .011108     .11615     .35545     .119567     .013       .011108     .11616     .11615     .25547     .013567     .013       .011108     .11616     .11615     .256547     .013567     .0135       .011108     .11616     .11615     .256547     .0109567     .0135       .011108     .11616     .11615     .256547     .0109567     .0135		m 959116.	.009838 7	• 008500	.001518	-011632	ىد
.07909     .02390     .2491     .15746     .0815       .01316     .00395     .01554     .0285     .0285       .01316     .00395     .01326     .0285     .0285       .01316     .01316     .01316     .0135     .0285       .01316     .01316     .01316     .0135     .0285       .01316     .01316     .1141     .0006     .0117       .01312     .01316     .1141     .00091     .00091       .01316     .1141     .12456     .11704     .026       .01316     .1141     .00001     .00091     .00091       .00011     .01316     .01316     .00001     .00001       .01110     .1141     .3545     .4109     .013       .01110     .1141     .3545     .4109     .013       .01110     .1141     .3545     .11704     .003       .1110     .1141     .3545     .1093     .001       .1110     .1141     .3545     .1093     .001       .1110     .1141     .3545     .1093     .001       .1110     .1141     .3545     .1093     .001       .1110     .1141     .11610     .11610     .11610       .1110 <t< td=""><td></td><td>10500+</td><td>0.000000</td><td>+ 05454# +</td><td>.501408</td><td>*000<b>-</b>05</td><td></td></t<>		10500+	0.000000	+ 05454# +	.501408	*000 <b>-</b> 05	
01316     00336     04156     02545     00345       00751     00744     02545     02545     000       00752     00740     00740     00745     00755       00753     00745     00745     00755     00755       00754     00745     00745     00745     00755       007575     00745     00775     007575     007575       007575     007757     007775     007575     007575       007575     01707     007775     007575     007575       007575     01707     007775     007575     007575       007575     01707     072745     007575     007575       007575     01707     07274     007575     007575       007575     01955     072745     01757     07527       00757     01955     07527     01757     07527       01110     11161     1122737     07527     07527       01110     11161     07557     07527     07527       01110     11161     07527     07527     07527       01110     1112231     07527     07527     07527       01110     1112231     07527     07527     07539       010155		.07909	062390	16945.	15746	.09213	
01116     00155     00155     00155     00155       01116     00155     00155     00155     00155       01110     01110     01151     01151     01155     00155       01110     01151     01151     01151     01151     01151       01110     01151     01151     01151     01151     01151       01110     01151     01151     01151     01151     01151       011110     01151     01151     01151     01151     01151       011110     01151     01151     01151     015250     010151       011110     01151     01151     01151     015250     010151       011110     01151     01151     015255     01161     01161       01151     01251     01161     01251     01161     01161       01151     01251     01161     01251     01161     01161       011051     01251     01251     01251     01161     01551       011051     01251     01251     01251     01161     01551       01051     01251     01251     01251     01161     01551       01051     01251     01251     01252     01551     01551		- <b>1</b> 2-1		-0 i	<b>\$</b>	40 G	
-00721     -00720     -00720     -00750     -00750     -00750     -00750       -00002     -00005     -00005     -00065     -00065     -00065       -00002     -00005     -00065     -00065     -00065       -00021     -0001791     -0005730     -0005       -00021     -0001791     -0005730     -0005       -00021     -011907     -0005730     -0005       -00021     -011907     -017916     -012270       -00021     -011917     -0001791     -0025730     -0005       -00021     -011917     -0001791     -0025730     -0005       -00021     -011916     -11610     -15277     -0019       -01108     -11610     -11610     -11610     -117105       -01101     -11610     -11610     -117105     -00573       -01021     -0193     -00127     -0193     -0019       -01021     -01010     -11610     -112737     -0019       -010221     -11610     -12628     -00257     -003577       -010221     -0103757     -01095     -003577     -003577       -010222     -010224     -0127337     -010964     -0013       -010225     -010147     -0127337     -		91510 ·	96000	44 I 40 •			
00000         00005         00005         00005         00005         00005         00005         00005         00005         00005         00005         00005         00005         00005         00005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005         0005 <th005< th=""> <th0005< th=""> <th0005< th=""></th0005<></th0005<></th005<>		100.100				202004	
Treeth         Treeth <thtreeth< th=""> <thtreeth< t<="" td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td></thtreeth<></thtreeth<>							
#TFEHP       91MCDME       1093289       115456       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111555       111655       111655       111655       111655       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       111553       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       1112233       11112233       1112233       1112233 <td></td> <td>-0406 -0406</td> <td></td> <td>1004N</td> <td>-157-5</td> <td>• 0.9213</td> <td></td>		-0406 -0406		1004N	-157-5	• 0.9213	
063629     003629     17450     0       003519     0226445     11619     005730     005730       003519     0257445     11619     005730     00130       013051     013051     000000     005730     00130       013051     013051     000000     005730     00130       013051     013051     013051     000500     013051       013051     013051     000000     005730     00130       01110     1110     01935     000327     01327       01110     01935     00327     00327     01367       01110     01935     00327     00327     01367       01110     01935     00327     01367     01367       01110     01935     00327     01367     01367       01110     01257     00327     01357     01367       01110     01257     01357     01357     01357       01110     01257     01257     01367     01357       01112     01112     02524     01035     0144       01757     011233     0     05574     01357       01757     01753     01607     01753     01645       01753     01551     055167 <td></td> <td><b>A</b>TYPEHP</td> <td>91×COME</td> <td>I OBÅZX &amp; C</td> <td>1 LF TNREF</td> <td>1 ZL DANAM</td> <td></td>		<b>A</b> TYPEHP	91×COME	I OBÅZX & C	1 LF TNREF	1 ZL DANAM	
-015297       -025367       -11686       -11533       -000100       -015260       -0005         -014477       -005345       -11616       -11616       -015260       -005         -01457       -005345       -01193       -00516       -005         -01457       -0193       -00713       -01527       -005         -01466       -0193       -0713       -00557       -013         -014721       -0193       -0737       -00557       -013         -0161       -0193       -0737       -00557       -013         -01721       -0193       -0737       -00557       -013         -01721       -0193       -00327       -0196       -013         -01721       -01915       -00327       -0196       -019         -011757       -00327       -01237       -0196       -0196         -011757       -00234       -01225       -01237       -0196         -011757       -00234       -01225       -01237       -0196         -011757       -0024       -01225       -01237       -01237         -011757       -00252       -01262       -01647       -02390         -011757       -00252		043420	SEACED.	124546	171450	022680	
000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       0000000       000000       000000		040401	027645		100541		
000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000       000000		526275	025921	9 662601.	.052360	•006250	
.005548       .00179       .005995       .019         .1108       .11616       .35545       .41996       .019         .11108       .11616       .35545       .41996       .019         .01457       .01935       .05727       .00595       .019         .01457       .01946       .05715       .01946       .019         .01457       .01946       .05715       .01946       .019         .01457       .00994       .05715       .01946       .019         .01457       .00914       .05715       .01946       .019         .01527       .00012       .01277       .01946       .0194       .019         .11108       .11616       .12638       .12233       .01946       .01946       .01946         .11108       .11616       .12233       .12233       .01752       .01946       .01946         .017527       .000242       .012223       .012224       .01946       .02345         .001946       .000242       .027652       .010262       .02762       .027656         .001946       .000242       .027652       .010262       .027652       .027656       .027666         .001946       .000262		.058202	L POE 10.	116000.	.005730	563500 ·	
.001510       .011615       .000000       .012426       .000         .1108       .11615       .05545       .41096       .063         .01935       .01935       .01935       .01936       .013         .01935       .01935       .05169       .0524       .001         .01935       .00010       .05169       .0524       .003         .01935       .00010       .0524       .0516       .0510         .01108       .01935       .05367       .019       .053         .01109       .11616       .42654       .003       .019         .01109       .11616       .42654       .003       .063         .017527       .000221       .00327       .00327       .01996         .017527       .000222       .000221       .01297       .0037         .017527       .000222       .000222       .00377       .00377         .017527       .000223       .00017       .00024       .01040         .017527       .00027       .00027       .00024       .00024         .017527       .000175       .000753       .000753       .000753         .010150       .000753       .000753       .000753       <		-00++77	• 0050 x8	100.	.005995	• 013854 ·	•-•
.11108       .11615       .35545       .41095       .063         .01457       .00934       .07109       .07524       .0010         .18721       .00910       .07109       .07524       .0010         .18721       .00910       .07109       .0554       .010         .18721       .00910       .07109       .05557       .0010         .1103       .11616       .42554       .01096       .00317         .11103       .11619       .42554       .01096       .00317         .11103       .11619       .42554       .01096       .00317         .11103       .11619       .42554       .01096       .00317         .11103       .11233       .11233       .01175       .00324         .001757       .00024       .11233       .005417       .01539         .001756       .00024       .000014       .12847       .01549         .001750       .00751       .00752       .005417       .05399         .001750       .00753       .00752       .005417       .05399         .001750       .00753       .00493       .05490       .05399         .001359       .001042       .010422       .05490		.03510	EES113.	000000-2	.092428	*00+927	
0.1851     0.1936     0.719     0.7527     0.0544     0.016       0.1872     0.0994     0.0715     0.7527     0.0764     0.076       0.1872     0.0994     0.0715     0.07527     0.0764     0.076       0.1872     0.0010     0.0237     0.07657     0.076       0.1103     0.11616     0.0237     0.07657     0.063       0.1103     0.11616     0.0237     0.01367     0.063       0.1103     0.11617     0.0264     0.01373     0.063       0.17527     0.01477     1.1283     0.01753     0.03347       0.017527     0.01753     0.01753     0.01753     0.05343       0.017527     0.01753     0.01753     0.05433     0.05433       0.017527     0.01753     0.01753     0.05433     0.05643       0.01753     0.02042     0.01753     0.05543     0.05243       0.01754     0.02042     0.01753     0.05543     0.05243       0.01755     0.02062     0.01753     0.05643     0.0534       0.01755     0.02062     0.05664     0.05214       0.01755     0.05757     0.0752     0.05754       0.01035     0.05664     0.06752     0.05754       0.02014     0.0753		.11108	.11610	, 35545	\$6014.	54690.	
•01851     •01956     •07198     •07198     •07198     •07198       •11108     •11614     •00315     •07527     •006567     •070       •11108     •11614     •03127     •0327     •07567     •070       •0110     •11614     •0327     •006567     •070       •0111     •0011     •00327     •0034     •0063       •01108     •01647     •11616     •25654     •063       •01757     •01647     •1233     •1096     •063       •01757     •00024     •112333     •10467     •05909       •01757     •00024     •01753     •08179     •05309       •01158     •000014     •07533     •112233     •10471     •05309       •010169     •07552     •07523     •08173     •07509       •000014     •07533     •07533     •10422     •10561       •055430     •0753     •07533     •07513     •07239       •010014     •07553     •07523     •07523     •07514     •1109       •020159     •0753     •07523     •07523     •07514     •07290       •010014     •07553     •07523     •07523     •07514     •07290       •020253     •07533     •07523 <td< td=""><td></td><td>9</td><td>J</td><td>ŝ</td><td>•</td><td>Ð</td><td></td></td<>		9	J	ŝ	•	Ð	
•1+57       •00994       •0515       •076       •076         •1103       •11616       •2554       •076       •076         •1103       •11616       •2554       •076       •070         •1103       •11616       •2554       •076       •070         •1103       •11616       •2554       •076       •070         •1103       •11616       •2554       •076       •073         •11103       •11253       •12547       •1096       •053         •01757       •00022       •112233       •10265       •10766         •01757       •00022       •112233       •10264       •10766         •014567       •00022       •112233       •10264       •10766         •014557       •000022       •01652       04717       •0799         •014567       •12875       •1046       •1076       •2390         •01558       •000017       •000017       •00713       •02390         •0100014       •0753       •0762       •02390       •02390         •010150       •0753       •0775       •0775       •02390         •010150       •0753       •0775       •0775       •02390		-01851	-01936	• 0 1 1 0 •	. 36849	1910.	
.762       .56160       .60387       1.10946      00         .1108       .11616       .2555       .10966      00         .1108       .11616       .2555       .1096       .000         .1108       .11616       .2555       .00227       .000         .1108       .11616       .2555       .00327       .000         .0001757       .00024       .1122347       .00346       .00346         .001757       .000242       .01122347       .01996       .00346         .001409       .000242       .01122347       .01996       .00396         .001409       .000017       .01122347       .01996       .02390         .001409       .000017       .011223       .01946       .01790         .001409       .00753       .000017       .010722       .00717       .02390         .001410       .00753       .00753       .00717       .02390       .116108         .0014135       .00753       .000017       .00717       .07390       .02390         .000140       .00753       .007022       .00717       .07390       .02390         .000140       .00753       .00753       .00717       .07391		+01+57	+ 6 6 0 0 ·	51150 ·	.07527	• 00635	
		1872;	-51610	60000°	1.19896	1+004.	
Inition     SUMMART TABLE       1017527     1560117     16668AT     SUMMART TABLE       017527     000544     112733     GREGONSTBUCTE       017527     000244     112733     GREGONSTBUCTE       017527     0000244     112733     GREGONSTBUCTE       017527     0000014     0000011     100001       017527     0000014     017733     GREGONSTBUCTE       0117527     0000011     000001     017533       011469     000001     000001     000001       011469     000014     000001     1048577557       0101916     00753     049333     5     00001       0201916     0753     046782     7     3     10641       0201916     0753     046782     7     3     10641       020191     0753     046782     7     3     10641       01025     10462     7     3     10641     12476       01025     110422     7     7     7     10795       01025     10462     7     7     7     10795       01025     110422     7     7     7     10795       01025     10462     7     7     10425       01025		12000.		1990.	19690		
ANECESS         ISEGUITY         ISCREAT         SUXMARY TABLE           -017627         -001647         -382847         G           -017527         -001244         -112233         G           -017527         -001244         -112233         G           -017527         -000244         -112233         G           -017527         -00022         -01735         G           -017557         -000022         -01735         G           -01755         -000017         Z         ABTTL         -07909           -000150         -000017         Z         ABTTL         -07909           -0001515         -000017         Z         ABTTL         -07909           -000170         -01753         -05433         -00742         -10742           -05621         -05017         -07612         -07933         -07914           -05012         -01042         -10452         7         RESAT         -26434           -010014         -05027         -10453         7         RESAT         -2654           -01035         -01045         -11045         7         RESAT         -20548           -01035         -01045         -12874		601716	> 1 B T T •			3.000	
NECES         ISCGUTY         ISCREAT         RECONSTRUCTE           001757         001647         382847         0           001757         001753         0         382847         0           001757         001753         0         90549         0         90557557           001757         001753         0         901775         0         90557557           001757         001753         001757         001757         0         90557557           001750         001753         001757         0         905437         0         905437           001016         00753         006682         9         9         90754         9         9           000014         00753         006782         9         9         9         9         9         9           010018         00755         10422         7         7         7         7         26491           01025         11642         7         7         7         7         1076           01023         00135         11642         7         7         7         7         26554           01025         11642         7         7         7					SUMMARY TABLE		
•017527       •001647       •382847       •0017527         •017527       •000244       •112233       •0543         •017527       •000244       •112233       •0543         •011527       •000244       •112233       •0543         •011549       •000241       •112233       •0543         •001916       •05643       •05643       •0543         •001916       •05643       •05643       •05433         •00014       •05643       •05643       •05433         •00015       •0753       •04933       5       •06414         •05643       •0753       •04933       5       •0744       •2493         •05543       •0753       •04933       5       •0744       •2493         •05543       •0753       •04933       5       •0744       •2433         •0554       •0355       •10422       7       7       *2433         •01025       •10422       7       7       7       *2433         •12674       •01042       7       7       *2433       *2433         •12635       •10422       7       7       7       *2433         •12636       •10422       7<	_	I ANECESS	JSEQUITY	16CRERAT			
001527       000244       -112233       0       RECONSTRUCTG         001459       000042       001125       0       1       RECONSTRUCTG         001459       000042       001125       1       0       0       0         001459       000014       000017       2       NAFT       07909       007909         001450       000022       000007       2       NAFT       07909       007909         00014       005433       000007       2       NAFT       07909       007909         00014       0753       000007       000007       2       NAFT       02390         00012       0753       000007       000007       2       NAFT       02390         010014       0753       0000753       000007       2       NAFT       02390         010015       0000753       0000753       000000       01246       11108       01074       05313         010018       000000       12874       0       12874       9       10044       05342         010018       000000       00000       010055       10455       10104       05342         0000018       000000       00000		.c370+8	- C01647	.382847 C			
001469     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     001465     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     011076     0110765     001755     001562     010562 </td <td></td> <td>• 017527</td> <td>• 0005 <b>1</b> •</td> <td>+112233 d</td> <td></td> <td>HECONSTRUCTED</td> <td></td>		• 017527	• 0005 <b>1</b> •	+112233 d		HECONSTRUCTED	
0019169     0010591     001001     105430       001916     075635     066825     10641     015450       070001     07533     066825     10641     02390       07001     07533     066827     3     10641     02390       0701     0753     06682     3     10641     02390       0701     0753     06682     3     10641     12390       01035     01012     10422     7     3     10621       01035     01012     10422     7     70916     11008       01035     01012     10462     7     7094     11018       01023     01025     10422     7     7094     11018       01023     010453     01042     7     7094     11018       01018     00012     00015     10465     116454     11018       00018     00000     01553     12854     054933     12864     05342       00018     00000     64933     12     12844     01744     05342       00018     00000     64933     12     12844     01744     05342       00018     00000     64933     12     12844     01744     05342       0		- 0 - 1 - 20	220000	SE/ 160+			
000014     075433     060007     0774     07339       075433     07543     07662     0764     02391       075433     0753     040714     12391       075433     07662     07662     07662       076635     0753     040714     12391       07104     0753     04033     07662       07105     01042     07664     08213       07105     01042     07664     076213       07105     01042     10422     1108       07105     01042     11664     01108       07105     01042     116964     01108       07019     01000     010425     07654       07019     0763     06433     116864       07018     00753     06433     117866       07018     00753     06433     117866       07019     0773     07733     05493       1     18064     118966     05342       1     18064     118966     05342       07019     07733     07733     05433       1     18064     118966     05342       1     18064     10108     05342       1     18064     118966     118966       <		691100.	120000	4 523623 V			
00001     0754J0     0752     04174     1604       07753     04174     1675     16433     1604     16376       01035     01035     00125     10422     7     7     1674       01329     00125     10422     7     7     75541     1053       01329     00125     10422     7     7     75541     1053       01329     00125     12874     9     7     75541     1053       01329     12874     9     7     7     7     7       01329     12874     9     7     7     7     7       01329     12874     9     10045     1107     1107       12874     12874     9     7     7     7       00019     12875     12185     9     12117     05342       000153     00000     94933     12     1444     05542       12864     10     11177     05342     1144     01552       13     12     10444     05342     15     10444       13     14     16     14     10552       14     14     14     16     16       14     16     16     16		914100		- 00000 -		40440*	
.0521     .0753     .64933     5     .00470     .2431       .001     6     6     .00125     .00422     .2431       .01035     .0125     .0422     7     7201       .01329     .00125     .0422     7     7203       .01329     .00125     .10422     7     7203       .01329     .00125     .10422     7     7203       .01329     .00125     .10422     7     7203       .0019     .00125     .10422     7     70621       .0011     .00125     .11012     9     10004       .00013     .01653     1.1864     9     10044       .00014     .00000     .66933     12     14864       .00013     .00153     .66933     12     14864       .00014     .00000     .66933     12     14864       .00171     .00153     .66933     12     .00144       .00171     .00171     .00171     .00175		+100D0-		- 0+0H25		347234	
001035 00125 010422 7 RESCAT 008213 01035 00125 010422 7 RESCAT 008213 01329 00129 01646 1042 7 RESCAT 009218 01620 118954 9 7 Иссык 11108 01620 101654 9 1 Иссык 01616 000018 00000 01653 01654 10 60000 00018 000753 06433 11 Г188F 017562 13 Иссе55 00621 14 Иссе55 00621 15 Сонтт 01562		.46211	.01753	5 C C C C C C C C C C C C C C C C C C C	A JORTIN		
.01035     .00125     .10422     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7     7 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>							
-01129 -01199 -12874 8 17954 -1109 -01129 -01199 -128954 9 INCAE -11615 1.28386 1-65027 1-18954 9 INCAE -11615 -00018 -00000 -01658 10 EANKAC -2655 -00018 -00000 -66933 11 F19876 -41056 -20018 -00053 -66654 10 F19876 -41056 -20018 -00054 10 F19876 -41056 -20018 -00054 10 F19876 -41056 -20018 -00054 10 F19876 -41056 -20018 -00054 10 F19876 -41056 -20018 -41056 -400564 10 F19876 -41056 -400564 10 F19876 -41056 -20018 -41056 -400564 10 F19876 -41056 -400565 -400565 -40056 -400565 -40056 -400565 -400565 -400565 -40056 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -40056 -400565 -40056 -400565 -40056 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -400565 -40		201035	34,00,	- 02#01-	7 RESCAT		
1.28306 1.55527 1.68954 9 INCOME 11616 •00018 •00000 •01658 10 54NKAC •1616 •00018 •00000 •01658 10 54NKAC •1655 •1718FF •41096 •05211 •01552 •0533 •17 •01562 14 NECES •05211 •0753 •05211 •0553 •05211 •0553 •05211 •0553		94113-		47961	R TYPENP		ď
-00018 -00000 -01658 10 БАИКАС2654 -00211 -000753 -64933 11 F198FF -41096 13 L0анин -015542 14 NECESS -06211 15 E011TY -00753 15 E011TY -00753		1.28385	1.55027	1-1895	9 INCOME	-11616	
-36213 -00753 -64939 11 FUNAF -41996 72 LOANTM -05342 13 LOANTM -01562 14 NECESS -06211 15 Equity -00753 -06561 -4433		.0008	.00000	.01658	10 BANKAC		2.0
12 LOANAM - 05542 13 LOANAM - 01562 14 NCCESS - 06211 15 E01177 - 00759 15 E01177 - 00759		.06213	.00753	.6493	1 FINAFF	41096	
13 COENTM +01562 14 NECESS +06211 15 EQUITY +00753 15 EQUITY +00753					12 LOANAM	5+540.	
14 McCC53 • 00421 15 EQUITY • 00453 • • (56684) • • • • • • • • • • • • • • • • • • •						• 0 ] 5 6 2	
						11200	•
					THE CREEK	2 C D 4 4 4	, ~

TABLE 4-1, BSS/TSST PROFILE

**ل**يدا.

1

**۳۵**ور کار د ا

, ,

.

e

5

^۸.

**\$** 

Â.

ي وي وي ر

BH:

. 5

, pen

=

n di serie d Serie di serie Serie di serie

. . . . .

o`_____ ≧_**#**`s*

* 1

*§* 

77.

1

1

!

•

i

J

																																									5.0	10-01	6 ° 9	2.0	•••	11.0	16.0			
фнеуття . 01213е		01210 012170					.27523	а :						+ 27620		MANAU 191	005560	• 0 1 5 1 0 5	C2160.					\$06LZ*	*1	594E0*	. 3 2 5 k 3	.75751	01000.	- 5060 ·			I	ECONSTRUCTEU	SUMASSITSSIA			26411+	. 63730			- 24 H C					60402 *			.6450%
5JnB11H		201940*		24:040	. 91001%	-01110-	30465	10100	0	.05078	50460 ·		.06121	.36465		, 1 C 1 1 0 6 6							CC3470*	.67173	4	11196						UMMARY TASLF		ă	•.	1 AGE	2 MARITL	3 LGCAL	1707d *							12 FINGER		TUAUT EI	14 NECESS	
* DH0%	.058844	.1:8372	.161468	.06198. AJ	.056104	.210526 fl		58730	Ð	.11455	• 15628	. 69135	1100.	66730			LOEALAL	• 1 2 4 5 4 5	• 148367 • 1	• 256349 CI	21 L + 00 +	• 01 1964	0.000000	52597	•		A 1 A 1 A .				<b>B1c1</b>	5					F40000	10 19145.		1.22245	ھ	19505.	.12433	.62277	[ielo.	1.22.85				
BLCCAL	-0044/4	010476	1,0000	048903	064935 ml	00000000		•11852	Ð	- 1975					760 7 .		9INCOHE	6082i0*	0f1970*	.045835	.058550	40335a6	£64100°		5 455 5 4	<b>6</b> !	• 65557	.92074	.37242	5 4 D C D 4	5 7 5 Ë E -		1100361	.01161			5.000		3275400	20402	<i>•</i>	.01761	4.01.04				300404			
1TI DAMO								. 26745					120754	F L C C C C	5619Z.		BILDEND								-24845.	÷		.1207	10567	. 20016	.24957		<b>IANECESS</b>	e+020r .	• 924246	101700°	• 01647		- 0 1 0 D 0 -	00545					11 # M. F. M. #	1000.	- C 4 4 5 -			
	2 A U E				000522	012400. 19:50				Ð	+6177+	0541Q.	a94675	. n 2028	9+9r I -		, , , , , , , , , , , , , , , , , , ,								0(3*4.	-0	2 4 7 4				000		TJL0ATTM	029405.	E935-C.	101101.	5566,00.	. 210100	841276.		104+C	-0		244004	* U 3 4 D *	- 5003Z	- 3 + 4 <b>6 1</b>			
	THIAL/GHT	-	en, 1	50 10	4	<b>ده</b> د	•		SCBSUM	TRIALS	4E A 1 8 5	C1 √	CE VAR	7 A R	RECSUN			1214L/GRP	~ '	~		*	ю, 10-1	0 47	Still C.M.	101415			200		AFCSUM		TD14: /39P	-			4	<b>ء</b> م	۹ ۹		Suesur	THIALS	ME 11.85	۰ ۲	CF VAR	A R	PECSUM			

TABLE 4-2, BSS/TSS(i) PROFILE

ø

1.5

一种小学学生 化化学学化学学 化化学学 化分子子 化分子子 化二烯二乙烯 化化学 医外外的 计计算机 计计算机 化分子分子 化合金分子 化合金分子 化合金分子 计计算机 计计算机

78

-----

further splitting of this group because none of the predictors can explain more than .01 of the total variability.

Step 2. In group 3, Financial Reference has the largest BSS/TSST of . 17^c in Table 4-1 and a BSS/TSS(i) of . 28^{c1} in Table 4-2. Note that the BSS/TSST remained the same for this variable but the BSS/TSS(i) increased markedly. This would indicate that Financial Reference is immaterial at a Credit Rating level of "poor", but that it is the single most important variable over the Gredit Rating levels of "unrated", "medium", or "good". Further, note that the predictive capability of Credit Rating (d1) decreased by a factor of two on the "local" (BSS/TSS(i)) basis after the split. This would indicate that there is a much smaller difference between the upper three levels of Credit Rating than there is between those three levels and the lowest level, "poor". By splitting group 3 on Financial Reference, a second homogeneous "twig", group 4, is formed consisting of 19 data units which have a mean criterion value of .00. Because of the similarity of their mean criterion values, group 2 and group 4 are combined together into subspace 1. A one term model (a binary indicator of group membership) is sufficient for this subspace.

Step 3. In group 5, Bank Account has a BSS/TSST of .11^(e) and a BSS/TSS(i) of .25^(c1). It is the basis for splitting group 5 into subgroups 6 and 7. Neither of these resultant subgroups is as homogeneous as were the previous "twigs". Each will require further modeling. Considering the BSS/TSS(i) profiles for group 6, note that both Phone ^(f, f1) and Credit Rating^(g, g1) show sizable values. However, the group 6 contains only 30 data units and a binary split on any one predictor at this level will result in groups that contain the minimum allowable number of data units and are still nonhomogeneous. At this

point, it is better to model group 6 with a first-order model containing both Gredit Rating and Phone. Thus, group 6 becomes subspace 2.

Step 4. In group 7, Credit Rating has a BSS/TSS(i) value of  $.10^{(h1)}$  which is only slightly larger than the values for Restime^(h2) and Phone^(h3). However, if allowed to split on Credit Rating, a homogeneous "twig" will result. Group 9 has 53 data units and a mean criterion value of .98. It comprises subspace 3 and is modeled by a one term model.

Step 5. In group 8, Loan Amount^(i, il), Rescat^(j, jl), Restime^(k, kl), Marital^(m, ml), and Local^(n, nl), all have BSS/TSS values reflecting predictive capability, but a binary split on any single one of these variables will not result in a homogeneous group. This group becomes subspace 4.

This walk-through of the heuristic logic employed in defining subspaces in an AID-Tree brings out two points that were observed during the analysis on the various AID-Trees in this research. First, reasonably sized groups which are homogeneous are prime candidates as subspaces, especially if they are split off in the very early stages of the AID-Tree. Second, in those parts of the AID-Tree where the groups are not homogeneous, predictors that cannot explain in excess of 10 percent of the variability within the subgroup will probably not result in isolation of new homogeneous subgroups of substantial size: in this case it appears better to fit local "continuous" linear models.

Building Local Models Within the Subspaces:

Once the subspaces are identified, the data in the BSS/TSS(i) profile, Table 4-2, can be of further use in hypothesizing the proper

terms for the models. In formulating the models for the individual subspaces, there are several possible approaches. If there is sufficient data available (in cases where at least one data unit per predictor category is available, and preferably on the order of 5-10 data units per predictor category), a binary regression model could be built by using either the MCA program or any standard regression package. In subspaces where the number of data units is not amenable to using binary or categorical models, the formulation of a "continuous" firstorder regression model is usually a reasonable alternative. Under the latter formulation, the investigator effectively imposes the assumption that a unit change in the predictor value will produce a fixed (constant) change in the criterion variable. He makes this assumption in exchange for using fewer degrees of freedom and predictors in the model.

In the definition of subspaces and local models for the five loan officers, various alternative formulations were attempted at the subspace level. For example, the approach of combining two neighboring subspaces into a larger subspace and formulating a configural model over this larger subspace was compared with the approach of formulating separate first-order models over the original subspaces. The general conclusion drawn from these efforts was that the approach of using first-order models over the individual subspaces was superior.

The Determination and Evaluation of Model Coefficients:

The task of obtaining regression coefficients for all of the models investigated was performed using an iterative determination/ cross-validation procedure. A similar procedure was originally suggested by a member of the research supervising committee,

Dr. E. Jennings, as a method of evaluating the stability of alternative models. The technique consists of the random selection of 80 percent of the data units from the input file, generation of the coefficients using the REGREJ routine of the EDSTAT-J system (Jennings, 1971), and cross-validation of these coefficients with the remaining 20 percent of the data units. This cycle was accomplished 10 times and resulted in a profile of coefficients for each predictor. Whereas Jennings uses this procedure solely to assess the stability of various models, in this research, the average values for each of the coefficients were actually used in the model. The objectives of this revised procedure were the simultaneous assessment of the stability of the model and the reduction of the idiosyncratic effect that any outlier data units would have on the model coefficiente The average coefficients were compared with the corresponding coefficients obtained in a single fit over all 224 data units for each of the five judges. They were found to be very similar. One interesting outcome of this comparison is reflected in Table 4-3. Note that the average values for predictive efficiency  $(R^2)$  for the 10 sets of coefficients, resulting from the iterative scheme, are slightly better than the corresponding values of  $R^2$  that were obtained in the single fit procedure. This would suggest that, indeed, some of the idiosyncratic effects of outlier points may have been overcome by the iterative procedure.

### TABLE 4-3

## COMPARISON OF PREDICTIVE EFFICIENCIES

Procedure			Judge		
	<u> </u>	2	3	4	5
Iterative Procedure	. 670	. 707	. 680	. 633	. 632
Single Fit Procedure	. 659	. 695	. 680	. 614	. 618

# FOR ALTERNATIVE COEFFICIENT DETERMINATION PROCEDURES

Comparative Results of Various Modeling Approaches:

## Installment Loan Officer Models

The quality of the various models investigated in this research were initially measured in terms of the standard statistical parameter,  $R^2$ , and the cross-validity from one subset of the 224 sample points to another subset. An 80/20 split between the determination sample and the cross-validation sample was used. The average crossvalidity for 10 trials per model of the first-order, second-order, and local models are shown in Table 4-4; the data for the AID splitting process consists of only one trial per AID-Tree.⁷ These data reflect

⁷The process of using an 80/20 split in the AID cross-validation process was found to influence the results in the cases of limited data sets since the groups in the subgroups propagated in the 20 percent sample become very small and susceptible to group means that were greatly affected by random sampling error. The cross-validity coefficient in AID is not generally monotonic and the values presented in Table 4-4 reflect the maximum cross-validity attained in the forced splitting.

Model			Judge		
	1	2	3	4	5
First Order	. 579	. 624	. 634	. 512	. 538
Second Order	. 649	. 645	. 653		
AID-Split	. 440	.621	. 675	.670	. 547
Local Model	.727	.659	.685	. 633	.558

# TABLE 4-4 COMPARISON OF CROSS-VALIDITY ON SPLIT SAMPLE

that the local models, derived from the AID-Tree, were consistently better than the first-order or the second-order models.

In actually evaluating the goodness of the various modeling approaches in predicting dichotomous decisions of the installment loan officers hit rate is a more relevant measure of goodness than  $\mathbb{R}^2$ . Since each of the models resulted in a predicted score in the normal range from .00 to 1.0,⁸ each possible value in this range could be used as the cut score, or discrimination point. Above this point the value of "1"(Yes) would be assigned to the data unit, and below this point, a value of "0" (No) would be given. Generally, at each possible cut point there are some data units that will be misclassified (either actual 0's given the value of 1, or actual 1's given the value of 0). The hit rate reflects the percentage of proper classifications, but does not reflect the type of misclassification that is being made. Data on cross-validation hit rates was commensurate with that shown in Table 4-4.

Actually scores slightly smaller than .00 and slightly larger than 1.00 can be obtained but in those cases the scores are rounded to .00 or 1.00 respectively.

The indifference of the magnitude of hit rate to type of error is extremely important since the decision maker may be more concerned with one type of error than he is with the other type. Hence, the performance of the model over a range of possible cut points is important. Figure 4-4 schematically depicts the profiles of hit rates for first-order



and local models for the installment loan officer data. It shows that even if the maximum hit rate of the first-order model is marginally better, the profile is generally inferior and the marginal superiority prevails over only a very small range of cut scores. Thus, the single parameter embodied in either the hit rate or the cross-validation coefficient is probably not the best criterion for determining overall sureriority, but rather, its magnitude and profile should be considered

jointly. Figures 6-14 through 6-18 in Chapter VI reflect that both the maximum hit rates and the profiles were superior for the local models in the 80/20 cross-validation process.

A completely separate cross-validation effort using 150 data units collected during a different time period was also performed. The maximum hit rates for the first-order models were slightly superior to those for the local models in this test. However, this reversal can be traced to a significant shift in the characteristics of the data set and a shift in the policy of the judges. The full details of this test are discussed in Chapter VI.

#### The Valenzi Data, an Independent Example

Whereas the approach of defining subspaces and local models was conceived as a result of the initial analysis of the policies of the loan officers and was developed primarily using this data, it might be criticized as being useful only in situations with binary criterion values. An insight into the generality of the local modeling approach was sought by applying it to an independent⁹ set of data for which a continuous criterion existed. The data used for this analysis was published by Valenzi (1970). The data consisted of the judgments made by employment counselors relative to the chances of applicants being hired for secretarial jobs. Five tricotomous predictors (typing speed, shorthand speed, experience, education, and social skills) were used to describe each applicant. The criterion was a scalar value from 0 to

⁹Independent in this context denotes that the investigator had not previously analyzed the data and was not able to intuit a good model as a result of familiarity with the process or the data.

10. A full 3⁵ factorial experimental design with two replications per cell was analyzed with both ANOVA and multiple regression models. Valenzi attempted to demonstrate and evaluate configural cue use by four judges; the analysis and comparisons in the following discussion considers only one judge, Judge 4.

Valenzi published the following results for Judge 4:

- ANOVA results with main effects and all two-way interactions resulted in 73 percent of the variability being explained by the main effects and 14.5 percent being explained by the second-order configural effects. A total of 87.5 percent of the variability was explained with some 41.5 percent being explained by the educational main effect.
- 2. The second-order multiple regression model resulted in 60 percent of the variation being explained by the first-order terms and 9.7 percent being explained by the second-order terms.
- 3. The largest interaction identified by either of these processes was Education by Social Skills which explained 8.5 percent of the variability in Valenzi's ANOVA model and 5.4 percent in his second-order regression model.
- The importance ranking for the predictors given by Judge 4 did not match the rankings derived from the statistical analysis models.

Valenzi concluded that Judge 4 did exhibit configural cue use and that ANOVA was a better approach to detecting it than were

standard multiple regression techniques or the "<u>a priori</u> partialling out" of the configural variance suggested by the judge's verbalization of his policy.

The results obtained by applying the AID4UT/AIDTRE computer programs and the local modeling approach to the same data were:

- An AID-Tree with 14 final groups explained 89 percent of the variability. Using an 80/20 split, a crossvalidity of 85 percent was achieved.
- 2. The first split in the AID-Tree was made between the lowest and the upper two levels of Education. This split explained 40.1 percent of the variability and resulted in a homogeneous "twig" which included 162 data units (1/3 of the total data set). This group had an average criterion value of .01 on a scale from 0 to 10.
- 3. The most important variable for the other 324 cases was social skills. Its interaction with all other variables was readily detectable in the detailed output from AID4UT.
- 4. Definition of two subspaces on the basis of the first split and the use of the following local models.
  (Subspace 1 included all cases with the lowest level of Education and subspace 2 included all others.)
  - **a.**  $\mathbf{Y} = \mathbf{b}_0 + \bar{\mathbf{Y}} + \mathbf{e}$ : for subspace 1
  - **b.**  $Y = b_0 + b_1$  (social skill)(typing)
    - + b₂ (social skill)(shorthand)
    - + b₃ (social skill)(education)

+ b_d (social skill)(experience) + e:

#### for subspace 2

resulted in a 6 predictor model which explained 81.8 percent of the variance, a gain of 12.1 percent relative to the 15 predictor second-order regression model used by Valenzi.

- 5. Changing the model over subspace 2 to a binary regression model which considered all two-way interactions,
  90 percent of the variation was explained.
- 6. The rank order of the variables as determined by the profile of BSS/TSS(i) from the AID4UT program was more consistent with the judge's verbalizations than the corresponding ranking achieved by Valenzi with ANOVA and regression analysis. Table 4-5 reflects the comparison. A complete reversal of the ranks of the first four variables arose from ANOVA. Only social skills was out of sequence with AID4UT.

#### TABLE 4-5

3

COMPARISON OF RANKED IMPORTANCE OF PREDICTORS

Rank	Judge's Verbalization	ANOVA	AID4UT
1	shorthand	education	social skills
2	typing	social skills	shorthand
3	social skills	typing	typing
4	education	shorthand	education
5	experience	experience	experience

The implications of the comparison of the results achieved on the Valenzi data are clear. The use of the AID algorithm to isolate homogeneous subgroups over which better fitting local models can be applied is potentially superior in situations other than those with dichotomous criteria. The superiority of the technique is not solely a function of having "lived" with the data and the decision process for a prolonged period of time. In this independent example, the model achieved was simpler and easier to reconcile with the verbalized policy of the judge.

Summary of Guidelines for Applying AID4UT/AIDTRE in Policy Capturing:

The experience gained thus far in applying AID4UT/AIDTRE to Policy Capturing is extensive in the sense that numerous models have been attempted and produced, but limited in the sense that these models were all for a particular decision process, except for the brief demonstration on the Valenzi data.

Table 4-6 presents a set of guidelines that should make future applications of the AID algorithm and the local modeling technique more efficient and effective. These guidelines reflect the experience of Sonquist (1970) as well as the experience gained in the current research effort. They are directed at using AID4UT/AIDTRE for the definition of subspaces and local models when there is limited data available.

These empirical guidelines for applying AID are somewhere between the stages of artistic and heuristic. This investigator doubts whether such guidelines will become more than heuristic in the foreseeable future. Considering that AID is a model seeking technique and its functions is to uncover relationships in data sets with unknown

structure, it is improbable that precise, deterministic criteria for defining subspaces or for specifying exactly which predictors are important, can be defined. If enough information were known about the data set to do this, the very need for the model seeking technique would be doubtful.

### TABLE 4-6

# GUIDELINES FOR EMPLOYING AID4UT/AIDTRE IN BUILDING POLICY MODELS

#### PREPARATION OF DATA:

- Segment predictors into 3-8 categories. A lower limit of at least 5 data units per category should be sought.
- If the criterion variable is badly skewed either divide the data set into subgroups or make a transformation on the criterion such as Y' = log Y.
- 3. Obtain a contingency table for the predictors. An association measure such as the Goodman and Kruskal  $\lambda$  as produced by Anderberg's (1971) GCORR program is helpful. Note those variables with very high associations since they may be measures of the same phenomena and substitute for each other in the splitting process. In such cases one of the variables should be eliminated from consideration as a predictor.

#### INITIAL RUNS OF AID4UT/AIDTRE

- 1. Run AID4UT with all stopping parameters disabled. (See Appendix C). This produces a "configural" model which provides an upper limit on the capability of the set of predictors under consideration to predict the variability in the data. For large data sets with many predictor categories, this run may be very time consuming and setting the minimum subgroup size to .01 of the total set may be advisable.
- 2. Refine/collapse the predictor categories if possible. If two or
more categories of a predictor never end up in different subgroups as a result of a split they can be combined into one category.

3. Analyze the residuals for obvious coding errors. Such errors are usually subtle, but can affect later results considerably.

## ANALYSIS RUNS OF AID4UT/AIDTRE

Ņ

- After corrections and modifications from the initial runs are completed, rerun the analysis with increased stopping parameters. In data limited cases where less than 10-20 data units per predictor category exist, set split reduceability = .01 and the minimum group size to ≥ .05 of the original sample size.
- 2. Analyze tree to identify large, homogeneous "twig" groups indicated by:
  - a. Mean criterion values of > . 9 or < .1 for dichotomous criteria
  - b. Standard deviations of the criterion values of < .2 of the standard deviation of the original group for continuous cviteria.
- Study the mean criterion values for each predictor category in each subgroup for the first 4 or 5 splits. Pertinent points are:
  - a. The difference in the magnitude of the mean criterion values in two subgroups resulting from a split reflect the "main effect" of the split variable over the parent group. Of particular importance are the trends of the mean criterion values throughout the AID-Tree. Plots of the mean criterion values versus the predictor categories for each of the subgroups in the AID-Tree will portray these trends. If the trends are similar then the predictor is likely to be independent of the variables used in the previous splits; if the trends for a predictor vary greatly, there may be interaction present between the predictor and the variables used in the previous splits.
  - b. If the standard deviation of the criterion values for a given predictor category is near zero and the category is small (< .05 of total sample) this represents a homogeneous</li>

group which may not be automatically identified by the AID-Tree because of its size and the intervention of more powerful predictors in the early stages of the splitting process. If there is a plausible theoretical reason for this group it could be removed from the data set in future analysis.

- 4. Analyze the BSS/TSS profiles to ascertain predictive capability of the individual predictors over the AID-Tree. Pertinent points are:
  - a. If BSS/TSS(i) is greatly (a factor of 2) different in the groups resulting from a split, two situations may exist; either there is an interaction between this variable and the split variable or they are measures of the same phenomena. A check of the contingency table would indicate if they are highly correlated and if substitution is likely. If the BSS/TSS(i) for the split variable changes, either groups with a single category of that predictor have been created or there is a higher order effect for the predictor.
  - b. If the BSS/TSS(i) value for a predictor remains the same after it has been involved in a split, but it loses the next split, a "continuous" interaction (as opposed to the discrete interactions represented by discrete hyperplanes) may be present. Separate analyses for each of the categories of the predictor involved would result in similar AID-Trees for each category, however, the mean criterion response values for corresponding subgroups in the various AID-Trees would be different.
  - c. If the BSS/TSS(i) for a predictor remains small (<.05), but relative constant over the entire AID-Tree, the predictor may have a small universal effect and could be included in each local model.
  - d. Comparison of the BSS/TSS(i) and BSS/TSST profiles will give further indications if a variable has a universal or only a local effect. Those variables which appear very low in importance on both scales can probably be discarded from the predictor set.
  - The identification of important variables for the various subgroups can be accomplished from the profile of BSS/ TSS(i). Lasting the most important variables on the skeleton version of the AID-Tree will provide a visual impression of groups in which the same predictors a confective.

#### DEFINITION OF SUBSPACE AND LOCAL MODELS

- 1. Subspaces can be identified on three bases.
  - **a**. Homogeneity relative to the criterion value.
  - Ь. Consistency relative to the predictors that are important within the subgroup.
  - Consistency relative to the effects that the individual prec. dictors have in the local subgroups.
- 2. If there are no homogeneous groups identifiable on the basis of the above characteristics, the total predictor space should be represented with one "continuous" model. In this case the hypothesis of "continuous" interaction terms of the type,  $b_1 x_1 x_2$ , can be accom-

plished by considering the BSS/TSST profile.

- 3. If there are homogeneous subspaces within the AID-Tree, the information from the splitting process provides the basis for the generation of these subspaces directly. The residuals list may be used to identify and physically separate the data cases in the various subspaces for separate analyses. Alternatively, the logical capabilities of the computer can be used to define binary subspace multipliers. CAUTION: It is extremely easy to define overlapping subspaces if the split process of the AID-Tree is not followed precisely.
- Once the subspaces are identified, the predictors within each 4. local model are hypothesized on the basis of the BSS/TSS(i) profile. If sufficient data units exist, binary models can be formulated. Otherwise, "continuous" first-order regression models appear to be the best approach. Those variables not inherently monotonic should be checked for this property over the subspace under consideration. This is an implicit assumption of "continuous" formulation and the BSS/TSS values do not necessarily reflect this restriction.

#### GENERATION OF REGRESSION COEFFICIENTS

Any available regression package can be used to generate the appropriate regression coefficients. The data transformation capabilities available in the DATRAN feature of the EDSTAT-J package are

particularly compatible with the logical definition of subspaces on the computer. The use of the iterative coefficient determination/ cross-validation approach is recommended since it allows assessment of the coefficient stability in the process of model building.

- A + A + # - W und

:: :: ..

, . , .

۰.

., .,

g!

e.

• *

••

रू ** ''.

: : ... : ...

a. "

# CHAPTER V: BACKGROUND, ENVIRONMENT, AND CHRONOLOGY OF CASE STUDY

Introduction:

5 .

This chapter provides the scenario and a brief history of the activities of this research that were specifically related to capturing and evaluating the policies of a group of installment loan officers. The results and implications are then presented in Chapter VI.

A Brief Resume of Numerical Credit Scoring:

The application of numerical rating systems to credit evaluation was first proposed by Durand (1941) in a study of several hundred accounts. He used discriminant analysis to categorize those accounts on the books as "good" and "bad" on an "ex-post" quality¹ basis. His study, and others that followed, generally indicated the potential utility of numerical credit scoring systems in screening out bad accounts. However, there was no meaningful implementation of any such systems until about 1960. Myers and Forgy (1963) attribute this lack of implementation to:

- a natural reluctance on the part of the experienced credit executive to abandon the time honored "judgmental" approach in favor of the newer and relatively untested quantitative methods;
- 2) the inability of statisticians to develop "foolproof"

[&]quot;"ex-post" quality is based on collection experience and "ex-ante" quality is based on prospective risk. The decision process considered in this research is based on the "ex-anie" quality of the loan application.

rating systems which consistently identify poor credit risks accurately enough to result in substantial net savings by the credit operation;

- 3) the difficulties involved in utilizing any such effective rating system in the operating situation; and
- 4) unwillingness on the part of the statistical consultants to invade the domain of the credit manager and do the selling job necessary to transform such an idea into a successful and useful operating tool. (p. 120)

In their own study, Myers and Forgy (1963) set about analyzing various statistical modeling techniques for scoring loans on mobile homes. Again they used the common procedure of scoring "ex-post" quality and did not consider the judgment process of the credit executive nor did they demonstrate the operational feasibility of the system. They did make one advancement by validating their models by means of cross-validation.

The expanding availability and use of computers within the credit industry during the mid-1960's propagated renewed interest in numerical scoring systems and considerable debate as to their utility has transpired in financial journals such as <u>Credit World</u> and <u>Banker's Monthly</u>. Continued research in the area has proceeded under the sponsorship of the National Bureau of Economic Research, the Credit Research Foundation, the First Pennsylvania Bank and Trust, and Morris Plan Company of California.

By 1968 some 35% of the top 200 banks in the United States were using numerical credit scoring of some form, and another 33% were considering such systems. Of those banks having systems, some 80%

97

had been borrowed by one institution from another. The borrowing of scoring systems was attributed to the extensive analytical effort involved in setting up a "tailored" system, and was responsible for significant implementation problems. (Wilt and Tierney, 1968) As a result of these problems brought about by using borrowed systems, the credibility and utility of numerical credit scoring has suffered even more in the eyes of the critics.

In a review of the first decade of use of numerical scroing systems, Zaegel (1971) noted that the benefits from use of such systems could not be accurately measured, but that losses were down approximately 20% in situations where the systems were being used. He did note that many improvements could be made especially with regard to the time it took to establish and validate a system, which requires about five years for the loan turnover cycle when the system is based on "ex-post" quality.

Although brief, this resume sets the stage for the research into applying Policy Capturing techniques in defining credit scoring models that are more "personal", "flexible", "understandable", and hence, more likely to be implemented in operational environments.

The Decision Process Considered in this Study:

The primary decision process modeled in this research is that of the installment credit officer in making judgments of whether to grant or to deny a loan to an applicant based on information contained in a written application. The decisions being modeled are limited to those involving "dealer paper", i.e., loan applications forwarded to the lending institution as a result of a dealer contract. Under a dealer contract, the lending institution provides the financing for a merchant's sale on an individual account basis, i.e., they agree to buy the installment contracts from a merchant on an individual basis if the applicant and the account meet the lending institution's standards of acceptability. Unlike more general credit decisions in which the interest rates and other particulars of the loan are negotiable parts of the decision, the dealer contracts under consideration — carry a fixed interest rate and provide that the merchandise will serve as collateral. Further, the amount and duration of the loan are not actually considered as items of negotiation, but as parameters of the loan. Thus, the decision is essentially uni-dimensional in that only a binary decision of "accept" or "reject" is made.²

A flow diagram of the "dealer paper" process is shown in Figure 5-1. Note that no personal contact exists between the loan applicant and the credit officer and the decision is made solely upon the basis of the input data contained on the written application and the applicant's credit rating that is obtained from the local credit bureau.

The original application form (Exhibit 1, Appendix B) is filled out by the clerical staff at the dealer location and is then telephoned to the lending institution. There the clerical staff of the installment loan department transcribes the information and calls the local credit bureau for a credit report. Upon receipt of this report, the information is attached to the application and forwarded to an available loan officer for his judgment. If he is a senior loan officer

²In special cases an "accept with full recourse" is made in which the lender grants a loan he would ordinarily reject if the merchant guarantees payment in case of default. In this study, these are considered rejects.



1.

FIGURE 5-1

his judgment is generally conclusive, but if he is junior, he normally makes a recommendation and obtains concurrence from one of the senior loan officers. When the decision is made, the dealer is notified. If the loan is accepted, he executes the contract and releases the merchandise to the customer; at this point the responsibility for maintenance of the account and collection of the loan devolves to the lender. If the decision is denial, the dealer must either seek financing elsewhere, carry the note himself, or vitiate the sale.

The entire process, from sale to a decision on the loan, involves approximately four clerks, two transcriptions of the data, and the judgment of one or more loan officers. It requires from one hour to several days. For purposes of later discussion, the "system" will include those elements from sale until notification of decision and the "credit system" will include those elements from receipt of the application by the lending institution until the ratification of the decision.

The Lending Institution:

The installment credit department participating in this study is part of Lamar Savings and Loan Association located in Austin. Texas. The department is undergoing a transition period of rapid growth. In the last three years their outstanding installment accounts have increased over 500% from less than \$2 million to over \$10 million. In 1970 and 1971 they began to concentrate their efforts on dealer contracts and that source of business amounts to approximately 68% of their current volume. They are currently processing approximately 600 applications per month with a staff of five loan officers and a clerical staff of 15 people.

The Decision-Makers:

The decision-makers who acted as the principle judges in this research were the five installment loan officers of Lamar Savings. Two other experienced loan officers from other lending institutions in Austin also participated in the effort to assess the transferability of the policy models. A short background sketch of each judge is given below:

- Judge #1 is an Assistant Vice-President and is the number two man in the installment loan department. He has nine years experience in installment lending and has been with this department for 3.5 years.
- Judge #2 is a Vice-President and is the manager of
  the installment loan department. He has 14 years
  experience and has been with the department for three
  years. He has had previous experience with numeri cal credit scoring systems.
- 3) Judge #3 is an Assistant Vice-President and is third senior man in the department. He has seven years experience, most of which has been under the supervision of Judge #2 and concentrated in the area of collection of delinquent accounts. He has been in the department two years.
- Judge #4 is a loan officer and has three years experience, all in this department.
- 5) Judge #5 is a loan officer and has two years experience,all with this department.

- b) Judge #6 was recently the manager of a competitive lending institution in Austin and was hired into the department near the end of the project. He has had experience with dealer contracts and numerical credit scoring and a total of four years experience.
- 7) Judge #7 is the manager of a local credit union and has 14 years experience. His organization does not have any dealer contracts and they do not currently use numerical scoring techniques.

Two Functional Objectives of the Policy Models, Tentative Approvals and Surrogate Judgments:

In reference to the decision process being considered, the objectives of the modeling effort are to provide decision models that can be used both for screening the loan applications on a "tentative approval" basis and also pre-scoring the applications on a "surrogate judgment" basis. The difference between these two functions lies in the availability/non-availability of the credit rating for use as a predictor.

As depicted in Figure 5-1, the ava "ability of a credit rating generally requires an input from the local credit bureau. At present, this input is obtained by telephone and is only available during normal working hours from Monday through Friday. If a merchant operates on the basis of six or seven days a week, this represents a significant impediment in his ability to sell and obtain timely financing for his merchandise. No action can be taken on a loan application taken after 4 p.m. Friday until at least 8 a.m. the

following Monday and the merchant must either delay delivery of the merchandise or assume the risk of possibly having to retrieve it if the loan is denied. This delay represents a tangible economic consideration to the dealer which he would like to minimize, especially in the case of those customers who appear to qualify for a loan on the basis of the other information available on the application form.

0

Whereas the credit rating is the most important single variable considered in the loan officers' decisions, they are generally unwilling to make a final commitment without it. However, it has been found that people with good credit ratings also generally score well on the other predictors. The problem is to determine if a policy, based on only the other predictors, can be defined that will identify a sufficient number of good accounts to justify its use without involving undue risk. If such a policy can be found, the credit executives would be willing to assume some increased risk in order to provide better service to the merchant. In this regard, a "tentative approval" would be an approval given in the absence of credit rating information which might require some modification of the contract such as "full recourse" if the credit rating was later determined to be unacceptable.

The pre-scoring or "surrogate judgment" application of the policy models would be accomplished with the use of the credit rating. In this case, the policy model would not suffer from the obvious structural deficiency of lacking the most important predictor. The rating from this model would be the expected rating that any application with that particular set of predictors would normally receive from a judge. Again, this could be used in two modes, if the judge finds no extenuating circumstances in the application, he can base his decision on a fixed threshold value and the predicted score; if he chooses to

consider other factors that are not accounted for in the policy model, he can use the predicted score as a starting point and base his decision on this score plus his evaluation of the extraneous factors.

#### Chronology of the Project:

6

The original impetus for this research came from a project in an advanced statistical applications course. Potential applications for policy capturing were being assessed and an attempt was made to model Judge #2's policy relative to a set of fictitious loan applications. This led to the discovery of four aspects of the problem that presented a challenge to defining viable policy mode's for loan officers' decisions. These were:

- The methodology of ranking decision cases as was the standard procedure in most past Policy Capturing studies was untenable in this situation.
- 2) The generation of fictitious data cases in a designed factorial experiment led to many ridiculous combinations of predictors and detracted from the credibility of the process.
- The problem of scaling and ranking of the predictors for use in standard statistical routines was not easily solved.
- 4) The modeling of an individual loan officer's decisions as a first-order function of the individual predictors neither produced satisfactory results nor was it intuitively satisfying to the loan officers involved.

The results of that initial effort did intrigue the participants enough to embark upon the research reported herein. The major activities of the project are now described.

#### Selection and Coding of Predictors:

The initial selection of predictors was suggested by the application form itself. The three senior loan officers (Judges 1, 2, 3) were each interviewed and asked how they defined and quantified the data on each of the input blanks. This resulted in the definitions and codes shown in Exhibit 2, Appendix B.

### Retrieval of Data:

\$

2 T P - 52

The category of loans under study consists of dealer paper on durable appliances. The original data base was obtained by randomly selecting 404 loan applications that had been processed by the loan officers during the period from May to September, 1971. Loan applications from two dealers were drawn from the files and coded according to the categories determined in the previous step. Those predictors such as age, income and equity which were directly transcribable were coded by the investigator. Those predictors that required some judgment in themselves were referred to the loan officer who had originally processed the application and he performed the coding. ³

### Analysis and Ordering of Predictor Calegories:

Initial analyses were performed on a set of 404 coded

³In this respect the exercise of expert judgment in the measurement function as advocated by Sawyer (1966) and Einhorn (1972), (See Chapter II) was inherently accomplished as part of the procedure.

applications. For those variables that were naturally monotonic such as age and income, AID analyses were used to determine at which breakpoints, category definitions were meaningful. For those variables which were not naturally ordered, the categories were assigned ranks based upon the average approval rate for all applications in that category. This procedure is discussed in sections 3, 2, 4 and 3, 2, 6, 2 of Anderberg (1971). At this point no assumptions were made as to the interval nature of the data and the predictor values were considered as ordered categories only.

#### Initial Analysis Efforts:

The categorized data from the sample applications was used in an initial series of computer runs that were directed primarily at answering the following four questions:

- Can predictor variables and decisions from historical files be used to capture the judge's policy without requiring him to make new decisions on different cases?
- 2) Can the historical data taken from the file be used to build a model that adequately predicts the actual approvals/denials without considering Credit Rating?
- 3) What is the structure of the decision process and can the AID-Tree adequately depict this structure?
- 4) Can the AID-Program be used to detect, and isolate inter-judge policy differences?

With respect to question two, the reason for the exclusion of Gredit Rating lies in the fact that Gredit Rating is the only predictor that is

not available immediately after completion of the application and requires an input from a source external to the dealer-lender subsystem. (See Figure 5-1.)

#### Definition of a New Predictor:

A major result of these initial efforts was the classification and delineation of the concept of Financial Reference. Originally this predictor had been stated as Credit Habit and was given the subjective categories of good and bad. The initial analysis indicated that it was the second most important predictor. Its importance led to a discussion of what constituted a good or bad Credit Habit in the opinion of the senior loan officers. After some debate, these officers agreed that the concept would be better labeled as Financial References and was a measure of the depth and quality of the applicant's borrowing habits. The specific categories they defined were:

<u>Major</u>--the credit references listed by the applicant include credit depth on experience with major lending and financial institutions which do not take undue credit risks and from whom a reliable credit reference might be obtained. This category includes banks, savings and loans associations, major chain department stores, and national based lending agencies.

<u>Minor</u>-- the credit references listed by the applicant include only local stores and usinesses from whom a direct credit check was not readily available or necessarily reliable. The credit references do not indicate significant depth of credit, i.e., experience of the applicant in mana, ing his t nances to meet a continuing repayment of a loan of size comparable to that being applied for. Examples of applicants in this category would be those listing only small revolving charge accounts, such as oil companies or those listing only local concerns which carry their own short-term credit.

Other--the credit references listed by the applicant indicated reliance upon lending institutions specializing in high-risk high-interest loans. The references indicated above average dependence upon borrowed money or lack of capability to manage one's money effectively to meet normal living expenses. Those people listing no references at all were included in this category.

#### Refinement of the Sample Data Set:

Ŀ

<u>1)</u>

A second refinement of the data set involved the concentration of effort on modeling the policy of the judges toward only one of the dealers, namely, Dealer A. This resulted in the reduction of the data set to 224 cases, of which 100 had been approved and 124 had been denied. These data are shown in Exhibit 4, Appendix B.

#### Generation of Codified-Predictor Data and Collection of Judges' Decisions:

Using the data entries for each of the 224 cases (See Exhibit 4.), codified predictor data for each application was generated in terms of the descriptors for each of the predictor categories. The predictor descriptors for each of the cases were displayed on a computer printout (Exhibit 5, Appendix B) and used by the judges to make new decisions for each application. The judges were given one set of codified predictor data which included all of the predictors except Gredit Rating and were asked to make their decision in the context of "tentative approval". Later, they were given a second set of data which included the Gredit Rating and were asked to make the decision in the context of granting final "approval" or "denial". As part of this activity, the two junior loan officers, #4 and #5, were brought into the project. They were informed of what had previously transpired and the objectives of the experiment; each of the three senior loan officers discussed his policy as depicted by the AID-Tree. The discussion that followed was possibly one of the more enlightening aspects of the entire project. The major points of this discussion revealed the following situation:

1) Since the installment department was in a state of rapid expansion, new people were hired and put to work immediately without benefit of any formal training program. Loan officers were generally assigned to collection activities initially and later allowed to evaluate applications under the supervision of the senior loan officers. However, in this latter phase the supervision generally took the form of getting a senior loan officer's approval on the final decision about an application. Although this procedure does provide some feedback in the cases where the junior and senior loan officers reach a different decision, it apparently does not suffice to provide the junior loan officers with a good understanding of how the senior loan officers evaluated each of the entries on the application form. Thus, the discussion of the applications to be rated appeared to provide the catalyst for the first comprehensive exchange of opinion among all five of these loan efficers.

2) The concept of Financial Reference as previously agreed upon by the three senior loan officers was not a part of the policy of the junior loan officers.

## Generation of Policy Models:

1

Upon completion of the evaluations by each of the loan officers, policy models for each of the officers were developed using th techniques discussed in Chapter IV. These models are presented and discussed in Chapter VI.

At this point a different problem became apparent in using these models as a predictive tool. The problem is that of which loan officer's model to use on which application. In the current operating scheme, the incoming applications generally are not routed to any particular individual, but are acted upon by whomever is available. This factor not only posed a problem for adequately evaluating the performance of a given loan officer's model but also has the effect of making an applicant's chances for getting a loan a function of which judge evaluates it, and what his idiosyncrasies may be. Therefore, several voting strategies were investigated. The strategy that best matched the actual decisions was determined to be one in which a loan was approved if four of the five of the individual loan officers approved it.

#### Validation by Field-Test:

Two of the original goals of this research were to analyze the problems that might be encountered in the implementation of these policy models in the real environment and to demonstrate the adequacy of the models by cross-validation. To this end, the appliance dealer who was the source of the 224 applications that were used in the modeling phase was approached and asked to participate in a onemonth field-test. This field-test took the form of having the dealer's clerical staff fill out the codified application form, Exhibit 6, Appendix B, in addition to the normal application form on each sale that required financing during July, 1972. In order to accomplish this task, the clerical staff was briefed on the project and given a set of instructions

111

as shown in Exhibit 6, Appendix B. Upon receipt of the original application, the loan officers also filled out the coded form but made their decision in the same manner they had always done. During the month, 150 usable applications were collected which were then analyzed in several ways. These analyses included:

- evaluation of model sensitivity coding differences between "exper: measurement" of the loan officers and the "non-expert measurement" of the clerical staff,
- cross-validation of the previously derived individual and voting policy models,
- evaluation of the average time and range of times the applications were in process.

## An Insight into Transferability:

ä

y.

One of the interesting findings arising out of the review of past experience with numerical credit scoring systems is the fact that such systems have been "borrowed" as opposed to "tailored" for new enviornment :. A natural question then is "are policy models captured in one environment amenable to being transferred to new environments?" Although an adequate answer to this question would require a major research project itself, an attempt was made to get an indication of the potential transferrability of the models. In this effort two experienced loan officers who were managers of other lending agencies in Austin were interviewed and asked to make decisions based on the codified predictor on the 224 applicat ons. One of these individuals, Judge #6 had recently been hired by the participating installment credit department but had not yet worked with them long enough for their policies to influence his opinion.

#### Summary:

8

e S

<u>)</u>,

.,

÷.,

્યુ

This chapter has briefly reviewed the past work in numerical credit scoring and has described the environment in which this experiment was conducted.

A short chronological description of the research activities relative to capturing and evaluating the policy models for the loan officers has been given. The various forms and instructions generated and used in these activities have been included in Appendix B. The results, conclusion and implications of this project are discussed in Chapter VI.

## CHAPTER VI: ANALYSIS, RESULTS, AND IMPLICATIONS OF LOAN OFFICER POLICY MODELS

#### Introduction:

In documenting the results of this research, it might be sufficient to simply list the final set of equations and their cross-validation hit rates. Fowever, many initial and intermediate results were obtained, each of which influenced the direction of the research activities that followed to some extent. Not all of the results were positive, but even those that were negative added to the knowledge of the investigator. It would seen somewhat pretentious not to discussall of these results for two reasons; a) they may answer questions of later investigators and help them avoid unnecessary steps, or b) they may provide the impetus for better ideas or decisions than those which were pursued by this investigator. Therefore, this chapter includes the intermediate results as well as the final equations in the hope of providing a better insight into what worked and what did not work. The breadth of the results presented in this chapter may create a problem in assimilating the information presented. To alleviate this problem, the presentation has been further segmented into topical sections and the following directory is provided.

Section	Topic	Pages	
I	Results and Decisions of Initial Analysis	116	
II	Analysis of Decisions for Codified Predictor Sets	126	
III	Presentation of Individual Policy Models	134	

IV	Comparison of Models and Judges	155
v	Discussion of Voting and Composite Models	163
VI	Results of the Cross-Validation Field-Test	167
VII	Implications for Implementation	180
VIII	Summary	189

à

ь , ,

•

٩

•. •

AND A STATE

2

s,

2

8

ШŚ.

Section I.

ķ

A,

5

Analysis and Refinement of the Original Sample:

The original sample of data included 404 applications which were randomly selected from the files of the lending institution. Of these, 270 applications had been submitted by a high volume dealer (Dealer A) who had been participating in a dealer contract arrangement with the lender for a period of approximately six months. The other 134 applications were from a lower volume dealer (Dealer B) who had been working on a dealer contract basis with the lending institution for several years.

The data on each application was coded in terms of one of the category descriptors shown in Exhibit 2, Appendix B. For those predictors that were not naturally monotonic, the categories were rank ordered according to the average success rates for all applications within each category. The data in Table 6-1 reflects the success, or approval, rate for each category of each predictor and was the basis for the category rankings.

#### TABLE 6-1

Variable \ Category	1	2	3	4	5	6	7
Age	. 50	. 57	. 58	. 79	. 89		
Marital	. 59	. 63	. 63				
Local Family	. 58	. 61					
Draft	. 00	.60					
Telephone	. 35	.63					
Average Time	.53	.62	.68				
Residence Cat.	. 25	. 51	. 72	.88			
Employ. Type	. 48	. 62	.64	.65	. 72		

## BY-CATEGORY SUCCESS RATES USED FOR RANKING OF PREDICTOR CATEGORIES

Variable \ Category	1	2	3	4	5	6	7
Income	. 44	. 49	. 64	. 48	. 59	.74	.74
Bank	. 33	.67					
Credit Habit	.12	.70					
Loan Amount	. 53	. 59	. 56	.66	.60	.70	
Loan Term	.68	.53	.55	. 57			
Necessity	. 54	.69					
Equity	. 56	.63	. 74				
Credit Rating	.10	. 55	. 65	. 94			

J.

ેન્પ્રે

Table 6-1 (cont.)

The initial analysis of the data set using both bi-variate contingency tables generated with Anderberg's (1971) GCORR program and output from the AID4 program led to several refinements and modifications to the predictor set. These included:

- Elimination of draft as a variable since draft eligibility automatically eliminated an applicant from consideration based on company policy.
- Elimination of debt as a predictor since useable data was available on less than 1/3 of the applications.
- Reduction in the number of categories required to describe the loan amount and income predictors.
- 4) Inclusion of self-employed and military officers into the executive category based on indifference of initial AID analyses to their classification in separate categories.
- Elimination of those applicants achieving loans through "full recourse" guarantees of the dealer.

These modifications resulted in a refined data set of 389 applications, 262 from Dealer A, and 127 from Dealer B.

Initial Analysis Disregarding Credit Rating as a Predictor:

The 389 data cases were used in exploratory efforts to build a model that would adequately predict the actual response without the use of credit rating as a predictor. Various modeling alternatives including multiple regression, multiple classification analysis, and AID-Trees were pursued.

These efforts were generally unsuccessful, resulting in first-order and second-order equations that could account for only 20-25% of the variability and AID-Trees that required 61 splits (62 mutually exclusive groups) to account for 60% of the variability.¹ The primary reason for the poor quality of these early models was assessed to be structural deficiency in the models. The actual decisions had been made considering the Credit Rating of the applicant. Attempts to model these actual decisions without using Credit Rating as a predictor represented omission of the single most important variable in the judgment process.

A second task of the preliminary analysis was to determine if a better fitting model (in terms of unexplained variability) might be obtained by using a continuous criterion. To this end, the loan officers had been asked to rate each of the applications on a scale

And a second second second second

¹ Throughout the initial analyses, discussed in Section I, Chapter VI, the square of the multiple correlation coefficient  $(\mathbb{R}^2)$  was used as the criterion of goodness for making comparisons and decisions. Later the hit rate (see Chapter IV, page 84) was adapted as a morappropriate criterion and used for the remainder of the research.

of 1 to 5. Figure 6-1 shows the comparison of the variability explained as a function of the number of mutually exclusive groups for both the binary and scaled criteria. Only Judge 1 and Judge 3 had processed a sufficient number of the original applications to perform the analysis, but in both cases their scaled criterion values were less predictable than their binary decisions. Review of the residuals for each of these approaches revealed that most of the added variance experienced in the scaled rating models was attributable to those applications that were denied, possibly indicating a lower thresholding effect in the judges' policies, i.e., a good application had measurable levels of "goodness" but a bad application was just "bad". These results would suggest that having the judge rate or rank the data cases in a manner different from their natural decision process would tend to produce erroneous policies models.

Another exploratory approach involved attempting to define "factors" entering into the judgment. In this effort Factor Analysis computer routines were used to combine 13 predictors into five "factors". These five factors were capable of explaining some 60% of the internal variability within the predictor set. However, the resulting combinations of "ariables in each of the factors were not particularly meaningful to the loan officers who stated that they really thought of loan applications in terms of the following four factors:

- 1) Capacity: capability to service debt
- 2) Character: stability of applicant
- 3) Credit: past credit record
- 4) Condition: particulars of loan and risk to lender.



FIGURE 6-1

Since one of the underlying objectives of the research project was to model known policies of the loan officers in meaningful terms, the "factors" that were mathematically defined were discarded and not considered further.

The analysis and comparison of the AID-Trees for each of the dealers, on an individual basis, did result in illumination of an important point. The trees showed an apparent difference in the policy of the lending institution toward each of the dealers. In the case of Dealer A, only 52% of the applications were approved while 75% of the applications from Dealer B were approved. Further, AID-Trees reflected that different variables were important in explaining the variability in each of the policies. For Dealer A, the most important predictors, as indicated by the first iew splits of the AID-Tree, include Bank Account, Age, Equity, and Income. For Dealer B, the most important predictors were Residence, Employment Type, Loan Term, and Local Family. This apparent difference was discussed with the judges and they reflected that there actually was a difference in their attitude toward the different dealers. They noted that this difference was not explicit but was implicit and arose from their working relationship will the individual dealers and the types of applications they received from each. Specific causes of the differences were:

0

1.12

- Dealer B has a more affluent and mobile clientele than did Dealer A.
- 2) They had been operating with Dealer B for a much longer time and they expected that some pre-screening of the applications, based on their accept/reject experience, occurred before the applications were forwarded.

121

----

3) Dealer A was more helpful in recovery and disposal of repossessed merchandise and so they were relatively more interested in the variables that reflected the applicant's "capability" to pay than they were in those reflecting his "credit" or "character".

Preliminary analysis for each of the three senior loan officers did indicate differences in the variables each considered the most important and served to convince them that the AID-Trees were a viable means of portraying and analyzing their policies. For example, Figure 6-2 reflects the success rates for each of the categories of the predictor Marital Status for the three senior loan officers. It indicates that being single is somewhat of a detriment in the eyes of Judge 1 while being divorced is a detriment from Judge 3's viewpoint. It is interesting to note that during the formulation of the predictor categories, all three judges had expressed the opinion that a married applicant had a better chance of getting a loan, but their responses reflected this to be true only in the case of Judge 1. However, a cautionary note is in order here! The assertion of a given attribute being "better" should be interpreted in the contexts of "all other things being equal". Hence, the zero-order correlation between "married" and "approval" may not reflect the true importance of the marital status predictor in the situation where there are other intervening predictors that are not independent of marital status. Nevertheless, the use of plots such as Figure 6-2 was found to be a good catalyst for discussion of the individual loan officer's policies.

Although differences between judges and dealers were detectable in the early analysis, the inability to use only historical



- 「東京市市である」

т. Д.



FIGURE 6-2

decisions to obtain models that were sufficiently predictive caused considerable consternation and re-analysis of the decision process and modeling methodology. Three basic problems were identified:

 $b_{12}x_{1}x_{2}$ 

 The transformation from an AID-Tree into a configural effects regression model requires a different conceptualization of interaction among the predictors than is indicated by the standard interaction term of the form

124

- 2) Decision processes probably cannot be modeled very successfully if all of the important variables that entered into the original decisions are not included in the model. For the loan officers, there is no way to partial out the effect of Credit Rating from the actual responses that had been made with it being considered.
- 3) In situations where the number of data points is relatively small compared to the number of predictor categories (389/53 in this data), uncontrolled variation in the input data used by the judges must be minimized to improve the "signal" to "noise" relationship.

The solution to these three problems was pursued in two directions, the first being the methodological development as discussed previously in Chapter IV; the second being the generation of the coded predictor data and the re-evaluation of this data by the three original senior loan officers and two junior loan officers (Judge #4 and Judge #5).

The re-evaluation of the applications based on the predictor "descriptors" was undertaken for two purposes:

1) to eliminate from consideration any extraneous

information that had appeared on the original application and thereby assess the effect of controlling the information on the application form, and,

 to facilitate interjudge comparisons by allowing comparison of each loan officer's opinion on each loan.

This procedure reflects adherence to Hoffman's (1960) principle of experimental control. Although the raw historical data did not permit definition of adequate policy models, it did provide a judgment data set reflective of the population of actual decision cases that are encountered in the real world.

Prior to regeneration of the predictor data, further refinement of the predictors was accomplished with the final set of predictors consisting of the 16 variables shown in Exhibit 3, Appendix B. The refinement was accomplished on the applications from Dealer A. Of the original 262 applications, only 224 could be retrieved from the installment company's files due to the fact that the files had been relocated and reorganized and some accounts had been closed out during the period of the initial analysis. This situation in itself reflected one of the difficulties of performing a Policy Capturing analysis with the use of actual data from a functioning organization. In order to cause a minimal disturbance on the on-going system, the original applications had been left in the system and were not totally accessible at a later date.

#### Decisions Based on the Codified Data Set: (Without Credit Rating)

Each of the judges was given a set of the Codified Predictor Sheets with all predic ors except Credit Rating and asked to either grant "tentative approval" or "reserve judgment". During this process, considerable discussion resulted relative to which predictors were important and which were not important. This was one of the particularly interesting and beneficial aspects of the study in that it represented the first time all of the five loan officers had extensively discussed their philosophies and policies relative to evaluating loans.

Since the objective of the decisions made without benefit of the Credit Rating was to identify only the best prospects for "tentative approval", the loan officers defined seven conditions, any one of which would automatically prohibit tentative approval regardless of the overall quality of the application. These conditions were:

- 1) under 21 years of age
- under 2 years residence time and under 2 years job time
- 3) over 5 years residence time and still renting
- 4) over 25 years old without major credit references
- 5) under \$400 income
- 6) no bank account
- 7) "other" financial reference (code 1)

These conditions reflect a conservative bias on the part of the judges since they were viewed as significant weaknesses in the applications that made a positive decision, in the absence of a credit rating, unduly risky. These rules were implemented in the models by defining a binary predictor that indicated presence of one or more of the seven conditions. For want of a better term, this predictor was labeled "Dings". Use of Dings as a predictor allowed much better fitting models for each judge's "without credit rating decisions". However, the match between the decisions predicted by the models and the actual approvals/denials was still not as good as desired.

ť.

ю.

There was considerable variation in which loans the judges individually gave "tentative approval" and only 66% of the 224 decisions were unanimous. More importantly, only 20% of the applications received unanimous "reserve-judgment-without-credit" decisions. This is partly attributable to denying the judges use of their primary predictor and partly due to the fact that, in making the "tentative approval" decision, they were really making a different decision than they were normally accustomed to making. In an effort to determine if a composite evaluation might be better for isolating just the very best prospects, various voting strategies were investigated. It was found that a strategy based on granting tentative approval if four of the five judges had individually given "tentative approval" to the application and did the best job of identifying the top 25-30% of the applications while minimizing the inclusion of applications which had actually been denied. The criterion of 25-30% was arrived at based on discussion with loan officers relative to what percentage of "tentative approvals" would be required to make a policy model beneficial to them. Voting models considering the "Dings" reflected some over-conservatism in that only 71 out of 224 cases would have received tentative approval and 11 of these 71 had actually been denied. Voting models that did not use "Dings" as a
predictor performed approximately the same since 74 of the 224 cases would have been tentatively approved with 12 mistakes. The regression coefficients for these models are presented in Table 6-2.

Again, one of the primary results of this effort came from discussion and clarification, in the judges' minds, of which attributes were indicative of abnormal risk.

Decisions Based on Codified Data Set: (With Credit Rating)

Each judge was given Codified Predictor Sheets which included the Credit Rating for each of the applicants. They were asked to make their decisions as if they were final approvals or denials on the loan. This set of decisions provided the data for comparison of the five judges on an inter-judge basis. Decisions on the Codified Predictor Sheets were made twice by each judge, thus providing information on the reliability (percent of decisions that were the same on both trials) for each judge. Comparison of the decisions made on each set of codified predictor data with the actual decisions, that had been made previously from the original application form, provided insight into the amount of "information" lost in the process of codifying the predictors. Table 6-3 contains the hit-table for the coded versus actual decisions and the reliability of each of the judges. Since each of the original applications had been evaluated by only one of the three senior judges, the matches between the original decisions of each judge and his decisions on the corresponding sets of codified predictors is also shown.

### TABLE 6-2

:

μ

Ì

÷

.1

•

er le

4'' ''

ч. ' . чи

a

0

, 54 . .

9

### TENTATIVE APPROVAL MODELS

PREDICTED SCORE=CONSTANT +  $\sum_{i}$  COEFFICIENT i * PREDICTOR

$\mathbf{Pr}$	edictor:	Coefficient	
1.	Constant	-1.377	
2.	Age	0125	
3.	Marital	0542	
4.	Local	. 0355	
5.	Phone	, 2311	
6.	Jobtime	. 0536	
7.	Restime	.0387	
8.	Rescat	. 0918	
9.	Type Employment	. 0230	
10.	Income	. 0164	
11.	Bank Account	.1200	
12.	Financial Reference	. 3002	
13.	Loan Amount	. 0328	
14.	Loan Term	~ .0352	
15.	Necessity	. 0371	
16.	Equity	.0275	
17.	Dings	1200	

Model without dings includes only first 16 terms.

۳ ۱
Ó
មុ
B
A
н

# CODED VS ACTUAL DECISION STATISTICS (C/A)

Щ Ц	
IAB	
HIT-	

Reliability

					ļ			ĺ			
Judge	No!	No	Yes	/No	No	/Yes	Yes	Yes	%Coi	rrect	
Trial(1,2)	1	2	1	2		2	1	2	Ī	2	
-	106	112	18	11	19	25	18	75	84	84	(%06)
2	101	06	61	34	12	4	88	96	86	83	(%68)
ŝ	102	1 00	22	24	18	11	82	89	82	84	(%06)
4	93 93	66	[3	2 2 2	31	21	87	79	80	80	(30%)
ۍ ا	100	011	24	14	13	24	87	-16	84	84	(85%)
"4/5": VOTE	110	1	14	:	17	:	83	;	85	;	

(Based on Judge Making Original Decision)

% Correct	86.5%	85.0%	87.5%
<b>Cased Decided</b>	112	24	5.8
Judge		2	ŝ

130

50

----

The key point derived from the comparison of the judges' decisions made on the codified predictors with their original decisions based on the full application is that codifying the data only slightly affected the decisions. This assertion is made since the reliability of each of the three senior judges was approximately 90 percent and the correspondence of their decisions based on the two forms of data ranged between 85 and 87.5 percent.

The inclusion of the Credit Rating predictor and the refinement of the categories of the other predictors resulted in attainment of a vastly improved fit to the data. Figure 6-3 shows the AID-Tree developed using the actual decisions as the criterion. In the total tree², nine splits occurred leaving ten mutually exclusive groups which accounted for 76 percent of the variability in the data.

Table 6-4 shows the ranked importance for the variables relative to their power to explain variability over the total predictor space and also over the predictor subspaces defined by the AID splitting process. The individual predictors are ranked according to their average BSS/TSST and BSS/TSS(i) values. The data in this table would indicate that CREDIT RATING, FINANCIAL REFERENCES, BANK ACCOUNT, LOAN TERM, TYPE EMPLOYMENT and PHONE are the most important variables in the decision process. Since they hold ranks 1 through 6 in both rankings, they are important for all applicants. MARITAL STATUS, LOCAL FAMILY, and NECESSITY hold very low ranks in both rankings and are the least important predictors.

Only eight of the splits actually appear on Figure 6-3 due to limitations in this version of the tree printout. The expanded version of the AID-Tree shown in later examples can portray 12 levels of the splitting process.



3.0

6



FIGURE 6-3

RANKED IMPORTANCE OF PREDICTORS

TABLE 6-4

55.4		12.0			16.0	с 5	11.0	14.0	7.0		9	9.6	2.0	0.0	0.0			15.0	10.0	-	•
•	24NV																				
<ul> <li>010T55T</li> </ul>		61-00-0	4	21	2	12	2	12			21	ž	0	<u>.</u>	~	•••	1	21	2		2
ROUND CRIT.	CON MEAN D			29.54225	7.55339	104.15052	47.59743	26.18343	74.046		45.21455	52,48242	24746.182	174.11977	44.03219		30 40 7 4 T T	16.31315	<li><li><li><li><li><li><li><li><li><li></li></li></li></li></li></li></li></li></li></li>		a.10++00+
•[6]534	540 1077 1077	LANJARCE FL	+ 0 0 0 0 °	.00002	.0000	.00022	.00015	ESUOD.		01004	000:2	.00009	.002.90	76100.	11000		22060*	10000.	10003		00013
IAND HEAN -		ULLE AND	1.03669	1.15499	1,32453	1,05099	1.91430	1.60114			.95840	1.33236	ACOLA.1	1.57607	1 4547		47104	1.17720			C+1+0*2
12 6A			.000620	£0400	00135	.01487	22107	14300		Sec 10.	660:0"	14900.	- 05064	13699			01410	.0259	A 7 8 7 4		.12452
UBGROUPS IS	RECONSTRUCTE	50×855/1551	P 2 7 1 0 *	.04772	01220	LAB24	01639			0712r*	.13766	.0847	+ 212				. 16458	26350.	0-24		567 <b>6%</b> .
VARIABLES, NO OF SI	SUB (VAR [ANT)		•07čl3	.04772	01220	1487	84470				.13766	34.460					.16920	02635			***[9*
LE 16				111		1				CAT	SF MD						M L Ng	.F 55			ERAT
RY TAU			AGE	44.9	i c				1 1 1 1	2	1 					ò	Ő			1	Ë
IVANDE				~		•		• •	Di	•	-	-		::	::	7		-		2	-

\$2°4	- M & Q N 4 - 1 4 0 N 4 E M 4 6 - 1 - M & Q N 4 - 1 4 0 N 4 E M 4 6 - 1 - M & Q N 4 - 1 4 0 N 4 E M 4 6 - 1
•	भ स र
0107557	64000 000 122 122 122 122 122 122 122 122
ROUND CHIS.	0000 1000 2000 2000 2000 2000 2000 2000
·510625	VARIANCE - 50 VARIANCE - 50 00047 00047 00047 00047 00059 00059 00059 00059 00054 00054 00054 00054 00054 00054 00054 00054 00054
RAND NETH ONE	COTFF VAR 1.07051 1.07051 1.07051 1.07051 1.07051 1.07051 1.07051 1.07051 1.07056 1.02056 1.02056 1.02056
21	4291 420 420 420 420 420 420 420 420
IRGPOURS IS	RECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONSTRUCT SECONS
JARTARLES, NO OF SU	918 1441 1441 1441 1441 1441 1441 1441 1
TANLE 16 1	2010 2010 2010 2010 2010 2010 2010 2010
LAVMANS	

ø

., /

: ; ,

ijste

e,e

. v

'n

¢,

. **.** *

**٦** .

<u>a</u> .

Section III.

596

₹*R*E

The Individual Policy Models:

Using the responses to the Codified Predictor Sets as the criterion, first-order, second-order, and AID-derived "local" models were developed for each³ of the five judges and for the four of five voting strategy.

The First Order Models.

The coefficients for first-order regression models of the form

Predicted Score = Constant +  $\Sigma$  coefficient_i x predictor_i

were obtained following the iterative model determination/cross validation procedure described in Chapter IV. Implicit in these models is the assumption that the predictor categories are ordered according to the codes given in Exhibit 3, Appendix B and that a constant change in the Predicted Score will result from a unit change in the level of a Predictor Variable. This assumption was made in an attempt to avoid idiosyncratic results that could result from fitting the 224 data points with a binary regression model which had 53 predictors. No attempt was made to test individual terms for statistical significance. However, it is apparent that any variable whose coefficient is smaller in magnitude

 $^{^{3}}$ Second-order models were only derived for Judge 1, 2 and 3.

than . 002 would have an effect on the total score of less than . 01 (since the range of values for the variables is between 1 and 5) and should probably be ignored. The first order coefficients for the 5 judges are shown in Table 6-5. The best cut score reflects that point where the maximum hit rate occurs.

The Second-Order Models:

Second-order regression models of the form

Predicted Score Constant  $\stackrel{!}{:} \Sigma \operatorname{coefficient}_{i} x$  predictor  $\stackrel{!}{i}$  $+ \Sigma \Sigma \operatorname{coefficient}_{ij} x$  predictor_i x predictor_j

were also analyzed. In the absence of substantive theory from which to hypothesize the terms in the second-order models, they were hypothesized by generating all possible first-order and second-order terms and using a stepwise regression routine to eliminate those terms not statistically significant at the .01 probability level. Due to computational limitations, the identification of potential terms was done incrementally with 16 first-order and approximately 45 second-order terms being considered in each of three runs. All terms identified by this procedure were included in the second-order model regardless of their "meaningfulness" or "interpretability". The iterative procedure was then used to determine the coefficients in Table 6-6 for the hypothesized model for each judge.

AID-Split Models:

AID-Split models of the form

:1

A Contraction of the second

TABLE 6-5

ţ

Ľ

### FIRST-ORDER MODEL COEFFICIENTS

<b>Predictors:</b>					
Judge	H1	#2	#3	#4	# 1
Constant:	-1.1178	7174	7337	6865	9802
Age:	01 98	. 0017	0183	0181	0298
Marital:	0672	0051	0070	0219	0485
Local:	.0810	. 0057	.0229	. 0773	.1168
Phone:	.1595	.1208	.0570	.1178	.0901
Jobtime:	. 0261	.0267	.0219	.0476	.0816
Restime:	0362	0051	0129	0449	, 0068
ResCat:	.0685	.0453	. 0161	.0427	.0563
Type Employment:	. 0327	135	. 0019	. 0261	.0184
Income:	0473	0208	0097	0336	0048
Bank Acct.:	.0506	.0753	. 0673	.0200	.0578
Financial Ref.:	.2033	.0794	.1086	.1102	.1645
Loan Amt. :	0002	0250	.0039	0036	.0087
Loan Term:	0403	0470	.0226	0601	0733
Necessity:	.0595	0526	. 0052	. 0387	.0104
Equity:	0013	.0121	0000	. 0330	133
Credit Rating:	.2400	.2950	.2873	.2820	.2274
Best Cut Score:	.4950	.47	.5052	. 37 41	.4041
Hit Rate:	16.	. 92	16,	. 88	. 90

••

136

< -- 1 m-

3 17 17 . . . . . · ... . J

> 0 =, :.

「「「「」」を

### TABLE 6-6

## SECOND-ORDER MODEL COEFFICIENTS

	4676 . 2872 . 0288 . 0144 0053	. <b>4748</b> .90
Judge #3	Constant Credit Rating Fin. Ref. * Bank Res Cat Res Cat	
Judge #2	Constant 5496 Phone . 0918 Bank Account . 0000 Financial Reference . 0139 Loan Term 0203 Credit Rating . 2890 Marital Jobtime . 0226 Marital Necessity 0215 Phone Bank . 0105 Bank Fin. Ref 0247 Loan Term Necessity 0213	.52
	<pre></pre>	. 43
Judge #1	Constant Phone ResCat Bank Account Financial Reference Credit Rating Phone Credit Phone Income Phone Income Phone Local ResCat Marital Loan Term ResCat Type Emp. Bank Acct. Necessity Fin. Ref. Credit Rtg. (Bank Acct. ) ²	Best Cut Score:

. 52

. 43

Best Cut Score: Hit Rate:

Predicted Score = 
$$\Sigma$$
 group mean_i x GM_i  
i = 1 if applicant is in group i  
where: GM_i = 0 otherwise

were obtained by analyzing the application data with the AID4UT program. The models as shown in Figures 6-4, through 6-8, represent the first 5 or 6 splits of the tree, at which point subspaces (Figures 6-9 through 6-13) were defined and local models were developed. In Figures 6-4 through 6-8, the group means are denoted by "A_i" and the split variable is denoted by " $X_j$ ". The subscript "j" represents the identification number of the split predictor per Exhibit 3, Appendix B. The super-scripts "(k,1)" denotes the categories of the split predictor that are in each group.

Local Models:

Local models of the form

Predicted Score = Constant +  $\sum_{i} GM_{i} \times (Local Model_{i})$ where:  $GM_{i} = 1$  if applicant is in subspace i = 0 otherwise

Local Model i = First or second-order model pertaining only to those applicants in subspace i.

were derived based on analysis of the AID-Trees. In the process of developing the local models, several alternative formulations within the various subspaces were investigated. The general result was that,

5

138

(



For Judge 1, group 2 and 4 were combined into a common subspace due to the homogeneity of their mean criterion values.

FIGURE 6-4

.50

Contraction in statute

÷,



•



1.

### FIGURE 6-6



2

FIGURE 6-7





 19 19 $A_{18} = 80$   $N_{18} = 20$ 



within the subspaces, first-order models tended to cross-validate better. The coefficients for the local models were obtained by the iterative procedure given in Chapter IV. The local models for the subspaces shown in Figures 6-9 through 6-13, are given in Tables 6-7 through 6-11.



، , ,

r

. .

<u>ب</u>

N

ť

### Table 6-7

Local Model Coefficients: Judge 1

Constant = -.6618

Subspace 1: "Poor" Credit Rating or "Other" Financial Ref. (GM1=1) local model 1: PS = .6687 (GM1) + constant

Subspace 2: "No Bank" and not in subspace 1 (SS1) (GM2=1) local model 2: PS = .3201 (Credit Rating) + .0722 (Phone) + .0600 (ResCat) + constant

Subspace 3:"Good" Credit Rating and not in SSI or SS2<br/>(GM3=1)local niodel 3:PS = 1,639 (GM3) + constant

Subspace 4:<br/>(GM4=1)Not in SS1, SS2, or SS3local model 4:PS = .2258 (Local) + .2044 (ResCat) + .0901<br/>(Type Employment) + .0818(CP1) + .3312<br/>(GM4) + constantwhere: $CP1 \begin{cases} 1 \text{ if Restime } \leq 2 \text{ and ResCat} \leq 3 \\ 0 \text{ otherwise} \end{cases}$ 

Best Cut Score: .58-.61 Hit Rate: .929 SUBSPACES: JUDGE 2

CONTRACTOR OF A

. š

4

١

ς.

· .-

7

. .

2

е. 1.

د. ا

. .

۰۰,

7.

11.11

, ²t

·· ,

ţı

1

.

...

æ



FIGURE 6-10

147

.

### Table 6-8

Local Model Coefficients: Judge 2

### Constant = .760

Subspace 1:	"Poor" Credit Rating
(GM1=1)	
local model 1:	<b>PS</b> = 7619 (GM1) + constant

Subspace 2:"Unrated" Credit Rating and No Checking<br/>(GM2=1)(GM2=1)Accountlocal model 2:PS = -.6642 (GM2) + constant

Subspace 3:"Unrated" Credit Rating with at least "Checking"(GM3=1)Accountlocal model 3:PS = .0910 (Phone) -. 3271 (Jobtime) + .0879(Loan Term) + .1632 (Financial Ref.) -.0341(GM3) + constant

Subspace 4:Credit Rating "Medium" or "Good" and<br/>(GM4=1)local model 4:"No-Bank" AccountPS = -.1639 (Type Employment) -.1278 (Income)+.1602 (Financial Ref.) + constant

Subspace 5:Credit Rating of "Good" or "Medium", "Savings"(GM5=1)or "Checking" and "Major" Financial Referencelocal model 5:PS = .211 (GM5) + constant

Subspace 6:Not in any of the other subspaces(GM6=1)Iocal model 6:Iocal model 6:PS = .2471 (ResCat) + .0159 (Loan Term)-.3475 Loan Amt. -.0099 (GM6) + constant

Best Cut Score:	. 46-	49
Hit Rate:	. 929	

光

N. 5'

in State

1



SUBSPACES: JUDGE 3

b

•

. .

149

FIGURE 6-11

### Table 6-9

Local Model Coefficients: Judge 3

Constant = -.2012

Subspace 1:	"Poor" Credit Rating
(GM1=1)	
local model 1:	PS = .2106 (GM1) + constant

Subspace 2: "Unrated" Credit Rating and "No Bank" (GM2=1) local model 2: PS = .2426 (GM2) + constant

Subspace 3:"Unrated" Credit Rating and "Checking" or<br/>(GM3=1)local model 3:"Savings"local model 3:PS = .0079 (Phone) + .1099 (Fin. Ref.)<br/>+ .2973 (Necessity) + constant

Subspace 4:	"Medium" or "Good" Credit Rating and "Other"
(GM4=1)	Financial References
local model 4:	PS = .1165 (Bank Acct.) + .0489 (Credit Rating) + .3720 (CP1) + constant
where:	$CP1 \begin{cases} = 1 & \text{if Jobtime is } 2 \text{ years} \\ = 0 & \text{otherwise} \end{cases}$

Subspace 5:	"Medium" or "Good" Credit Rating and "Minor"
(GM5=1)	or "Major" Financial References
local model 5:	PS = .2323 (Phone) + .0571 (Type Employment)
	+ .1641 (Fir. Ref.) + .0180 (Credit Rating)
	+ constant

Best Cut Score: .53-.65 Hit Rate: .92

1. . . . .

÷

e

مع الم

.

۰.

٢.

•••

.

*"*,

viða. L

ł.

5.0



Statute - ALIA STREET



1

4

ð.

**.** 

1516.1

ļstir ļ

0

5

### Table 6-10

Local Model Coefficients: Judge 4

Constant = .0546

Subspace 1: (GM1=1)	"Poor" Credit Rating			
local model 1:	PS =0199 (GM1) + constant			
Subspace 2:	"Unrated" Credit Rating and "3 year" Loan			
(GM2=1) local model 2:	PS = .0600 (GM2) + constant			
Subspace 3: (GM3=1) local model 3:	"Unrated" Credit Rating and "2 years" or less Loan Term PS =1647 (Local) + .1537 (Income) + .0374 (Bank Acct.) + .2132 (Fin. Ref.) + constant			
Subspace 4: (GM4=1) local model 4:	"Medium" or "Good" Credit Rating and "Other" Fin. Ref. PS = .1479 (Type Employment) + .0761 (GM4) + constant			
Subspace 5: (GM5=1) local model 5:	"Medium" or "Good" Credit Rating and "Minor" or "Major" Financial Reference PS = .4337 (Phone) + .0553 (ResCat) + .0692 (Credit Rating)3508 (GM5) + constant			
Best Cut Score: Hit Rate:	. 53 57 . 92			

. **4**11

- 14

,*#F7?

advie

SUBSPACES: JUDGE 5

• • •

o .,

.

t.

. .

*з*,

ši r

۰,

÷

а,

. . <u>.</u>

. . .

<u>,</u> .,

ک نو

ь÷,

р . ц

. •

t

..



FIGURE 6-13

and and a second se

153

### Table 6-11

Local Model Coefficients: Judge 5

Constant = -.5003

"Poor" Credit Rating

"Unrated" Credit Rating

Subspace 1: (GM1=1) local model 1: where:

PS = .5132 (GM1) + .6865 (CP1) + constant  $CP1 \begin{cases} = 1 & Jobtime = 5 & years and "Major" Fn.Ref. \\ = 0 & Otherwise \end{cases}$ 

Subspace 2: (GM2=1) local model 2:

:  $PS = -.2696 (GM2) + .2144 (Marital) + .0452 (Jobtime) + .0927 (Fin. Ref.) - .0707 (Phone) + .2936 (CP2)(Phone) + constant CP2 <math>\begin{cases} = 1 & \text{if "checking" or savings for Bank Acct.} \\ = 0 & y \end{cases}$ 

where:

Subspace 3: " (GM3=1) o local model 3: F

"Medium" or "Good" Credit Rating and "Minor" or "Other" Financial Ref. PS = .1153 (Local) -.033(Phone) +.1616(Restime) +.0950 (Credit Rating)(Fin. Ref.) +.0220 (ResCat) + constant

Subspace 4:	"Medium" or "Good" Credit Rating, "Major"
(GM4=1)	Fin. Ref. and "Less than 2 years" Jobtime
local model 4:	PS = . 5221(Phone) + . 1260 (Type Employment)
	0131(GM4) + constant

Subspace 5: "Med (GM5=1) Fin. local model 5: PS =

"Medium" or "Good" Credit Rating, "Major" Fin. Ref. and "Over 2 years" Jobtime PS = 1,5011 (GM5) + constant

Best Cut Score: .91 Hit Rate: .44-.47

٢\$

### Section IV:

### Comparison of Models:

For the first-order, second-order, and local models, the model coefficients shown in Tables 6-5 through 6-11 were used to produce a prediction hit/miss table of the type shown in Exhibits 16 through 18. Appendix B. This provided the data to construct error curves reflecting the number of false approvals versus false denials for each possible cut score between .0 and 1.0. The curves in Figures 6-14 through 6-18 reflect the profiles of these error rates for each model type and each judge. The points plotted on the curves reflect the error rates at cut scores where both the level of false approvals and the level of false denial change. Thus, each point on the curve is a local optimum cut score since it reflects the minimum false approval rate attainable at a particular false denial rate, or vice versa. The important thing to note from these curves is that the curves for the local models dominate the curves for the first-order and second-order models; that is, at any given false approval rate, the local model for the judge will result in the same or fewer false denials than his first-order or secondorder model.

In this sense the "goodness" of the model is being measured relative to the model's predictive capability over the whole spectrum of possible cut scores and not solely on the model's maximum hit rate.

Comparison of Judges:

Comparison of the judges' policies solely on the basis of





ί.,

FIGURE 6-15



ERROR CURVES FOR JUDGE 3



-



a .

.

the magnitude of the coefficients for the predictors is not very informative when the predictors are correlated, as is the case in this decision process. Models with greatly different coefficients can result in very nearly the same set of decisions. For example, Table 6-12 presents the similarity ci the decisions of the five judges in terms of the percentage of matches. Thus data indicates that all of the judges' decisions are the same between 87 and 91 percent, yet Table 6-5 indicates a fairly wide disparity among the coefficients. Note that for Judge 2 and Judge 5, four of the coefficience a = 1 for a and b are the the form b and b and

ي. دور

*

### TABLE 6-12

INTERJUDGE	SIMILARITY
Income	(ant)

	محمد المراجع المراجع	الكارتين المنتدما			entre test materies		55	
		2	3	4	5	4/5 Vote		
1	100	89.3	87	87	87	94		
?		100	89.2	89.8	91	94.5		
3			100	88.5	86.4	91.7		
1				100	89.8	90.6		
5					100	93		

Similarly, comparison of the s-coad-order (Table 6-6) coefficient does not reveal much useful information.

Comparison of the Aid-Trees for each judge (Figures 6-4 through  $\leq 3$ ) reflects that Judges 1 through  $\leq$  are primarily concerned with Gredit Rating  $(X_{16})$ , Financial References  $(X_{11})$  and Bank Account

 $(X_{10})$ . However, Bank Account is not a primary variable for either Judge 4 or Judge 5 while Loan Term  $(X_{13})$  is important to Judge 4 and Time on Job  $(X_{r})$  is important to Judge 5.

Further, the Aid-Tree identifies a Financial Reference of "Other" as a sufficient condition for denial, regardles of Credit Rating, for Judge 1. Yet Judges 3 and 4 approved 47 percent and 53 percent of applications if they had a "Medium" or "Good" Credit Rating. Careful study of the Aid-Tree can suggest many such differences. However the question must always be asked, "Is the difference due to policy or is it due to the computational scheme?" During the research it was found that a very useful technique was to present each judge his tree and have him verbalize or explain the splits. This led to exposure and discussion of many subtle differences in opinion among the judges. Section V:

 $\overline{2}$ 

The Voting and Composite Models:

As was the case in the "without credit" models, a voting strategy of 4/5 appeared to best match the actual approvals and denials. The last column of Table 6-12 reflects that the decision of each of the individual judges matches voting policy better than they match the decisions of any of the other individual judges.

An alternate composite model was also obtained by the more traditional process of clustering first-order models of the individual judges with the EDSTAT-J version of the JAN hierarchial clustering routine. The coefficients for the 4/5 vote and clustered composite models are shown in Table 6-13. Figure 6-19 shows the merge diagram from the clustering process. This diagram reflects the policy comparison technique that has been generally used in the past. It reflects that Judge 1 and Judge 5 have the most similar policies, yet their hit/miss ratio shown in Table 6-12 is one of the lowest.

This paradox is due to the nature of the data and the fact that the criteria for closeness are different. In the clustering approach, the sum of squared errors between actual score and predicted score is the criterion. In the hit/miss approach, the fact of a difference is important, but the magnitude is not (all "miss" magnitudes are 1 and "hits" are 0). The whole point of presenting and discussing the merge diagram is that, with binary decisions, the use of continuous criteria such as the squared multiple correlation coefficient ( $\mathbb{R}^2$ ) or the sum of squared errors is probably not desirable and can lead to vastly different results.
## TABLE 6-13

## COEFFICIENTS FOR 4/5 VOTE AND COMPOSITE MODELS

Predictors:

<u>,</u> **F** 

... ,

,е --

••

, , ,

s , je

.

•

۰. .

·

ar e Q

.

.

.

8.1

# Coefficients

		Cluster ed
	4/5 Vote	Composite
Constant	-1.0469	8615
Age	0268	0201
Marital	0383	0344
Local	.0700	.0610
Phone	. 1393	.1053
Jobtime	. 0322	.0456
Restime	0308	0279
ResCat	.0830	. 0527
Type Employment	. 0151	. 0174
Income	- ,0225	0276
Bank Account	. 0655	. 0584
Financial Reference	.1538	.1279
Loan Amount	. 0072	.0128
Loan Term	0585	0571
Necessity	. 0061	0146
Equity	.0066	-,0093
Credit Rating	. 2647	.2622

Best Cut Score:	. 46	. 27	
Hit Rates	. 94	- 92	

Since the 4/5 vote policy model actually represented a set of substitute responses for the actual decisions or the individual judge's decisions, it was used to build an Aid-Tree (Exhibit 13, Appendix B) and to define subspaces and local models. Although the superiority of the local model relative to the first-order model still in 1d up in terms of the error curves shown in Figure 6-20, this superiority was not as great as it was in the case of the individual judges. This decrease is understandable since the 4/5 vote policy tends to average the policies and models of the individual judges, and in 40 doing, obscures the configuralities and idiosyncracies of the individual policies. Thus the power of the local models in modeling the various subspaces is diminished since the boundaries of these subspaces are not as well defined.



(mint



í.

8.

FIGURE 6-20

#### Section VI.

° . .

•••

h

 ${\mathfrak a}^{\mathfrak a}$ 

## Results of the Cross-Validation Field Test:

## A Shift in the Sample Characteristics and the Policy:

Of the 150 new applications collected during the field test, 75 percent were given loans without the stipulation of "full recourse". This reflects an increase of 30 percent in the approval rate between the 224 applications used in the model building phased and the 150 applications of the cross-validation population. This increase is attributable to two basic factors that are detectable in Table 6-14.

#### TABLE 6-14

Credit Rating	Original Sample			Cross-Valination Sample		
	#	Percent of Sample	) 'erceot Approval	ŀ	Percent of Sample	Percent Approval
Poor	76	34	0	27	18	7.5
Unequal	4.0	17	40	27	18	66
Medium	46	2.2	5 fg	3.6	11	8)
Good	62	27	95	-8≎	۰ <u>,</u> ۵	99
Totals	224	100	45	156	4.0.2	75

#### APPROVAL RATE vs. CREDIT RATING CATEGORY

First, there was an apparent increase in the caliber of the applications relative to the primary decision variable, Credit Rating. Second, there was an increase in the approval rate for all categories of Credit Rating.

There are two probable explanations for the increase in the caliber of the applications:

- sampling differences between the random sample of
  224 and the total July 72 population of applications used
  tor cross-validation
- 2) an actual shift in the credit credentials of the potential borrower during the month of July 72.⁴

The increased approval rate reflects a change in the relationship between the dealer and the lending institution. This change is attributable to the longer time that they had worked together by the time the cross-validation sample was taken. Whereas the original sample reflected their relationship after only 4 to 5 months of the dealer contract, they had been doing business approximately 14 months at the time of the second cample and the trust and understanding between them was greatly improved. During this period, the volume of business from Dealer B had grown by some 400 percent and the loan officers expressed a willingness to accept some what riskier applications in order to keep the dealer happy and continue the expansion of business. This change is especially noticeable in the applications with

^{*}Discussion with Judge 2 revealed there is a functuation of the credit quality that is a function of the general economic conditions. When business is slow, a general decrease of credit quality prevails.

"Unrated" and "Medium" Credit Rating.

## "With Credit" Policy Models vs. Decisions on the Codified Predictors:

One result of these shifts was to improve the performance of the policy models and lower the cut scores corresponding to the optimum hit rates. As can be seen in Table 6-15, the cross-validation hit rates for the judges' decisions, based on the codified predictor sets, were generally be, er than the hit rates on the original sample. This

# Table 6-15

## MAXIMUM HIT-RATES

Equation	Original Hit-Rate	Cross Validation Hit-Rate
Judge 1first order	. 91	. 96
Judge 1second order	. 92	. 96
Judge 1local model	. 93	. 93
Judge 2first order	. 92	. 96
Judge 2second order	. 92	. 97
Judge 2local model	. 93	. 93
Judge 3first order	. 91	. 95
Judge 3second order	. 90	. 95
Judge 3- Jocal model	. 92	. 95
Judge 4first order	. 68	. 95
Judge 4) - local model	. 92	. 94
Judge 5 - first order	. 90	. 96
Judge 5 - local model	. 91	. 93
4/5 Vote stirst order	. 94	. 97
4/5 Vote local model	. 94	. 97

table also reflects that the first-order and second-order models tended to do slightly better than the local models in terms of cross-validation. Given the observed shift in policy, concentrated in the Credit Rating categories of "Unrated" and "Medium", this is not surprising. These are the predictor categories that defined the subspaces over which the local models were the most complex and influential. The superiority of the local models (as shown in Figures 6-14 to 6-19) was mainly due to the better fit of the local hyperplanes over these subspaces. This superior fit would naturally be most affected by perturbation of the policy in those subspaces.

#### "With Credit" Policy Models vs. Actual Decisions:

ĺ.

5

心理

л.

ъ.,

The various composite models were cross-validated against the actual approvals and denials that the loan officers had made using the original application forms (Exhibit 1, Appendix B). Exhibits 16, 17, and 19 present the appropriate hit/miss tables and Figure 6-21shows the error curves for the various policy models. The first-order "4/5" vote and composite models each attained 92 percent accuracy. The local "4/5" vote model was 91 percent accurate. Comparison of the first-order voting model's performance in predicting decisions that were made on the codified predictors with its performance in predicting the actual decisions based on the original application form reflects that there is only a 5 percent difference. This is somewhat less than the difference between using codified predictors and actual applications on the original 224 as reflected by the data from Table 6-3. The decrease probably reflects the judges' ad ptation to the use of single category descriptors through their participation in the project. In any event, the difference between the results based on the codified predictors



and the results based on the natural language applications does not appear to be a significant problem in construction of policy models for this decision process.

### Clerical vs. Expert Measurements:

1

¢.,

ţ

÷

-1

•

ź.

2

ş

....

One of the objectives of the field test was to ascertain the capability of the clerical staff to properly codify the application data. To this end, the set of instructions in Exhibit 6, Appendix B, were explained to the clerks and left for their reference in filling out the applications. Their questions were solicited and answered the first day of the field test. After each clerk expressed confidence in her understanding of the instructions, there was no further monitoring of the process of translation from the original application into the codified predictors. However, the judge who made the decision on the application was also asked to fill out the coded application form. This provided a means of comparing the "clerical" measurement with the "expert" measurement. An item-by-item comparison on each of the applications was made and the results are displayed in Table 6-16. The wide disparity in the assessment of the predictors is rather amazing, especially in the case of those predictors which should have required absolutely no judgment. A check of both translations with the original applications revealed the loan officer was generally correct in cases where a difference existed, but not always. Figure 6-22 reflects the effect of this disparity on the decisions predicted by the "4/5" voting model. The lack of a large effect in the predictions is attributable to the fact that the primary predictor in the model is Credit Rating, and this was only available to, and assessed by, the

#### Table 6-16

# EXPERT VS. CLERICAL MEASUREMENT DIFFERENCES ON APPLICATION VARIABLES

÷

÷.#

1,

ь

. . .

. 10 ^{- 15}

. .

ч Л.,

·, ç,'

۰.

į.

`о ,

4

; 'e 3

șt.

Predictor	Cases Assessed Differently*		
Age	10		
Marital Status	7		
Local Family	25		
Telephone	12		
Job Time	29		
Residence Tim .	29		
Residence Category	26		
Type Employment	61		
Income	52		
Bank Account	40		
Financial References	69		
Loan Amount	32		
Loan Term	16		
Necessity	18		
Equity	35		
Credit Rating	Only assessed by loan officers		

*At least one item differed on 141 of the 150 oplications.

loan officer. In the case of the "tentative approval" model. Credit Rating was not used. A comparison of Figures 6-23 and 6-24 reflects that misassessing the predictors does have a substantial effect. In either case, it would appear that some formal training of the clerical staff would be required to insure that their assessment of the predictors is congruent with that of the loan officers.

### "Without Credit" Policy Models vs. Actual Decisions:

The shift toward leniency in the policy of the judges had a positive effect on the observed performance of the policy models used



と言語がない

to predict "tentative approval". Figure 6-23 reflects that if everyone scoring .85 or above had been approved, over 48 percent of the applications could have been "tentatively approved" with only a 3 percent error rate. This is well above the 25-30 percent criterion set by the loan officers as a minimum limit for usability. Further, in assessing the mistakes that occurred, it was found that 2 of the 3 false approvals, made by the model at the .85 cut score, had actually been granted under the "full recourse" option. Figure 6-24 indicates the performance of the model with clerically translated predictors is significantly worse and would probably be unacceptable. Figure 6-25 shows that the model which includes "Dings" is somewhat more conservative, but does perform well in terms of error rate. As for implementation, the use of the more conservative model would probably be the better option in order to assure acceptable performance when the overall credit quality is lower.

### First-Order vs. Local Models:

Ł

5

The reversal of the first-order and local models in terms of cross-validation hit rate would indicate that the local model is more sensitive to policy changes, especially if the change only affects certain subgroups of the applicants. However, the local model is more robust in the sense that it has a better profile of hit rates over the entire range of application scores, and would be less sensitive to general changes in the policy affecting the entire spectrum of applications. Figure 6-26 presents the profile of hit rates on the crossvalidation sample for all possible cut scores. The first-order models enjoy a local superiority at a given cut score; the local models do a better "average" job of predicting the decisions.



÷¥

FIGURE 6-23



FIGURE 6-24

.







4^d

Section VII.

Implications for Implementation:

The results of this modeling effort indicates that there are several applications of the Policy Capturing technique that would be beneficial to the installment loan decision process.

#### Training:

The discussion and interaction that the process stimulated among the loan officers was felt to be very beneficial. In a post project interview, there was general agreement among the loan officers that the models helped them to understand how their colleagues thought and to illuminate some of the more subtle aspects of their own policies. One specific example of this improved communication was the definition and clarification of the concept of Financial Reference. A second example was best expressed by one of the junior loan officers, who said:

> I really did not know how the three senior loan officers evaluated the applications, other than on past credit, until they discussed their 'trees'.

It would appear that having a new employee make decisions on a standard set of loan applications and then capturing his policy to compare with those the existing group of loan officers would be a simple and effective method of evaluating his compatibility. An example of this was encountered when the transferrability effort was pursued. The policy of a new employee with considerable past experience as manager of another loan company was captured. The results reflected that he would have given some 38 percent of the applicants with "Poor" Credit Rating loans, while the corresponding success rate for the "4/5" vote was 0 percent. Further, his understanding of the newly defined predictor of Financial Reference and his concern with Bank Account were not commensurate with the attitudes of the other loan officers. The policy capturing exercise made early identification of these differences possible.

#### Mechanization of Prescoring or "Surrogate" Judgment:

The performance of the "with credit" models appears to be sufficient to allow their use either as an aid to the loan officer, or to automatically approve loans. Although Dawes and Diller's numerical criterion for indicating "bootstrapping" will occur does not appear to be directly applicable in the case of binary decisions, the comparison of the hit rates of the models (See Table 6-15.) with the reliability of the judges (See Table 6-3.) would indicate that any of the models would provide some aid.

Procedures for implementing the models could be manual or computerized. Any of the equations previously presented could be accessed through a computer terminal or stored on a programmable desk calculator to provide the application scor  $\ldots$  Figures 6-27 and 6-28 present the risk curves that could be used by the loan officer as he considered an application. On those applications that do not contain significant extenuating circumstances, a fixed cut score could be used for automatic approval or denial. Figure 6-28 indicates that by evaluating applications with the local "4/5" voting model, using a rule of granting loans which scored over . 06 and denying loans that scored less than . 04 would represent minimum risk and yet leave a very small



FIGURE 6-27

۰,

15 2.11

<u>к</u> 5

8 6

"<u>:</u>

11.12 1.51

ð.

κ.



range of ambiquity.

One ancillary advantage of using the policy models is that the loan officer could get a consensus evaluation from his colleagues by using their models along with his own.

## "Tentative Approval" During Non-banking Hours:

The performance of the "tentative approval" models indicates that a definite improvement in the level of service the lender can provide to the dealer could accrue. The motivation for use of these models lies in the speed advantages that the use of the models provides. Figure 6-29 shows the distribution of times that the cross-validation applications spent in the "system" and the corresponding times they spent in the "credit system". The difference in the average times, 14.3 hours, is largely due to the lack of access the dealer has to the lender during non-banking hours. The tentative approval of applications could have eliminated this waiting time for 48 percent of those "after-banking-hours" applications submitted during the field test. This is particularly applicable to sales made on Friday afternoon or Saturday, but which must wait until Monday for consideration.

Again, either manual or computerized techniques could be used. In the simplest system, a manual of predictor categories with application scores (similar to a telephone book) could be provided to the dealer. This approach would be rather inflexible to policy changes. A more sophisticated system could utilize a programmable desk calculator or a remote computer terminal for real-time evaluation of the application.

The difference between the models for "surrogate" judgment and "tentative approval" lies in the current non-availability of a



. ....

÷ð

本語 そう

100



credit rating at the dealers showroom. With current trends in automation, this condition will likely change with the credit history of an applicant being stored in the credit bureau computer and being accessible electronically via a remote computer terminal. In this situation, much better evaluations could be made and the connotation of "tentative approval" could be removed. To implement a system of this type, the policy model of the lending institution could be stored in a time-sharing computer. At the time the application is made, it could be typed into a remote terminal such as a teletype, the credit rating could be electronically obtained from the credit bureau, and the application score and an appropriate decision could be rendered. An ancillary benefit would be the electronic transmission of the original application to all appropriate parties and the elimination of much of the manual effort currently required from the clerical staffs of the lender and the dealer.

Transferrability of Policy Models:

Two experienced loan officers from other lending institutions were used to assess the potential transferability of the predictor set and the models developed for the loan officers at Lamar Savings. In initial interviews, both of these judges reviewed the predictors and their categorizations, and generally agreed with them. Although each expressed some differences in the exact categorizations of various predictors, these differences were no greater than the differences that had existed among the three senior loan officers from Lamar Savings at the beginning of the project. The most important thing noted was that the only variables suggested by these judges that were not included in the 16 predictors shown in Exhibit 3, Appendix B, were Sex and Number of Dependents. Both of these were suggested by Judge 7. He noted that his clientele was composed mainly of salaried employees with large families from minority groups and was somewhat more restricted and less affluent than the clientele of Lamar Savings. Neither of the predictors were viewed by Judge 7 as critical omissions.

Ć

.

.

The AID-Trees for the two judges are shown in Exhibits 14 and 15 of Appendix B. A comparison of their decisions with the vote decisions of Judges 1 through 5 is given in Table 6-17.

# TABLE 6-17 COMPARISON OF "4/5" VOTE AND DECISIONS OF JUDGES 6 and 7

Judge	No/No	Yes/No	No/Yes	Yes/Yes	Percent Match
6	61	4	66	93	69
7	98	8	29	89	83.5

Both the AID-Tree and the hit table reflect that Judge 6 apparently has a different policy than that reflected by the majority of the original loan officers. The lower predictive capability of the AID-Tree would suggest that the policy of Judge 6 may not be as well defined or consistent as those of the other judges. The data in the hit table reflects a greatly different policy toward denials. As previously discussed, Judge 6 had just been hired by Lamar Savings from his position as manager of a smaller, more speculative loan company. The high number (66) of approvals by Judge 6 which were denials by the 4/5 criterion would indicate that Judge 6 was accustomed to assuming more risk by granting applicants with lower Credit Rating loans. His results represent a fortuitous demonstration of the potential use of the Policy Capturing technique to uncover differences in philosophy and background of new employees.

đ,

The AID-Tree and hit table for Judge 7 reflect that his policy is much closer to the "4/5" vote of the original judges, but there are still some differences, primarily in the importance of Bank Account and Financial References. These differences are probably reflective of the different clientele since Judge 7's policy is better explained by variables such as Residential Category, Time at Residence, and Income, than it is by variables such as Financial References and Bank Account. It is important to note that the predictive capability of the AID-Tree for Judge 7 is comparable with those of the other five judges. This indicates that the omission of Sex and Number of Dependents as predictors does not greatly degrade the model and confirms his assertion that they are not critical to his policy.

Based on this rather limited insight into transferability, the practice of using the weights of a policy model, derived for one environment, in another environment does not appear to be the best approach. However, the redefinition of these weights by using a standard set of predictor cases does appear feasible.

## Summary:

This chapter has presented the major results of the effort to model the policy of installment loan officers under two conditions-with and without access to a Credit Rating on the applicant. Results of both local and first-order models suggest that they could be effectively employed to aid the loan officers in making more efficient and timely decisions. The use of the AID-Tree was found to be a particularly good catalyst to interjudge communication, particularly when the judges are not accustomed to conceptualizing decision processes in terms of algebraic or statistical models.

The results suggest that the local models are more idiosyncratic than the first-order models and are sensitive to policy changes that affect only certain portions of the decision space. However, this sensitivity could be reduced by using a larger set of data to define the local models. The local models do provide a better "average" fit over the entire space.

Models that were able to predict binary decisions, using codified predictors, with 97 percent accuracy were developed. In predicting decisions made with the natural language applications, 92 percent accuracy was attained. In either case, these levels of predictability are better than the 90 percent reliability of the judges. When compared to the 75 percent accuracy that would result if every applicant during July 72 had been given a loan, the improvement in accuracy obtainable with the policy models ranges between 17 and 22 percent. Whereas giving everyone who applies a loan is clearly unwise in the viewpoint of the loan officer, this latter comparison is not really too informative. The actual economic advantage of using the policy models can only be evaluated after they are used in a decision making role and appropriate cost and profit data has been collected. Comparison of such with the costs and profits for the current mode of operation will then reflect the economic viability of using the models.

All Street

υ

é

.5

<u>نځ</u>

ŗ.

-

320

The feasibility of identifying the top 30 percent of the applications in the absence of the primary predictor was demonstrated.

Finally, evidence of the utility of the Policy Capturing technique to assess interjudge differences and evaluate the policy of new employees was obtained.

# CHAPTER VII: CONCLUSIONS, COMPARISONS, AND RECOMMENDATIONS

Viability of Applying Policy Capturing to Operational Decisions:

. 5

The primary goal of this research was the adaptation and demonstration of Policy Capturing in operational environments. Toward this end, the analysis was performed entirely with data taken from an operational environment and the criteria for success were defined relative to the needs and conditions within that environment.

The ultimate product sought was a technique that could be used by the Operations Research practitioner to define, evaluate, compare, and predict the judgments of decision makers in the business, academic, and administrative fields. In order for a modeling technique to be viable in these environments, it must be reasonably straightforward to use and compatible with the stimulus data (in amount and format) as it exists in the environment being modeled. It must produce models that are accurate and understandable.

The results embodied in Table 6-15, Figure 6-23, and Figure 6-28, and the remarks of the junior loan officer, (page 180) are probably the best evidence that can be offered for the utility and viability of Policy Capturing in the operationally valid environment that was studied in this research. They reflect that policy models were obtained which could predict the decisions of installment loan officers with 92 percent accuracy when all of the important predictors are available. Models were also produced which could meet the criterion of identifying the top 25-30 percent of the loan applications even when the most important predictor was not available. The

remarks of the junior loan officer, and the continued participation of the already burdened staff of the installment credit department attest to the believability and communicability of the model.

Comparison of a judge's policy as expressed by any of the algebraic models presented in Chapter VI, with the "structural image" model as expressed by the corresponding AID-Tree, leaves little doubt as to the benefits of graphically displaying the policy. This is especially important in communicating the concept of Policy Capturing to decision makers who are not mathematically inclined.

Modeling Configural Judgment Processes with Local Models:

2

. ہو

_h. _.....

5

As for the conceptualization and modeling of configural decision processes as a series of local hyperplanes, the comparison of the models obtained for the loan officers (see Figures 6-14 through 6-18) would appear to support the superiority of this approach from both the accuracy and the interpretability standpoints. Collateral support for the contention that interaction in judgment processes might best be modeled as a discrete phenomene is provided by the comparison of the results obtained using local model with Valenzi's results for "continuous" models.

The Character of the Model Seeking Technique, AID:

In assessing the robustness of the model seeking procedure embodied in the application of the AID4UT/AIDTRE programs, several pertinent points can be made. First, categorical variables are used; this is the most general of all possible representations of the individual predictors. Second, there are no underlying restrictions or assumptions imposed in the model seeking phase of the process. Third, the procedure does not require designed experiments or independent predictors. Finally, the capacity of the computer program is large enough to handle up to 80 possible predictors and does not require an "a priori" culling of the variables, or hypothesis of the interactions.

On the other hand, the procedure does not produce a completely definitive answer, but rather, requires the application of considerable judgment on the part of the analyst in the interpretation of the output data. In this sense, the technique merely provides the analyst with the capability of choosing where he wants to inject his wisdom and experience in the modeling process. With more traditional analysis techniques, he makes the decision by choosing the technique and implicitly imposing the appropriate assumptions. In the use of AID4UT/ AIDTRE and local models, he makes his decisions in terms of explicit interpretations of which subspaces and local models to use.

Comparison of Results of this Research with Those of Past Research:

.

يە ئ

ं स्

<u>,</u> 1

 $\mathbf{v}$ 

1.1

ş

<u>i</u>,

δ.

8

1. ....

There are some interesting comparisons that can be made between the results of this research and the conclusions of past investigators. These comparisons generally reflect that the policy models developed for the loan officers are in harmony with the results and expectations of the clincial psychologists.

In reference to past attempts to define configural models, the results of this research reflect an improvement that is chiefly attributable to the use of the AID technique for determining where to "fish" (Green, 1968) for configural relationships and the modeling of

these interactions as discrete phenomena.

Ĵ

44

Other specific comparisons include:

Power of First-Order Models, vis-a-vis, Configural Models. Hoffman (1960) and Yntema and Torgerson (1961) demonstrated the power of first-order paramorphics in modeling judgment processes. The results of this study reflect the same situation. Purely on the basis of statistical significance, there is probably no difference between the performance level of the various types of models investigated in this research. However, given the choice of two equally good models (in terms of either explained variation or hit rate), the model that appears to be more reflective of the underlying process would be the better choice since it would be more intuitively satisfying to the judge.

Relative Shrinkage of Configural Models Due to "Overfit". Goldberg (1968) found that configural models of judgment processes tend to be far less stable than their corresponding first-order models. Ward (1954) indicated that configural models generally tended to "overfit" the idiosyncracies of the data, as indicated by much greater shrinkage in the cross-validation coefficient. From these findings, one could infer that if the configural model held up as well as the first-order model upon cross-validation, some evidence of legitimate configural effects exists. The shrinkage for the first-order models of the loan officers averaged 8.7 percent while the shrinkage for their local models averaged 9.7 percent. This difference of only 1 percent would indicate that the "overfit" phenomena is minimal. It would offer evidence that the configurality is real, especially since the cross-validation rates themselves were uniformly higher for the local models (see lable 4-4). Diminished Configurality in Group Policies. Slovic (1968) noted that models for group judgments tended to be less configural than those for

individuals. This same situation is reflected by comparison of the differences in predicted capability between the first-order models and local models for the "4/5" vote policy with the corresponding differences for the individual loan officers. (See Table 6-15.)

AID-Tree Representations of Einhorn's Non-compensatory Models. The lexicographic model suggested by Einhorn (1970) is conceptually similar to the process depicted by the structure of an AID-Tree. His conjunctive model, in which strong attributes are most important in determining the total score, is analogous to the substitute advantage model of Sonquist (1964) which results in an upper terminating trunktwig structure in AID-Trees. Similarly his disjunctive model correlates to the substitute disadvantage model as reflected in the lower terminating trunk-twig structure.

Recommendations for Further Research:

CARDINAL MARKED

This research has served to demonstrate the potential utility of applying Policy Capturing to a practical decision process, but it does not represent a final answer to the applicability of the concept. The project has probably generated more ideas and questions than it has resolved. These new questions can provide the basis for further research projects in both the areas of 1) development of new applications, and 2) improvement of the current modeling techniques and capabilities. Specific projects recommended by this investigator are:

Implementation of Policy Models as Decision Aids for Loan Officers: The models presented in this dissertation could be implemented on

either a programmable desk calculator, or a remote teletype hookup to a time-sharing computer. The primary tasks of such a project would include:

ŀ

- Development of the computer code or calculator routine to implement the appropriate models and evaluate incoming applications.
- 2) Redesign of the luan application form to facilitate collection of the appropriate categorical data.
- Training of the clerical staff for proper evaluation and coding of the predictor data.
- 4) Collection and evaluation of data to evaluate actual performance of the policy models in terms of the "ex-post" quality of the approvals and the lost revenue of the denials.
- 5) Expansion and modification of the current models to allow implementation for other appliance dealers and the development of models for other categories of dealer contracts.

Sequential Generation of Stimulus Data: At the beginning of this research, the incremental generation and consideration of single cases of stimulus data was envisioned as a possible method of constructing policy models with a minimal set of data. At this point it is still unclear how significant changes and trends in the policy could be detected by addition of a single new decision at each stage of the Policy Capturing process. However, a modified version of this concept of incremental data generation does appear to have merit when used with the AID4UT/AIDTRE computer programs in the definition of subspaces and local models. In development of a policy model, initial analysis should be performed with a sample of actual stimulus data of at least 10 cases per predictor. The identification of homogeneous subgroups would eliminate subspaces within the predictor space which do not require further investigation. In the non-homogeneous groups, past experience indicates that there are often too few data units to determine a stable model. This problem could be solved by generation of an incremental set of synthetic stimulus data which falls entirely within the nonhomogeneous subspace. In this manner, the data set would be augmented in those subspaces of the predictor set where the policy of the judge is most complicated and in which there is a requirement for more predictors and more complicated local models.

The task of incrementally generating realistic stimulus data within selected subspaces could be approached by using a synthetic data generation routine such as the one developed by R. F. Fallis of the Ohio State University Psychology Department. This routine was used by Naylor and Wherry (1964) to demonstrate the feasibility of capturing policies with synthetic data. The correlation matrix obtained from the original set of actual decision data would be used by the data generator to assure realistic data was generated within the subspaces under consideration.

÷

੍ਹਤ

A specific research project in this area would include the following tasks:

 Acquisition and modification, or development, of a synthetic data generation routine.

- 2) Identification of an appropriate decision process and application of the proposed procedure for defining local models. For example, synthetic data for the non-homogeneous subspaces of the loan officer's AID-Trees might be generated and new models generated.
- Evaluation of the change in the model quality resulting from the synthetic augmentation procedure.

Estimation of Time to Repair Electronic Equipment: During the past decade, the process of estimating the mean time to repair (MTTR) of various electronic equipment has become of considerable concern to design, reliability, and maintainability engineers. The need for this data is growing increasingly important for use in optimization mcdels for modularizing equipments and determining repair strategies. The problem lies in the fact that the MTTR data is usually needed in the early stages of equipment development, but is not available until after the equipment has been produced and become operational. Only after the equipment has failed and been repaired can observational data be gathered and analyzed. Even if this process were efficient. which it is not, the resultant data is obsolete by the time it is available. The past procedure for obtaining MTTR data needed in the design process has been to infer values for new equipments, based on experience with similar equipments. This process has not been as accurate as is required for use in sophisticated optimization and design routines.

. .

A recent study was sponsored by the U.S. Air Force Materials Laboratory, WPAFB, Ohio, in which the subjective estimates of MTTR values for 23 various pieces of electronic equipments were obtained from experienced maintenance technicians. Although this study had some of the aspects of a Policy Capturing effort, it did not model the technicians judgments as a function of the basic characteristics of the equipment. Instead, estimates for each individual piece of equipment were collected and compared with actual maintenance experience for that piece of equipment. The results indicated a .86 correlation between the estimates of the repair times and the actual repair times (Smith, 1971). This reflects that technicians can make valid estimates of repair times on particular equipments, but it does not improve the situation relative to predicting the repair times for new and different equipments.

It would appear that this situation offers a natural area for the application of policy models. A project in which the estimated repair times for each equipment were modeled as a function of the basic characteristics of the equipment instead of specific equipment's names or stock numbers might provide the model that could be used for predicting MTTR values for new equipments. The project would consist of modeling the MTTR data in terms of parameters which were common among equipments and building models with these parameters as predictors. Specific tasks to be performed in such a project would include:

<u>ن</u>

 Determination of the basic characteristics of a class of equipments such as function, size, shape, type of assembly, accessibility, type of failure mode, repair procedure, and required test equipment.
Acquisition of existing judgmental data from the appropriate agency and description of the individual equipments in terms of these basic characteristics, or generation of new judgment data on new equipments.

Ł

3) Definition of models for estimating MTTR values and comparison of the predicted values for similar equipments, not used in the modeling phase, with actual maintenance data that is available in such publications as the FARADA Handbooks.

Prediction of Successful Rehabilitation of Military Prisoners: A project that could be of direct benefit in further evaluating the potential of the local modeling technique proposed in this dissertation is suggested by a study performed by Smith, Gott, and Bottenberg (1967). In that study, various trait and background data was collected on U.S. Air Force prisoners and first-order models for predicting successful return to duty were developed. They achieved predictive efficiencies of 77.4 percent and suggested that better models could probably be obtained with further analysis. Their primary objective was the demonstration of feasibility of the Policy Capturing approach.

During the course of the current research, the need for such a predictive system surfaced as a result of discussions with the personnel from the U.S. Army Correctional Training Facility (CTF), Fort Riley, Kansas. In their case, accurate prediction of potential failures and successes of retrainees could be used to modify the length of stay and the course of instruction required for individual retrainees. Such predictions would conceivably be made on the basis of instructor evaluations obtained in the early part of the rehabilitation program.

This project could also provide insight into the proposed

use of Policy Capturing techniques to evaluate differences in policy between similar governmental agencies. The experience of interacting with two separate governmental agencies would undoubtedly draw upon the managerial talents of the researcher as well as his technical talents and would be a particularly appropriate project for an operation researcher or management scientist.

Specific tasks of this project would include:

- Acquisition of the appropriate data used by Smith, Gott, and Bottenberg. This data is still on file at the Personnel Research Laboratory, AFHRL, Lackland, AFB, Texas.
- Analysis of data with the AID4UT/AIDTRE programs and the development of local models.
- Comparison of the local models with these first-order models obtained in the previous study by Smith, Gott, and Bottenberg (1967).
- Acquisition and modeling of judgment and actual data from the CTF, Fort Riley, Kansas.
- 5) Comparison of the data and models on an inter-agency basis.

### Final Comments:

The ultimate goal of this research effort was the "demonstration of the potential benefits and the viability of applying the Policy Capturing concept in ecologically valid environments". To be totally complete, such a demonstration must include more examples than were reasonably accomplishable during the 16 months spent on this research. Fortunately, the examples attempted were successful, knowledge was gained, and tools developed which should make further demonstrations of the concept's viability less laborious.

The projects suggested for further research represent areas in which this investigator believes significant results might be produced in a relatively short time. Other applications are somewhat more esoteric and will require considerably more conceptual innovation. Examples of further applications of Policy Capturing are the development of models for the objectives and desires of voters, preference models and consumer policies in marketing, and the modeling of the objective functions for research and development planners.

The results of this experiment have increased the interest in Policy Capturing and the research will undoubtedly continue. However, there is much to be accomplished before Policy Capturing becomes a completely "canned" technique.

Whatever the final outcome of these proposed projects, some benefit will inevitably be realized for these applications of Policy Capturing. As a minimum, the process of trying to quantify subjective human judgment forces the participating decision maker to think about, and verbalize, his policy. In this respect, the perception that benefits could be derived from mathematically modeling decision processes is not new. As early as 1889, Lord Kelvin recognized the potential of quantifying human judgment by stating:

> I often say that when you can measure what you are speaking about, and express it in numbers, you know something about it, but when you cannot measure it, when you cannot express it in numbers, your knowledge is of a meager and unsatisfactory kind; it may be the beginning of knowledge, but you have scarcely, in your thoughts, advanced to the stage of science, whatever the nature may be.

# APPENDICES

ł

/'2'

١

1. Mar 700 an

. . .

#### APPENDIX A

Definition and Comparison of Mathematical Models

Purpose:

In the literature covering the past research in Judgment Analysis, various mathematical models are discussed but they are seldom defined in explicit mathematical terms. One of the goals of the present research is the introduction of the model seeking technique, AID, as a tool for improving the policy-model hypothesis process and transforming that process into more of a directed search and somewhat less of a "fishing expedition". Since the many subtle aspects of linear modeling theory cannot all be discussed in this appendix, the objective of this appendix is to merely discuss the various models and computational methodologies considered in this and past research and to relate them to the general linear model.

The basic function of any model is the <u>parsimonious</u> description of nature. Models can be classified as being either deterministic or stochastic. In Policy Capturing we are concerned with stochastic models and the use of various statistical techniques to build and evaluate stochastic models. The model itself represents the relationships that are presumed to exist in the population. Any assumptions that are made as adjuncts to the model pertain to the population.

The General Linear Model: (GLM)¹

The reader's familiarity with vector and matrix notation and terminology will be assumed throughout this discussion.

 $Y = X\beta + \epsilon$  or Expected Value (Y) =  $\hat{Y} = X\beta$ 

where:

 $\beta = \begin{bmatrix} \beta_1 \\ \beta_2 \\ \vdots \\ \beta_p \end{bmatrix}$ 

Y = y₁ ; an 'n x l' vector of random observations from the population of all possible Y values y_n

 $\mathbf{X} = \begin{bmatrix} \mathbf{X}_{11} & \mathbf{X}_{12} & \cdots & \mathbf{X}_{1p} \\ \mathbf{X}_{21} & \cdots & \cdots & \vdots \\ \mathbf{X}_{n1} & \cdots & \vdots & \vdots \\ \mathbf{X}_{n1} & \cdots & \mathbf{X}_{n1} \end{bmatrix} : \text{ an 'n x p' matrix of fixed}$ 

: a 'p x l' vector of unknown parameters

: an 'n x l' vector of random errors

### The GLM for a Hypothetical Three-Predictor Problem:

nen J X

, SHE

One can always formulate a problem as a general linear model which has unique binary predictors for each possible combination of the predictor variables. For example consider a problem in which there is a criterion that is a function of three predictor variables,  $Y_{ijk} + f(R_i, S_j, T_k)$ , two of which (R, S) are ordinal, and the third (T) being categorical. Further, consider that there are three distinct levels associated with each of these variable and that the data to be analyzed consists of one point² having each of the possible combinations of these predictor values. Table B-1 shows the criterion value for each of the 27 possible combinations (patterns) of the predictor variables.

		R i=1			R i=2			R i=3	
	T k=1	T k=2	T k=3	T k=i	T k=2	T k=3	T k=1	T k=Z	T k=3
S j=1	¥ ₁₁₁	Y ₁₁₂	Y ₁₁₃	¥ ₂₁₁	¥ 212	Y ₂₁₃	Y ₃₁₁	¥ ₃₁₂	Y ₃₁₃
S j=2	Y ₁₂₁	Y ₁₂₂	¥ ₁₂₃	Y ₂₂₁	¥ ₂₂₂	¥ ₂₂₃	Ү ₃₂₁	¥ ₃₂₂	Y ₃₂₃
S j≈3	Y ₁₃₁	Y ₁₃₂	¥ ₁₃₃	Y ₂₃₁	Y ₂₃₂	Y ₂₃₃	Υ ₃₃₁	Y ₃₃₂	¥ ₃₃ 3

TABLE B-1

The most general linear model that could be written for predicting the value of a data units possessing one of these 27 predictor combinations would be of the form:

² In the general case, multiple data points would occur for each predictor-combination. In this case, each criterion value would have an additional subscript denoting that it was the one of several in the (i, j, k)'th cell.

$$Y_{ijk} = \hat{\beta}_{1} X_{111} + \hat{\beta}_{2} X_{112} + \dots \quad \hat{\beta}_{26} X_{332} + \hat{\beta}_{27} X_{333} + \epsilon_{ijk}$$
(A)

where:

- Y = the criterion value for that data point having the i'th level of predictor R, the j'th level of predictor S, and the k'th level of predictor T. (the criterion value for that point in the (i, j, k)'th cell)
- $X_{ijk} = \begin{cases} 1 & \text{if the point is in the } (i, j, k)' \text{th cell} \\ 0 & \text{otherwise} \end{cases}$
- $\hat{\beta}$  (n=1, 27) = the 27 unknown parameters of the model

ijk = the error in the predicted value of the point in the (i, j, k)'th cell.

In vector notation corresponding to that given in the definition of the GLM. Y would be a vector of 27 criterion values, X would be a 27 by 27 identity matrix, and  $\hat{\beta}$  would be a vector of 27 unknown parameters.

Note that the model that we have presented is not very interesting in that it has exactly the same number of parameters as there are data patterns. Further, there is only one point per cell and each data point can be predicted exactly; therefore, the error vector is null. This model is not particularly desirable in terms of the parsimony with which it describes the data but it does serve as an example of a true model; a true model being defined as:

> A model that does not impose any relationships upon the expected values of the criterion variable which do not exist in nature.

Although a model which has one binary predictor per variable pattern is a <u>sufficient</u> condition for a model to be "true", it is not a <u>necessary</u> condition. A more parsimonious "true"

model might be attainable if certain regularities exist in the relationships between the individual predictors and the criterion values. Such relationships represent restrictions on the GLM that can be explicitly imposed on the model, but which are more commonly implicitly "assumed" to exist in the process of model formulation.

Equivalency and the Regression Formulation of the GLM:

à

- <del>//</del>

Before preceding with the discussion of the restrictions that are imposed on models as an alternative to representing each possible predictor pattern by its own binary vector, the concept of equivalent models will be introduced. Two linear models (M1 and M2) are said to be <u>equivalent</u> if they meet the following conditions:

- 1) Both models have the same criterion vector: (Y)
- 2) Both models have the same number of linearly independent predictors: (rank of  $X_{M1}$  = rank of  $X_{M2}$ )
- Every vector in the first model is a linear combination of the vectors in the second model.

The multiple regression model is, in fact, equivalent to the general linear model. In the regression formulation, the X-matrix is composed of an n-dimensional unit vector and a 'n x (p-1)' matrix of predictor values. The most general formulation of the regression model is the "categorical" or "binary regression" model. The binary regression model results when the elements of the X-matrix only takes on the values of 0 or 1, and the predictor vectors are interpreted as classifications or attributes. In this model, the x_{np} values of 0 or 1 reflect the absence or presence of the p'th attribute in the n'th data case.

The binary regression model that is equivalent to equation (A) for the hypothetical problem is:

$$Y_{ijk} = \beta_0 + \beta_1 X_{11} + \beta_2 X_{112} + \dots + \beta_{26} X_{332} + \epsilon_{ijk}$$
(B)

where:

Contraction of States of Contraction of the

÷Ż

and the second of the second second

 $Y_{ijk}$  = as defined previously  $X_{ijk}$  = as defined previously

and:

In model (B), the GLM of model (A) has been reparameterized and the new parameters have been referenced to the value of 
$$Y_{333}$$
 as a result of the elimination of the  $\beta_{am}$  coefficient. There are still 27 parameters

 $\beta_0 = \hat{\beta}_{27}, \ \beta_1 = \hat{\beta}_1 - \beta_0, \ \beta_2 = \hat{\beta}_2 + \beta_0 \dots \text{ etc.}$ 

27 with which to estimate 27 data points, thus allowing the prediction of the Y values without error.

The regression model given in (E) could be written in terms of the vector notation given in the definition of the GLM. In that case, Y would be the same vector of 27 criterion values, X would be a partitioned matrix of the form

 $\mathbf{X} = \begin{bmatrix} \mathbf{U} & \mathbf{I} \\ \mathbf{O} \end{bmatrix}$ 

where:

U = a 27 by 1 unit vector I = a 26 by 26 identity matrix O = a 1 by 26 null row vector

1078-001-001

209

uit

Because of the generality of the multiple regression model and its equivalence to the general linear hypothesis, all other linear models or computational techniques can be viewed as special cases of the regression model upon which various restrictions have been imposed. Each of these special computational techniques can be accomplished by applying multiple regression procedures to equivalent regression models that have been obtained by reparameterization of the general linear model. Although the mathematical equivalence of various experimental designs and special computational models to reparameterized regression models has long been recognized, the specific techniques and procedures to effect such reparameterizations have only recently been addressed by such authors as Bottenberg and Ward (1963), Draper and Smith (1966), Jennings (1967). Mendenhall (1968), Weber (1971), and Ward and Jennings (in press).

Restrictions Imposed on the GLM:

Ň

As mentioned previously, there are several types of restrictions that are commonly placed on the general version of the regression formulation of the GLM. Searle (1971) has pointed out that any restriction can be viewed from three different viewpoints. These three viewpoints are:

- The restriction is arbitrarily imposed as a sidecondition solely for the purpose of obtaining a unique solution to the parameters.
- 2) The restriction is naturally generated as a result of the hypothesis being tested and the presumed state-of-nature.

3) The restriction is part of the basic model.

In this sense, restrictions placed on the GLM to achieve parsimony would fall into category 2. The two most common of these restrictions are:

<u>Constant Difference Restrictions</u>: This restriction implies that the difference in the expected value of the criterion value  $(\widehat{\mathbf{Y}})$  is a constant (K) for any unit difference in the value of the predictor variable. In terms of our hypothetical model this would mean that if there were a unit difference between the 1st, 2nd, and 3rd levels of the variable R, the following relationships would hold:

1  $\hat{Y}_{111} - \hat{Y}_{211} = \hat{Y}_{211} - \hat{Y}_{311} = K$ 2  $\hat{Y}_{121} - \hat{Y}_{221} = \hat{Y}_{221} - \hat{Y}_{321} = K$ 9  $\hat{Y}_{133} - \hat{Y}_{233} = \hat{Y}_{233} - \hat{Y}_{333} = K$ 

This restriction allows the representation of all levels of the R variable by a single predictor and thus reduces the number of predictor variables in the model. By assuming both of the ordinal variables (R&S) conform to the constant difference restriction, the model for the hypothetical problem could be written:

$$\mathbf{r}_{ijk} = \beta_{0}^{+\beta_{1}R_{i}^{+}\beta_{2}S_{j}^{+}\beta_{3}T_{k=1}^{+}+\beta_{4}T_{k=2}^{+}+\beta_{5}R_{i}S_{j}^{-} \\ +\beta_{6}^{R_{i}T_{k=1}^{+}+\beta_{7}R_{i}T_{k=2}^{+}\beta_{8}R_{i}^{T_{k=3}^{+}+\beta_{9}S_{j}^{-}T_{k=1}^{-} \\ +\beta_{10}^{S_{j}T_{k=2}^{+}+\beta_{11}S_{j}^{-}T_{k=3}^{+}+\beta_{12}^{R_{i}S_{j}^{-}T_{k=1}^{-} \\ +\beta_{13}^{R_{i}S_{j}^{-}T_{k=2}^{+}+\beta_{14}^{R_{i}S_{j}^{-}T_{k=3}^{+}+\epsilon_{ijk}^{-}$$
(C)

where:  $R_i = the actual value of R at the i'th level$  $<math>S_j = the actual value of S at the j'th level$  $<math>T_{k=1} = \begin{cases} 1 & for data units at the lst level of T \\ 0 & otherwise \end{cases}$   $T_{k=2} = \begin{cases} 1 & for data units at the 2nd level of T \\ 0 & otherwise \end{cases}$  $T_{k=3} = \begin{cases} 1 & for data units at the 3rd level of T \\ 0 & otherwise \end{cases}$ 

If the assumption were true, then equation (C) would still represent a "true" model.

No-interaction (Additivity) Among the Predictors: This restriction implies that the difference in the expected value of the criterion value as a result of a difference in the value of one predictor is independent of the value of the other predictors; that is:

$$\hat{\mathbf{Y}}_{111} - \hat{\mathbf{Y}}_{211} = \hat{\mathbf{Y}}_{121} - \hat{\mathbf{Y}}_{221} = \hat{\mathbf{Y}}_{131} - \hat{\mathbf{Y}}_{231} - \dots = \hat{\mathbf{Y}}_{133} - \hat{\mathbf{Y}}_{233} = \mathbf{K}\mathbf{I}$$
  
etc.

If this type restriction is imposed on the model for the hypothetical example, the resulting model would be:

$$Y_{ijk} = {}^{\beta}_{0} {}^{+\beta}_{1} {}^{R}_{i=1} {}^{+\beta}_{2} {}^{R}_{i=2} {}^{+\beta}_{3} {}^{R}_{i=3} {}^{+\beta}_{4} {}^{S}_{j=1} {}^{+\beta}_{5} {}^{S}_{j=2} {}^{+\beta}_{6} {}^{S}_{j=3}$$

$${}^{+\beta}_{7} {}^{T}_{k=1} {}^{+\beta}_{8} {}^{T}_{k=2} {}^{+\epsilon}_{ijk}$$
(D)

where:

2

 $R_{i=1}, S_{j=1}, T_{k=1} = \begin{cases} 1 & \text{for data units at the first level} \\ 0 & \text{of the predictor} \\ 0 & \text{otherwise} \end{cases}$ 

$$R_{i=2}$$
,  $S_{j=2}$ ,  $T_{k=2} = \begin{cases} 1 \text{ for data units at the 2nd level of} \\ \text{the predictor} \\ 0 \text{ otherwise} \end{cases}$ 

$$R_{i=3}, S_{j=3} = \begin{cases} 1 \text{ for data units at the 3rd level of} \\ the predictor \\ 0 \text{ otherwise} \end{cases}$$

If both the constant difference and additivity restrictions are imposed upon the GLM, the model takes on its most parsimonious form:

 $\mathbf{Y}_{ijk} = \mathbf{\beta}_0 + \mathbf{\beta}_1 \mathbf{R}_i + \mathbf{\beta}_2 \mathbf{S}_j + \mathbf{\beta}_3 \mathbf{T}_{k=1} + \mathbf{\beta}_4 \mathbf{T}_{k=2} + \mathbf{\varepsilon}_{ijk}$ 

where:

. A . .

ŧ

R _i	-	actual value of R at the i'th level
s _j	=	actual value of S at the j'th level
T _{k=1}	-	$\begin{cases} 1 & \text{for data units at the 1st level of T} \\ 0 & \text{otherwise} \end{cases}$
T.	=	1 for data units at the 2nd level of T

The Estimation of Model Parameters and Evaluation of Model "Truth" from Sample Data:

The hypothetical example that was previously given considered a population that had only 27 data points, and a "true model" could be written very easily. Generally, models are used to describe the relationships in much larger populations of data with 213

in white and shall

many more predictor patterns. A parsimonious model that is also <u>known</u> to be true may not be readily apparent. In this case, the analyst must rely upon probability and statistical theory to evaluate the hypothesis that any particular model is "true". This involves the evaluation of the parameters of the model itself as well as the statistics associated with the particular analytical technique that is being employed.

In order to estimate the parameters of a model, sample data is taken from the population and analyzed with the appropriate statistical techniques. The model itself specifies the structural relationships among the variables. The sample data is used to infer the values for the parameters  $\frac{3}{3}$  of the model. The quality of these inferences is a function of the deviation of the hypothesized model from a "true model" and the degree to which the sample data resembles the population. Neither the degree of "truth" of a model nor the amount of sampling error can be assessed exactly. However, use of appropriate analytical and experimental procedures can obviously help to avoid the imposition of "untrue" relationships and can help to minimize sampling error. In the first case, avoidance of restrictions and assumptions which impose arbitrary relationships on the variables will reduce the chance of imposing erroneous relationships. In the second case, the use of randomization and the selection of samples that are sufficiently large will help to reduce sampling error.

: 4

In order to distinguish between the parameters for the population and the estimates of these parameters derived from the sample, the convention of using Greek letters to designate population parameters and the corresponding English letters to designate the sample estimates will be followed.

Given a model and sample, the analyst generally has a choice of which statistical analysis technique he wishes to use. This choice is tempered by the amount and type of data and the particular information that the analyst wants.

The specific statistical analysis techniques to be discussed in this appendix include: Multiple Regression (MR), Analysis of Variance (ANOVA), Multiple Classification Analysis (MCA), and Automatic Interaction Detection (AID). All of these formulations are actually restricted cases of the general linear model, and, in that sense, they might best be viewed simply as alternative analytical techniques, each of which is applied to data fitting a particular experimental design and each of which most conveniently achieves some analytical goal.

The most important difference among these analytical techniques is not in the basic character of the underlying models; rather, it is in the way in which those models are obtained. In this context, the differentiation of the techniques into "model seeking" and "model hypothesis" categories is appropriate. MR, ANOVA, and MCA fall into the category of "model hypothesis" and AID falls into the category of "model seeking".

A "model hypothesis" technique is one in which a model of the structural relationships among the individual variables of the problem is hypothesized on an <u>a prior</u>i basis and the analysis is performed to determine the relevance of the individual variables.

A "model seeking" technique is one in which the model is not hypothesized on an <u>a prioribasis</u>, but rather, the data is analyzed and displayed in a form so as to allow model hypothesis to proceed. Model seeking techniques are generally heuristic and mathematically

informal, such as in the case of trial and error approaches. The AID technique represents a step toward formalization of the model seeking process based on the analysis of the data with a sequential variance-reducing procedure.

The Least-Squares Computational Process:

The estimation of the parameters for a multiple regression model has generally been confined to the procedures of least-squares in which the sum of the squared residual errors (ESS =  $\sum_{i=1}^{n} \epsilon_{i}^{2}$ ) is minimized. This procedure involves the explicit or implicit solution of a set of 'p' simultaneous equations, known as the normal equations, for 'p'⁴ parameters (b_i), that are the estimates of the model parameters ( $\beta_{i}$ ).

Standard computational formulae for the parameters are derived and given in many texts such as Draper and Smith (1966). They are:

Given:  $Y = X\beta + \epsilon$ 

the least-squares estimator of  $\beta$  is:

ŝ

 $b = (X'X)^{-1} X'Y$ 

⁴Here 'p' refers to the number of parameters and includes the constant  $b_0'$  and the ' $b_i'$ ' weight of each of the predictor v: riables of the problem.

the predicted values of the criterion are:

the residuals are:

æ

8

۰. آ

$$e = Y - Y$$

and the coefficient of determination is:

$$R^{2} = \frac{b'X'Y - ((\sum_{i=1}^{n} y_{i})^{2}/n)}{Y'Y - ((\sum_{i=1}^{n} y_{i})^{2}/n)}$$

and the Error Sums of Squares is:

$$ESS = Y'Y - b'X'Y'$$

There are several algorithms for solving these normal equations, each of which has some particular advantages. The most direct of these algorithm involves explicit solution of the above equations as given in their matrix form. This procedure has one major drawback, if the X'X matrix is singular, there exists no inverse and the equations cannot be solved. This situation exists where there are linear dependencies in the set of predictor vectors and the number of parameters in the model (p) is greater than the number of unique patterns in the data set (m). In cases where  $p \ge m$ , the X-matrix is said to be of less than full rank and a unique solution for the parameters estimates

is unobtainable. A less direct approach embodies the iterative determination of the parameters and does not require matrix inversion. If the predictors are independent, this procedure will result in the same parameter values as the explicit solution, and if the predictors do contain dependencies, it will produce one of the infinitude of non-unique solutions that corresponds to the minimum ESS. Thus, use of this latter procedure will allow acquisition of the ESS information required for hypothesis testing relative to full and restricted models even when the individual parameters cannot be uniquely determined.

A computer code that accomplishes the first procedure is the STEP 01 routine of The University of Texas Center for Highway Research, and a code that accomplishes the second procedure is the REGREJ routine of the EDSTAT-J Library (Jennings, 1971) at the UT Computation Center.

## A Numerical Example for Comparing the Various Analytical Techniques:

In the comparison of the various analytical techniques, a small numerical example is now presented. The data for this example is given in Table B-2. The general linear model corresponding to equation (A) is given in Table B-3. The corresponding data for the binary regression model is given in Table B-4. Table B-5 reflects the data for the model in which the predictors  $(X_1, X_2, X_3)$  have been restricted to by the "constant difference" relationship discussed previously. It does not reflect imposition of the "additivity" restriction. Note that the very small change in the error sum of squares (ESS) between the models in Table B-4 and Table F 5 would indicate that the restriction does not substantially detract from the "truth" of the model.

DATA FOR NUMERICAL EXAMPLE

	x,	=1	X	1=2
	X ₂ =1	x ₂ =2	X ₂ =1	x ₂ =2
$X_3 = 1$	12.5	20.5	3	9
$X_{3} = 3$	24.5	36.5	-1	9
X ₃ =4	30.5	44.5	- 3	9

## GENERAL LINEAR MODEL FOR NUMERICAL EXAMPLE

$$\mathbf{Y}^{\perp}$$
 = (12, 5, 24, 5, 30, 5, 20, 5, 35, 5, 44, 5, 3, -1, -3, 9, 9, 9)

0 0 0 0 Û Ű Û Û Ð X = υ 0 0 0 1 0 0 0 0 0 0 0 1 10 0 0 0 0 0 0 0 0 0 0 0 

$$b^{T} = \beta^{T} = (12, 5, 24, 5, 30, 5, 20, 5, 36, 5, 44, 5, 3, -1, -3, 9, 9, 9)$$

$$c^{T} = (0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)$$

ESS = 0

Note that p=12(number of predictors)/n=12(number of data points)

# BINARY MULTIPLE REGRESSION MODEL FOR NUMERICAL EXAMPLE

 $\mathbf{Y}^{\mathbf{T}} = (12.5, 24.5, 30.5, 20.5, 6.5, 44.5, 3, -1, -3, 9, 9, 9)$ 

		1	1	0	0	0	0	0	0	0	0	0	0
		1	0	1	0	0	0	0	0	0	0	0	0
		1	0	0	1	0	0	0	0	0	0	0	0
		1	0	0	0	1	0	0	0	0	0	0	0
		1	0	0	0	0	1	0	0	0	0	0	0
		1	0	0	0	0	0	1	0	0	0	0	0
Х	=	1	0	0	0	0	0	0	1	0	0	0	0
		1	0	0	0	0	0	0	0	1	0	0	0
		1	0	0	0	0	0	0	0	0	1	0	0
		1	Ũ	0	0	0	0	0	0	0	0	1	0
		1	0	0	0	0	0	0	0	0	0	0	1
		1	0	0	0	Û	0	0	0	0	0	0	0

 $b^{T} = \beta^{T} = (9, 3.5, 15.5, 21.5, 11.5, 27.5, 35.5, -6, -10, -12, 0, 0)$  $e^{T} = (0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)$ 

ESS = 0

p. Saintasar itati i

# REGRESSION MODEL WITH CONSTANT DIFFERENCE PREDICTORS FOR NUMERICAL EXAMPLE

$$(Y = \beta_0 U + \beta_1 X_1 + \beta_2 X_2 + \beta_3 X_3 + \beta_4 X_1 X_2 + \beta_5 X_1 X_3 + \beta_6 X_2 X_3 + \epsilon)$$
  
$$Y^{T} = (12.5, 24.5, 30.5, 20.5, 36.5, 44.5, 3, -1, -3, 9, 9, 9)$$

								-	
		1	1	1	1	1	1	1	
	]	1	1	1	3	l	3	3	
		1	1	1	4	1	4	4	
		1	1	2	1	Z	1	2	
		1	1	2	3	2	3	6	
		1	1	2	4	2	4	8	
X	=	1	2	1	1	2	2	1	
		1	2	1	3	2	6	3	
		1	Z	1	4	2	8	4	
		1	2	2	1	4	2	2	ļ
		1	2	2	3	4	6	6	
		1	2	2	4	4	8	8	

$$\mathbf{b}^{T} = (1.24, .038, 7.19, 11.79, -1.70, -7.99, 2.12)$$
  

$$\mathbf{e}^{T} = (\mathbf{e}_{1}, \mathbf{e}_{2}, \mathbf{e}_{3}, \mathbf{e}_{4}, \mathbf{e}_{5}, \mathbf{e}_{6}, \mathbf{e}_{7}, \mathbf{e}_{8}, \mathbf{e}_{9}, \mathbf{e}_{10}, \mathbf{e}_{11}, \mathbf{e}_{12})$$

ESS(Error Sum of Squares) * = .1433

Ł

* The data for the b^T and ESS were calculated with EDSTAT-J computer program--see Table B-8. The ESS is explained on the following pages. The Analysis of Variance Technique:

The Analysis of Variance (ANOVA) technique is a computational methodology which directly apportions the total variance within a set of experimental data into that part of the variance that is attributable to differences in the levels of the predictor variables (treatments) and that part that is due to random error within the treatment levels. ANOVA is a model hypothesis technique that is usually directed at the testing of the null hypothesis that the mean responses for each of the treatment levels ( $\mu_i$ ) are equal, i.e.:

H:  $\mu_1 = \mu_2 = \mu_3 = \dots = \mu_p$ 

<u>There is no single model formulation for ANOVA</u>; the model tested in any given analysis is a function of the underlying experimental design of the data and the hypothesis being tested. Since a discussion of the many aspects of experimental design is not within the scope of this appendix, the discussion henceforth will be confined to the simplest of all ANOVA formulations. The case to be discussed is the fixedeffects model for a single factor experiment in a completely randomized design. Excellent references on both Experimental Design and ANOVA are Kempthorne (1952), Hicks (1964), and Mendenhall (1968).

#### The One-Way ANOVA Formulation:

Since the objective of analysis of variance procedures is to test the hypothesis of the equality of means, the linear model for the analysis is formulated in terms of deviations of the mean response for each of the treatments from the overall mean response or grand mean. These differences are known as "effects". The model for the experiment under discussion is:

$$Y_{ij} = \mu + \alpha_j + \epsilon_{ij}$$
(E)

where:

Y_{ii}

: response of the i'th observation at the j'th treatment level

$$\mu \qquad : \text{ grand mean} = \sum_{i=1}^{n} \sum_{j=1}^{n} \frac{Y_{ij}}{n}$$

 $\alpha_j = \mu - \mu_j$  : effect of treatment "j"

• ij : random error of the i'th observation at the j'th treatment level

If the numerical example problem were to be formulated as a one-way ANOVA problem on the variable  $X_1$  (by ignoring the  $X_2$  and  $X_3$  dimensions of the predictor set) there would be 6 data points at each level of  $X_1$ . The appropriate vectors for the model are shown in Table B-6.

### TABLE B-6

Note that there are only two unique data patterns with which to estimate the three parameters,  $\mu$ ,  $\alpha_1$  and  $\alpha_2$ . Hence m < p and the situation exists in which a unique value for the minimum ESS can be found but unique values of the parameters cannot be found. Whereas the specific purpose of ANOVA is generally the comparison of  $\alpha_1$  and  $\alpha_2$ , this is an unacceptable situation. In actuality, the model that was presented in equation(E) is not complete. The additional restriction of

$$\sum_{j=1}^{p} \alpha_{j} = 0 \qquad (F)$$

is also imposed on the model. This restriction is listed by many authors as an assumption and its actual function is not realized by many students. In reference to Searle's (1971) categorization of restrictions, both viewpoints 2 and 3 appear to be proper interpretations and the ANOVA model is actually composed of both equations E and F.

Since the ANOVA technique is directed at testing of hypothesis and ultimately results in the application of statistical significance tests, the statistical assumption that  $\epsilon_i \approx \text{NID}(0, I\sigma)$  is generally imposed.

## Considerations in ANOVA for Multiple Predictor Experimental Designs:

ð

孟

Ċ,

ø

In the application of ANOVA, certain simplifications in interpretation accrue when the total variance in the data set can be divided directly into independent portions. This can be accomplished with simplified mathematics if the experimental design for the data set is factorial with equal cell frequencies. Although the absence of this computational nicety does not prohibit application of ANOVA, it does diminish the advantage of testing the hypothesis of equality of means via ANOVA. In the latter case, the test of hypothesis by comparisons of the ESS obtained from full and restricted regression models is a much more attractive alternative. Again it is to be emphasized that the relationship between an ANOVA formulation and a specific regression formulation of any problem is dependent upon the hypothesis to be tested and the experimental design for the problem

As an example of the equivalence of the data obtained by the traditional calculational procedures for ANOVA and the use of full and restricted regression models, consider the data for our numerical problem. Table B-7 shows the ANOVA table that resulted from hand calculations with the formulae from Hicks (1964). Table B-8 shows the output from the EDSTAT-J computer program for the full model shown in Table B-5 and a restricted model in which the  $\beta_4$ ,  $\beta_5$ ,  $\beta_6$ , coefficients were restricted to be 0. This restriction tests the hypothesis that the interaction effects are all 0. The full model is:

$$Y = \beta_0 U + \beta_1 X_1 + \beta_2 X_2 + \beta_3 X_3 + \beta_4 X_1 X_2 + \beta_5 X_1 X_3 + \beta_6 X_2 X_3 + \varepsilon^{(1)}$$
  
the restricted model is: 
$$Y = \beta_0 U + \beta_1 X_1 + \beta_2 X_2 + \beta_3 X_3 + \varepsilon^{(2)}$$

Note that the difference in ESS for the two models is:

$$ESS_{R} - ESS_{F} = 320$$

which is the same as the difference between the Total Sums of Squares and the Main Effect  $(X_1, X_2, X_3)$  Sums of Squares shown in Table B-7.

## TABLE B-7

ANALYSIS-OF-VARIANCE TABLE

Source	D. F.	S. S.	M. S.
x ₁	1	1704	1704
x ₂	1	320	320
x ₃	2	168	82
x ₁ *x ₂	1	٩	9
x ₁ *x ₃	2	292	146
x ₂ *x ₃	2	18	9
x ₁ *x ₂ *x ₃	2	J.	. 5
Total	11	2513	

*Subtotal = 320

Š.

Ŷ

ġ,

* *

្សូរណ៍ ក្រុំ ខ្ញុំ

367

25

The Multiple Classification Analysis (MCA) Model and Procedure:

The MCA model was developed by the personnel at the Institute for Social Research at the University of Michigan to assess the relative importance of individual predictor variables in the situation where the predictor variables were correlated. It is a specialized categorical or binary regression model in which the elements of the X-matrix represent group membership of data cases within a particular category of the variables. Whereas Ward (1962)

i

*

ų

Å.

ι.

1

: A

аны 1.5 1

∙.∦ _}}

> di A

ું ર

ý.

÷.,

. .

40

. . . . . . . NOUEL NUMBER 1 ITENATION TULERANCES _ UDOUDLUDOD ....... .... CRITENION 9 PREDICTORS VAR-NO. DELTA A20. BTHANY BTHANRS ENHON MSG PROR **ITF**M .67817 .6782 .8595 .6782 18595 .0010 80.867 1 -.82341 1 39.237 • +4257A Z 6 30+728 27+829 24+240 5 -+16320 .88307 .9163 +9508 12395 3 . •91140 •92282 +33986 .6364 +4444 -1484 1 •33986 5 =•07960 4 5 . +4780 •3081 . 6642 . -231 #54.# .99994296 . .1433 ENNOR GUM OF SQUARES. Standanu Enror of Estimate .1693 BASEU ON 7 PREVICTORS. . VAN-NO. RAW WT. STO. WT. ENROR ENTRY VALIDITY .001312 .0000 1 1 .246603 .0005 -3574 7.19+8 5 2 .2500 3 11-7929 1+016438 +0004 . -1-7020 --128170 -7-9944 -1-315117 -- 3290 ···0009 -128170 6 -- 3484 • 0 u o 3 3 5 201242 ž ++528 .350263 -.0004 6 1.2432 UNIT VECTOR WEIGHT . . . . . . HODEL NUNBER 2 .....AIU .... 9 CRITERION 1 - 3 PREDICTORS PROR ITEA VAR-NO+ DELTA STRANY STHAND ENHON HSQ P59. 6782 8057 1 -. 82351 2 . 35705 .6782 .8057 .0010 80.867 54,259 67817 80566 12 .25857 87252 .0725 +8725 40.042 .0767 3 3 3 KRAGE SUM OF SOUARESS Stangand Error of Estimate 320,3333 . 6-3278 BASED ON 4 PREDICTORS. VALIDITY VAR.NO. RAW WT. STD. WT. ERROA ENTRY -23.6333 10.3333 3.0000 -.#23514 .357048 .258571 0000 1 2 3 123 3570 . 28.5000 UNIT VECTOR WEIGHT €2 .

had pointed out that Hoffman's (1960) measure of Relative Weight was meaningless in the case of correlated predictors, they sought a measure that would reflect an adjustment for intercorrelation among the variables. Toward this end their procedure calculates "adjusted means"⁵ for each category of each predictor.

**و** 

a**l** 

. A

18

<u>A</u>

ئة. 195

5. 6

de.

One difference in the MCA formulation than that commonly used in binary variable regression exists relative to the restriction that is used to insure a full rank X-matrix. In the case of a binary regression formulation where the inclusion of the constant term causes the X matrix to be less than full rank, the model is usually reparameterized so as to include only p-1 categories, making the constant term identical to the mean response for the omitted category. This has the impact of referencing all of the other category "effects" to the mean response for the omitted category instead of the grand mean. In the case of MCA, a full rank X-matrix is attained by restricting the weighted sum of the category means to zero, i.e.:

$$\sum_{j=1}^{p} n_{j} \alpha_{j} = 0 \qquad (G)$$

The algorithm used to solve the parameters of this model follows an iterative procedure known as the "sweepout" method

Adjusted means are estimates of what the mean value for each category would have been if the predictor vectors had been orthogonal. In this sense, MCA adjusts for non-orthogonalities in the data (Andrews, Morgan, Sonquist, 1967).

(Anderson and Bancroft, 1952) wherein each of the coefficients of the normal equations is sequentially adjusted until further adjustments are below some predetermined threshold level. In describing the relationship between MCA and ANOVA, Andrews, Sonquist and Morgan (1967) describe MCA as a computerized version of ANOVA with unequal cell frequencies. The major difference again is in the restriction of the MCA model, equation G, versus equation F of the ANOVA model. It is pointed out that the difference in this restriction does affect the value obtained for the "treatment" effects in the case of unequal cell frequencies, but does not affect the reduction in the total sums of squares. MCA does not calculate "F-test" values but does output the appropriate sums of squares for such calculations.

The MCA output for the numerical example is shown in Table B-9. Sonquist and Morgan's (1967) description of the output parameters is as follows:

η;

- a : the deviation of the j'th category mean of the i'th predictor, adjusted for the effects of the other predictors
- n_i : the correlation coefficient indicating the ability
  of a predictor, as categorized, to explain variation
  in the dependent variable
  - : the proportion of the Total Sums of Squares explainable by a given predictor "i"

 $\beta_i$ : a coefficient analogous to  $\eta_i$  except based on the adjusted means for each category

5+19+500 262*000 S U M OF Y SQUARF 1705.750 3975.750 \$66.500 2010-500 3000-500 S U H OF Y SQUARF S U M OF Y SGUARE 2512.750 2192.417 320.333 +11-51 195.000 5491.500 ¢6.500 128.500 50 KUS 26-000 SUM OF 000-1**0** 000-69 000-54 ETA B Beta B \$∪H Q£ 1 67A = 857A = ETA 9 ETA 8 19961. ► MULTIFLE R-SQUARE . COEFFECIENT A: 1.J) COEFFICIENT A1 2.J 719.11--5.167 5.167 -5.000 1.000 4.000 COEFFICIENT A1 21J TOTAL SUM OF SOUARES Explained sum of Squares Restourt sum of Squares .67817 .67817 .12745 .06656 .06665 STANDAND DEVIATION SUM OF Y-SUUARE DEV. FROM Grand Mean CEV. FROM Grand HEAN 11.417 DEV. FROM Grand Mean -5,147 5,167 -5.000 1.000 SUR CT V ETA-SOUARE . BETE-SOUARE . ETA-SGUARE = Meta-Square = .93409 ETA-SQUARE -UETA-SQUARE -389424 **TABLE B-9** 440 ADJUSTED FOR DEGREES OF FREEDUM - MULTIPLE M = ●●● UNADJUSTED FOR DEGREES OF FREEDUM - MULTIPLE M -71 î С L A 5 S H E A K 12.000 С L A S S H E A N 28.147 C L A S S K E A N 11.250 17.250 20.250 16.250 11.683 21.417 1764.08333 1764-08333 526.3355 520.3255 142.96004 168.93999 PER CENT 90°9 PERCENT 50.0 50.0 PEA CENT NO. OF USTA UNITS ANALYZED . . WEIGHT VARIARLE SUBSCHIFT DEP. VARIAN E SUBSCRIPT SUN OF #EIGHTS 9.0 9 SUM OF HEIGHTE SUM OF WEIGKTS ê.9 6.9 0 0 0 4 4 4 . . . . . UNADJUSTED DEVIATION 35 ADJUSTED DEVIATION 55 UNADJUSTER DEVIATION 55 ADJUSTED DEVIATION 55 UMADUNSTED DEVIATION SS AQUUSTED DEVIATION SS SUM OF WEIGHTS NO OF CASES GRAND MEAN -0-0 ND OF CASES NO OF CASES ۰ ï N X ŝ C00E X( 1.J) C00E X1 2+J5 X( 3.J) PRECICIOR PREGICTOR PREDICTOR ------

.82351 .82351

.35705

ં હુ

i at ix

. .

a

4

ð

Ą ŝ

1

230

.87252

MULTIPLE R-SUUARF #

β²_i

ol yrum : a measure of the ability of a predictor to explain variation in the dependent variable after adjusting for the effects of the other predictors: Note: This is not in terms of percent of variation explained and the sum over all predictors is not constrained to be less than or equal to unity

R²: the multiple correlation coefficient which indicates the proportion of variance in the dependent variable explained by all of the predictors together

The net advantage of using MCA instead of using a standard regression routine to do binary regression appears to lie in the mechanical convenience of defining the predictor matrix and in the fact that the deviations of the mean category responses for all categories are referenced to the grand mean as opposed to 'p-l' categories values being referenced to the p'th category mean.

The Automatic Interaction Detection (AID) Technique:

The AID technique fulfills a different role in the modeling process than the previously discussed techniques. AID is a "model seeking" technique and as such does not presume a specific structure to the data. AID partitions the data set via a stepwise one-way analysis-of-variance procedure and results in a model of the form

$$Y_{j} = \sum_{i=1}^{n} b_{i} X_{ij} + e_{j}$$

where:

s ₽

87.1

# 605

Y : response of the j'th case
b : the group mean of 'n' mutually exclusive groups
X : a binary indicator of group membership
e : the random error of the j'th case from its group mean, b i

The unique feature of this model lies in the implicit application of the restriction:

$$\sum_{i=1}^{n} X_{i} = 1 : \text{ for all } j$$

In terms of Searle's (1971) categorization of restrictions, this restriction is an inherent part of the model and results from the procedure of sequential "de-clustering" the data units until "n" mutually exclusive groups exists. This de-clustering takes the form of a binary split of the data set based upon that division that will result in the greatest reduction in the error sums of squares. Anderberg (1971) aptly characterized AID as a "monothetic divisive cluster analysis technique" in that it defines clusters by performing sequential binary splits on one variable at a time.

It is important to recognize that AID is nothing other than a particular restricted version of the general linear model. The claim by the authors, Sonquist and Morgan (1964) and Sonquist (1970) that the model is different because it does not require "additivity" could better be stated as:

> AID is a specific case of the general linear model in which the assumption of additivity is trivially satisfied due to the restriction of  $\sum_{i=1}^{n} X_{i} = 1$  for each of the 'j' data units.

Figure B-1 shows the AID-Tree for the numerical example problem and Table B-10 shows the least squares estimates for the equivalent regression model. In order to reparameterize the AID model into an equivalent regression model, the vector for group 7 was omitted. Therefore, the constant term of the regression output (1) reflects the group mean for group 7 and the other coefficients (2, 3, 4, 5) reflect the deviation of group means of groups 8, 9, 4, and 6 from the mean of group 7.

Being a "model seeking" technique, AID functions to analyze and display the data set in such a way that the latent characteristics and predictor relationships of the data set are emphasized. This emphasis is accomplished by the construction of a binary tree, the branches of which reflect the predictors, and categories of those predictors, that explain the most variance in the response variable. The structure of this tree represents a visual presentation of the structural relationships among the predictors. The analysis of such trees and the hypothesis of models there from is discussed extensively in Chapter IV.

The output of the AID technique is a set of mutually exclusive clusters of similar data units which best explains the variance in the data. Other statistical parameters relative to the variance explainable by each variable in each branch of the tree are also output.

_

Note that there is a factor of 10 difference in the numbers appearing on the AID-Tree and those in the corresponding regression output. A restriction to integer numbers in AID required multiplication of the criterion vector of the numerical problem by a factor of 10.

ССВОЦР 7°651441 МЕАNE 405,00 RSG = .908 6680UP 7°651441 МЕАNE 405,00 RSG = .908 002 5 %0.= 40.00 РКОВ: .002 6 %0.= 8418 COUES 1 2 %2 4 %0.= 73.64 РКОВ: .0140 8 %0.= .0140 0045 2 3 %0.= 73.64 РКОВ: .0140 0045 2 3 %0.= 73.64 РКОВ: .0140 0045 2 3 %0.= 30.60 РКОВ: .002 0045 2 3 %0.000 РКОВ: .002 0045 2 3 %0.= 30.60 РКОВ: .002 0045 2 %0.002 0045 2 %0.= 30.60 РКОВ: .002 0045 2 %0.= 30.60 РКОВ: .		00V 4-FINAL MEANS 105 85 0 85 0 85 0 85 0 85 0 85 0 85 0		ску Гарексана (1975) 1017 9 Г [ Ма]. ЖЕ АМ. ГР. С. В. С.		иктерствинатерствина. ЭШР Пактика Пактика Пактика Б.С.а. 24.04 РОПа 5200 КСПСТОК 2 Хд ОШБ5 О Хд С.С.С.5 24.04 РОПа 52205 С.С.С.5 24.04 РОПа 52205 С.С.5
		•6580UP •680UP • PREUICI • COUES				2000 2000 2000 2000 2000 2000 2000 200
	201.67 950 = 201.67 950 = 104.03 9208=		८ स २३ 0 0 2 2 2 2 2 3 3 3 3 3 3 4 4 4 4 4 4 4 4 4			÷ • • • • • • • • • •
	• • • • • • • • • • • • • • • • • • •	• • • •	● ● ● ■ ■ ● ■ ■ ● ■ ■ ■ = ■ = ■ = ■ = ■ = ■ = ■ = ■ = ■ =	••••		
	64000 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		GROUP 1 680UP 1 6 N 12 12		668000 3 66800 3 8 8 8 98601 CTOR 1 • CODES 1	

The statust assessed the Aline and the Review of States (States)

.....

.

9

0

NAMES OF TAXABLE PARTY.

6

234

4

... ...

ş

ø

Ø

MODEL NUMBER

fin the statistic data is not

••••• JIEHATION TULERANCE= .0000010000 9		PSQ.       BTRANU       BTLANRB       EMMOM MSQ       PROF       ITER         #3778       #378       #4378       141,272       0191       1         *3718       #376       #4378       141,272       0191       1         *3718       #376       #4378       141,272       0191       1         *3018       55055       #3955       75.333       0123       2         *88964       5043       .3579       34.601       0093       3         *94425       5109       .2764       20.011       .0347       \$         *95948       .4514       .2124       14.547       .1689       5	93761 MULTIPLE R97976406 Våres= 100.6668 Of estimate = 3.7922 båseu om 5 predictors.	T. STO. WT. ЕННСК ЕNTRY VALIDITY 38 1.222202 .0005 16616 1159420470007 22893 1876180700000 30077
ITERATION 9	ът 1	PSQ. BTRANV 43778 4376 73018 55055 68964 55053 94425 5109 95948 4514	3761 MULTIPLE H 1res= 1res= : estimate =	<ul> <li>ST0. WT. EF</li> <li>B -1.222202</li> <li>942047</li> <li>618070</li> </ul>
The objectives of the application of this analysis technique are different than the objectives of those techniques in which a "true model" is presumed and then statistics are generated for the testing of hypothesis. Consequently, there is really no direct compairson between it and the other techniques. AID is best described as a predessor to the application of these other techniques. However, AID does result in a valid model that is a restricted version of the general linear model. If the mutually exclusive groups into which AID divides the data set are found to be valid within the theoretical context of the data being analyzed, it can be viewed as a binary regression model about which various hypotheses may be tested.

A computer listing and user's description for the AID4UT/ AIDTRE program is included as Appendix C of this dissertation.

#### Appendix B: Exhibits 1-19

127

Ð

. Ŋ

60

ķ

ं . है *

"Х.

à

đ

Ľ,

۰.

.

- 2

_____. _______ This appendix consists of various forms, instructions and computer outputs used in this research and referenced in the body

of the dissertation. They are included here as supplementary exhibits.

Exhibit Title Page 1 Original Application Form 238 2 Initial Variables and Categories 239 3 **Revised Variables and Categories** 240 4 Data for 224 Sample Applications 241 5 Coded Predictor Data Sheet 243 6 Field Test Instruction Set and Coded Application Form 244 7 Data for 150 Cross-Validation Applications 249 8 AID4UT Output for Judge #1 251 9 AID4UT Output for Judge #2 256 10 AID4UT Output for Judge #3 261 11 AID4UT Output for Judge #4 266 12 AID4UT Output for Judge #5 272 13 AID4UT Output for 4/5 Vote 278 14 AID4UT Tree for Judge #6 281 15 AID4UT Tree for Judge #7 284 Hit/Miss Table--4/5 Policy with Credit Rating 16 288 17 Hit/Miss Table -- Composite Policy with Credit Rating 290 18 Hit/Miss Table -- Tentative Approval without **Credit Rating** 292 19 Hit/Miss Table--4/5 Policy (Local Model) with **Credit Rating** 294

237

#### Exhibit 1 ORIGINAL APPLICATION FORM

标志

ð

1

Latmar Savi International Savi Net Amount Required 8	ilga Asanciation at of Asservices) a loss under the Consumer				
Lamar Savi Inaroby upply for membership and Not Amount Required a Lifentification: Name Addom	ing a As ancia tion				
Addem	a loss under the Consumer		Date:		
Net Amount Required 8 Lieutification: Name Address(Securit) All present address years, To		<b>Credit Code for</b>	the purposes a	nt aut below:	
Liestlication: Name		1	Number of Mou	sta#	
Address					
Al present address					and the second
-	(Lity)	An		mied 🖸	
Treview address					
Nume of valo				N	unber of ependums
M Millioner: Serial No.	If Civil Service, Bede	e Ne		(Age) lec. No	
Name and Address of Relatives Not Li	ving with Applicant:				
My Employment or Business in	Self-unployed by []	Manas at 7		I	Post
Bushum Address				niegkens	
Kipsi of Rusings		·····			
Present Salary or Not Income from in	dama 8	per wennte 🖸 🛛 per	ymer ()		
Wile's Salary 8	per month (1. ser ver (1		- Cihar 1-	anna (Pist) A.	
Win's Residence				<b>for</b> u	et
Sundander England bis					~
	Here it and it a	a und Adderes	•		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
I have [] reasived credit [] c	redit applications preding (	n the following			
l		S			
· · · · · · · · · · · · · · · · · · ·					
My Bank is					
I have debts outstanding as fails					
Danadas Daita	To When Ladabad	Date	Orienal	Prosent.	Manadaba Paramata
			1.		
			1		
<u>l</u>				l	1
The proceeds of this loan will be	s used:				
To improve seal property on which	a one to four fundy dwalling [	3 is (3 is not (3	will be inceted a	i fellown	
(a) Improvements to be state					
(h) Preparty Identification: Lat	Black	•			
Address		· · · · · · · · · · · · · ·	Type of Structu	Ni	
My this is by C contract to p	urchase dated	(peen			
(a) Process Liens against property	. Amount &				

-

#### Exhibit 2

### ORIGINAL PREDICTOR VARIABLES AND CATEGORIES

#### Variable

#### Categories

< 23, 23-30, 30-50, 50-63, > 63
Married, Divorced, Single
No, Yes
Eligible, Ineligible
Yes, No
< 2, 2-10, >10
Rent, Buying, Own, Rent Free
Unskilled, Self-employed, Military, Executive, Skilled
<\$200, 200-400, 400-500, 500-600, 700-800, 800-1000, > 1000
Yes, No
Good, Bad
< 500, 500-1000, 1000-2000, 2000-3000, 3000-4000, > 4000
< 150, 150-250, 250-400, 400-600, 600-800, > 800
6-12, 15-18, 24, 36
Necessity, non-necessity
< 10%, 10-20%, > 20%
Poor, Unevaluated, Medium, Good
Bad (0~2.4), Good (2.5-5)
Approved, Disapproved

*Variables that were later redefined or re-categorized.

ک

74

. .

AP.

. 5

er.

4.6.1.

Exhibit 3 **REVISED PREDICTOR VARIABLES AND CATEGORIES** 

	Variable	Categories
1.	Age:	(1) (2) (3) (4) (5) <21, 22-24, 25-30, 31-40, > 40
2.	Marital Status:	(1) (2) (3) Married, Single, Divorced
3.	Local Family:	(l) (2) No, Yes
4.	Telephone:	(1) (2) No, Yes
5.	Time on Job:	(1) (2) (3) <2, 2-5, >5
6.	Time in Residence:	(1) (2) (3) < 2, 2-5, > 5
7.	Residence Category:	(1) (2) (3) (4) Rent free, Renting, Buying, Own
8.	Type Employment:	(1) (2) (3) (4) Misc., Unskilled, Skilled, Executive
9.	Income (monthly):	(1) (2) (3) (4) <\$400, \$400-\$600, \$600-\$900, > \$900
10.	Bank Account:	(1) (2) (3) (4) None, Savings, Checking, Other
11.	Financial Reference:	(1) (2) (3) Other, Minor, Major
12.	Loan Amount:	(1) (2) (3) (4) < \$300, \$300-\$600, \$600-\$900, >\$900
13.	Loan Term:	(1) (2) (3) 1 year, 2 years, 3 years
14.	Purchase Type:	(1) (2) Non-necessity, necessity
15.	Equity:	(1) $(2)$ $(3)<10%, 10-20%, > 20%$
16.	Credit Rating:	(1) (2) (3) (4) Poor, Unrated, Medium, Good

1

/inclosed.

n utr

MAN'

R

w. 

124

. 🕻

Exhibit 4 -- DATA FCR 224 SAMPLE APPLICATIONS

ĥ

ń

	000000000000000000000000000000000000000
	-0
	~~~~~~~~~~~
\$\$\$\$#######\$#\$#\$#####\$#\$U\$############	0000000000 000 0
$\bullet \bullet $	000
MAMOOROHUMAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	
	222999999~020~0209999999999999999999999
\bullet	00000000 00 00
· · · · · · · · · · · · · · · · · · ·	0
000****00000*0*0000*0*****************	000
	~~~~~
ちゅうちゅうしょう しょうしょう しょうしょう しょうしょう くろうちょう くょうろしょ	~~~~~~
	NNN
***************************************	~~~~~~~~
	~~~~~

	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
	~~~~
	N-N
ម្មសុស្ថាមសុស្ថាកាការដល់អ្នកស្រុកសុការដល់កាលការលេងក្រោះ	***********
「「」」「「」」「」」「」」「」」「」」「」」「」」」」」」」」」」」」」	
The second supportant of the second s second second se	**************************************

مستعدية أسالهم والا معام الم

ldentifi...thon 2/2/2/1/1. ote 4/5 vole w/o credit Judge #7 Judge #7 Refrial Judge #1 w/credit Refrial Judge #4 w/credit Refrial Judge #5 w/credit Refrial Judge #5 w/credit うくひょう ひはのし ちょりょうじょう しつりつ よしひょう -------Culumn J. **** * 7 9 3 THE REPORT OF A dentification Purbicane Type Equity Equity Credit Rating Gredit Rating Gredit Rating Judge 15 w/o credit Judge 15 w/o credit Judge 15 w/o credit Judge 12 w/credit Judge 12 w/credit Judge 13 w/credit Judge 15 w/credit Judge 15 w/credit Judge 15 w/credit Judge 15 w/credit Judge 16 w/credit J/credit 「うしちででももてかすぎましょう。 ひつり ひつちをす こうちゅうてい のうてい ちょうえん ちょうのくし のいうしょう *********** しょう うちょう しょう く う ご よう うり ご りゅう ご こうく う じょう うち う ごう しょう ごう ~~~~~ <u>Hertificetton</u> Case NR Actual Reapuree Aartial Statua Local Phone Jobinne Reatima Reacar Type Entployment NMAMMANNA ANNANA NANI AMMA NAI NANANANANA ANA Financia' Ref. Loan Am.. Loan Time ******** come うう アダリ しょう アボア ウェーク アジョン フラリー しろう しょう こう アン・アン・アン ank. ****** Column 10 イェンドー クリット しませい ひま 非常 ひまええる うくうきん ひままえや ****** - 0 \mathbf{R} \mathbf{N} \mathbf{n} - + 64 B. - **- - - - - -**

2

.

4,1

٩,

Ţ

,

100 - 100 -

. .

1.1

8

ð 6

, 5 _____) ______

.--nİj , **Ø** j

ar * **F**

. 1

Ż

de la

ł

Exhibit 6

Procedures for Lamar Data Sheets

Field Test Instruction Set and Coded Application Form

3

During the month of July we will be running a fieldevaluation of a proposed computerized credit evaluation system. Based on extensive analysis of credit applications previously submitted to Lamar, the attached application sheet and explanation of categories have been developed.

After a regular application blank has been filled out completely, we ask that the person who would normally call in the credit report completely fill out the special credit form, being sure to put down the date, time and case number. The case number should also be put in the top left corner of the original application and an entry made in the notebook. If questions should arise relative to which category an applicant really belongs to for a given item, you should use your own best judgment based on the category definitions provided in the attached sheet.

These sheets will be picked up every few days by Mr. Gooch, who is doing the evaluation.

Case #____Date and Time of Appt._____Salesman #____ Age: 21 or less 22-24 25-30 31-40 over 40 Marital Status: Married ____ Divorced ____ Single ____ Local Family: No Yes Telephone: No Yes Time on Job: less than 2 years _____ 2-5 yrs. _____ over 5 years _____ Time at Residence: Less than 2 years __2-5 yrs. __ over 5 yrs.__ Residence Category: Rent free ___ Renting ___ Buying ___ Own Outright ___ Type Employment: Misc. __ Unskilled __ Skilled __ Executive __ Bank Acct.: None ____ Savings ___ Checking ___ Ck/Sav.___ Finan. Ref.: Other ____ Minor ___ Major ____ Income: Under \$400 4-\$600 6-\$900 Over \$900 Amt. of Loan: Under \$300 ____ 3-\$600 ____ 6-\$900 ____ Over \$900 Equity: 10% or less ____ 10-20% ___ Over 20% ____ Loan Term: 1 year ___ 2 years ___ 3 years ___ Purchase type: non-necessity ____ necessity ____ TO BE FILLED OUT BY LOAN OFFICER Credit Rating: Poor Uneval. Med. Good Decision: TD ____ Okay ____ Loan Officer Initials ____ Date and Time of Decision Special Consideration?

Category Explanations

- Case #: A sequential number assigned to this application for bookkeeping purposes.
- Date and Time of Application: Date and time application would be called in for Lamar approval.
- 3. Salesman #: Salesman making sale.
- 4. Age: Age of applicant.

a

1

- 5. Marital status: Current status.
- Local Family: Close family, mother, father, children living within 50 miles of Austin. Yes, otherwise, no.
- 7. Telephone: Self explanatory.
- 8. Time on Job: Total time in current job or current job plus last previous job if previous job was the same type job.
- 9. Time in Residence: Total time in current home or in current home and last previous home in Austin.
- Residence Category: Rent-free--living with parents or relatives; rest are self explanatory.
- 11. Type Employment:

<u>Misc.</u>: Jobs which require minimum training or may be seasonal or somewhat unsteady. Examples are entertainers, retired sople, people on welfare or social security, sports people, bellhops, circus and carnival workers, dishwashers, farm laborers (itinerant), garbage collectors, hod carriers, hospital aids, janitors, laundry workers, military E-3 and below, parking attendants and porters. Include pensioners, or students with an income under \$250 monthly. <u>Unskilled</u>: Jobs which require little training but which are steady. Examples are apprentice tradesmen, bartenders, cab drivers, clerks, clerk-stenos, collectors, cooks, dancers, factory assemblers, factory machine operators, hair dressers, laborers, longshoremen, maintenance men, merchant seaman, military enlisted E-4 and E-5, miners, moulders, musicians, oil field roustabouts, painters, paper hangers, radio announcers, salesman (under five years), service station attendants, truck drivers (short haul), typists, waiters, warehousemen and watchmen.

<u>Skilled</u>: Jobs normally involving considerable training skill or education. Supervision responsibilities limited. Examples are insurance adjusters, artists, auto and airplane mechanics, bookkeepers, bricklayers, bus drivers, cabinet makers, cable splicers, carpenters, chefs, civil service workers (grade WB-9 and above), credit managers, dental and laboratory technicians, die makers, draftsmen, electricians, engineers, manufacturer's representatives, firemen, food checkers, glaziers, heavy equipment operators (cranes, bulldozers, etc.), inspectors, instrument repairmen, insurance agents, interviewers, lathe operators, lithographers, machinists, mechanics, military enlisted E-7 and E-6, policemen, pressmen, salesmen (over five years), secretaries, truck drivers (long distance), barbers and carpet layers.

<u>Executive</u>: Normally persons with line authority supervision five or more persons. Examples are accountants, airplane pilots and navigators (commercial), architects, chemists, crew chiefs, dentists, department heads, dispatchers, editors, electronic specialists, foremen, head chefs or maitre d'hotel in large restaurant or hotel, law enforcement officers, lawyer, manager (branch, office of sales), military officers, E-3 and above, oil drillers, physicians, programmers, project engineers, railroad engineers and conductors, supervisors, superintendents and teachers.

12. Bank Account Self explanatory.

23.5

ple

ie^{re}

13. Financial Reference: Previous and current creditors are listed and the following categories apply:

Code 3 (major): Credit references include major chain stores

and major banks or other large financial institutions. References do not indicate reliance on short-term, high-interest financing. Example references include: Household Finance, AVCO, GAC, Allied Finance, Sears, Wards, Scarbroughs, Joske's, Dial Finance, all banks, savings and loans and credit unions.

<u>Code 2 (minor)</u>: Credit references include only local stores and businesses. References do not indicate reliance on shortterm, high-interest financing. Examples are Term Plan, American Finance, Aetna, Republic, local department stores, jewelry stores, oil companies, etc.

<u>Code 1 (other</u>): Credit references include only out-of-town sources, no reference at all, or heavy dependence on shortterm, high-interest financing.

- Income: Self explanatory--use total monthly income listed for husband and wife.
- 15. Amount of Loan: Total amount of contract.
- 16. Equity: Percentage down.

5

, M

÷

- 17. Loan Term: (0-17 months) = 1 year, (18-27 months) = 2 years,
 (28-36 months) = 3 years.
- Purchase Type: This tells whether major item in contract is necessity in a normal home or not. Examples are: <u>Necessities</u>: stove, refrigerator, washer, dryer, air conditioner, television and dishwasher.

Non-necessities: Sewing machine, stereo.

- Credit Rating: Judged credit rating from direct and Credit Bureau reports.
- 20. Decision: Approval or disapproval.
- 21. Loan Officer Initial: Initials of loan officer making decision.
- 22. Date and Time Approval or Rejection sent to:
- 23. Special Consideration? To be annotated if special consideration such as recourse or other unusual circumstances affected decision on this application.

Exhibit 7 -- DATA FOR 150 CROSS-VALIDATION APPLICATIONS

à

દ (સ

,

Ĵ

3 9 1

2.5	2 2		<u>ۍ</u> -	••	•	•	1 IN	٠	-	ŝ	-	n u	۱ - (w 4	W -)	N -		-	-		n +			• •	~ 0	•		i w		u	~	• •• •				
			•												N	~								~ ~			•	^	i N	•	•	•	Ū.	~					•
22	2.5	-	<u>.</u>	•	21	2.5		•	e 4	•	ų,	p w		n 4		-0	• •		0.0	-	• •		1.0	n #	~	• • •		- 0		~		~ •		~					
		-	•					Ŷ	4 4	4	4	•			~	rų.	^	1	~ ~	- 14	-			* 14		N	ġ.		m	Ň		~	i i						
			o ~		υ,			_								•~			• •				_		_						_			_					
-	N: NJ	-			- (.	• •	~	~ ~	, n,	~	- → ∧	, <u> </u>	~ ~	~		~				• •	~ ~	-	— — ⊷ ∿	~ ~			~~~				~~~		~			~ ~		
NN	NN	-	^	-	•	. -	• •4	A.	~	• ~•	NI -	- 0	1		~		~	• ••			• •	~	-		~	-	;	~ ~			_	~		NA			N -		
~~	~ ~	-	→ ^		N/	N -	- 14	N	~ ~	i Ni	Ν.	• 14			N f	a 💊	N -	• ^•		-	• ••	(N) =	-	- ~	~ ~	-		2	-	~~~	-	~ ~	1	~	1.11	4 14	N		
~ ~	~ ~	-	- ^	-	(× -	• ^4	N	~ ~	• ••	Ν.	- ^		~ ~	N; f	y N	Ν-	• •••			- 14	~ ~	-	~ ~	~ ~			~~	-	~ -	N	~		~	• • •	• ••	~ -	~	u
NN	~ ~	-	~ ^	17	- 1	× -	• ••	N.	~ ^	1	N 4	• •		~ ^	N I	4 M	~-	****	-		• ••	N:	-	N N	~ ^	-	å #1	~ ~	N	NN	-	~ ^	1 14	N 6	1 14 1	4 A	~ -	~ ~	- ن
	72		е п 		<u>,</u>		-	-	-	• •	0.		•			••••		• ••	~ •		• •				• *	1 48 44	10	~ ~	N.	n n	N	N 4	• **	**	• •	• ••	N N	~ *	u N
				1				1	4 6	• ^	N 4	* •		~ 4		-	~ 4	-	~ 4	~ ~ ~	• •	* ~	t fig	N 4	- ^		• •	- ^	N.	4 4	m	~ •	* *	0 0		*	4 N	~ .	•
11	1	Ľ.			~ .		Ľ	Ξ		2	1		1	25		17	11	1	11	1	1	1	1	99	99	11	-	-	0	? 1	9	11	1	-	-	-	ទុទ្ធ		2 69
NN	~~	-	~ ~		~ (• •	~	~ ~		<u></u>				~ ~		~ ~										1.67			* "	-	5	1.00	40.0	•••	••••	<u>۳</u>	6 1	р —
-		-		-	Na 1	• -			4.4				-	-				1.04										- NA - NA				~ ~		~ ~			~ ~ ~	2	
NN	w ~4	~	~ ~	ŝ	•	v n	• ~•	N 1	~	• ~•	~	u ~u	-	~	-	• •	N	-	- ~	- 40	• N	N					-	~ ~		~ ~	N.	~ ~		NA			N	-	
~ ~ ·	~ ~	~	- ^	N	∾.			-	- -	. –	- •	• ~	~	N -	n -	• ^•		• ••	N	- w	• ^1	NN	-		NN	-	1.01	n -	-	~ ~	•	n N		NA		• •	- 14	N -	
NM	~ ~	-		-		- ^		n,		• ~	-		-	~ ~1	~ '	۰	~ ~	•	-		•			N M			•		-	~ ~	~	-	-	-		•	~ ~	~	n 🗕
NA	- 14	N.	~ ^	i Ni	NI I	u -	• •	-	~ -	• ••			-			••		• ~ •	~ ~		• ^•	~ -	-		- ^	-	-	~ ~	~ (•-	٩,	NN	• ••	N٨		• •			• ••
***	~ ~	~		Ni	2		• •	2				u m	- 10	N #	~	1 11	A. A	-	<u>AI ()</u>	1 Au -	1	n N		~ ~	m 4	2			-	~ ~	-	~ +	e 🖷	**		• • •	~ -	**	2
	~ ~			-	-		1	2	~ "		2	1		~ ~	-	1 -	-	1	NN		5	<u>n</u> M	1 14		-		•••	m m	~	~	2	~ ~	-	-	-	• ••	m N	~	• •
								<u> </u>				:]			-		2-							~ ~	~ ~	N 4	-	n	*	~ ~	2	~	• ••	1	• M 4	•			~
AL IN C		-							~					~	-		~ ~	4 NI 4 AL	1. CO			~				21.5	1	~ *	***	N N	-	N 11	N	N N		1.0	4 N	-	4 m
- N -	n m	~		. Ne				~		_	A			-					-		-	1949 1949 1949 - 1949									~	***				•			
	~	-		-	~ •	م د	-				<u>.</u>						~-	-	NN	-									~ ~			~	~	2					
-	- 14	-	NN	-		u n	•	ny i	~ ~	1 14	N 4	• ••	-	~~	-	-	~			-	-	~ ~	-	N N	NN		-		A .	~~~	4	N N	-	NN		1.74	~ ~	AL	• ^4
- ~ ~	4 N	•	- ~	•	N 1		-	ny (~	• ••	~ ~	يہ ي	-	N a 199	~		N N	-	-	N -	• ••	N N	-	-	~ ~	-	- 14	A		- ~	•	-	-	-		-	~ ~	~	-
		-	.	n	-	- ^	i m	-		•			-	-4	-						• •		-	n n					-		N,		-	~ -		• •		-	• ••
~ * *	n in	W ·		~	~ *	• •	-	"		5	~ *	n sh	N y	ŝ	N 4	*	~ ~	• •	N +	ŝ	-	n '1		n vi	~ v	-	м.	4.4	m 1	^	2	N 17		vî 4	4.	n n	-	~ ~	•
22	2.2	•	22	2	2		-	-			9	1	23	31	21	1	33	-	~ *	• •		* !		è.	22	29	2	• 🖬	=:	==	23	<u>~ 1</u>	2	<u></u>	••	1	.	• *	0
22:		2	33	2			2	2		1	-	12	-			-		-			<u> </u>	20	1	202	22	1.2	2	ねた	2.		*	**	2	22	121	12	2 A N N	2.6	25
40	0 0	ŏ	22	-			-	2	~ 6	3		55				5	0 0			6 3		80		64	0. d		2	22	20	. <u>.</u>	16		5	22	23	15	2 82	2	23
																									P 1 1		-				_	~ ~		~~	-				•~•
.			<u>0</u>	•	<u>-</u> *	2	2	• •		-		-			9	-		2.	به د.		-		in s		n W			- 2		•		- =	-	22	4 -		• •		
- 2.					-		-		• •									•						•				•••				_						_	
- 2		-		•	•	1																																	
			. <u>.</u>	•				• •		ų,	m m	-	-	• ••	• 0		- 7	-	0 W	* *	-	4				un vo		- 10			~ •		ala i		• -		e m.	~ 0	
- 5- 853	35		51 502	00	, .	4	5	e v 4	n 165 7 17	5	m m	-			92		-02	31	9 N	.		1	្អូន		* ~	\$\$	33	33	- •	• •	~ 1	- =	1	39	:-	•	• 2	~ 3	
	35		212		*	; ;	\$ 2	• •	;	\$	m m	-					-1	51	9 ~	•••	~	11	្រុះ		**		::	;;	~ ^	~~	~ •	-			: -	•	• ~	~ 2	
			91 922 0 1	0 60			21 50			5 4 0 1	• •		n	•••	2 1 20	•		15			-		05			2 1 45		3		~	~ ~ ~				•			~ 51	
	221 22	2 2 1 48	91 902 911	1 0 90		54 122	221 50		5 5 0 4 8	5 * 0 1 1	* • • • •	2 2 1 3	221 3	•••	2 2 1 23		1 2 2 2 2 2 2	15 1 1 2		* 1 * *		2	2 2 1 2 2 2 4 1 5 2		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	221 45	2 2 1 44	25 1 52		2 2 1 2 2	2 1 2 2		2 2 1 16 2 2 1 16		2 1 7 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	•	2°2°	2 2 1 19 2	•
		2 2 2 1 48	91 912 91 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	06 0111		2 2 2 1 49	2221 50			5+ 0 1 1 1	5 1 7 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 1 3	2 2 2 1 2 2 2 2 1 1 1 1 1 1 1 1 1 1 1 1	••••	1 1 1 0 16	1 1 1	1 1 1 0 1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2	22 1 1 2 2		2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		2 2 2 1 54	2 2 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2			2 2 2 1 45	2 2 2 1 44	29 1 2 2 2		2 2 2 1 2 2	2 2 2 1 2 2 2 3		2 2 2 1 1E		2 2 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		5 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 1 19 2	•
	22221 47	2 2 2 2 1 48		0 0 1 1 1		2 2 2 2 1 49	22221 50		2122048	S* 01111	22221 3	2 2 2 2 2 2 2	22221 3		2122120	1 1 1 1 1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	22211 51	2 9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	22221 6		2 2 2 2 1 54	2 2 2 2 1 20 2 2 2 2 2 20			2 2 2 2 1 45	22221 44	2 2 2 2 1 6	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 1 2	2 2 2 2 2 2	2 2 2 2 2 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 1 1E		2 2 2 2 1 19		5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	2 2 2 2 1 1 6 2 2 2 2 2 1 1 6	
	2 2 2 2 2 2 1 47 1	2 2 2 2 2 1 48				64 12222	22221 50			111110 45	222221 3	2 2 2 2 2 2 2 2	222221		2 2 1 2 2 1 2 3 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5	1 1 1 1 0 1	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 1 1 51	111116 2	222221 4		2 2 2 2 2 1 54	2 2 2 2 2 1 20 2 2 2 2 2 1 50			2 2 2 2 2 1 45	22221 44	222221 6	222221	2 2 2 2 2 3 2	2 2 2 2 2 2 2	2 2 2 2 2 2 1 1 1 2 2 2 2 2 3 1 3 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 2 2 2 2 1 1E		2 2 2 2 2 2 1 19		2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 2 1 19 2	
	2 2 2 2 2 2 2 4 1 4 1	2 2 2 2 2 2 1 48				0 2 2 2 2 2 1 49	2222221 50		6 1 5 1 5 5 0 48		2 2 2 2 2 2 2 2 3 2 3	122221	222221 3		51 1 1 1 1 0 16 1 2 2 1 2 2 1 2 2 3	3111110	3 2 2 2 2 2 2 2 3	2 2 2 2 2 1 1 91		3222221 4		2 2 2 2 2 2 2 1 54				1 2 2 2 2 5 1 42 0 5 2 2 5 5 1 45	9 2 2 2 2 2 1 44	1222221		322221 2	1222221	41 1 2 2 2 2 2 0 5 5 5 5 5 5 5 5 5 5 5 5 5 5	1222221 JE	2111110 43	522221 19		2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 1 1 1 1 0 2 0 1 1 1 0 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
						2022222149						11222221				2311111	<pre></pre>	191922221		13222221 4						2122222145		21222221		2 [2 2 2 2 2 5 4] S	21222222		11222221 JE		4 2 5 5 5 5 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			2 0 1 1 1 1 1 4 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2	
						120222221 49						11122222111														-0 - 1 - 2 - 2 - 5 - 1 - 6		29 1 2 2 2 2 2 1 2 6-		2 2 2 2 2 3 4	- 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		-011222221 JE					-0 4 3 3]]]] 0 2	
						112022222149					2 2 4 2 1 1 1 1 0 4	E 12222111E3	7 2-0 4 2 7 7 2 2 2 1 3		2 1 - C 4 5 1 2 1 1 1 0 26 2 1 - D 1 1 2 2 1 2 2 1 23	P 2-C 2 3 1 1 1 1 0 1	5 0 0 1 5 5 5 5 5 5 1 5 1 5 1 5 1 5 1 5			2 2+2 1 3 2 2 2 2 2 1 4 2 2+0 2 3 2 2 2 2 2 1 2						2 1-0 2 1 2 2 2 2 5 2 1 45			2 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2-0+3-2-2-2-3 2	2		5-011222221 16					2 - 0 4 3 3 1 1 1 1 0 2 5 - 0 2 0 2 2 2 2 2 1 19	
 2 3 - 0 4 3 5 5 5 5 7 6 5 3 6 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 5 7 6 7 6						• 5 1 1 2 0 2 2 2 5 5 1 40 ·	3 2 3 6 4 2 2 2 2 2 2 2 1 50				2 2 5 4 2 1 1 1 1 1 0 3	E 12222211122.	* * * * * * * * * * * * * * * * * * * *		2 2 1 - 0 4 5 1 1 2 1 2 4 2 6 2 6 2 6 2 6 2 6 2 6 2 6 2 6 2 6	1 2 5-6 2 3 1 1 1 1 0 1	1 5 6 6 6 7 3 1 1 1 1 1 1 0 1 4 7 7 7 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	2 2 1-9 1 2 2 2 2 2 1 9		3 2 3-6 1 3 2 2 2 2 2 1 4			• • • • • • • • • • • • • • • • • • •			6 5 1-0 5 1 2 5 5 2 5 1 6 5 6 5 4-0 6 6 5 5 5 2 2 7 1 6 5	6 2 1-0 4 9 2 2 2 2 2 1 44		4 2 4 1 4 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 2 2 2 3 2 5 2 2 3 2 5 4 3 2 5 4 3 2 5 2 2 3 2 5 2 5 2 5 2 5 2 5 2 5 2 5	6 2 6-5 2 1 2 2 2 2 2 1 2	• 5 2 - 5 5 5 5 - 5 1 1 5 1 1 5 5 - 5 1 9 5 5 5 5 1 9 5 5 5 5 5 5 1 9 5 5 5 5	5 - 0 1 1 2 2 2 2 2 1 1 E	2 1 0 1 2 1 1 1 1 1 0 1 2 E E E E E E E E E E E E E E E E E E	<pre></pre>			[- 0 4]]]]]] 0 2 \ 2 5-0 2 0 2 2 2 2 2 1 19	
		2 4 2 3 0 1 2 2 2 2 2 2 9 0 1 2 4 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 1 3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		2 3 2 3-0 1 4 2 2 2 2 2 2 2 1 46	242112022222149	1 3 2 3 6 4 2 2 2 2 2 2 1 50		2211-04012122048		2 1 2 5+0 4 2 1 1 1 1 1 1 6 3	2 4 2 3 2 2 2 2 1 1 1 2 2 4 2	2 4 2 5-0 4 2 2 2 2 2 2 1 3	2221340111110	2 2 2 1-0 4 5 1 7 1 1 1 0 26	2125-623111110	2 1 5 6 6 6 2 3 1 1 1 1 1 0 1 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 1-0 1 2 2 2 2 2 1 9 9		2 4 2 5-6 1 3 2 2 2 2 2 2 1 4			2 4 2 2-0 5 0 2 2 2 2 2 2 2 3 50			2 4 2 1-0 2 1 2 2 2 2 2 1 45	2 4 2 1-9 4 9 2 2 2 2 2 1 44	24 21-2 2 2 2 2 2 2 - 2 4	2 4 2 6-2 2 2 2 2 2 2 2 2 1 1	2 4 2 2 4 3 2 2 2 2 3 3 2 4 2 4 2 4 3 4 3	2426-021222221 2	2 4 2 2 4 1 4 2 4 2 4 2 4 2 4 2 4 2 4 2	2 4 2 5-0 1 1 2 2 2 2 2 1 1E	2114-01311114 16 43	2425-45222221			2 1 1 1 - 0 4 3 3 1 1 1 1 0 2 2 4 2 5 - 0 2 0 2 0 2 2 2 2 2 1 19	
		2 2 4 2 3-0 1 2 2 2 2 2 2 1 48		06 0 1 1 1 1 2 2 0 2 2 2 2	22323-0-0142222221 46	12421120222221 49	31323-64222222150				2 2 4 2 5+0 4 2 1 1 1 1 4 3	E 1222221112421	12425-04222221 3		2 2 2 1 - C 4 5 1 1 1 1 0 26 1 2 2 2 1 - D 1 1 2 2 1 2 2 1 20	12125-02111110	2 2 1 5 6-0 2 3 1 1 1 1 1 0 1 1 1 2 1 2 3 1 2 3 1 2 3			8 8 3 2 3 4 6 1 3 2 2 2 2 2 2 1 6 1 2 4 2 3 4 0 3 3 2 2 2 2 2 1 2			2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 1 2 4 2 5-0 1 3 2 2 2 2 2 1 50			2 2 4 2 1-0 2 1 2 2 2 2 2 1 45	12421-049222214	29 1 2 2 2 2 2 1 2 5 - 1 2 5 2 2 3	1 5 4 7 4-9 2 5 5 5 5 5 5 1 1 1 2 4 3 5-6 2 1 2 2 2 2 2 1 1	22425-043222221 2	12426-021222221 2	1 2 4 2 5-0 1 0 2 2 2 2 2 2 1 18	22425-011222221 JE	2 2 2 1 2 1 1 1 1 2 1 4 1 4 2 5 2 1 2 1 4 1 5 2 1 6 4 3 2 5 2 1 2 2 4 3 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2	2 2 4 8 5 1 4 5 2 2 2 2 2 1 19		2 1 4 2 5 - 0 1 3 2 2 2 2 2 1 2 3	2 2 4 2 5-0 2 0 2 2 2 2 2 1 1 2 0 2	
			2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	06 0 1 1 1 1 1 2 0 2 2 2 2 2 2		1 2 4 2 1 1 2 0 2 2 2 2 2 1 49 49	1 3 1 3 2 3 6 4 2 2 2 2 2 2 1 50 2 1 2 2 1 2 2 2 2 2 2 2 1 50				122425+0422222221 3	E 12222211122+215	3 2 4 8 5-0 4 2 2 2 2 2 2 1 3		1 2 2 2 1 - 0 1 1 2 2 1 2 1 2 0 26	312125-62311110	1 2 2 1 1 6-0 2 3 1 1 1 1 1 0 1 3 3 1 2 3 2 2 2 2 2 2 3 3 3 3 3 3 3 3	312221-91222221191		2 1 1 3 2 3 5 1 3 2 2 2 2 2 2 1 4 5 1 1 2 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5			1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 1 1 2 4 2 5-0 1 2 2 2 2 2 1 50		322115150111116 70	2 2 2 4 2 1-0 2 1 2 2 2 2 2 1 45	3 1 2 4 2 1-0 4 9 2 2 2 2 2 1 44	39 [2 2 2 3 2 1 2 6 -1 2 4 2 3]	2 2 2 4 2 6-9 2 2 2 2 2 2 2 2 1 1 2 4 2 6-9 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	122425-04322223	1 2 4 2 6-5 2 1 2 2 2 2 2 1 2	312425-0102222222	2 2 2 4 2 2-0 1 1 2 2 2 2 2 1 1e	121210122111110 15			2 5 7 4 5 2 - 0 1 3 5 5 5 5 5 5 1 5 3 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	1 2 2 4 2 5-0 2 0 2 2 2 2 2 1 1 2 0 2	
		2122423-0122222226	24 12 2 2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7		1 2 2 2 2 3 4 0 1 4 2 2 2 2 2 2 4 4 4 4 4 4 4 4 4 4 4 4	2 1 2 4 2 1 1 2 0 2 2 2 2 2 1 49 49	3131323-642222221 50	<pre>1 C C C C C C C C C C C C C C C C C C C</pre>			1 1 2 1 2 5 4 7 5 4 7 2 1 1 1 1 1 6 3	E 122221112242121	2312425-04222221 3		2112121-04511110 16	1 3 1 2 1 2 5-6 2 3 1 1 1 1 1 0 1	2 1 2 2 1 1 6-0 2 3 1 1 1 1 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1		1 2 1 2 1 2 0 1 0 1 1 1 1 0 2	<pre>4 2 8 1 3 2 3~6 1 3 2 2 2 2 2 1 4 1 3 1 2 4 2 5~6 2 3 2 2 2 2 2 1 2</pre>			1 1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 2 1 1 2 4 2 5-0 1 2 2 2 2 2 1 50		332211515011111	2 2 2 2 4 2 1-0 2 1 2 2 2 2 2 1 45	1312421-0492222144	29 1 2 2 2 2 1 2 1 2 5 2 5 1 1 1 1 1 2 5 2 5	2212426-22222221	2 1 2 2 4 2 3 4 3 2 2 2 2 3 1 2	2132426-021222221 2	1 1 2 4 2 2-0 1 0 2 2 2 2 2 2 1 1 8 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 2 2 2 4 2 2-0 1 1 2 2 2 2 2 1 1e	2 3 2 1 1 4 9 1 3 1 1 1 1 1 9 16 16 17 1 3 1 2 1 3 1 3 1 3 1 3 1 3 1 3 1 3 1 3	1 1 2 2 4 8 5 1 4 5 2 2 2 2 2 2 1 19 3 1 2 2 4 2 1-0 0 1 2 2 2 2 2 1 1			2 1 2 2 4 2 3 4 2 3 1 1 1 1 0 2 2 2 3 2 1 2 4 2 3 4 2 3 4 2 3 4 2 3 4 2 3 4 2 4 2	
12122111102111201112021211 5 45 122225020425251 1 12222502042042551		4 2 1 2 2 4 2 3-0 1 2 2 2 2 2 2 1 48 1	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		1 1 2 1 2 2 2 3 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	12112421120222281 49	33131323-642222221 50	5 4 1 1 1 4 9 1-9 1 6 2 2 2 2 2 3 1 40 2 4 1 1 1 4 9 1-9 1 6 2 2 2 2 2 9 1 40			31312125=04211116 3	E 122222111E2+21E15	4 2 3 1 2 4 2 2 0 4 2 2 2 2 2 3 1 3 3 4 5 1 2 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		2112121-04511110 1645145122120	41312125-62311110	12122116-033111110 1	21312221-01222221 91	1112121-0362-05255551126 2	6 6 2 1 1 3 2 3 6 1 3 2 2 2 2 2 1 6 3 1 3 1 2 6 2 5-6 2 3 2 2 2 2 2 1 2		32132421-012222222154	3 1 1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 4 2 1 3 2 4 2 5-0 3 2 2 2 2 2 3 1 50			3222421-02122222145	4 3 2 4 2 -0 4 9 2 2 2 2 2 44		4 2 2 1 2 4 2 6-3 2 2 2 2 2 2 2 1 1 4 1 1 1 2 4 2 5-6 2 1 2 2 2 2 2 3 3	4 5 1 5 5 4 3 5 5 0 4 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	32132426-021222221 2	4 1 3 1 5 4 5 2 0 1 0 2 2 2 8 2 3 18 4 1 3 1 2 4 2 5-0 1 0 2 2 2 8 2 3 18	22222425-011222221	221321214-913111110 15	4]]22485[452222] 2312249]-03122222]			2 8 1 1 1 1 1 8 4 3 - 1 1 2 1 1 2 2 2 2 3 2 3 2 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 4 2 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 5 4 2 5 5 4 2 5 5 5 4 2 5 5 4 2 5 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 4 2 5 5 4 2 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 4 2 5 5 5 4 2 5 5 4 2 5 5 5 5	
		3 4 2 1 2 2 4 2 3-0 1 2 2 2 2 2 2 1 48 1	312232113-02113-12222		31122323401422221 48 4	212312421120222221 49	- 3 3 1 3 1 3 2 3 6 4 2 8 2 8 2 8 2 8 1 50 	324111491-012222214			3311225425-042222221 3	3213124231112242113	- 3 4 2 3 1 2 4 2 2-0 4 2 2 2 2 2 5 1 3 1 3 1 5 1 2 1 1 1 1 2 2 1 1 1 1 6 1		22112121-04511110 16 34211221221251 25	3 4 1 3 1 2 1 2 5 - 6 2 3 1 1 1 1 0 1	2121212455-0432222121			3442132354513222221 4		332132421-012222222154	3 3 1 1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 5 4 2 1 1 2 4 2 5-0 1 2 2 2 2 2 1 50		313322115150111116	3 3 2 2 2 4 2 1-0 2 1 2 2 2 2 3 42 3 4 5 3 4 5 3 4 5 4 5 4 5 4 5 4 5 4 5 5 5 2 2 2 2 2 2	341312421-94422221 44		. 3 4 2 2 1 2 4 7 9-7 2 2 2 2 2 2 2 1 1 1 4 1 1 1 2 4 3 6-7 2 2 2 2 2 2 2 2 1 1	3 4 5 1 5 5 4 3 5 5 6 4 3 5 5 5 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6	332132426-021222221 2	341312425-010222222	32222425-011222221 le	122132114-913111110 16	94]]22485]452222] 32312249]-031222221			2 2 1 1 2 2 4 2 3 4 3 1 1 1 9 2 3 3 2 3 1 3 2 4 2 3 4 2	
				06 0 1 1 1 1 1 2 2 2 2 2 2 2 1 2 2 1		1212112421120222221 49	. 3 3 3 3 1 3 1 3 2 3 6 4 2 2 2 2 2 2 1 50 				. + 7 3 1 3 1 2 1 5 2 6 4 2 1 1 1 1 1 6 3 . 1 3 3 1 1 2 2 4 2 5 0 4 2 2 2 2 2 2 1 3	E 1321211123421112411121	.]] 4 2]] 2 4 2 2 - 0 4 2 7 7 2 2 2]] .] 1 1 1 1 1 2 1 1 1 1 2 2 1 1 1 1 0 1		3342112121-045112110 16	1 3 4 1 3 1 2 1 2 5 - 6 2 3 1 1 1 1 1 9 1	<pre>% L Z L Z Z L I 6 -0 2 3 1 1 1 1 1 0 1 (4 3 4 1 3 1 2 4 5 5-0 4 3 2 2 2 2 2 1 23</pre>	1 2 2 2 2 3 - 2 3 2 3 2 3 3 3 3 3 3 3 3 3		2 3 4 4 2 1 3 2 3 4 5 1 3 2 2 2 2 2 2 1 4 - 3 3 3 1 3 1 2 4 2 5-6 2 3 2 2 2 2 2 1 2			- 2 3 3 1 1 2 2 4 2 2-0 5 0 2 2 2 2 3 50 3 5 4 5 1 3 2 4 2 5-0 1 3 2 2 2 3 1 50				- 3 3 4 1 5 1 2 4 2 1-9 4 4 2 2 2 2 1 44 	39 2 5 5 5 1 - 2 5 1 - 5 5 5 5 5 5 1 1 5 5 5 5 5 5 5 5 5]] 4 2 2 1 2 4 7 6-V 2 2 2 2 2 2 2 1 4 1 4 1 4 1 7 1 7 4 9 5-0 7 1 7 2 7 2 7 2 9 1 9	3 3 4 5 1 5 2 4 5 2 4 5 2 5 5 5 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6	+]] ? } ? + 2 + 2 + 0 2 2 2 2 2 2 2		3 3 2 2 2 2 4 2 2-0 1 1 2 2 2 2 2 1 PE	1122132114-913111110 15				1 2 2 1 1 1 2 4 2 2 4 2 3 1 1 1 0 2 1 2 2 4 2 3 1 1 1 0 2	
				222212222222222222222222222222222222222	2 2 2 3 3 1 1 2 1 2 2 2 3 - 0 1 2 3 - 0 2 3 - 0 2 3 4 7 4 5 4 5 4 5 5 5 5 5 5 5 5 5 5 5 5 5	113212312421120222281 49	5 3 3 3 3 1 3 1 3 2 3 4 6 4 2 2 2 2 2 2 1 50	1 1 1 1 2 2 4 1 1 1 7 9 1 0 1 0 2 2 2 2 2 2 1 7 8 1 7 1 7			3 2 4 3 2 1 3 1 2 1 2 5 4 2 5 0 4 2 1 1 1 1 6 3	E 12222111224216136114	3] 3 4 2 3 2 4 2 2 0 4 2 2 2 2 2 2 1 3 2 3 1 3 1 5 1 2 1 1 1 2 2 1 1 1 0 1		2 2 1 2 2 1 1 2 1 2 2 1 -0 4 5 1 1 1 1 1 0 26	131341312125-62311111	2 2 1 2 1 2 2 2 1 5 9-0 2 3 1 1 1 1 1 0 1] 2 3 6 6 2 3 2 3.54] 2 2 2 2 2 2 2 1 6			2 2 3 3 1 2 2 4 2 5-0 5 0 2 2 2 2 2 1 50 \ 4 1 5 4 2 1 2 4 2 5-0 1 2 4 2 2 4 1 2 4 2 5-0 1 2 5 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2			1 3 3 3 3 3 5 5 5 4 5 1-0 5 1 5 5 5 5 5 5 1 45 1 3 3 3 5 5 5 1 5 5 5 -0 6 6 0 5 5 5 5 5 1 45	143341512421-91922223 44 143351512425-91925223 44		1 1 1 3 4 8 2 1 2 4 7 9-7 2 2 2 2 2 2 1 1 7 4 4 1 4 1 7 1 7 4 9 5-0 7 1 7 2 7 2 7 2 7 7 7	2 1 2 2 2 2 5 4 3 2 5 2 5 1 2]] 4]] 2] 2 4 2 6-5 2 1 2 2 2 2 2 1 2	3 3 3 4 3 7 4 5 4 5 5 4 9 5 7 4 7 4 7 4 7 4 7 4 7 4 7 4 7 4 7 4 7	33322222425-011222221		12334112248514514322221 19 114323122421-001222221 1			2 0 1 1 1 1 1 4 0-1 1 7 7 1 1 2 7 2 7 2 1 1 0 2 7 2 7 2 1 1 0 2 7 2 7 2 7 1 1 2 7 7 7 7 7 7 7 7 7 7 7	
2221112112911222111291		1 3 4 3 4 2 1 2 2 4 2 3 0 1 2 2 2 2 2 2 1 48 1		22 1 2 2 1 2 2 2 2 2 2 2 2 2 1 1 1 1 0 90	< < < >	1 31 32 1 2 31 2 4 2 1 1 2 0 2 2 2 2 3 1 49 4	3 2 3 3 3 3 1 3 1 3 2 3 6 4 2 2 2 2 2 2 1 50	3 1 3 × 1 1 × 0 -0 × 0 × 0 × 0 × 0 × 0 × 0 × 0 × 0 × 0			3 2 4 3 2 1 3 1 2 1 2 5 4 7 5 4 4 2 1 1 1 1 4 3	E 1222211122+2121321+1	2 2 3 3 3 4 2 3 1 2 4 2 5-0 4 2 7 2 2 2 1 3		1221221111221-01-04511110 16	1331341312125-5251211110	2 2 2 1 2 1 2 1 2 1 2 9 - 0 2 3 1 1 1 1 1 0 1 1 1 2 3 4 3 4 5 4 3 2 2 2 2 2 2 2 3 3 4 3 2 2 2 2 2 2 2			33323442113254513222221 4 133331112425-023222251 2			9 3 2 2 3 3 1 1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 1 9 4 3 5 4 9 1 1 2 4 9 5-0 1 9 9 9 9 9 9 9 1 50			2333332222421-02124242122222142	1 3 4 3 3 4 1 3 1 2 4 2 1-9 4 9 2 2 2 2 2 1 44 1 3 3 3 3 3 1 3 1 4 2 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5		2 3 3 3 3 4 2 2 2 2 4 2 6-2 2 2 2 2 2 2 2 2 1 1 2 4 1 2 4 2 6-2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	122334212242544322223	2] 4] 3 2 5 2 4 2 6-5 2 2 2 2 2 2 2	33334131242541642222223	233322222425-011222221	221122132114-91311114 16	[323]4] [2242[45]45] [452222] [9			2 0 1 1 1 1 1 4 0-1 1 7 1 1 2 1 1 2 2 2 2 2 1 1 2 1 1 2 2 2 2	
222112212221162111221212121212121212121		3334342122423-012222222148			2 2 2 3 3 3 1 2 2 3 2 3-0 1 4 2 2 2 2 2 3 3 3 1 2 2 3 2 3-0 1 4 2 2 2 2 2 3 1 4 4	1 31 32 12 1 2 4 2 1 1 2 0 2 2 2 2 2 1 49 49	3 1 3 2 3 3 3 3 1 3 1 3 2 3 6 4 2 2 2 2 2 2 1 50				3 3 4 3 1 3 1 2 1 2 2 4 2 5 4 4 2 1 1 1 1 4 3 3	E 1222221112242121222221	1 2 1 3 1 3 4 2 3 1 2 4 2 5-0 4 2 2 2 2 2 2 1 3		1 2 2 1 2 2 1 1 2 2 2 1 -0 4 5 1 1 1 1 1 0 26	1 3 1 3 1 3 4 1 3 1 2 1 2 2 - 2 2 - 2 2 3 1 1 1 1 1 9 1	3 2 2 2 1 2 1 2 1 2 2 2 1 1 6-0 2 3 1 1 1 1 1 1 0 1 2 2 2 2 2 2 2 2 2 2 2 2			3 3 3 3 2 3 4 4 2 1 1 3 2 5 4 6 1 3 2 2 2 2 2 1 4 1 5 1 3 3 3 3 1 3 1 2 4 2 5 4 2 5 4 2 3 2 2 2 2 2 2 1 2			2 3 3 2 2 3 3 1 1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 1 1 8 4 3 5 4 9 1 1 2 4 2 5-0 1 3 2 2 2 2 3 1 50			3 5 3 3 3 3 3 2 2 2 2 4 2 1~0 2 1 2 2 2 2 2 1 45 2 2 2 3 3 3 3 5 6 2 1 2 6 2 4 6 4 6 6 6 6 6 6 6 6 6 6 6 6 6 6	2 1 3 4 3 3 4 1 3 1 2 4 2 1-9 4 9 2 2 2 2 2 1 44		1 2 3 3 3 3 4 2 2 1 2 4 2 6-9 2 2 2 2 2 2 2 1 1 1 1 1 1 2 4 2 6-9 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 2 2 3 3 4 2 1 2 2 4 2 5 - 0 4 3 2 2 2 2 3 1 2	3 2 3 3 4 3 3 2 1 3 2 4 2 6-8 2 1 2 2 2 2 2 1 2	133334131242541672222223	212333222222425-011222221		3 5 3 2 3 3 4 4 1 2 2 4 8 5 1 4 5 2 2 2 2 2 1 3 4 1 1 5 4 5 4 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7			2 3 3 3 3 3 3 4 3 4 5 4 5 5 4 5 5 1 1 1 1 8 5 5 5 5 5 6 5 6 5 5 5 5 5 5 5 5 5 5 5	
1 2 2 2 1 1 2 1 2 2 1 1 9 2 1 1 1 7 9 2 2 3 9 3 1 3 2 3 2 3 3 3 1 3 2 3 2 3 2 3 3 3 3		2 3 3 3 3 4 3 4 2 1 2 2 4 2 3 0 1 2 2 2 2 2 2 1 48 1			2 1 2 2 2 3 3 3 1 1 2 2 3 2 3 •0 1 4 2 2 2 2 3 3 3 1 1 2 2 3 2 3 •0 1 4 2 2 2 2 2 3 1 4 4	111313212312421120222221 49	2 3 1 3 2 3 3 3 3 1 3 1 3 2 3 6 4 2 2 2 2 2 2 1 50	<pre>% []]]]] [] []]]]]]]]]]</pre>			211324331312125425-0422121140 3	E 1 2 2 2 7 1 1 1 1 2 4 2 1 1 1 3 2 1 1 1 3 2 1 2 1 2 1 2 2 2 2	2 1 2 1 3 1 3 4 2 3 1 2 4 2 5-0 4 2 2 2 2 2 2 2 1 3		2122122112121212045112110 16	2313313413121255623111110	1 3 2 2 2 1 2 1 2 1 2 2 1 1 6-0 2 3 1 1 1 1 1 0 1 2 2 2 2 2 3 2 2 2 2 2 2 2 2 2 2 2 2 2			2333323442113254513222221 4			2 2 3 3 2 2 3 3 1 1 2 2 4 2 5-0 5 0 2 2 2 2 2 1 50			2 3 2 3 3 3 3 3 3 2 2 2 2 4 2 1-0 2 1 2 2 2 2 2 1 45 7 2 2 2 3 3 3 3 5 2 1 1 2 4 2 4-0 5 0 2 2 2 2 2 1 45	2 2 1 3 4 3 3 4 1 3 1 2 4 2 1-9 4 9 2 2 2 2 2 1 44	25 1 3 3 3 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 2 2 3 3 3 3 4 2 2 1 2 4 2 6-9 2 2 2 2 2 2 2 1 9	211223342122425425212	2 3 2 3 3 4 3 3 2 1 5 2 4 2 6-0 2 1 2 2 2 2 2 1 2	2 1 1 3 3 3 3 4 1 3 1 4 7 1 6 7 7 7 1 1 1 1 1 1 6 7 7 7 8 8 7 8 8 7 8 8 7 8 8 7 8 8 7 8 8 7 8 8 7 8 8 7 8 8 7 8	221233322222425-011222221	2 1 2 2 1 1 2 2 1 3 2 1 1 4-9 1 3 1 1 1 1 1 0 16	2 [3 2 3 3 4 4] 2 2 4 2 5 4 5 2 2 2 2 2]] 9 2 [2 3 4 5 2 3 1 2 2 4 2]-0 3 1 2 2 2 2 2 2 1 3			2 0 1 1 1 1 1 2 0 2 0 2 0 2 0 2 0 2 0 2	
222222211221212212122211222122221222212222		2233334342122423-0122222221 48 1	24 12222101-20-011202210000000000000000000		2 2 1 2 2 2 3 3 3 1 1 2 1 2 6 7 3 4 7 2 2 2 2 2 3 3 3 1 1 2 2 3 2 3 4 0 1 4 2 2 2 2 3 3 1 4 4 4	2111313212412421120222281 49		2 2 1 1 2 3 3 2 2 1 2 1 2 1 2 1 - 1 2 2 2 2 2 2 2 3 4 4 4 4 4 5 5 5 5 5 5 5 5 5 5 5 5 5 5			2211324331312121245542511114 3	E 1222211122+211132141322	221212131342312425425-04222221 3	22122211111222114011114	3 2 1 2 2 1 2 2 1 1 2 1 2 2 1 -0 4 5 1 1 2 1 9 16	1 2 3 1 3 3 1 3 4 1 3 1 2 1 2 5 -6 2 3 1 1 1 1 1 9 1	2132221212122146-023111150 1221234341312455-0432222123	221121321312221-01222221		22333323442113254613222221 4	2 2 3 (4 5 3] 4 2 2 2 2 4 2 5-6] 3 2 2 2 5 1 6 2	2233423332132421-012222222154	2 2 2 3 3 2 2 3 3 1 1 2 2 4 2 5-0 5 0 2 2 2 2 3 50 2 2 1 9 4 3 5 4 3 1 3 2 4 2 5-0 1 3 2 2 2 3 1 50			2232333332222421-021222221 45	122134341312421-04922221 44	an [2222] [222]	2 2 1 2 3 3 3 4 2 2 1 2 4 2 6-9 2 2 2 2 2 2 2 1 1 1 1 1 1 2 4 2 6-9 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2211233342122425425212	2 2 3 2 3 3 4 3 3 2 1 2 2 4 2 6-0 2 1 2 2 2 2 2 1 2	22113333413124254102222221	L 2 2 L 2 3 3 3 2 2 2 2 2 4 2 5-0 1 1 2 2 2 2 2 1 16	221221122132114-013111110 16	221232334122485124214322221			2 2 1 2 2 3 2 2 1 1 2 2 1 2 2 4 2 1 1 1 2 6 2 7 2 8 2 8 2 8 2 8 2 8 2 8 2 8 2 8 2 8	
222222112212212122221122122221222222222					1 2 2 1 2 2 2 3 3 3 1 1 2 2 3 2 3 0 1 4 2 2 2 2 2 3 3 3 1 1 2 2 3 2 3 0 1 4 2 2 2 2 2 3 1 4 4	32111313212412421120222281 49		1 2 2 1 2 2 3 2 2 1 2 1 2 1 2 1 2 1 2 2 2 2			1 2 2 1 1 3 2 4 3 3 1 3 1 2 1 2 3 4 2 4 2 1 1 1 1 6 3 3 3 3 4 2 3 3 3 4 3 1 3 3 1 2 2 4 2 5 4 2 5 4 2 5 7 4 2 2 2 2 2 2 7 3 3	E 1222211122+2111222141321412222	32212121313423124251242504222221 3	128122219111222114011114	1 2 1 1 2 2 1 2 2 1 1 1 2 1 1 - 0 4 5 1 1 1 1 0 16	1 2 2 1 3 3 1 3 4 1 3 1 2 1 2 1 2 2 2 2 2 1 1 1 1 1 1 1 1	221322212121212146-02311119 11221234341312425-04322221 23		32211123111121212120300111116	122333323442113234613222221 4		12233423332132421-012222221 54	2 2 2 3 3 2 2 3 3 1 1 2 2 4 2 5-0 5 0 2 2 2 2 3 1 50 2 2 1 9 4 3 5 4 2 1 3 4 2 5 0 1 3 4 2 5 0 1 3 2 2 3 1 50			122323333332222421-021222221 45	11221343341312421-04922221 44	a) [2,2,2,2,2,3,3,4] [2,2,2,2,3,4] [2,2,2,3,4] [2,2,2,3,4] [2,2,3,4] [2,3,4] [2,3,4] [2,3,4] [2,4]	12212333342212426-922222221	12211223342122425425252223	12232314337112426-021222221 2	122113333413124254254262222223	1 2 2 1 2 3 3 3 3 2 2 2 2 2 4 2 2-9 1 1 2 2 2 2 2 1 1 E	1221221122121212121012211110 15	1221132334112242514524214			2 0 1 1 1 1 1 4 0 - 1 1 7 3 1 1 2 3 5 3 5 5 5 5 5 7 3 3 1 1 2 3 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	
1 2 2 2 2 2 2 1 1 1 2 2 1 1 1 2 2 1 1 2 1 2 2 1 1 2 1 2				512111221221221222229999	8 4 6 6 4 1 6 6 6 7 1 1 2 1 6 6 6 3 4 6 2 6 6 6 6 1 8 8 9 9 9 1 2 2 2 2 3 3 3 1 1 2 2 3 2 3 4 0 1 4 2 2 2 2 2 3 1 4 6 4	5 3 2 1 1 3 1 3 2 1 2 4 2 1 2 4 2 1 1 2 0 2 2 2 2 2 1 49 4	311231323333131313234642222221 50	2 2 2 2 1 2 3 3 2 5 1 5 1 5 1 5 2 1 4 2 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5			51221132433131212542544211110 3	E 12222111242111341325221	2 2 2 2 2 2 2 2 3 3 4 2 3 1 2 4 2 3 2 4 2 2 2 2 2 1 3 2 2 2 2 2 2 2 2 2 2 2 2 2	3128122215515122215011119	2 2 2 2 2 1 2 2 1 2 2 1 1 2 2 1 - 0 4 5 1 1 1 1 0 16 3 1 2 2 1 2 4 3 3 4 2 1 1 2 2 2 1 - 0 1 1 2 2 1 2 2 1 20	4132313313413121255-C23111110	4 2 2 1 3 2 2 2 1 2 1 2 1 2 2 1 1 6-0 2 3 1 1 1 1 1 0 1 3 1 1 2 2 1 2 3 4 3 4 1 3 1 2 4 5 5-0 4 3 2 2 2 2 2 1 23			5122313323442113234613222221 4 51121133331112425402322221 2		512233423332132421-012222221 54	5 2 2 2 3 3 2 2 3 3 1 2 2 4 2 5-0 5 0 2 2 2 2 3 50 4 1 2 2 1 9 4 3 5 4 3 1 2 4 2 5-0 1 2 4 2 2 2 3 1 50			5 1 5 5 3 5 3 3 3 3 3 2 2 2 4 2 1-0 2 1 2 2 2 2 3 3 3 3 3 4 2 4 2 4 2 4 0 4 0 2 2 2 2 2 3 1 4 6 4 5 5 5 5 2 2 2 2 2 2 2 2 2 4 5 4 5 5 5 5	4 1 2 2 1 3 4 3 3 4 1 3 1 2 4 2 1-0 4 9 2 2 2 2 1 44	a a <th>51221233342212426-02222221</th> <th>31221122334212242242242252223</th> <th>3 1 2 2 3 2 3 3 4 3 3 2 1 1 2 4 2 6-0 2 1 2 2 2 2 2 1 2</th> <th>3122113333413124254162222223</th> <th>2112212333222222425-011222221 JE</th> <th>JI Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z</th> <th>) 22 [32])4 22425 452222]9 4\22 21 52]12242 -02 22222 </th> <th></th> <th></th> <th>2 0 1 1 1 1 2 4 2 1 1 2 2 1 2 2 2 2 2 2 2 3 1 3 1 3 2 3 2</th> <th></th>	51221233342212426-02222221	31221122334212242242242252223	3 1 2 2 3 2 3 3 4 3 3 2 1 1 2 4 2 6-0 2 1 2 2 2 2 2 1 2	3122113333413124254162222223	2112212333222222425-011222221 JE	JI Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z Z) 22 [32])4 22425 452222]9 4\22 21 52]12242 -02 22222			2 0 1 1 1 1 2 4 2 1 1 2 2 1 2 2 2 2 2 2 2 3 1 3 1 3 2 3 2	
(a) 1<					1312212223331122233312222233340142222233344	1532111313212412421120222221 49		5212215215215561515651114215252551 4			0 5 1 2 2 1 1 3 2 4 3 3 1 3 1 2 1 2 3 4 0 4 2 1 1 1 1 6 3	E 1222211122421132131242141322221	13 3 2 2 1 2 1 2 1 3 3 3 4 2 3 1 2 4 2 5 1 4 2 2 2 2 2 1 3		9313212212211112212110 10 931323124334211221-011221221 23	1 0 1 1 1 1 2 2 3 4 1 3 1 3 1 3 1 3 1 4 1 3 1 5 1 5 2 5 2 3 1 3 1 3 1 3 1 3 1 3 1 3 1 3 1 3 1 3	15 4 2 2 1 3 2 2 2 1 2 1 2 1 2 2 1 1 6 - 6 2 3 1 1 1 1 1 0 1		10 2 3 2 2 1 1 1 2 3 1 1 2 1 2 1 2 1 2 1 2	2 5 1 2 2 3 3 3 3 2 3 6 6 2 1 1 3 2 3 6 6 1 3 2 2 2 2 2 2 1 6 1 5 5 1 2 5 1 2 5 1 2 5 2 2 2 2 2 2 2 2		9512233423332132421-012222221 54	9512223322331225425-050222221 50 9412211943542132425-013222231 50			10512232333332222421-021222221 45 1051122233333422112424-0902222221 46	0411221343413124212421443222144		5 2 2 2 2 3 3 3 4 2 2 2 4 2 6-0 2 2 2 2 2 2 2 3 3 4 2 2 1 2 4 2 6-0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 1 2 2 1 1 2 3 3 4 5 1 2 5 4 2 5 6 4 3 2 5 2 5 1 2 5	13 3 1 2 2 3 2 3 3 4 3 3 2 1 1 2 4 2 6-0 2 1 2 2 2 2 2 1 2	2 1 1 1 1 2 1 2 1 2 1 1 1 1 1 1 2 1 2 2 2 2 2 1 1 1 1 2 1 2 2 2 2 2 3 1 3 1	15 2 1 2 2 1 2 3 3 3 2 2 2 2 2 2 4 2 5-9 1 1 2 2 2 2 2 1 9 F	94 12 2 1 2 2 1 1 2 2 1 3 2 1 1 4-9 1 3 1 1 1 1 1 9 18 9 9 9 9 9 9 9 9 9 9 9 9	9]] 2 2] [3 2 3 3 4]] 2 2 4 2 5] 4 5 2 2 2 2 2]]0 0 4 1 2 2 1 1 2 3 1 5 2 3 1 2 2 4 2 3-6 3 1 2 2 2 2 2 1 3		23322123433422223425-01322222123	2 0 1 1 1 1 1 4 0 - 2 1 2 2 1 2 2 1 2 2 2 2 2 2 2 2 4 5 1 5 2	
16 17 27 27 17 27 27 17 27 27 17 4 27 17 17 17 27 4 27 27 27 27 27 27 27 27 27 27 27 27 27 2		310 4 1 2 2 3 3 3 3 4 3 4 2 1 2 2 4 2 3 0 1 2 2 2 2 2 1 48 1			311 31 2 2 1 2 2 2 3 3 3 1 1 2 2 3 2 3 40 1 4 2 2 2 2 2 1 2 40 40 40 40 40 50 50 50 50 50 50 50 50 50 50 50 50	311 5 3 2 1 1 3 1 3 2 1 2 4 2 1 2 4 2 1 1 2 0 2 2 2 2 2 1 49	313 3 1 1 2 3 1 3 2 3 3 3 3 1 3 1 3 2 3 6 6 2 2 2 2 2 2 1 50	315 2 1 2 2 1 5 3 3 5 5 1 5 1 5 5 5 5 5 5 5 5 5 5 5	2164122113133121212214012122048		510 5 1 2 2 1 3 2 4 3 2 1 3 1 2 1 2 2 4 2 5 0 4 2 1 1 1 1 6 3 5 5 5 5 6 4 2 1 3 3 4 3 1 3 3 1 2 2 4 2 5 6 0 4 2 2 2 2 2 2 2 3 3 5 5 5 5 5 5 5 5 5 5 5	512 5 1 2 2 2 2 1 4 1 1 3 2 1 3 1 2 4 2 3 1 1 1 2 2 2 2 2 1 3	513 1 3 2 2 1 2 1 3 3 3 4 2 3 1 2 4 2 5 0 4 2 2 2 2 2 1 3 5 4 2 1 2 2 1 1 2 2 1 1 1 2 2 1 2 1 1 1 1	515 31 2 2 1 2 2 2 1 1 1 1 1 2 2 2 1 1 4 0 1 1 1 1 1 0 3	6 7 2 2 1 2 2 1 2 2 1 2 1 2 1 2 2 1 -0 4 5 1 1 2 1 2 0 6 9 3 1 3 2 3 1 2 3 1 2 3 4 2 1 1 2 2 2 1 2 2 1 2 3 1	612 413 2313313413121255-62311110	015 4 2 2 1 3 2 2 2 1 2 1 2 1 2 1 2 1 4 6 6 2 3 1 1 1 1 1 9 1 1 7 19 3 1 1 2 2 2 2 2 1 2 3		710 2 3 2 2 1 1 1 2 3 1 2 1 2 1 2 1 2 1 2 1	712 5 1 2 2 3 3 3 3 2 3 6 6 2 1 1 3 2 3 6 1 3 2 2 2 2 2 1 6 7 315 5 1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2		10 0 2 1 2 2 3 3 0 2 1 2 2 2 1 3 2 4 2 1-0 1 2 2 2 2 2 2 1 24	10 9 5 1 2 2 2 3 3 2 2 3 3 1 1 2 2 4 2 2-0 5 0 2 2 2 2 2 1 50 10 9 4 1 2 2 1 9 4 3 5 4 2 1 1 2 4 2 5-0 1 2 2 2 2 2 2 3 4 50		1010 4 1 2 2 3 1 2 3 4 3 1 3 3 2 2 1 1 5 1 5 1 5 0 5 1 1 1 1 1 4 16 17 17 17 17 17 17 17 17 17 17 17 17 17	1010 5 1 2 2 3 2 3 3 3 3 3 2 2 2 2 4 2 1-0 2 1 2 2 2 2 2 1 45 1010 5 1 1 2 2 2 3 3 3 3 5 2 2 1 2 4 2 4-0 5 0 2 2 2 2 2 7 1 45	1010 4 1 1 2 2 1 3 4 3 3 4 1 3 1 2 4 2 1-0 4 9 2 2 2 2 2 1 44		10 5 2 2 2 3 3 3 3 4 2 2 2 4 2 9-9 2 2 2 2 2 2 2 1 1 10 2 3 1 2 1 2 4 4 3 4 1 3 1 2 4 2 5-0 3 1 2 2 2 2 2 3 3	1012 3 1 2 2 1 1 2 2 3 3 4 2 1 2 2 4 2 5 4 4 3 2 2 2 2 3 1 2	1013 3 1 2 2 3 2 3 3 4 3 3 7 1 1 2 4 2 6-0 2 1 2 2 2 2 2 1 2	1015 3 1 2 2 1 2 3 3 4 1 3 1 2 4 2 5 4 1 9 2 2 2 2 2 3 18	1015 2 1 2 2 1 2 3 3 3 2 2 2 2 2 2 4 2 5-0 1 1 2 2 2 2 2 1 16	1000 31 2 2 1 2 2 1 1 2 2 1 3 2 1 1 4-9 1 3 1 1 1 1 1 9 10 10 10 10 10 10 10 10 10 10 10 10 10	0] 2 2] [3 2] 0 4] 2 2 4 2 5 4 5 2 2 2 2 2 10 10 4 1 2 2 1 2 3 1 3 2 3 1 2 2 4 2 -0 3 1 2 2 2 2 2 1			2 0 1 1 1 1 1 4 0 - 1 1 7 7 1 1 2 7 5 2 7 2 7 2 1 2 1 1 2 1 1 2 7 5 7 7 7 1 1 2 1 2 1 2 7 7 7 1 1 2 1 2 1 2	

249

William Martin Martin

Identification	CARE NR.	Day Sold	Hr. Seld	Age	Marital	Local	Telephone	Time on Job	Time in Residence	Residence Category	Type Employment	Bank Account	Financial Ref.	Income	Loan Amount	Loan Term	Purchase Type	Credit Rating	Decision	Loan Officer Making Decision	Sverial Considerations	Salesman	Clerk Coding Data	Judge #1 w/credit	Judge #2 w/credit	Judge #] w/credit	Judge #4 w/credit	Judge #5 w/credit	4/5 vote w/credit	Hrs. in System	Hrs. in Credit System	
Column	-	~	4-8	••	-9	~	-	0	54	=	12	13	-	15	lt	17	16	19	20	21	22	23	*	25	26	27	35	29	30	31-32-33	14-35-36	

NANNANNANNANNANNNNNNNNNNNNNN
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
******
emeening nan name een eesen een na
ANNA-AREAANN AARAARAAAAAAAAAAAA
N-N-OHONANAK, NNAAOAOOOOO

Ø

f

6

. . Ľ,

, έ,

19 1-

h - 42

<u>,</u> ر• ì к." .

. .

**;**. ` 1 ÷ . .

,M M Í يعور ا . Ú, jín 1 0 1.

**4**66

. آدر

۹. . . ·#

22 1 2 . ļ , nn^{34.} . 5 . .....

۸.

r d^{e l}

/

Į.

250

A STAT

GROUP I SPLIT GROU SPLIT GROU	۳ ۲ ۲ ۲	107AL 224 <b>.0</b>									:				
SPLIT GROU	e -		- WE 1GHT	-	SUM OF Y 99.600000	SUR SUR	: Y-SQUARE DODGDO	55+24	T S S 5536	4	NE AN 195429	4	10. DEV. 1662043		
SPLIT GROU	ď						SPL I T	SUMMARY							
SPLIT GROU		ON PRE	DICTOR	16	CRERAT		INTO GROUP	2 MITH 5	CODES	, , ,	_				
SPLIT GROU	P 2	N= 140000	76 MEA	Ar File	.0] €=068944682446	GROUP		148 MEAN=			11.74	RSG#	.38285 F-R5	E = =05	7.7
	<u>م</u>	ON PRE	DICTOR	11	FINREF		INTO GROUP	411- 4	CODES						
GROL	4 4	N	19 HEA		0.00	GROUP	AND 640UP 5 N=	5 BITE 129 HEAN=		1 2 5 T=	1.698	R50=	.55585 F-R5	50= 86.	.08
SPLET GROU	- -	JN PRE	DICTOR	101	BANKAC		INTO GROUP	411H 9	CODES						
10H3	9 d	N =	30 KEA		.37	GROUP	AND GROUP	7 8115 99 MEANE	CUDES 66		6.617	R50=	+66515 F-RS	14 =05	
SPLIT GROL	۰ م	ON FRE	DICTOR	16	CRERAT		INTO GROUP	H114 8	CODES						
6RUL	9 8	-N	46 HEA	N=	.16	GROUP	AND GROUP	9 HITA 53 MEAN 2012	. CODES		3.520	RS0=	*68678 F-RS	50+ 15.	.12
SPL11 6901		ON PRE	DICTOR	12	LOANAH		INTO GROUP		CODES						
10H9	(P 10	N=	16 MEA	N	.94	GROUP		-NA3M 06			2.104	R59=	-70063 F-85	50= 10.	.09
SPLIT GROL	9	CN PRE	DICTOR	2.	CRERAT		INTO GROUP		CODES						
GROL	JP 12	= N	19 MEA	Ň	.16	GROUP	AND GROUP	-NV34 []		- - -	3.665	800 8	.74151 F-AS	50= 34.	.32
SPLIT GROU	П. 4	AN PRE	DICTOR	1	HARITL			# 14 # # # # # # # # # # # # # # # # # #					* * * * * * * * * *	***	
1049	P 14		5 MEA		-20	GROUP		25 MEANE			2.613	₽50=	.76516 F-R5	ig= 21.	.75
	•														
					*******			*	* * *						****
NCF TC	ITAL TS	55	TOTA	L 855	TOTAL	NSS N	R-SQUAB	03	œ				F-ANDVA OF	1 043	N
8 55.	245536	م	42.27	1928	E79.21	606	.76516	459 .E	4245				200-5419	7 21	16

•

ю.,п

ş.,

:..

4

<b>47251</b> 3M	444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 444614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 446614 4466144444444	- 2005 -	.822489 .827729 .827729 .823459 .823459 .823459 .823459 .823459 .923459 .849754 .849754 .849754 .849754 .849754 .849754 .849754 .849754 .849754 .92059	4562** 51 51 5252 5552 65195 60195
SJOBT1K	482148 482148 482148 481114 581828 481118 481118 481118 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41189 41188 41188 41188 41188 41188 41188 41188 41188 41188 41188 4111	6446 •6423 •69307 •69307 •69307	.171458 .28842 .28842 .284455 .29545 .205555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .255555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .2555555 .255555555	542.46. 5175. 64579. 701519. 101519. 56519.
40 HCHE		.05589 .09163 .00558 .00558 .00548 .0054844C	442421. 76282. 76282. 842422. 842422. 842422. 842422. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 84242. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8424. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 8444. 84	. 51401 . 65146 . 65146 . 65146 . 65146 . 63181 . 85183
MOCAL	44498 44498 44498 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 44698 446988 446988 446988 446988 4469888 446988 446988 446988 446988 446988 446988 446988 446988 446988 446988 446988 446988 446988 446988 446988 4469888 446988 446988 446988 446988 4469888 446988 446988 446988 4469888 446988 446988 446988 446988 446988 4469888 446988 446988 446988 446988 4469888 446988 446988 4469888 4469888 4469888 4469888 4469888 4469888 4469888 4469888 4469888 4469888 4469888 4469888 4469888 44698888 4469888 446988888 4469888 446988888 44698888 446988888 44698888888 44698888888 44698888888 44698888888 44698888888 44698888 44698888888888		78952 86044 86044 88564 88664 88664 88664 728457 728457 86664 86669 86669 86699 86699 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 866998 86698 86698 86698 86698 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 86698 86698 86698 86698 866988 86698 86698 866988 86698 86698 86698 86698 866988 86698 86698 86698 866988 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 866988 866988 86698 866988 86698 86698 86698 86698 86698 866988 86698 86698 866988 86698 866988 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 86698 866988 866988 866988 866988 86698 86698 86698 86698 86698 86698 866988 86698 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 8669888 866988 866988 866988 866988 866988 866988 8669888 866988 866988 866988 866988 866988 866988 8669888 8669888 866988 866988 866988 866988 8669888 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 866988 8669888 866988 866988 866988 866988 866988 866988 8669888 8669888 866988 866988 866988 866988 86698888 866988 866988 866988 86698 86698 86698 866988 86698 86698 86698 866988 86698 86698 8	.43274 11 21175 241951 241951 241624 241624 2416245
294.01TL	24724 24724 24242 24242 24242 24242 24242 24242 24242 24242 24242 24242 24242 24242 24242 24242 24242		9205-90 65255-90 855350 170520 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 252550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 255550 2555500 2555500 2555500 2555500 2555500 2555500 2555500 2555500000000	
2	4012652 4012652 401265 401252 4012525 401265 401265 412525 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412665 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 412655 4126555 4126555 4126555 4126555 41265555 41265555 41265555 41265555555 4126555555555555555555555555555555555555		\$00000 \$1210000 \$1210000 \$121000 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12100 \$12000 \$12100 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$12000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$10000 \$100000 \$10000 \$100000 \$100000 \$100000 \$100000 \$100000 \$1000000 \$100000000	.67926 13 .85225 .85227 .85223 .8222 .67526
14	TALA 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	A CC VIA	・ ネーショック ちゅう ゆやり ほかう キーショー ひょうし	SUBSUN TRIALS MEANDS CF VAR CF VAR RECSUN

ANALYSIS & MILE SISTING

**4** 

,

.

5 0 **5** 

.

Ġ.

		ANALTERS WITH	(1)\$51/35Ū	
.K.I	.OAMTH	14460555	15EGULTT	1 & CRERAT
-1		849768.		14256.
~	298968.	.029246	19509.	.187271
-	.001793	104500.	.40053	241212.
4	202500-	-007647		102011.
ŵ	.019136	12650		140008.
	. 01440	.601000	.043063	
-	.01000	461928*	\$*\$0000	
•	,\$26864	- 6 1 4 5 6 5	.\$50258	
•	. 866965	.1045.35	0.0005	
10	.022222	6.00050		
11	.02728Q	149600.	861+98.	6.90000
12	167160.	141010.	-50831¢	
2	.051852	8 - 00000	.085714	.966667
MUSBU	.28144	.26783	. 14445	1.24843
RIALS	1	1	1	0.0
EANES	.02165		[[[]].	.12983
<b>v</b>	-91679	.02785	******	13451.
F VAS	. 86821	1.15195	Z=14827	1.63798
K A P	.00035	. 60678	.00059	.01017
ECSUM	-28144	.26783	.14445	1.4584.8

- 7

8

Ŀ.

ئ. س

4

3

14

ð

. <u>2</u> . 1

j.

1

. . . . . . . . .

55.2

.0101557

Calt.

GMUCH 969119"

4

REAN

GRAND

3

5

SUNGROUPS

NO OF

VARIABLES.

:

TABLE

SUMMARY

253

ł

GROUP 6 NEAME 137 Ma 38 S-0-8 AS ссост 2 2 3 соста 12 2 3 соост 2 2 3 соост 2 2 3 соост 2 3 3 соост 3 2 соост 3 2 соост 3 

الأثلاث والمتلاد الت

2 . .

.

а,

े ज डॉ.

. * ..

**N** 

Ľ

.

254

				1997. 1997.	
				••••••••••••••••••••••••••••••••••••••	
			20183100011000012220000 Roup 1505[Hal 4244 Na 25 5[Hal 4244 Prejicur 2 447[1]	соста е . 6016 1396 [МАЦ КЕАМ: 6016 1396 [МАЦ КЕАМ: 6116 5.0.4	
			• • • • •	102	
				47 RSC =	
		ана Мац. Ибант 5.0.= 2. Цокная		46494 5.0.8 2.004848	
				11 30 5 1 2	
• Va		6000 6000 6000 6000 6000 6000 6000 600			
- 14 PROPE	<b>4</b> 8		- 46 - 450 - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 4 - 4	9 2 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	
S.D.C.	KEAK= S+D+=		HEAN- HEAN- S.D.= Crerat		
		•			
HEDICI					

. .

. #

. A ²¹

*

.*

<u>9</u>1,

ð

and the second state of th

4 **4** 

¢

# Exhibit 9 -- AID4UT OUTPUT FOR JUDGE #2

Ŷ

...

570° DEV. 49950151
NE 30 161676.
5 × 1 52.886.22
SUM Y-SQUARE 101-00000
SUM OF Y 187-00800
Total Weight 224.489385
= ţ
en oce

# ASYMMYS 137dS

2PLIT SPLIT	1013	- 6	M PRE	DICTOR	£ .	CRERAT .13	GROUP	UND CHA	GROUP GROUP	2 K]1- 3 K]1H 6 M[AM	CODES 6			15-54	#20-	-52244 F	-250= 	2+2-9
LI N	anca.		ž.	C1C100	e .	BANKAC •53	dincaro	INTO AND	GROUP GRUUP		CCDES 1	Ni Constant	71	1964	R\$0.	- 55953 -	-RS0-	18.45
54011 54011	GR OUP GROUP	1 1 1 1 1		51CTOR 75 MEA	1	+3+ 4 = 2 # 4 # 4 # 4 # 4 # 4 # 4 # 4 # 4 # 4 #		NU NU NU	GROUP GROUP		CODES (	• • • •		4. g 4	R505	.62592 F	-950-	N7. AC
5P.17	eeore Cerru		N PRE	DICTOR				INTO	GROUP GROUP		CODES CODES	-~		.715	R\$Q-	1 (8783	0 9 9	[+-[+
SPL II	660UP		0 K 1 A E	DICTOR 23 MEA	: .	F 1425F	6800	140 140	0 6.85 UP 6.83 UP 8.		CODES	-		3,585	\$50+ \$	.70438 F	-850	12.15
L V	GROUP	9 2		DJC7CR	•	JC811A •85	64006		0 64008 64008 ##		CODES CODES	-	16	1.0.¢	nso=	.72822 F	# 0 9 # -	\$.}G
i i	GROUP		0 F FE	010168		9H0041	90039	124	90089 6 8 9 0 6 8 8 9 0 6	15 6 17 15 6 17 15 6 17	CODES		1	2.464	RS0-	7 2005 1	-R50=	15.87 ****************
i i	6A DUP		4 1	11 164	- ;		ancar5		90083 10080 110	10 m15	C0055	-	N 1	C.12		. 76216 7	-R50+	14 <b>.95</b> •••••••
	Creek		40 40 4				GROUP				CODES		1	2.023	R50=	1 40467.	- 2010 	11.57
								- u. -										
¥	1014	۲۲ <b>18</b> 1		101	15 2 1		01.AL #55	æ	361052-1	a	×				-	F-81048	0f 1	DF 2
•	55-04	(1133		43.21	44721	21			. 1 7 4 350.	. 12	1799%					21.6014	•	412

256

1

		ANAL 57 NAME	6357189181			
-	AGE	ZHARITL	JA.GCAL	SHOKE	5.00FTH	<b>4</b> REST1#
						100010
2		401124	.96065	. 850055	-102436	
•• •		417744	921154"	. 844359		
•••		100000	.027489	. 605933	.025640	
÷.		590 IU.	.055645	.050397		
•			204100	185160.	•05420°	
n			14746	6.00000	£2410E.	<b>19790</b>
-0	P200C9.		401100	9.00000	113611.	6.60000
-	627740		LOAL TO.	0,00030	.072465	
	.164637		224210	0.00000	.015625	81//201
<b>8</b>	LECC80.		000000	0.000030	.312500	
#39 + 1	0.00000			.063954	120101.	[234]]
i i	.018690	100°-0		\$52632	201220,	.263158
( 14 1-4	203158		10000	0.00000	CECEYO.	12120
	*****					
	,	14445	21.121-	.34855	1.44029	1,9C1.
		11	51	:	5	
A TALAL	, , ,		-01+72	.02683	.11079	
MEANBS				21060.	262'0'	
	19971°	1.1959	1.66808	1.12238		04400
		00105	.90060	15000 .	50801 °	
	11110-1	24461	20161.	.34855	A2044.1	
		AND YSES BITM	855/155(1)			
					,	
	74ESCAT	ATTREMP	9 JNC ONE	188ANKAC	IIFINKEF	TAKAU 12.1
TRIAL			186674	422771.	67 <b>48</b> 11.	.019966
-	•C[130"	577720°		151902	.152692	.014644
~	538510.		11-937 T		.053465	.009729
ň	957043.	Ca1918-		269069	.216000	. C16835
*	250190"		112120-	.004096	.126165	554640"
Ś	125860"		011567	0.00000	. 0 0 4 5 8 0	
••	CCODSI.	164141.	.268984		S21400"	911089 ·
- 1		1022161	0.000000	0.00000		
0 9	e.000000	.066667				000000
	0.00000	00000000	.022262			.014433
	.038384	.067621	956107		109520	.095694
2	0.00000	146810.	424219°	000040.0	.004286	EEEE.0.
2	CELEVO.					
	.64244	.65442	19672.	.76528	21586"	(1589. []
A P I I I		2	6		10000	61220.
	0+9+2	.05634	94460		104140 ·	.05607
2 2	46620.	.04679	A40404 .		0299	1.46329
CF VAS	1,09141	64959.	0.000°.		46404	AIC00.
VAR	10200.	61200-	- 400°	52800 522800	1,06614	.68547
RFCSUM		4オオのの"	***			

•

	HILA SISATKK	ES5/155(1)	
I JL DANTH	14ME CE 35	15604171	1 & CREPAT
.987748	.237478		522442
A44.160.		624118.	8E4C41"
C279C0.	.002290		.262178
.18254	.608269		0.56555
C\$+1+0.	.002506	66780º.	828668.
.92476	. 663813	**2*18.	411114
193200	8.00000		162505.
4.80084	0.620000	521110.	
0.00000	6.05000	000000	
.022222	0.00000		. 312500
92/010.	.014529	. 004040	.007621
1+5810.	.015156	291769.	0.00000
412411.	0.00000	0.00000	E24030.
M .68444	.16624	.10407	A1627.1
5	1	2	•
5 . P4646	1800.	9120 <b>9</b> .	((91.
D .05017	19410.	.009.	16521.
R 1.07510	2.04957	1.010.1	43408"
P .60252	67080°	. 60365	.02328
A004.	.19675	.10407	2.44161

55.1				13.0	14.0	12.6	2.0	ę. 0	9.6		11.0	9.4 4	4.4			15.0	1	1.0
•		ANAR																
		erours	3	2		2	2	3	2	2	2	13	21	3	2	3	3	•
ROUND CRIT.		Y NEAN CHE	144.64072	32.92275	25.12964	46.91424	42249.241	102.03561	86.39702	68.00723	77.1440	124.80699	959/6.641	97.18413	61.56251	14.24762	14.24502	135.96227
145641		VARIANCE PCT	.69911	20160.	.00060	18809.	.00845	.00468	10200.	FICO.	.00547	\$1210.	-6408.	41004.	.40252	67030"	.68089.	5262 0°
A MD HEAK +		COEFF VIR	1.12459	1.71494	1.44894	1.12230	\$2668.	[ 429] . ]	14140.1	(+6/7.	1. 15895	1.58150	4870¥.	1.06329	1.97518	2.04957	1.67493	. 29435
3	9	0 9 (1)		92269.	. 82455	21060.	86268.	. 86786	. 05 39.	. 0.674	45C/9"	06218.	<8,9,9J.	.05607	.050.	19418.		16251.
Uncroups 15	RECONSTRUCT(	\$UM655/155	1.0711	19++2-	26141.	. 34885	1.44029	[1851.	++2+9.	2+144.	19612.	. 22826	1.0661-	1.585.	. 59654	.10624	.10697	2.46101
IANLES NO DE S	(THAESANT) OUS	11551/558HOS	1.07717	1822.	56301.	. 34165	1.44629	C1951.	44244.	21459.	19615.	.28528	51486.	1.284.7		.10624	19991.	1.70378
LE IN VAR				11L	4	¥	134	154	Cal	C M D	OME	KAC	#E.F	Na.K	M C M	E\$5	117	RAT
SLAMARY TAB			1 AGE	2 RAR		AHA .	8	4 RES	7 RES	9	- INC	NAB BAN	11 11	12 LOAD	13 LOAD	Dia Marci	15 190	14 CHE

6

A

ł

. .

л г.

•••

ىتە

. . .

ંક્રે

. .

1

. (f^{r. 4}

Autoria Autoria

ŝ

258

						•••		1	ME AN-	27	150 -	H
			Prepictor 18	5.0.5 BJAKAC			PAEDICI	2 TI 2	1 × 46 f			
			C00E5 1 2	•							00	
+ CROW 3 HEAM		225.				• • • •	CUDE 5	-	2			
C00ES 2 3		•••				•••	1		KEAN- S.D.		1 808	
			6404P 4 6404P 4 74010104 10 00065 8						MC 946			••••
						Ĭ	1 000 14			53	- 90a	
c.Roup 1 MEANs	•••					•••	CCORS	-				•••
• Ns 234 5.0.1 • LEVEL 1 •	3					• • • •	RECT.		KE3N- 5.0. 1486C			
			686UP 7	- 10° \$	. 38 750 . 11 720		C not s	2				•••
••••				CAELE		• • • •	ROUP PREDIC	A I RO	#EAN 6.0.0	2		
6680.0 18								-		*****		
			сессе 6°Г17. Van Va Ряслістон 16 Сосеб 1	LL REAS		*****						

*

吾

Ĉ

. .

N.

E Lin Lin Lin

. . • essectant and the second and the second second is the second and the second and the second se - 6800P 14 NEAM • N> 11 5-55 • N> 12 5-55 • • • • еволе 10 меаки .74 640.е 10 меаки .74 Ма 23 5.0.е .44 CROUP 12051444 WE AND CROUP 1205144 ME AND + * 17 5.0.4 A PEDICION 5 409118 . .

\$

· ·

_

2

Exhibit 10 -- AID4UT OUTPUT FOR JUDGE #3

\$10. DEV-MEA3 172852404. 5000 Y-500ARE 75.71+286 SUM OF T 104.00000 TOTAL NEIGHT 224.0000 N 22 erour

## SPLIT SUMARY

		;	• 11 • • • • • • • • • •	;		-			******	,	*****
-0		1		:				:		:	
20-0		i	F RS	1				-		i	
			\$00000. \$00000				02264		22121×		
	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		350 =		<b>8</b> 50 		1099 H		1500 e e e e e e e e e e e e e e e e e e		850.
:	17.65 		5.365		\$ 		3,210 80000000		3.547 24400185		1.936 
					-	-			-	2	
•*	85	• -	21	o -	:	•	-	• -	<b>1</b>	0-	
CODES		CODES		CODES		CODES		CODES		CODE S	
111	H		N		E Ave	11		HE 14	TAN S	1	EANE
~	Ĭ	••		0		60	100	22		22	
6.40U	N= N	6KOUP CROUP		GROUP		CROUP CROUP	N=	680UP 640UP	N	GROUP CROUP	
AND ON		0.141	5	UT UT	-	N N		0141	=		2
	66709		GROUP		1005		GROUP		GROUP		GROUP
RERAT	÷0•	INREF	11	REAT	1	JANKAC	•05	PHONE PHONE	•63	TYPENP	.22
									1		
	EAN=	Ĩ	E AN=	4	- 14 31	×	IE AN=	g	HEAN		W.
01010	11		15	01010	2	0101	2	1010	•		
K PRE	-	R PREI	:	N PRE	ł	1886 N	÷,	+ = = + = = = = = = = = = = = = = = = =	ž	334 W	ł
•	~	:"	+	* 0	a		•		10		2
GROUP	GROUP	district of	GROUP	ekôue -	GROUP	GROUP	Georg	GROUP	GROUP	GROUP	GROUP
5M.11		5PL []		SPLIT		SPLIT		2PL 11		2PL 17	

## FINAL SUMPANY

C F

\$

* *

.

. . . .

a.

**.** .

. * † .

新

		HILA SISATWW	653/75211;			
ž	¥	244R11L	JL DCAL	SHOH 31	\$-0611×	642 57 8M
TRIA.	.627367	. 122536	941888,	696-19	137844	A14354
•	623226	241644		7478741 747789		661499"
-	#11E18-					141144 242480
•	. 026633	-020753	40250J			202060.
'n		542664.	165019.	.121480	.025562	946418.
•	-015152	000000	.000.02	9.00000	PT 9563.	566449°
► (		21111.	.02050	6.00000	- C1 0082	C19410.
	. 592976	400000°°	.002476		1<285×1.	
Þ	29/989*	**************************************	600000 e	9.00000	.994762	.004762
		95235	.015619	. 059048	.610233	.026889
	.125521.	4+[825.		4~1168.	.122807	• 04 J062
SUESUES	.28799	23622	91176.	19514.	10-20	C011C
TREALS	-	11	-			
NEANSS	. 12616	.62129.	1400.	04195	. 64564	74010-
0 8	VICCS.	93098	.00456	42570.	04970	01454
CF VAP	1.27081	1.45477	1-42561	1.07811	- 5751	42.27
VAR	.00110	9000	900 <b>9</b> 91	1000		
RECSUM	. 22705	-23422	61213.		1202	21192
		MILH SISA TONN	(1) 22/122(1)			
¥.	25047	81 TPEMP	\$ INCOME	i ebakk ne	1 15 [ MBEF	IZI OANAN
TETAS						
	. 441671	5 5 5 5 4 T				
•					******	190244.
	2222.00.				[26402.	
4	465346.	211418.	467.746			AA77144
۶.		.050367	.649205	101214		050210.
، ب	56505 <b>8</b> .	111111	6.6008.9	9.00096	.084175	251510.
-	95515 <b>9</b> .	545460.	1++250.	.652441	.027770	.0001.
•	600004°8		.002976	.0.7619	0.000060	.005132
			800036 8	6.90000	0.000000	0,00000
		052210.		.155556	ecctio.	.009697
11	**5128*	290640*	.052632	0.00000	.022556	1052632
MUSBUS	11162.	.59549	. 38815	L475.	25605.	. 15+08
TRIALS	=	61	11	10	10	-
	. \$2107	•0240 <b>2</b>	93559.	.05796	260/0.	.0176.
	-0110-	.04258	\$C828.	.04531	.06105	E1910.
	102C0.	-72262		24241.	- B¢ 740	62*16*
	20407K	20102	. 60051		64600.	.00026
	464677	46648.	. 18515	.95769	88677.	.19408

?

\$

	1 6 CRE RAT	54468S"	300290.	-147642	0000 <b>00</b> 0000	.036727	84000018	.03210	A17260.	0.00000	0,00000	0.055090			. 14585	.20336	10400.1	.04136	16+04.1	
(1) 551/558	15500157	.905472	.022545	.012953			.054175	161500.	0.00000	0.000000	.004138	++5410"	A/JAJ*	1	97814°	92CSB.	1.27963	.60056	\$120Z*	
AMALYSIS WITH	1.4467555	615418	104245	.015403	954519		144953	.604676	0.0000	000000	110100	.122607	 91159.	11	-0-106	.05+54	1.32712	19200		
	. Dawtw	142413.		000440	168416.		-684175		.0120			60500	19152.	1	. 02283	29850.	214A5.1	00044	.25167	
			• 1		• 4	•	* •	•		n e			Suesum	TRIALS	RF ANGS		17h 3-	den a	RECSUM	

55.7		
	¥ 4	
I	5 3 3 3 4 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9	
ROUND CRI	MAM CM	
.46525	KARAKAKAKAKA 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011 6011	
D MEAN -	K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K44C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K444C.1 K44C	
CR 14		
=		
ugarows 15	RECOMSTRUCTE SURBSS/1550 629469 629469 619479 61949 619497 621497 621497 621497 621497 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621607 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 621407 6214000000000000000000000000000000000000	
ARIABLES. NO 07 5	Sug Start and 11 Sug Start and 11 Sug Start and 22 Sug Start and 23 Sug St	
-		
1 ABI.E		
SLWART		

2.

a İr

, , ,

.

**đ** 1

· 6

.

٠

		6400.P S REAN 640.P S REAN P N CODES 1 2 FIMBEF	• • • • • • • • • • • • • • • • • • • •	НООР 115 [ИАК КЕИ Н. 65 5.5. РЕСПСТСИ РИОКЕ СООЕ 1 СООЕ Е 1 СООЕ 1 СООСОСОСОЕ 1	. 95 ASU . 21 PROB . 63 RSC -	
CODES 2 3 CERAT CODES 2 3 CERAT CODES 2 3 CODES 2 CODES 2 CO		- 583- - 584- 				
		CROUP APTIMAL MEAND A 15 5.0. Predictor 11 finger Codes 0	8.4 2.2 1.80 2.6 2.6 2.6 2.6 2.6 2.6 2.6 2.6 2.6 2.6			
есесь 1 не син-	4 <b>2</b>			0680,07 9 86.04 0680,07 9 86.04 14 20 5.07 14 5.07 15 10 10 10 10 10 10 10 10 10 10 10 10 10	.55 756 =	
		CARON 7 NEAN 9 N 0 5.0. 8 FEDICION 16 CREMAT 6 CODES 1 16 CREMAT	-23 850	CUCC & C & C & C & C & C & C & C & C & C	••••••••••• •65 850 = •22 8408=	
CGOUNT 2 REAL	- 055 - 0 °					• • •
****		CACUP 4.FIML MEAN- CACUP 4.FIML MEAN- N 76 5.D. Predictor 10 Creat Cores 0	10,22,11, 11, 10,22,11, 10,22,11, 10,22,11, 10,22,11, 10,22,11, 10,22,11, 10,22,11, 10,22,11, 10,22,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11, 10,2,11,11,11,11,11,11,11,11,11,11,11,11,1			

₹

j,

いいいいのかい

.

.

2 0

5

*



S.

.

.

...

. . . .

÷.

); 71, 7

ŀ

ti.÷ 

÷

14

ť о. •

F-AACVA DFJ DF2	64.3359 B 215
x	
8-50U4BED	
TOTAL #55	15.761855
TOTAL BSS	164270-44
TOTAL 755	55.639286
MCF	¢

## FINAL SURMASY

Exhibit 11 -- AID4UT OUTPUT FOR JUDGE #4

ţ, '

. e ł . . .

ан К.

Â

ц.

а. 2005 5

•

HAL T T KANAT R

. And

÷,

. 1

، بهر ک

یند : چرک ا

•

510. DEV. .49928801 NEAN •52678571 1 5 5 55.639286 SUM Y-SQUARE 114.00000 50% 0F Y 218-50200 1914L KELGHT 224**-8889**0 и 224 erour

594.[1 GROUP ]3 OM PREDICTOR ]1 FINREF INTO GADUP 16 MITN CODES 0 1 And Gadup 16 Met 7 MEAMe .57 Group 17 Mitn Codes 2 And Gadup 16 Met 7 MEAMe .57 Group 17 Ne 11 Refer 1.00 1.2 2.700 R544 .31373 F-R544 10.72

à

÷,

// · · · ·

a (

erestim erestim	0.912337     •10.912337       0.912337     •10.127337       0.91237     •10.27733       0.91723     •10.27733       0.91723     •10.27733       0.91723     •10.27733       0.91723     •10.27733       0.91723     •10.27733       0.91713     •10.27733       0.91713     •10.9273       0.91713     •10.9273       0.91713     •11.2773       0.91713     •11.2773       0.91713     •11.2773       0.91713     •11.2773       0.91713     •11.2773       0.91713     •11.2773       0.91713     •11.2773       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173       0.91713     •11.9173  <	NANADIS	14695 14605 11278 14278 14278 14278 14278 14259 142519 142510 142511 142511
4.PHONE	45160 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46149 46	•5339 188464C	84189 84189 841828 841828 841828 841828 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 841848 8418
B55/15541J 24.0CAL		.26916 <b>656/15</b> 5 (1) 91MCOME	
ANALTSIS VI'N . Zmaritl	Seperation (C. C.	426926. MILY 2124 MIM	522220 522220 50129 50129 50129 522299 52129 52229 52229 52229 52229 52229 52229 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 522220 52220 52220 52220 52220 52220 52220 52220 52220 52220 52220 52220 52220 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 52200 5200 5200 5200 5200 5200 5200 5200 5200 50 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 5000 50000 50000 50000 5000000
ž	4444 4444 4444 4444 4444 4444 4444 4444 4444	.22772 .22772	1212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 212579 2125779 2125779 21000000000000000000000000000000000000
-	1414 1414 1414 1414 1414 1414 1414 141	recson 1	1 

.

267

	16CRERAT		2010010			.075800	5-000-5	.008929	.066595	0.00000	0.000000	.96090	¥1648.	•	.14152	.15983	1.12938	.02555	1.55674
(1)\$\$1/\$58	ISEOUTTY				9228-2A .	.00197 <b>8</b>	291160.		.024555	*68293 <b>*</b>	- 1 8000	4 ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	23863.	3	.02171	42460.	1.5771+	1100.	E8862.
- HILA SISATANA	LANECESS	102224*	1928204		9[9000'	.60193.	42024.	00000000	.000062	.181995	9-90056	.010227	8504Z*	11	101201	.05133	2-3+286	.00263	66012.
	LOANTR		•633698	921588*	1/6592*	-005629	.135857	<b>•3356</b> .		.020145	.10209	6.00060	.60826	1	.66083	ee770.	1.28226	.00408	.64909
	HT .	-	N 1	•	4	¥A	•	~	•	•	-	=	MUS BUR	TRIALS	MEAMOS	0 9	CF VAR	AAR	RECSUM

55.4

GRAND MEAN .

3

SUMMARY TABLE TO VARIABLES. NO OF SUBGROUPS IS

S.7         COFF VAL         V           777         0.200         0.200         0.200           777         0.200         0.200         0.200           717         0.200         0.200         0.200           718         0.200         0.200         0.200           719         0.200         0.200         0.200           710         0.700         0.200         0.200           711         0.111         1.0100         0.700           700         0.011         1.0100         0.700           701         0.700         1.0100         0.700           700         0.011         1.0100         0.700           700         0.0100         0.700         1.0100           700         0.0100         2.4120         0.700           700         0.0100         2.4120         0.700           801         0.0100         2.4120         0.700           801         0.0100         2.4120         0.700           801         0.0100         2.4120         0.700           801         0.0100         2.4120         0.700
7/72         92.00         1.25.00           91         92.00         1.25.00           91         92.00         1.25.00           91         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.25.00           92         92.00         1.04.05           92         92.00         1.04.05           92         92.00         1.04.05           92         92.00         2.04.12           92         92.00         2.04.12           92         92.00         2.04.12           92         92.00         2.04.12           92         92.00         2.04.12           93         93.00         2.04.13           93         93.00         2.04.14           93
9125         -23454         1.2285           911         -0244         1.2285           1362         -0241         1.2262           1362         -0354         1.2262           1362         -0354         -0355           1132         -0356         -0355           1132         -0356         -0355           1133         1.1133         1.1135           1135         -0356         -3355           1135         -03113         1.1255           1135         -0412         58156           104         2.4753         2.4756           10513         2.1512         2.05156           104         2.4773         2.4756           10513         2.34756         2.4756
5918         .024,3         1.2028           1362         .02748         1.528           1362         .02748         1.528           1362         .02748         1.528           1365         .02748         1.528           1365         .02748         1.528           1365         .0214         1.528           1251         .0244         .9365           1262         .0113         1.1755           1365         .07142         1.04675           140         1.0465         .5865           192         .0715         2.8156           193         2.1779         2.4756           109         2.4779         2.4756
1305 (07441 1.55542 1305 (03719 1.55542 1305 (03719 1.7514 1655 (11330 1.1755 1655 (11330 1.1755 1655 (11312 1.04575 16455 (1132 1.04575 16455 (1132 1.24555 1693 (112 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1693 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.24555 1694 (113 1.245555 1694 (113 1.245555 1694 (113 1.245555 1694 (113 1.245555 1694 (113 1.245555 1694 (113 1.245555555 1694 (113 1.2455555 1694 (113 1.2455555555 1694 (113 1.245555555555) 1694 (113 1.24555555555555555555555555555555555555
150.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120.         120. <th< td=""></th<>
2237 22540 1.72418 5562 20164 1.7559 5562 11130 1.17559 2675 0.7142 1.04575 0.04550 0.5416 2692 0.0509 2.4755 293 2.47750 1.28256 293 2.5739 2.7750 1.28256
03004         •90505           5523         03504         •9355           612         11130         11135           615         •0142         58165           745         •04451         •58156           615         •0442         58156           61740         2.4756         5815           619         •0513         2.4756           613         2.1779         2.3426           613         2.3426         65131           613         2.5426         58156
555
2050 .071.2 1.04075 .505 .04650 .58106 .09412 .62508 .09412 .62508 .0513 2.4785 .0513 2.94850 .0513 2.94860
7945 .04650 .58166 2622 .07412 .62368 6915 .00009 2.47856 5919 .07799 1.28226 599 .05131 2.38286
5692 .094.12 .66256 6915 .00.059 2.47856 5909 .05139 1.26226 1093 .05133 2.34286
6915 .06069 2.47856 5909 .07799 1.20226 6093 .05133 2.3*286
5409 .07799 1.28226 599 .05133 2.3~286
1099 .05133 Z.3+286
1833 .03424 1.57714
5676 .159 <b>61 1.1</b> 2938

÷

5.

* ⊡(

 $\mathcal{D}$ 

					***********		********	
					CROUP 1	ME ANN	• 92 R\$Q •	-115.
					Ce av a	5.0°	.26 PR030	
								•
		• N= 188	5,0,4	36 PROS 0.600				•
		- PREDICTOR 16	CRERAT		•			•••••
		- LUUES 2 ]					- 21 660 -	•
		•		-	SI -4	5.0.	-50 PACH-	.000
		**************	***********		· FREDICTOR 1	L FINGEF		•
*GROUP 3 MEAN=	- 258 ST.	-+84-		-	- CODES 0			٠
F Nr 148 5.0.F	42 PP08= 0			•	•			•
· PREDICTOP 14 CREMAT		•		•				•
• CODES 1 2 3		•						
•		0		•	6 400H9	= M2 3H	. 20 R50 #	
		•			1 Mar 27		-** PH38-	.000.
								• •
				-56 #50 # -555	• COULS • •			•
. 4			2.U.C					•
• •			LKERAT					
• 1				- •				
		•						
	******				1 401111144.	S LUANDA		•
•	•			-	couts 2			• •
								•
	•							
	*********							
•								
•								
•								
¥ 1								
4 005015108 14 505017								
		•						
		•						
•		•••						
*****************************								

-

. . .

,

2.0

"." ,

. В Ц Ц Ц Ц Ц С С С С

* * *

8 ð 8

**. ....** 

,, L

۰.

.

-

conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservations conservati сяюце "lo-flwar wfawz .56 R59 = .64] 690це "lo-flwar Mfawz .56 R59 = .64] в ж в Reductok . Рыомб • ٠ ٠ севои с исали .53 севои с исали .53 Ми 15 5.05 .55 Level 2 .... .....

1 - **1** 

.

3 4

9 2 4

.

.

. ì

22 • 6800,0 1 KEAN= • N= 224 5+0== • LÉTEL 1

· · ·

.

ۍ توري

*

lacent

269

ł

;


, 1²

.

. . .

-- 2

.....

270

1

-

.....

4 PREDITOR 9 INCOME . . . . • 24 B REANS 27 Sedan LEVEL 2 ٠ ٠ +GROUP . . . .

Ŧ

. 5

. : . , •

.

r ,

- ;

:

- 14

Exhibit 12 -- AID4UT OUTPUT FOR JUDGE #5

.

.

2. ...t " A.

...

, , , , ,

ł,

•

510. DEV. .49998007 #EAN 17863524. 1 5 \$ 55,995536 SUM 7+50UARE SUM OF Y 111-00000 TUTA! WEIGHT 224.00000 422 8 GROUP 1

											:	ţ
	23.8	5	6.76	21.1	5.23	6.68		84	*	3.15		
		Ň				-			-0	Ň	_	
	054	F50	-BSD	05 d -	C 5 6	-85	5 2 2	929	0.01	-RSO	224-	
	196	0			8	1 66	2		064		Q N	
	.35	2	50	ů,	5	-59	1	S,	9	5	6 6	* ?
	=05E	50-		50	10	su-	- - 	- 0S	58=	-05	250	
	_	_	_	_	_	_				_		
		1	121	3	217	.302	3	22	624	Ę,	9 9	
		, n		- •	Ň			-				*
				~	-	-			~	-		
	• N	a-v	0 N 19	0 7 2	-~~	0 N - 1	0 N 0 N	0 1 0	<b>-</b>	0 N <del>-</del>	<b>G</b> N O	
	CODES	CODES	CODES	CCDES	C0065	codes	COCES	CCDES	C00E5	2005	CODES	•
R,	ter	EEE	<u>ees</u>	i ĉi		112	EEZ	111	E.E.E		662	
A MH J			3 A U 4 N 4		24		12	223		20 21 21 21	S S S S S S S	A A A
11	35	3	35	22	33	33	35	<b>a b b c c c c c c c c c c</b>	34	non An		5
3	33.	89.	8.3 ±	3932 0			331	893 <b>x</b>	e e e e e e e e e e e e e e e e e e e	383	38 z	
	THA C	INI	T GNA	14	197	122	H 2 5	1984 1987		NN N	R P R	
	GROUP	6ROUP	GROUP	GROUP	68300	GROUP	GROUP	69069	_n0e9	GROLF	9005	
		:										
	1	le e	5		н "	Ξ.~		E a	Ξ	5	¥	
	CRER	CREA -	FING S	BANK 2	RESC 1	19 19 19	46E	â	180	F 1 4 8	REST	
	4	9	=	-	7-	v5	- ,	s .	۰ .	= _		
		5		. I	; 3; • 5 7	a r		a u	5	a w	5	
	DICT	110	34	D1:CT	D1C1	DICT 60		0167	01CT 20	0101	2	
	BXG	Б. М	Pag 1	PRE	PRE	PRE -	ä	384	FR.	389	Wa .	i
	ð i	ð Í	ő Í	5 5	i i i	č ž			6	5 1	5 I 0 N	Ī
										N		
		CROUP	GROUN	0000	1089	GROU	6600	c ROU	0000	GROU	10 49 10 49	
	5		5	-	L I	5	-11 -11	5	5	1	5	U U
	- E - S		i di	<b>1</b>	- <b>G</b>	5	•	19	12.	<b>3</b>	: .	

272

0F2 212

F-ANOVA OF 1

61EE8.

R-5QUABED .69420297

œ

101AL **#55** 17.123264

TOTAL 855 38.872267

101AL 155

134 E

68EST JN	<b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>21221</b> <b>22262</b> <b>22262</b> <b>22262</b> <b>22262</b> <b>22262</b>	124 CANAM 912592 912552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 922552 9225552 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 922555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 925555 9255555 9255555 925555555 92555555 925555555555	0.00000 0.000000 0.000000 0.000000 0.000000
5_00T1K	<pre># 1 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 =</pre>	LIF INCEF . 164088 . 164088 . 605217 . 050999 . 635714 . 001013 . 172057 0. 010000	.584416 .584416 .100000 .000000 .000000 .110143 .2102 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .27108 .271
4 PHONE	440 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481443 481445 48145 481445 481445 48145 48145 48145 48145 48145 48145 48145 48145 48145 48145 48145 481	153,445,45 15,161,2 15,161,2 15,162,2 11,155,2 11,155,4 11,155,4 11,155,4 11,155,4 11,155,4 11,155,4 12,000,600,400,400,400,400,400,400,400,400	+61620 •000000 •000000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •00000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •0000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •00000 •000000 •000000 •000000 •000000 •000000 •000000 •0000000 •00000000
34.0CAL	286418 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 48128 4	(1) 281/828 3MD2016 3MD2016 4E2%09 615280 615280 750250 750200 0000000	.10185 (1511) .010000 .00000 .00000 .01009 .0109 .0109 .0109 .0109 .0109 .0109 .0109 .0109 .0109 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101 .0101.0101 .0101.0101 .0101.0101 .0101.0101.0101.0101.0101.01001.01001.01000000
ZMARJTL	244280 244280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 2462800 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 246280 2462800 2462800 2462800 2462800 2462800 24628000 246280000 246280000 2462800000000000000000000000000000000000	AMAL 7515 WITH AMAL 7515 WITH 6177540 184-99 184-99 125019 125019 015192 0515455 051555	.019182 .019182 .01919 .01919 .01919 .01919 .01919 .01919 .01919 .01919 .01919
1 A GE		RESCAT • 042749 • 042749 • 1312795 • 1312795 • 133472 • 133472 • 05305 • 015373 • 053073 • 055073 • 055075 • 05507	
-	781 AL 781 AL		16 11 12 15 15 15 15 16 16 16 16 16 16 16 16 16 16 16 16 16

۰. ۱

. . .

,

.

ANALYSIS WITH BSS/TSS41)

273

18 A.

		HALA SISAJAHA	(1)32/132(1)	
Ϋ́.	OAN ST	14NECESS	15EQU (TY	<b>LECRERAT</b>
-	186930"	.021474	.001264	.357964
~	.035580	.039245	.001556	.109559
-	.036167	.00094	050400.	.145261
-	667660.	496E 10-	.00000	0,000060
in	.976523	£135CO.	.065291	.136364
	7275EQ.	.056880	.006936	000000000000000000000000000000000000000
~	.007756	044200"	-926159	158170
•	142457	.015873	.015873	0,00000
æ	.003253	021129	918R00"	.041250
•	.096182	0.00000	.016195	0.00000
11	.0+0+0.	0.00000	0.00000	*3*0*0*
12	C+1701.	.047619	.063333	0,00000.0
2	000000.0	0.00000	0.000000	6-00006
11	*20520°	0.00000	0.05030	0,0000
15	228100.	0.0000	0.00000	.001852
16	\$1+B08.	.075501	. 304468	6.00000
SUBSUM	. 76581	.34109	.20415	1.07632
TRIALS	•	*1	16	•
REANDS	.04786	56150.	.01276	.11965
3	.0+365	<b>6</b> 622 <b>0</b> °	.02042	.10740
CF VAR	.91207	91878.1	1.63054	.69765
VAR	16100.	.0003	2.000.	£211º.
RECSUM	.76531	-3414G	-20415	1.41435

ali na statistic i sa she

- 1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1 1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、1999年、199

c

54.0			5.0	12.0	10.0	13.0	6.9	3.0	C.B	7.0	11.0	4.0	2-0	14.0	9.9	15.0	16.0	1.6
		RANK																
. 1221910		<u>ک</u>	16	51	1	91	1	16	*	16	16	15	*	16	16	16	•	•
•		1 GROUD																
CALT.		AN N	52	ŝ	6	*2	90	:	26	56	16	\$	\$	55	60	5	20	=
KOUND		GRU ME	1444821	260.12	03,533	+3.679	116.316	116.573	411.05	101-852	77.046	001-011	206.130	38.765	61.673	37.802	22.625	212-164
		<b>YARIANCE PCT</b>	<b>teC</b> 00.	-00148	.00477	.00146	19500.	· C1+93	*n£00*	+9200-	.00156	E4200-	-02552	-00192	16100-	.0003.	24000-	-01153
HEAN .		84A .	90108	33605	46553	54583	64863	11755.	07688	59446	598643	29195	80145.	12+00.	91207	97819	60054	.09765
GRAND		595	•	-	-	-	•	-	-	•		-	-	Ň	•	÷	-	Ì
16	a	1) s C	.04145	-03450	.06934	.03825	.05062	01221.	£1550-	.05138	- C 3949	-07366	00041-	-96+0-	-04365	.02299	-62042	.10748
UBGROUPS 15	RECONSTRUCTE	1551/SS8#05	1.12284	-46102	titer.	[6596.	1.05758	1.59322	.81906	10616"	.69519	09671-5	16658-1	- 3+ 999	.76581	651+[.	-20415	1-91+35
. NO DF SI	VARIANTI	55/15511)	1.12284	6226+-	E1621.	:0505.	EI4E9.	1.59322	.71667	10616.	61569°	1-10053	1.62742	66642"	.76581	-34109	-20415	1-07662
VARJABLES	5 805	SUN8:																
1																		
TABLE			A GE	MARITL	LOCAL	PHONE	100118	<b>RESTIM</b>	MESCAT	17PEMP	INCOME	BANKAC	+ INREF	LOANAN	LUANTW	NECESS	EUULTY	CRERAT
SUMMARY				N	n		'n	•	~	•	۴	10	1	12	2	1	ņ	91

274

				•	**********************		
	•••••	68200 7 68200 7 85 7 85 7 85 7 85 7 85 7 85 7 85 7 80 8 10 8 10 8 10 8 10 8 10 8 10 8 10 8	ME4NE ME4NE 1.4D.= F [WREF		GROUP IS FINAL NEANS REDICION 5 JOETH CODES 1 2	1.06 750 + 0.07 70 4 0.07 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 4 0.01 10 10 4 0.01 10 10 10 10 10 0.01 10 10 10 10 10 10 0.01 10 10 10 10 10 10 10 10 10 10 10 10 1	
лаевлаевесессессессе ЮШ ⁶ 3 НЕАМ. 108 3.6.5	-800*0 =8024 8**				6800P 19*1 PAL PEAN Nº 20 5.0. PREDICTOR 5 JOBTIM CODES 8		
PREDICTOR 14.CRERAF EODES 2 3					аевсекововановеровено GROUP 1106 [MAL MEAN= N 11 55.0.0 Defiting 7 56.6.0		**************************************
		GAOUP 6 N= 34 PAEDICTOR 11 CODES 0 1	REAN= 5.C.= FINRE	-56 450505 -500505 	CJACS 2 J		
84.544.544.444.444.444.444.444.444.444.4		6 4 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7			PHEDICICH T XESCAL		•••
Nu 224 5-0-4	44 44 44	* * * * * * * * * * * * * * * * * * *	4 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9		(52000 9 MEANT WE 20 5.0.1 PREDICION 10 DANKAC CUDES 2 2 1	- 10 HSG - 46 PK085	• 55 • • 600 •
		54004 5 144 40 1 PREDICTOR 16 1 CODES 1	MEAN® 5.0 - 7 CRERA.	.41 RSG # .422 .50 PROBE <b>B.98</b> 0.		••••••••••••••••••••••••••••••••••••••	0.600*
			•		CODES 0 CODES 0 GROUP 13 NEANE MEANE MEANE MEANE MEANE	- 154 15" - 1 HSG -	• 100 • 665 • 9
14436436465608646664446		ебяоце 4 Мж 76	94454444444444444444444444444444444444	.27 PR05 = .422	CCDES 2		• • • •
		PREDICTOR 10 CODES C	- CAERAT		GKOUP 12-FINAL MEAN- N= 60 5.0=114 Predictor 5.00=114 Confe d 1	- FORD E1 -	100 665
							••

ويعتبنه ويعتدون

...

. .

.

ς,

. <del>.</del>

a a

<u>بع</u>ر :

įst.

....

275

· · .

-----

د دو کرد. در دو کرد کرد. مرکز دو کرد.

				40 40 90
				0.60 FR08
		800		ARUP 22011MAL NENN- ARU:10 5.0.5 PRC1:108 6 8573M CODES 0 1
4 2 2 4 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		• • • • • • •		• G • • • • • • • • • •
	• • • • • • •	6 - · · · · · · · · · · · · · · · · · ·	0 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	
	¥3	•	*** N/ ·	
MAL MEANH Mal MEANH 5.0.5 1 AGE	 HEAN= 5.0.1			
	. <b></b>			
CODES				

A CONTRACTOR OF ANY ANY ANY

14.14

100

and the state of the second second second second second second second second second second second second second

an a

276

COLUMN TH



-

# Ezhibit 13 -- AID4UT OUTPUT FOR 4/5 VOTE

F -----

ŀ

•

510. DEV. NA3M NA3M 17268664. 9E5564.+5 **Su**'4 T-50UARE **9" .e**84060 51.020005 101AL WEIGHT 224-80000 *22 enoue 1

									SPL11 S	MAARY									
111	6R0UP	ð 	PREDUI	Light	16	CPERAT		ND 6	9008 2009		H CODES	• N							
	GROUP	~		5 MEAN		-69	GROUP	-	N= 10	HEAN				14.91	A SQ=	-50041 F		222-4	
1.4	c.Rour	ð 	BOJE .	ctox		F 1NREF		N 10	GROUP		CODES	<b>a</b> -							
	GROUP			NEAN 2	1	-20	CROUP	, vi	6 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	NY 3H				7.614	R50=	7 E0E14.	-850=	26.40	:
11 45	GRÔUP	ð N	PREDI	CTOR	9	CRERAT		0.4	GROUP	1	L CUDES								
	GROUP		1	MEAN S		0.00	GROUP			MEAN		-2	-	5.323	RSUR 1	1 10040.	-850=	22.59	
PL IT	GROUP	5	10384	CTOR		PHONE		N10	6HOUP BOUP	111	CCDES	- o							
	GROUP			B MEAN		-50	GROUP	•	69 K M	NY SH			1-1	4.024	RS0=	1 82029.	-R50=	16.45	
SPL IT	skôup	ð •	PREDI	108	2	BANKAC		NTO NTO	GROUP	0 471	CODES		0 1						
	GKOUP	10 4	15 - 21	I NEAN		-05	GROUP		1	-H 3H		.25	<u>, 1</u>	3.907	#20=	151515	-950-	18-16	
SPL 11	GROUP	6	PREDIC	C10R	•	RESTIN		N10	GROUP	11= 2	0005	۵.							
	680.10	12	100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100	WEAN S		. 17	SROUP			ME AN		- 62	-	3.532	RSur	72840 F	-856-	10.59	
SPL 11	uROUP	10	PREDIC	CTOR	=	FINREF		N 10	6k0UP		CODES		_						•
	GROUP	*		5 MEAN	19	.20	GROUP	15	Na l	NE AN		N 4 9	T.	1.749	850=	1 63121.	-950-	14.93	-
11745	GROUP	12 01	PREDI	C T OR	1	F INKEF		011	GHOUP	1	COPES								
	ekone	16		6 MEAN:			GROUP					¥ # #	1-	3.745	R59=	7 7050 F	-850+	61.75	5
5 PL 1	GROUP	ō	PREUI	C T OR		TYPENP		0 I V	GROUP	9	CODES		_						
	GROUP		=	B MEAN		0.00	croup	9 6 T				~ ? :	T=	2.280	R59=	. 18297 F	-850=	12-29	:
	:																		
	-							1	UNUS TWO	AP Y									

DF 2 F-ANOVA DF1 a. 8-5004850 101AL 455 224 10101 707AL #55 TOTAL TSS Ň

278

214

.

65.7800

.8485

. 78296579

************

629653.64

965596.45

10

			64049 5 64049 5 85010103 1 85010103 1 7 2025 1 2	1000000000 10000000000000 100000000000		*0400 *0400 *** PFEDIC		82444444 MEANA 5.D.+		
						GHOU	PF INAL			**19,
CHOUP 1 NEAKE A 106 5.0.4 PREDICION 16 CREMAT C CODES 2 3	- 80 8 0				* * * * * * * * * * * * * * *	• • • • • • • • • • • • • • • • • • •	4 4 9 9	A DAG		• • • • •
÷						GHOUF	1945 IMAL	• • • • • • • • • • • • • • • • • • •	. 05N E4.	
• • • • • •			GROUP 4 W= 15 PREDICTOR 12 CODES 6	F LUREF	**************************************	PREDICI	2 6 T	\$ 0 • •	8084 54.	****
	***********	•••				Anda News	B°F IMAL B	**************************************		-192
+ 6.ROUP I NEANE ↑ Na 2.24 5+3+#	•••• 75	1				PREDICT	5 7 8 1	dk3d)		• • • •
	* * * *	ľ				ERDIG N		5.0.5		, 115- 9, 868-
••••			PREDICION 16	MEAN# 5.0.4 5.2.4		•• CUDES				• • • •
	*************	• • •	1 53053			640UP 1	- INAL	ж ам. 	- 15 RSQ -	-515- 0,000-
• 6870UP 2 HEAM: • W: 116 5.0.* • Predictor 16 Crerat	.09 R30 .	.500.		••••••		CODES				• • • •
**************************************		• • •				•				
			**************************************	REANE 5.0.1 RERAT						
		. :								

. го . **4** • . • . . . ***** * 

i. o

÷

φ

<del>بھ</del>

ö

e .

ар Х.

`, •

1

IN FINAL REAM IN FIN	PREDICTOR 11 FINGE PREDICTOR 11 FINGE PREDIC	99 RSG - 72# 112 FROM - 072 2 CODES 2 11 FINEEF PREDICTOR 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 CODES 2 11 FINEEF 2 FROM - 15 FINEE 2	Wi KEAN 98 69 - 728 5.0 12 PROM - 01 5.0 12 PROM - 01 5.0 22 COUCS 2 1 FINE REAN - 0 EEAN - 91 5.0 20 EEAN - 91 5.0 20 EEAN - 91 5.0 20 EEAN - 91 5.0 20 EEAN - 91 EEAN - 91 5.0 20 EEAN - 91 EEAN - 91					RS0 = 742	•
Ist PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR IN PREDICTUR IN PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR PREDICTUR	<ul> <li>*72*</li> <li>*PREDICTUR</li> <li>*PR</li></ul>	-98 E59 - 729 -12 PROM - 001 - PREDICTOR - PREDICTOR - PREDICTOR - PREDICTOR - 15 - 16 - 16 - 16 - 16 - 16 - 17 - 16 - 17 - 16 - 17 - 17 - 17 - 18 - 18 - 19 - 19 - 19 - 19 - 19 - 19 - 19 - 10 -	Wi FEAN 98 R50 722 S.O.M 12 PAGE 72 S.O.M 22 PAGE 72 REAL 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 MEAN 91 M	INAL WEAN	•••••	MEAN= .5. 5.0.= .5.	· · · · · · · ·	800.0000000000000000000000000000000000	•
IPFINAL MEAN IPFINAL MEAN INFINAL MEAN INFINAL MEAN ISFINAL MEAN IS	-72# -72# -72# -42# -43# -43# -43# -770 -770 -770 -770 -770 -770 -770 -77	98 850 - 722 -12 PROBA -001 -12 PROBA -001 -26 -26 -26 -26 -26 -27 -27 -27 -27 -27 -27 -27 -27 -27 -27	Mi EAM. 98 ESG - 728 5.0. 12 PROM -001 5.0. 201 MEJM. 91 EL 2 5.0. 26 EL 2 5.0. 27 EL 2 5.0. 26 EL 2 5.0. 27 EL 2 5.0. 27 EL 2 5.0. 27 EL 2 5.0. 27 EL 2 5.0. 27 EL 2 5.0. 57 EL 2 5.0. 57	CORS 2 CORS 2				68000 1404 11 68201 1404 11 68201 12 68201 12 6001 12	
IPFFINL SEAR Information Information Information Information Information Information Information	-728 -0015 -0016 	94 850 - 728 -12 PROB -001 	ML KEAN- 98 850 - 728 - 128 5.0					• • • • • • • • • • • • • • • • • • • •	
	• • • • • • • • • • • • • • • • • • •	-91 FSG - 1728 -12 PRG - 1728 - 93 - 26 - 26 - 26 - 42 PRG - 1728 - 1728	ML HEAN				11361944000000000 113611441 REAN= 16 5.0.5 16 11 FINREF		16-FINAL M5400 16-FINAL M5400 5-0- 5-0- 1 1 1 1

Exhibit 14 -- AID4UT TREE FOR JUDGE #6

. • #

5

.

... ⊈,∮.

;

, **x** 

...

, **.** 

, [84 - 5

_ •***** 

ی هن پ

نۇ چ

		••••••••••••••••••••••••••••••••••••••	6 RESTIN		***				
	2		• • • • • • • • • • • • • • • • • • •			11 12	неессериени МЕ Ама 5 « О « н		• 400 •
		68000 55 8 8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	- + + + + + + + + + + + + + + + + + + +	41 PROB		S 20H	LI TINREF B B B B B B B B B B B B B B B B B B B	1 053 04.	•••••••••••••••••••••••••••••••••••••••
• • • • • • • • • • • • • • • • • • •				***	CODE	s L	TINGET		• • • •
••••••••••••••••••••••••••••••••••••••	\$**				GROUP ARE ARE	15-11 15-11		- 004 00 0	
•••		• PREOFCTOR	MEAN= 5.0.= 6 restr	- 70 RSD -	-900.4				•••
		CODES 2			•680UP • N# • P860	* 125 * 125	REAKE S.D 7 RESCAT	. 46 850 =	
•6A046 2 HEANS H 7 5.0.4 • PEDICT7A 16 CREALT • CODES • 16 CREALT	15 105 1. 16 1008 1.						SEEAME SEEAME S.O.E.E		
		• 640UP 4 • 640UP 4 • Ns 53 • PREDICTOR				, ,			•••
		* CODES 0 1				1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	HEAN= 5.0.= • INCOME	-14 R34 -	
									•••

				9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9
			•	- 20 850 8 - 49 PR08 - 40 PR08
	• • • • • • • • • • • • • • • • • • •	и Е Ант К Е Ант Е L - 2		IAL MÉAN= 5.Dc= ■ INCOME
21=11 21=11 1 2				
PREDI CODES		• • • • • • • • • • • • • • • • • • •		PRE COCE
	•			
		••••••		
		е в М		E T T
	• • • • • • • • • • • • • • • • • • • •			IS EQU
••••••••••••••••••••••••••••••••••••••		• • • • • • • •		
	*	3 3 3 3 2 3 2		
. 185	•			* * * * * * * * * * * * * * * * * * * *
	0 9 1 4	•••••		
		22		<b>1</b> 0
1919-99-99-99-99-99-99-99-99-99-99-99-99	• • • •	D N N		
11XX				INAL ERST
DICTOR				
5 30 8 8 4 0 9 8 4 0		3		

:: CHARTER CONTRACT CROW 13*FIMAL REAN MA PREDICION 15 EQUITY CODES 1 2 604049 & MEAR • M • 2 5.0.4 • LEVEL 2 • • ********* ... 1 PHEDICION 6 RESTIM UP 9 NEAN= 11 S-D-0 LEVEL 2 • • -680UP

ė

•

. .

đ

्त. क.**ल**5

e)

×,

Exhibit 15 -- AID4UT TREE FOR JUDGE #7

1

Æ

		••••••••••••••••••••••••••••••••••••••						
COLES 2 3 CREAR					6680UP 11	#6495		0150 0150
	•	66000 N= Predictum Codes >	MEAN+ 5.0.4 CRERAT	- 72 RSC - 95 PRSC - 96 PRSC				• • • • • • • • • • • • • • • • • • •
GROUP I REAM					CODES 0 1 CODES 0 1	у Ілгинс есеесеение я:В.# 3 LO:AL		
	•	CAROLE S A CAROLE S A CAROLE S A CODE 1 CCDE 1	6 CAEAA	200 90 90 94		не Ана Не Ана 5 « D « в	- 1545	• • • • • • • • • • • • • • • • • • •
GROUG 2 MEAN 4 R. 114 5.0. • PREJETOR 16 CHERAT • CODES 5 1	9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9						=80%d 2♀* = 358 E2*	
			NEANE 5.C. 6 CRERAT	. 29 RSU *	.509 600 600 600 600 600 705 705 705 705 705 705 705 705 705 7			

 CUDES 6 1
 Cudes 6 1 * * * * * GEROMP'ISPINL MEALS M. J. S.O.4 Predictor & restim Cours 2 • 6800P 10 MEAN • Na 20 5.0.4 • LEVEL 2 ٠ 560.0° 2561att REAm .57 PSG 1 .714 560.0° 2561att REAm .57 PSG 1 .714 5.0.5 5.0.5 .0.0 Predictor 13 Loantm .59 PR08: .800 Codes 2 CODES 1 LONNY CONTRACT 1.00 RSO - 7340 CODES 1 CONTRACT 1.00 PEOES - 360 PREDICTOR 13 LONNY CODES 1 CODES 1 CGROUP 1, MEAN= 8 N= 26 5-0-0 1 LEVEL 2 .

Contraction of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second seco

(

. . .

285

ć

			· · · · · · · · · · · · · · · · · · ·		
					- 190 ASL -
			астоссоссоссоссоссоссоссоссосс NP 270FINAL И£ИН 5.0.2 €.01C10R в түРЕМР 00CS //		100 2641Mal Mean 1 3 5.0.4 201004 6 179549 2055 1 3
• • • • • • • • • • • • • • • • • • •					
		î,	9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9		
		MEANS 5-D-F EL 2			
68646 ⁹ 1797				6 POUP 16 N N 18 CODES 0	
• • • • • • • • • • • • • • • • • • •	•			• • • • • • • • • • • • • • • • • • •	*
		-::			
		NEAN= S.D.M		HAL HEAME HAL HEAME 5 - D - * 6 RE ST IM	
CODES 1 2		state		GROUNT ZUNTI NE PREDICTOR CORES	

г.ч "Я

.

đ•,

x

· *

£

به بنه

· .北田里 · · · · · · · · · ·

/.....

286

<u>م</u>

н - Муна - М

. _____

Exhibit 16 -- HIT/MISS TABLE--4/5 POLICY WITH CREDIT RATING

••

î.

Å

່ ອ₁ ກ

Ņ

			1	אוב ז-כוושיוני	11vE Facouer	CV COUNTS				
PREUJCTEN	A UNCUP	MJNCH	B B B B B B B B B B B B B B B B B B B	RUNGER	1012E4	Rijestije Billion		A38mu#	*CHBEB	OHJECTIVE
SCORE CUT-OFF	NUK 14 Sateral	ACT 1 14 JATERVAL	ACT 1	MRED L	PHEO 1	PHEN B	8 (1) 8 401 8	DJEN 0	roekse1	FUNCTION
1.00	1	**	4	¢	•	ç	"	16	ī	1.
		<b>,</b> -						5	44	50
	•		ā		č	3		64	•	:
	,		92	¢	29	Ţ	11	4	:	;
•••	0	0	24	۴.	29	1	76	Ş	-	•
- 15	-		5	¢	ê.	÷	1	76	201	001
	•	•	5	•	3	4	5	2	194	101
-		•	2	•	2	*	37	74	11	
24.	•	£	**	c	2	2	Ģ	74	11	
Ŧ	G	e	2	¢	*	91	÷	42	1	
. 90	~	~	£.	c	2	5	1	-		2
84.	•	e .	4	¢	:	5:	ç ;	::		
			2;	C		đ :	5;	5;		
	•		:;			£ 1	;;			4
		9			::	5 1		: :		
6	•			6 4		1				
::	•	•				: :				
		~				<b>,</b> 2	: 5			
		<b>-</b> •	2				: 2		411	411
	-	-	2	- •		:	10	a de		•
	• •				12			Ţ	•	11*
		-		• •		, c		5	124	120
		• •				2	2	4	121	123
	•	• •		•	3	23	17	49	123	E21
		e		e	1	\$	:	3	121	123
	4	•	4 R	~		<b>د</b> م	ž	Ĩ	122	122
Ę	-	<del>ن</del>	•	-	67	27	2	5	122	122
-12	-	•		•	23	2	<b>2</b>	<b>S</b> :	121	121
				<b>~</b> 1		€ 2	2 ;	ē		128
0	•	e ·		•			23			101
				•		5	1 <u>1</u>		121	123
		c			16		*	5	122	122
	-		5	F	7	23	2	5	122	122
51.	•	e	5	-	7	ť,	<b>;</b> .	5	122	122
44	-		3.	- ,	26	2	#:		2	21
							1;		5	1
	- •	-			; ;			5		
1041	• 0	: 0				2	: 2	1	121	125
	• •••						2	5	124	126
.54	•	æ	:	-	•	•	2	5	124	128
-57	•	ſ	:	•	67	•	2	ş	124	129
•5•	~1	-	5	-	2	Ĩ	*	ç,	623	6 Z 1
- 55	. ••	~	5	-	100		2	e i	5	
45			2	- 1	101	2	×;	5	201	261
Ş	•	0	19 4 19 4			::	;;	, c		261
2	9:				191	. :	:;			12
	<b>}</b> 1	3 **			102	: :	: 2	i i i	111	-
	•	•		•	1					

288

							•			
SCORE		ACT 1 IN		1 0384	PAED 1	0 0374	C U3M	D URA	L'ORM'CT	NO LONIA
CU7-CFF	JUTERVAL	LATERVAL	4CT 1	ACT 6		ACT 1	4CT A			
•••	•	•	z	-	102	1	٠E	84	133	11
		• •		-	162	<u>•</u>	:	Ţ	131	3
122	•			-	201	1	=	44	1	=
44	•	-	5	-	1 <b>a</b> 2	-	1	Ş	1	
.45	-•	60	5	•	103	<b>!</b>	Ē	41	1 32	22
			100	•	401	2	= ;	4		22
7	~	~	201	•		₫.		:		1
245	~* <b>4</b>	-	103	•	187	5	6	Ŧ	<b>4</b>	
•	-	ò	[4]	r	104	<b>-</b>	2	5	5	<u> </u>
04.	-	e	E 0 1	•	108	61	32	42	5	5
•2•	-	1	104	ď	101	•	22	4	114	
	•	¢	104	r	747	•	2	Ţ	1	
	-	c	•	•	101		2		<u> </u>	
	•		471	ur i	144	<b>7</b> 1	21		2	
.15	-	-	5	ť	071	<b>a</b> 1	ê i	5		
	~	-	105	æ	211	•	1	<b>a</b> :		
5	-		:07	4		£	= ;	2		1
22.	-	•	107	¢	=	4	F	2		1
11.	•	•	101	4	1	•	7	1		5
<b>9</b> .	•	¢	107	4		-	F	1		
92.	•	-	107	ď		~	F	2		
.28	•	Ð	101	4		•	7	11	Ě	
12.		-	107	•	•11	4	ş	ž	117	161
42.		•	107	•	115	£	R	ž	5	<b>9</b> 71
52.		67	107	æ	÷11	•	L	2	¥01	961
		•	107	•	-1-	•	T.	*	135	135
.23	•	-	107	•	114	æ	<b>د</b>	ŕ	ŝ	135
		-	101	•	e11	~	T.	*	134	135
	~	-	104	5		•	1	2	135	135
	-	-	104	-	110	•	14	2	135	135
			101	5		-	2	1	114	136
	-			12	121	•	X	r	104	13
	-		104	5	122	•	2	¥2		661
		-	110	2	[2]	-	2	27	114	•
51.	-	£	110	11	124	-	ĩ	24	1.51	661
•	~	-	110	16	126	-	ĩ	*	Ē	1
		•	110	17	127	•	2	2	134	130
21.	•	c	11¢	21	151	-	4	•	124	4 N N
.11	-		111	2	201	•	£	-	121	121
-10		<b>f</b> 2	111	<b>3</b> 3	6	•	5	17	124	•2
<b>B</b> O.	•	•	171	23	[1]	•	Ā	1	124	92
<b>6</b> 0 .	<b></b>	•	111	2	134	<b>R</b>	2 :	£ ;	2	621
• 0.7	. <b>'</b> • <b>9</b>	62	111	23		•	2	4	121	123
• 0 •	. y	-	112	2	2	-	2:	2:	[2]	21
6	•	<b>e</b> 1	211	52			= :	1:	29	22
• •		-	11		121	-	<b>_</b> :	- :	221	
2			1	2	871	•		-	5	2
20.							5.	5	5	33
	•	8	=	2		6 (	2.			33
80^ I	8	E	113	11		F	r	6	:	

¥

Ŋ

.х ,

.

Ø

2

ų

Ņ

Т. Б

÷.,

:

1,

. đ

. . .

a ∛ 7 7 11 ₹

t

Exhibit 17 -- HIT/MISS TABLE -- COMPOSITE POLICY WITH CREDIT RATING

· • * ٢

/_____

	DEJECTIVE FUNCTION	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$
	NUMBER CORRECT	\$ 5 5 7 2 7 6 6 5 6 5 5 5 7 7 5 7 5 5 5 5 7 5 7 5
	4 2384 Number	<u>NN </u>
	NUMBER PRED B ACT B	£6266555555555555555555555555555 <b>335655</b> 84584534344444
CY COUNTS	NUMBER Pred 0 ACT 1	x eel 2 c 3 c 3 c 3 c 3 c 7 c 7 c 7 c 7 c 7 c 6 c c c c c c 2 c 2 c 2 c 2 c 2 c 2 c
TIVE FREqUEN	ми <b>нес</b> Ряде <u>I</u>	NYN NY4 4 486 49 49 49 40 AN AN AN AN AN AN AN AN AN AN AN AN AN
#1 5-COUCH	NUN929 FRED 1 ACT 8	
76	NUMBER PAED 1 ACT 1	オ り こ む で み み み み み み う う て て て て て て ち み み み み み み み み み み み み み
	NONCUM ACT I IN SNTERVAL	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$
	NONCUS NUM IN INTERVAL	은 사실 수 위해 정확은 시 위수 은 시작은 전체가 내 위한 방과 참 것은 것 것은 것 같이 것 같이 가 것 가 가 가 가 가 가 가 가 가 가 가 가 가 가 가 가 가
	MEGICTED SCORE OUT-OFF	

COUNCITY FUNCTION	496966969666666666666666666666666666666	121
MUNGER CORFET	81171111111111111111111111111111111111	125
MURGER		:=•
1 100000 1 100000 1 1000000	aaaaaa ann phing cungg figgaraa <b>talaa a</b> alla talacha talacha ta	12*
NUMBER 8860 6 867 1	***************************************	140
NUMBER PRED 1	$\begin{array}{c} \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet \bullet$	51
1 (384 1 (384 8389		:::
NUMBER PAEC 1 ACT 1	<b>*************************************</b>	193
404004 407 1 10 14758446	# 8 m m w 2 8 8 8 8 8 8 4 4 m N ON 4 8 N m 9 2 8 4 9 9 8 m 5 8 7 9 m 8 8 8 9 1 m 9 8 8 5 9 7	- # 2
404504 414 24 415 24	*****************	• <b>1</b> 1
AEDICTED SCORE CUT-OFF	************	

Ĭ,

Exhibit 18 -- HIT/MISS TABLE--TENTATIVE APPROVAL WITHOUT CREDIT RATING

_ + _ - **- -**-

5

NR JECTIVE FUNCTION			21					6		2				1					2					Ξ	111	<b>C</b> 11	115	114	21		1	115	115	•					121	122	123	123	123	123	621		124	124
NUMBE CORRECT		-	2		54	<b>U1</b>	C F		40	1	1	2 :	F	::							5					611		•16	=			115	115	114						122	123	121	[2]	123	123		1	
MUNBER MUNBER		• 17	691	107	164	105	103	102	8	6	40	56	26			;	:;		; ;	:;				i vi		3	5	ş	E T	<b>;</b>		::	3	÷.	Ŧ	Ŧ			5	15	14	3	f	*	32	N P	5	
NUMBER PRED A ACT B	į	*	ź	ň	\$	46	ţ	5	2	¥	<b>1</b>	<b>£</b> :	3	1	5	<b>,</b> ;	,	22		21	2					2	1	2	r	2	21	22	52	23	2	2	21	07	25	ເຕ		2	2	22	21	2	23	Ş
1000000	i		5		7.	D¥	R 4	63	ł	:	ł	54	2	5	5		7			•	c . • r	; ;	1			1	2	4	2	22	22	52	5	20			<u> </u>	5:	::	<u> </u>	=	:2	-	~	=	=•	•	,
NUMBE BHED 9		•	•	*	:	<b>4</b> 4		;	51	-	5	55	7	3	44		21	<b>*</b> ;	2;						9 9		100	101	241	101	0	901		107	109	109	2			113		•11	:	•11	3	110	5	
901284 2010 1 201		-	-	_	-			~	. •	. ~	•	•	~	-	-	•	~	•	•	•	•	r 1		• :		: :			12	1	2	1:	. :	-		-	2	1:	<b>-</b> .	11	2 2	2 2	1	5	5	<b>1</b>	5!	- 1
NUMBER BMED : ACT :		۳,	•		5				5	4	5	5	- PG	с <b>.</b>	-	-9 • 1	5	5	21	5	21			:	: :			3	<b>.</b>	•	ي <b>ا</b>	24		5	Į.	ž.	£.						100	101	107	102	40	• • •
40%CLM 407 1 14 145844L		1 1 1	-	~		-		•	. •	. a	•		₽,	•	~	-	•	•		-	-	e.	-	• •		- •	<b>i</b> a <b>p</b> i	-		~	£	- (	FC		~	ø	•	- 1	•	• •		- 6	• •	) g	_	•	~	
16464 14 14 14 14 14 14 14 14 14 14		*					•			1 4	-		-	•	~	-	•	-	-	-	~	•			<b>N</b> -	-		<b>d</b> =-1		ų	2	P4 4		,	- 64	•	ų		~	<b>o</b> -	• -	- •	• •	- ~	~	•	-	•
REDICTEU \$CORE Cut=OFF		1.50	. 68						1				69.		.87	ļ	.45	÷	. <b>.</b> .	28.	19.									1	.70				-	44,	Ţ	20.						55.	•3•	5.	A	16.

292

------

20	NUN IN INTERVAL	ACT 1 TH INTERVAL		9 1 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	PHED	- 134-	ACT A	PRED	CORRECT	FUNCTIO
	•	•	1	5	[2]	L	1	27	124	124
	-	•	105	5	124	e		26	123	123
	•	•	1 05	:	z	æ	×.	¥.	121	21
	9	÷	105	<b>P</b> .1	421	•	<b>E</b> .	×2	121	621
	•		105	<b>1</b>	421	•	5	ŝ	2	
		•	501	5.	21		2:	( ;	124	
	~ 4		50,	24		• •	2	:;		121
			501			•	5:	6,2		
	•		5	Ç 4						121
	,			5					1	121
		-		52			: =		2	121
			101		126	•			21	121
		•	105		ž	•	: :=		5	121
	• •	•					: ≤		121	121
	•			1	Le	•		1		122
	• •	- •				- •		3		122
	•	•	101					2	2	121
	•						. 2	Ĩ		122
	• •	- •								100
	•	> <		;;				;;		122
				2:			::			10
	• •			3:			::			
	~ 4			17		Pa	::			
		-				• •	. :	2		
	• •	- •		11		•	: :			1
		4		27		• •	1 1			12
					12	• •	2 2			2
	• •	D (		.;	32		2:			
				:;		• •	-		2	
				67		10				
	•	6		::		•	2:	-		
		с.		6;			2:	11	221	
	••			ξ 2		n #				101
	4 4	•				•	2	2:		
	•	•						12		
		• •		.;		•	5.5			
	• •	- 1					5.5			
	•	•		ù ř			<u>t</u> -	-		
	•			. :			53			
	01					<b>u</b> 1	-	2	5	2
		e .		ũ,			÷.	21		
							• 0	::		120
	• -						•			
			=	5.					112	
	> •			2 9		.,				
				5		• 1	6 1	5 :		
		•		22		<b>N</b> i 7	4 .			
		-					r •	5		
	-			23		•	Ŧ	8	2	
			•							

ر کې د کې د سال

> . ابر . الروب

÷.

, **R** 

ŝ

29,3

Exhibit 19 -- HIT/MISS TABLE--4/5 POLICY (LOCAL MODEL) WITH CREDIT RATING

....Ę

7

.

ч жо

• - e

5

e A

ģ

, *1* 

L-

-----1 8

(-----

л	
COUNT	
FREUULNCY	
2-CUMULATIVE	
ARLE	

UBJEL I IVE FUNCTION	0 *	<b>c</b> <b>s</b>			120	121	121		נזו	123	143	146	166	126	120	125	146	148	128	128		125	871					151	177	161	261	27		201		201	135	221	564	132	261	775	561	132	2F1		201	751
NUMBEN CORRECT	04	40	* C -	001	001	221	12		123	121	123	120	126	126	126	126	126	128	128	128	144	126	128	IEI	161	191	131	IEI	131	151	134	751	Zrt			251	132	132	132	261	261	132	261	132	251	251	271	136
лика£ н Р.нЕ.0 0	145	145		n u 9 4				5.0	50		2.9	5	53	25	65	59	56	57	57	<b>-</b>	57	52	57	55	3	25	25	25	Ş	52	5	7.1	<b>1</b>	1.5	57		5		15	15	51	51	51	51	<b>.</b>	15	77	10
NUMEEH Pred 0 Act 0	36	36	36	5		9 ×		5 4 7 m	2 <b>x</b>	2.4	9	1	<b>9</b> E	36	æ	36	36	9e	36	ж.	36	36	Ъ£	ŝ	35	ŝ	35	ΞĴ	55	<b>3</b> 5	÷	5	£ :	35	5 4	ζ₹	1	5	ڑی د	21.	35	35	5	35	35	ي ئ	ť,	5
NUMBER PRED 0	109	104	5	59		5	240	5	9 4 9 7	2 4	9 40 9 40	ריי ו ריי ז	23	53	23	23	<b>23</b>	23	21	21	21	21	21	11	11	-	17	17	17		36	16	9	9.		2 4	4	91	16	lo	16	16	36	16	16	16	91	lt
	ŝ	5	61	0 0	n . 0 a	α Γ	0 0 0 0	0 0	0 a	0 Q	0.93	33	16	16	16	1.5	6	66	E 6	6	66	5	3	98	96	96	86	85	6 B	Э.Р	5.4	6.	55	0° 1	r	10		55	5.5	5	99	55	99	5	55	5	6 C	66
ALE CTUMULA NUMBER PRED 1 ACT 0	-	4				- ••			-	• •	- 1	• -						1		-	-	7		ณ		N	2	ny.	<b>ب</b> ر.	<b>∼</b> ∎	ۍ ۲	ŝ	2	vr	<b>v</b> 1	4.9	• ~	i N		~	- ~4	2	~	~	2	~	[,] ب،	~
NUMBEK PRED 1 ACT 1	ł	t	78	4 80 1	30	45		20					60	06	6	90	90	56	26	56	5 <del>0</del>	56	56	96	<b>9</b> 6	96	96	96	96	9£	ĺ6	16	16	97		24	10		26	67	16	16	16	16	15	67	16	97
NGHCUN ACT I IN INTERVAL	*	9	2	e ·	<b>.</b>		<b>.</b> ,	5	<b>ت</b> د	<b>.</b>	5 0	• •		• •	. 0			~	6	¢	a	a	a	4	0		3	a	a	Ċ	-	0	0	0	9 (	50	<b>,</b>		• •						ø	จ	0	9
NONCUM NUN IN INTERVAL	s	0	74	Ð	0	9		<b>D</b> (	-		•	<b>,</b>		a		- <b>G</b>	. 0	2	9	0	0	ŋ	G	- un	a	a	0	0	Ċ		-4	o	ø	0	0 4	<b>.</b>	י כ			. 0	• •	. <b>'</b> 3	. 6	. J	0	c	0	c
PREDICTED SCORE CUT-OFF	1.50	66*	96.	163	•95	- 95	<b>46</b>	E6•	26.	14.			7.8.	49	- 92-	46.			191	.60	. 19	.18	12.	.76	. 75	.74	E7.	. 72			0	-67	• 66	. 65	+ <del>1</del> + + + + + + + + + + + + + + + + + + +		20								.53	•52	15.	-50

She with a set of the set of the set

294

: 1

•

1.1

Succession First of

PREDICIED	RONCUM	NONCUM	NUMBER	NUMBER	NUMBER	NUMBER	NUMBER	Pumber	NUMBE 1	UBJECT . V:
SCORE	NUM IN	ACT I IN	PRED 1	PHEI) 1	PHED 1	PHED 0	FRU 0	P.4.0 0	CONHECT	FUNCTION
	INTERVAL	INTERVAL	. ACT 1	ACT D		ACT I	ACI 0			
ç	4	<	07	n	65	. 91	35	51	132	261
х а 4 •	2 4	• •	10	4 N	6	-9	50	١٢	132	251
	- 4		90	y n	000	i L	5	50	661	CE1
	- 4	- 0	0	4 0			5	50	551	661
	•	2 4	0 0	; n	100		55	50	661	CEI
	2 6	• •	10	4 N	100	15	35	50	133	551
		• •	86	1 ~2	100	15	35	50	133	133
•	• •	• •	86		100	15	35	50	133	
14			96		100	15	35	50	133	CE 1
	• ~	• •	86	~	100	15	35	50	EE 1	661
			5	1	100	15	ŝ	50	CC1	133
			96	~	100	15	35	50	<b>[3</b> 3	[[]
		• •	с С		100	15	35	50	661	133
	• =		65	1	100	51	35	50	661	[ <u>[</u> ]
		. 9	6.6	~	160	15	35	50	661	CEI
4	•	•	66	. 01	. 161	*1	5	*	134	4E I
55	. 9	. 0	66	~	101	14	35	4	134	45.6
	• •		50	~	101	14	35	4 6	134	134
			100	~	102	13	35	H 4	135	135
	• <	• =		•	102	13	35	10 f	135	135
	<b>.</b>	• •	100	1	102	1	35	<b>4</b> 8	135	135
	•	• «			162		55	4 B	135	135
10 P		> <		<b>u</b> ^	101	2	50	4	135	1.5
24		• •		101	201	1	35	<b>4</b> 6	135	135
			100		102	5.3	35	48	1,35	135
			100	. •	102		35	10 1	135	115
	• C	• •	100	. 0	102		35	48	135	135
		• •	100		102	61	35	48	135	135
22.		• •	001		102	61	35	48	135	135
	• •	a	100	1	102		35	8047	135	1.15
			100		102	:1	35	44	561	3C L
	• <	• •	100	• ~	102	13	35	10 <b>4</b>	135	1.15
217			100	~	102	13	ŝ	10 17	135	561
			100	2	102	2	35	10 4	135	135
	-	•	100	2	102	51	3C	6) 4	135	551
41.	9	3	001	~	102	2	<b>32</b>	80 ·	51	5° 1
		•	100	ŝ	162	1.1	ŝ	<b>8</b>	135	1.15
- 12	0	0	100	~	102	51	S.		<b>1</b>	
		0	100	ŝ	102		ŝ	10 •1	<b>5</b>	771
01.	•	•	100	2	1 ù 2	13	35	Ð (	51	1 n n
60-	ø	•	100	~	102	13	5	80 S	<5 T	11
.08	9	9	100	~	201		1 1	0 ( 7		<u><u></u></u>
-07	•	a	100	~	132		ξ, j	60 a 17 -	251	577
- 96	c	•	100	2	102	5	35	10 L 37 -	51	551 571
· 05	-1	rų.	102	-	105	. 11	*	Ĵ.	071	1.00
+0+	11	•	111	3	122	~	56	9 Q		1.1
E0.	•	<b>Q</b> 1	111	3	122	~ ~	9.2	50	101	121
-02	0	\$		3	271	•		6 J 0 U	- F.C.	101
-21	9	3	111	31	221	~ <	ខូ <	6 C	22	
00	28	2	611	1E	561	>	>	3	<b>D 4 4</b>	5 = 7

1. j N

⊐ ,

د 1...

່. ເ

. .

ъ.

. . .

2

295

•

# APPENDIX C

# Users Description and Listing for AID4UT/AIDTRE and Ancillary Programs

### Introduction:

: ;;

` A 🔅

AID4UT/AIDTRE are a pair of computer programs which implement the Automatic Interaction Detection (AID) Algorithm on the University of Texas CDC-6600 computer. They are adaptations of the AID-4 computer program developed by the Personnel Research Division of the Air Force Human Resources Laboratory, (AFHRL) Lackland AFB, Texas.

The basic AID algorithm was developed in 1964 by John A. Sonquist and James N. Morgan at the University of Michigan. They developed and exported the AID-2 program to various universities (including UT). The AID-4 version of the program from which the AID4UT/AIDTRE programs were developed represents an expansion of the AID-2 program by the AFHRL staff to include the following optional capabilities:

1. An option to select randomly a sample from the original group of observations and to process only that sample.

2. An option to select randomly a sample A and a sample B from the original group, to process sample A first and maintain a history of the splits, and then to force sample B to make the same splits as taken by sample A; i.e., single cross-validation by forced splitting.

3. An option to do the same as indicated in item 2, but in addition, to process subsequently sample B freely, and then to force Sample A to make the same splits as taken by sample B; i.e., <u>double</u> cross-validation by forced splitting.

4. An option to allow sample A and sample B to be provided by the user; i.e., no random sampling with AID-4, and to provide for single cross-validation. Actually double cross-validation can be accomplished by two separate runs and the switching of samples A and B.

5. An option to exclude cases with out-of range predictor

values.

An option to specify the number of iterations to be 6. printed in detail.

An option to print only a brief summary of the splitting 7. involved.

> 8. An option to print at each iteration each of the following:

A t value testing the significance of the difference of a. the means of the resultant groups of a split.

 $A R^2$  value at each split indicating the percentage b. of variance explained up through the current iteration.

4

5.

يد تر ال י איי י

к.^д. Х

¥.

. -у

1.12

A F value testing the significance of the increase c. in  $R^2$  from the previous iteration.

A F value for a one-way analysis of variance d. conside ing all final groups at this iteration.

9. Various format changes for control cards, reports, etc., including additional messages explaining the direction of the split.

> 10. An option to plot the resultant splits.

The AID-4 program was operable on an IBM 7040 computer system. It was documented for use by the AFHRL in an internal technical report.

Modifications to the AID-4 program required for its implementation on the U.T. CDC-6600 included:

Development of a Fortran IV version of the plot routines. a.

Conversion from the program linkage used on the IBM b. 7040 into two separate programs with a tape file interface.

Further modifications were made to improve the utility of the program and included:

Addition of an optional skeleton tree plot that includes a. only the first eight plot levels.

Addition of a routine to display BSS/TSS profiles and b. to calculate a coefficient of Potential Explanatory Power (PEP) as

proposed by B. M. Finifter of Michigan State University.

Purpose:

**.** .

..`

1

٠.

•___

. . .

...

.... ,

e,

.

21

۰.

•

_____

0 10.0

The purpose of this appendix is to provide a User's description of the AID4UT/AIDTRE program and to provide information on the ancillary programs used with AID4UT/AIDTRE in the process of Policy Capturing.

With the express permission of the Commander, AFHRL, Brooks AFB, Texas, the description of the input to the AID-4 program has been taken directly from the AFHRL technical report entitled "Automatic Interaction Detection, AID-4", July, 1971.

Additional information has been added to reflect the modifications made in the AID4UT/AIDTRE version of the program.

# THE AID4UT ALGCRITHM

The basic steps of the AID4UT algorithm for the "normal splitting" of groups into subgroups; i.e., no forced splitting, can be summarized below. The only additional logic required for forced splitting is concerned with the generation of a history file of normal splits to be imposed upon a different set of observations.

## Initialization. Steps 1-4.

Ł

. )

ċ,

 ${}^{\rm e}$ 

ŗ.

6

÷

1.1

i q

ъ

Ф.,

Ş,

. .

e

i.

÷,

*

1. Read in all control cards (title card, format card or cards, description card, predictor cards, and criterion card) which define the predictor variables, the criterion variable, the stop criteria, etc., and establish all of the options to be processed. If an end-of-job card is read, terminate the job.

2. Print a summary of the control cards and all the variable definitions under "Control Card Parameters."

3. Read in the original input file, according to the options specified, and generate the recode categories (classes, responses, codes) for each case (observation). Edit each case and print each case with an out-of-range predictor values are to be excluded, also print any such cases under the same report. All cases listed under this report are eliminated at this point.

4. Identify all the remaining cases (assuming that no forced splitting or random sampling or partial use of the original input file is called for) as belonging to group number one. Group number one is the current candidate group for splitting and, indeed at the start, the only one. Continue to Step 8.

Test for Termination of the Algorithm. Step 5.

5. If either the current number of final groups (includes the candidate groups and the groups that cannot be split) equals MAXGP (a maximum number read in from the description card), or the number of candidate groups equals 90, terminate the algorithm by going to Step 25.

Determine Which Group Should be Selected for Attempted Partitioning. Steps 6-8.

6. Considering all groups obtained so far, select one of

A STATE OF THE OWNER.

them, group i, such that

на на на на на

110

a. The total sum of squares of that group (TSS₁) is greater than or equal to Pl (percentage read in from description  $c_{a:d}$ ) percentage of the original total sum of squares (TSS_T); i.e., group number one.

b. The number of cases in group i is not less than twice the value of NMIN(the minimum number of cases required to be in both resultant groups for a candidate group to be split, read in from description card).

c. The group i has not already been split into two other groups.

d. There has been no previous failure to split the group i.

e. The total sum of squares of group i is not less than the sum of squares for any other group that meets the above four criteria (a-d).

7. If there is no such group i, terminate the algorithm by going to Step 26.

8. The group i selected is the current candidate group i which will be used for an attempted splitting (partitioning). Identify it with its group number (i) and print

 $N_i, W_i, \Sigma Y_i, \Sigma Y_i^2$ , TSS_i, and  $\overline{Y}_i$ , and S.D._i.

Partition Scan Over All Predictors. Steps 9-14.

9. Set j, the predictor index, equal to one, BSS jMAX' the maximum between sum of squares over all predictors of group i, equal to zero, and skip to Step 11.

10. Increment j by 1. If j is greater than the total number of predictors, go to Step 23, for the partition scan is complete. Otherwise, continue to Step 11.

11. For each category k of predictor j of group i, compute

 $N_{ijk'} W_{ijk'} \Sigma Y_{ijk'} \Sigma Y_{ijk'}^2, Y_{ijk'} S. D_{ijk'}$ 

12. If there are at least two categories k such that  $N \neq 0$ , continue to Step 13. Otherwise, print an appropriate message and return to Step 10, for predictor j has only one category.

13. If predictor j has been defined as monotonic, skip Step 14 (do not sort the Step 11 statistics), and go to Step 15 instead.

14. Sort the statistics produced in Step 11 together with the category identifiers for predictor j, into ascending order of the means  $(\overline{Y}_{ijk})$ .

Partition Scan Over the k Categories of Predictor j. Steps 15-19.

15. Set p, the category index, equal to zero, BSS the maximum between sum of squares over all categories of predictor j, equal to zero, and skip to Step 17.

16. Increase p by 1. If p is as large as  $k_j$ , where  $k_j$  is the number of categories within predictor j, then print the statistics for category  $k_j$  and go to Step 20, as all possible feasible splits to be examined are completed.

17. If either

1

. ý

<u>_</u>____

.

×.

RIN G  $p \qquad \Sigma \qquad N_{ijk} < NMIN \text{ or}$   $k=0 \qquad N_{ijk} < NMIN,$   $(N_i - \sum_{k=0}^{p} \qquad N_{ijk}) < NMIN,$ 

go back to Step 16, as this split cannot be made. Otherwise, compute BSS, the between sum of squares for the attempted split of group i on predictor j between the categories  $(0, 1, \ldots, p)$  and the adjacent categories  $(p+1, p+2, \ldots, k_j)$ , and print the statistics for category p according to the print options specified.

18. If the BSS computed in Step 17 is less than  $BSS_{kMAX}$ , the current maximum between sum of squares, return to Step 16. Otherwise, continue to Step 19.

19. The BSS computed in Step 17 is the largest encountered so far for predictor j. Set  $BSS_{kMAX} = BSS_{p}$ , store p, and return to Step 16.

Determine the Best Predictor. Steps 20-21.

ų,

 $\mathbf{a}$ 

è

в

्राः _____ 20. If  $BSS_{kMAX}$ , the maximum between sum of squares over all categories of predictor j, is equal to or larger than  $BSS_{jMAX}$ , the maximum between sum of squares computed for all previously scanned predictors, continue to Step 21. Otherwise, return to Step 10.

21. BSS_{kMAX} for predictor j identifies the best predictor and category tested so far of group i. Set BSS_{jMAX} = BSS_{kMAX} and return to Step 10.

Determine Whether to Make Split. Steps 23-24.

23. If the maximum between sum of square,  $BSS_{jMAX'}$  obtained after the scan of each category of each predictor of candidate group i is equal to or greater than P2(percentage read in from description card) percentage of the original total sum of squares,  $TSS_{rr}$ .

continue to Step 24. If not, eliminate group i as having failed in a split attempt, compute residuals as the options specify, and return to Step 6.

24. Split group i into two new groups, group NCF+1 and group NCF+2, where NCF is the current number of candidate and final groups, according to the predictor and category identified in Step 21. Print all information concerning the split including BSS iMAX,

 $BSS_{jMAX}/TSS_{T}$ , and the t value testing the significant difference between the means of group NCF+1 and group NCF+2. In addition, print a "Current Summary" indicating at this point the total number of candidate groups (including group NCF+1 and group NCF+2) and groups that cannot be split,  $TSS_{T}$ ,  $BSS_{T}$ ,  $WSS_{T}$ ,  $R^{2}$ , the total proportion of the variance explained,

an F value to test the significance of the reduction of the error sum of squares due to the new split, and an F value for a one-way analysis of variance for all final groups so far. Add group NCF⁺¹ and group NCF⁺² to the number of candidate groups and list all candidate groups, according to print options specified, before returning to Step 5. Termination of the Algorithm. Steps 25-27.

۹ ۲ ۲

τ,

÷, *

r .

**

2 1.45]

. **9**94

191

> 4 3

8. **Å** 

. .

1 I

ŗ

25. Either the current number of final groups equals MAXGP or the number of candidate groups equals 90. Print an appropriate message, terminate the algorithm, and go to Step 27.

26. There are no more groups eligible for attempted partitioning. Print an appropriate message and go to Step 27.

27. Compute any residuals, according to options specified and return to Step 1.

# The Program:

11

The compiled AID4UT/AIDTRE programs currently exists on Permanent File 4409 at the UT Computational Center. The AID4UT program processes the data set and determines the binary splits that best explain the variability in the data. On option, it prints out the results for each iteration and prints or punches a residuals list corresponding to the final groups. On option, it produces tape files for use by the AIDTRE program in graphically displaying the skeleton and detailed AID-Trees. The AIDTRE program operates in tandem with the AID4UT program.

# Input Data:

The input for the AID4UT/AIDTRE programs consists of system control cards, program control cards and source data (either on cards or on tape). An example of a deck setup is shown in Figure C-1.

In the itemized description of the input data, a symbol of "R", "O", or "*", is given by each field. These symbols mean:

- $\mathbf{R} = \mathbf{field} \mathbf{required}$
- O = field optional
- * = exception described in field description.

# ITEMIZED INPUT DESCRIPTION

# AID-4 TITLE CARD

Card	
Columns	

# Description

- 1
- R The card type, punched equal to 1, which identifies the title card and initiates each AID-4 run. A title card is required at the beginning of each run, and if not, a terminal error will occur.

AID-4 runs can be "stacked," i.e., more than one run made within the same job, only under the following conditions:

- a. Each run in the stack does not require an "unaccessible file" such as a tape file that is not currently mounted on a tape drive or such as a tape file that cannot be correctly positioned with the rewind option.
- b. The runs do not contain conflicting options such as an IRUN equal to 3.
- 2-49 O The title which is printed at the beginning of the output under "Control Card Parameters." Forty-eight alphanumeric characters are provided for the user to identify a particular run.
- 50

. . . . . . R IRUN, the run option, which can be summarized as the following:

Value of IRUN

0

1

### Meaning

- Use every case in the original input file for a normal AID-4 run; i.e., no forced splitting, provide BSS/TSS profiles.
- Select a random sample A from the original input file, and use only the cases that belong to A for a normal AID-4 run. Then force those cases that belong to B to make the same splits as taken by A; i.e., single cross-validation.
Select a random sample A and a random sample B from the original input file, and use only the cases that belong to A for a normal AID-4 run. Then force those cases that belong to B to make the same splits as taken by A. Then use only the cases that belong to B for a normal AID-4 run, and force those cases that belong to A to make the same splits as taken by B; i. e., double cross-validation.

Given sample A and sample B, use only the cases that belong to A for a normal AID run. Then force those cases that belong to B to make the same splits as taken by A. Note that double cross-validation can be accomplished by submitting a second job with A and B switched.

## 51-56 R* NCPERM, the number of cases in the original input file.

3

4

Ì

ÿ,

 $\sim \sqrt{2}$ 

ij,

,

5 19 19

. ⊥×ku

ð.,

5

民志

د .. ک ق NCPERM is optional if IRUN equals zero and the input file is a tape file ending with an end-of-file mark; i.e., NCPERM can be left blank, in which case all records in the input file will be processed until an end-of-file condition is reached.

NCPERM is also optional if IRUN equals zero and the input file is a card file providing an "end-of-file card" is placed behind the last data card.

NCPERM is not applicable if IRUN equals four.

57-58 0 NREELS, the number of reels for NCPERM.

K NREELS equals blanks or zeros, NREELS is assumed to be one.

If IRUN = 4, NREELS for sample A and B must be equal.

# 59-64 R* NSAMA, the number of cases to be randomly selected in sample A.

, h

1 Con

. مەر

÷.,

à

57.8¥

.

ំង

ų

NSAMA is optional if IRUN equals four; i.e., NSAMA can be left blank in which case all cases in the sample A will be processed until an end-of-file condition is reached.

NSAMA is not applicable if IRUN equals zero.

65-70 R* NSAMB, the number of cases to be randomly selected in sample B.

NSAMB is optional if IRUN equals four; i.e., NSAMB can be left blank in which case all cases in the sample B will be processed until an end-of-file condition is reached.

NSAMB is not applicable if IRUN equils zero or one.

71-78 R* NUM, an eight digit starting random number to be used in the random selection process when IR UN equals one, two, or three. NUM should not be a multiple of 17 or 5882353.

NUM is not applicable if IRUN equals zero or four.

79-80 R IFMT, the number of variable format cards to be read in next. The range of IFMT is  $1 \le \text{IFMT} \le 4$ .

307

#### AIR-4 FORMAT CARD OR CARDS

## Card Columns

## Description

 1-78 R The variable format beginning with a left parenthesis and ending with a right parenthesis specifies the integer type (I) conversion of the numerical data for the input file or files. Only integer type (I) fields can be defined within the parentheses, with X conversion to skip characters and the slash (/) to indicate the beginning of new records.

> The maximum number of input variables to be read in (as specified by NV in the description card) is 83 (80 fields for predictors, one field for the criterion, one field for identification, and one field for weights). Each predictor field read in, however, need not be used. Therefore, for certain sequences of runs, it may be advantageous to use the same format for all runs and control predictor selection with the predictor cards only.

> The only permissible characters in the input files to be read by the integer (I) type format are  $\emptyset - 9$ , +, -, and blank. Otherwise, a FORTRAN terminal error will occur.

The number of format cards expected by AID-4 depends on the value of IFMT in the title card. At least one format card is required for each run with a maximum of no more than four.

For IRUN equal to zero through three, the variable format card (or cards) specifies the arrangement of the data in the original input file.

For IRUN equal to four, the variable format card (or cards) specifies both the given sample A and the given sample B. Therefore, the arrangement of sample A and sample B must be identical.

79-80 Not used.

ۍ,

1820

1.1

#### AID-4 DESCRIPTION CARD

Card	
Columns	Description

5

....

5

1

4

8

ante Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo Santo S

.

16 Y

1

- R The card type, punched equal to 3, which identifies the description card. A description card is required for each AID-4 run.
- 2-6 R P1, the percentage of the original total sum of squares that must be contained in any given group if that group is to become a candidate group for splitting. That is, the total sum of squares in any given group must be greater than or equal to P1 percent of the original total sum of squares for that group to become a candidate group. Five decimal places are assumed (read with an F5.5 format). For example, to specify one percent, the user would punch 01000 in columns 2-6. Note that the decimal point is not actually punched in the card. The range of P1 is .00001 ≤ P1 < 1. P1 punched equal to 00001 essentially deactivates this stopping criterion. (Cannot be left blank.)
- 7-11 R P2, the best split of a given candidate group must reduce the original within (unexplained, error) sum of squares by P2 percentage of the original total sum of squares if that group is to be split. That is, the largest between sum of squares obtained after the scanning of all predictors of a given candidate group must be equal to or greater than P2 percentage of the original total sum of squares for that candidate group to be split. Furthermore, once a candidate group has filled the F2 requirement, that group will not be considered for splitting again, even though the group meets the P requirement. The range of P2 is such that P2 ≤ P1. Ordinarily P2 is set equal to P1. (Cannot be left blank.)

Note that Pl and P2 define an upper bound on the number of possible splits.

12-16 R MAXGP, the maximum number of final groups (includes candidate groups and groups that cannot be split), regardless of the values of Pl and P2, that can be obtained before the algorithm will terminate.

Contract of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second sta

If forced splitting is called for, i.e., IRUN greater than one, the maximum value for MAXGP is 90.

Ordinarily MAXGP is set high enough so that the stopping criterion is determined by Pl, P2, and NMIN.

17-21 R NMIN, the minimum number of cases required to be in both resultant groups for a candidate group to be split. NMIN must be  $\geq 2$  or a fatal error will result.

्र भूर -

... ...

£

a,

Δ.

₽₿

2. . . **. .** 15

": <u>'</u> ;

_, 🎘

22-26 R* KSTOP, the number of iterations to be printed in detail, i.e., the details of the scanning of all predictors for the candidate groups. After KSTOP is exceeded, only summary information is printed.

> If KSTOP is left blank, KSTOP will be set to zero and only summary information will be printed.

If KSTOP equals -1, only a "Split Summary" is printed. (Minimum output)

- 27-29 R* NV, the number of variables to be read in from the input file. This number includes the predictor variables, criterion variable, the identification variable (if applicable), and the weight variable (if applicable). <u>NV is required, must be less</u> than or equal to 83, and must agree with the format as specified in the format cards.
- 30-32 R* NR, the field number of the case identification variable as determined from the format card or cards.

If residuals are called for (IRESID greater than zero), <u>NR</u> must be specified ( $0 < NR \le 83$ ). Otherwise, NR can be left blank.

33 0 KRW, the rewind option which determines whether or not to rewind the original input file (a tape file) before the processing for a given run begins.

> If KRW is equal to a zero or a blank, the original input file is not to be rewound.

If KRW is equal to one, the file is to be rewound.

If the original input file is a card file (ICARD equal to one), KRW is not applicable.

If IOPT is equal to a blank or a zero, the cases with out-ofrange predictor values <u>are not excluded</u>. Predictor values less than the specified minimum values are assigned recode category 00 and predictor values greater than the specified maximum values are assigned the largest recode categories.

If IOPT is equal to one, the cases with out-of-range predictor values are <u>excluded</u>. The excluded cases are printed in the "Reject Summary" if the summary is called for; i.e., KREJ equal to one (see KREJ description).

IOPT is extremely important in that IOPT determines the format of the predictor cards (see description of AID-4 predictor cards).

0 KREJ, the reject summary print option which determines whether or not the "Reject Summary" is printed. The summary lists all those out-of-range cases that are excluded because of invalid predictor values or invalid criterion values.

If KREJ is a zero or a blank, the summary is not printed.

If KREJ is one, the summary is printed.

36

35

34

£

.e.#

с . _В

" **(**"

-25

> ری رو بر هم

(100)

0 IRESID, the residual summary option which determines whether or not residuals (the criterion values minus the final group mean) are computed.

If IRESID is a zero or a blank, no residuals are computed.

If IRESID is one, the residuals are printed in the "Residual and/or Reject Summary." This summary is printed each time a final group is obtained; i.e., a group that cannot be split any more. Approximately 200 residuals are printed per page. If IRESID is two, the residuals are punched into cards. Also IRUN must be less than two.

If IRESID is three, the residuals are written on Tape 8. Also IRUN must be less than two. With this option Tape 8 can be rewound and printed at job completion to give a concise list of residuals.

37 0 ICARD, the card input option which determines whether or not the original input file is cards.

5

4

• •

۰. ۲

**,**,

9. 4 S

d y

_i1

ъ

. .

8

v.;;

If ICARD is zero or blank, the original input file is presumed to have been prepositioned on Tape 25.

If ICARD is one, the file is cards. However, IRUN must be less than three.

38 0 ITREE, the plot option which determines whether or not the normal splitting phases (1 and 3) are plotted. The forced splitting phases (2 and 4) cannot be plotted. If ITREE ≥ one, Tape 10 and Tape 14 are generated for use by AIDTRE.

If ITREE is zero or blank, no tapes are generated it ITREE is zero or blank, no tapes are generated.

If ITREE is one, only the detailed / ID-TREE will be generated.

If ITREE is two, detailed and skeleton AID-TREES will be generated.

39 0 NOGO, the control card checking option which determines whether or not the run terminates after the printing of all control cards.

If NOGO is a zero or blank normal processing is assumed.

If NOGO is one, the run will terminate after the printing of the control card summary.

NOGO runs do not require any files and can be stacked with other runs.

40 0 IESS, the error sum of squares option which determines whether or not the error sum of squares for forced splits is printed.

an' famin

If IESS is a zero or a blank, the error sum of squares is  $n \circ f$  printed.

If IESS is a cne, the error sum of squares is printed.

41-80 Not used.

. . . . .

4

) **(** 19. 1 19. 1 19. 1 19. 1

े **ह** 

-12 A

## AID-4 PREDICTOR CARDS

## Card Column

## Description

- R The card type, punched equal to 4, which identifies the predictor card. At least one predictor card is required for each predictor (specified categories can require more than one card).
- 2-19 R The name of the predictor which is printed at the beginning of the output under "Control Card Parameters" and each time a split is made on the predictor named. Eighteen alphanumeric characters are provided for the name.
- 20-22 R The field number of the predictor as specified by the format card or cards.
- 23 R KBL1, the first predictor specification variable, which determines the type of the predictor.

If KBL1 is zero, the predictor is <u>free-floating</u>; i.e., the categories within the predictor are not sorted and remain in original recode order when the scanning process begins.

If a predictor is free-floating, the between sum of squares obtained by the scanning process will be maximized over all possible splits for that predictor. If a predictor is monotonic, the between sum of squares obtained will be maximized only over the monotonic splits for that predictor. This does not, however, imply that all predictors should be typed as free-floating, for certain predictors make little sense as such. The type is dependent upon the nature of the predictor.

R KBL2, the second predictor specification variable, which determines how the predictor is to be defined. The values of the predictor (as read by the variable format) must be positive or negative integers with a range of -99999 to 999999. AID-4 recodes the values of the predictor into categories numbered 00, 01, ..., 39. The maximum recode value is 39; i.e., the total number of recode values for any one predictor cannot exceed 40. Two methods of defining recodes are provided.

If KBL2 is zero, the predictor is defined to have <u>equal intervals</u>: i.e., the recode categories are determined from a minimum value; a maximum value, and an interval length. See description of columns 25-80.

If KBL2 is one, the predictor is defined to have <u>specified</u> categories; i.e., the categories are determined from specified input and recode values. See description of columns 25-80.

25-80 The format of columns 25-80 is dependent upon KBL2 and the exclude option, IOPT (column 34 of the description card), and can be summarized as the following four possibilities:

11.

ţ,

I. KBL2 equal to zero and IOPT equal to a blank or zero; i.e., equal intervals and cases with out-of-range predictor values not excluded.

- I 25-30 R The minimum value, which determines the predictor values to be included in the first category; i.e., category 00. Any predictor value less than or equal to the minimum value will be included in category 00. Also any predictor value less than one interval greater than the minimum value will be included in category 00.
- I 31-36 R The maximum value, which determines the predictor values to be included in the last category. Any predictor value greater than or equal to the maximum value will be included in the last category.
- I 37-42 R The interval length, which together with the minimum and maximum values, determines the number of recode categories. The number of categories is computed by

 $((\max-\min)/(\operatorname{interval}) + 1,$ 

which must be less than 40. Note that remainders after division are truncated.

For predictor values greater than the minimum value and less than the maximum value, the recode categories are computed by

(predictor value - min) interval

Note that remainders are truncated.

I 43-45 R Specified recode category 1, the value of which is assigned to any predictor value equal to specified input value 1 (columns I 46-5i). Provision is made for reassigning up to three values of the predictor (using specified recode categories 1, 2, and 3). Note, however, that the specified input values 1, 2, and 3 should be greater than or equal to the minimum value and less than or equal to the maximum value, as the range check for minimum and maximum values is performed first. If a predictor value is less than the minimum, the value is set equal to the minimum. If a predictor value is greater than the maximum, the value is set equal to the maximum.

If the recode is not used, a -1 must be punched in columns I 44-45.

I 46-51 R Specified input value 1.

á.

8

5

**4**5

I 52-54 R Specified recode category 2, the value of which is assigned to any predictor value equal to specified input value 2 (columns I 55-60).

If the recode is not used, a -1 must be punched in columns I 53-54.

- I 55-60 R Specified input value 2.
- I 61-63 R Specified recode category 3, the value of which is assigned to any predictor value equal to specified input value 3 (columns I 64-69).

If the recode value is not used, a - 1 must be punched in columns I 62-63.

I 64-69 R Specified input value 3.

70-80 Not used.

**II.** KBL2 equal to zero and IOPT equal to one; i.e., equal intervals and cases with out-of-range predictor values excluded.

- II 25-30 R The minimum value by which any predictor value less than the minimum value is excluded.
- II 31-36 R The maximum value by which any predictor value equal to or greater than the maximum is excluded.
- II 37-42 R The interval length, which together wit the minimum and maximum values, determines the number of recode categories. The number of categories is computed by

((max-min)/interval),

which must be less than 40. Note that remainders after division are truncated.

For predictor values greater than or equal to the minimum value and less than the maximum values the recode categories are computed by

> (predictor value - min) interval

Note that remainders are truncated.

II 43-45 R Specified recode category 1, the value of which is assigned to any predictor value equal to specified input value 1 (columns II 46-51). Provision is made for reassigning up to three values of the predictor (using specified recode categories 1, 2, and 3). Note, however, that the specified input values 1, 2, and 3 should be greater than or equal to the minimum value and less than the maximum value, as the range check for minimum and maximum values is performed first. If a predictor value is less than the minimum or greater than or equal to the maximum, the case is excluded.

If the recode is not used, a -1 must be punched in columns 44-45.

- II 46-51 R Specified input value 1. Note that this value cannot be greater than or equal to the maximum, or less than the minimum.
- II 52-54 R Specified recode category 2, the value of which is assigned to any predictor value equal to specified input value 2 (columns II 55-60).

If the recode is not used, a -1 must be punched in columns II 53-54.

- II 55-60 R Specified input value 2. Note that the value cannot be greater than or equal to the maximum, or less than the minimum.
- II 61-63 R Specified recode category 3, the value of which is assigned to any predictor value equal to specified input value 3 (columns II 64-69).

If the recode value is not used, a -1 must be punched in columns II 62-63.

- II 64-69 R Specified input value 3. Note that the value cannot be greater than or equal to the maximum, or less than the minimum.
- II 70-80 R Not used.

**a**#i

30°.

. .

***** *'755'

ິ 🙀 🕹

III. KBL2 equal to one and IOPT equal to a blank or zero, i.e., specified categories and cases with out-of-range predictor values not excluded.

- III 25-27 R Specified recode category 1, the value of which is assigned to any predictor value equal to or less than specified input value 1 (columns III 28-33).
- III 28-33 R Specified input value 1.
- III 34-36 R Specified recode category 2, the value of which is assigned to any predictor value greater than specified input value 1 and less than or equal to specified input value 2 (columns III 37-42).

III 37-42 R Specified input value 2.

- III 43-45 R Specified recode category 3, the value of which is assigned to any predictor value greater than specified input value 2 and less than or equal to specified input value 3 (columns III 46-51).
- III 46-51 R Specified input value 3.
- III 52-54 R Specified recode category 4, the value of which is assigned to any predictor value greater than specified input value 3 and less than or equal to specified input value 4 (columns III 55-60).
- III 55-60 Specified input value 4.
- III 61-63 R Specified recode category 5, the value of which is assigned to any predictor value greater than specified input value 4 and less than or equal to specified input value 5 (columns III 64-69).
- III 64-69 R Specified input value 5.
- III 70-80 Not used.

÷.

If more than five categories are required, additional predictor cards can be used by duplicating columns III 1-24 and continuing in columns III 25-69 respectively. As many cards per predictor as necessary can be used as long as the maximum of 40 recode categories is not exceeded; i.e., 00-39.

The last specified input value must always equal 999999 and any predictor value greater than the previous specified input value is assigned the last specified recode category. For example, if specified input value 4 equaled 999999, any predictor value greater than input value 3 would be assigned recode category 4.

IV. KBL2 equal to one and IOPT equal to one; i.e., specified categories and cases with out-of-range predictor values excluded.

IV 25-27 R Blank (except for continuation cards; see next to last paragraph below.)

- IV 28-33 R The lower bound by which any predictor value less than or equal to the lower bound is excluded.
- IV 34-36 R Specified recode category 2, the value of which is assigned to any predictor value greater than the lower bound and less than or equal to specified input value 2 (columns IV 37-42).
- IV 37-42 R Specified input value 2.

11 N

ð.

': A

ې د د ا

- IV 43-45 R Specified recode category 3, the value of which is assigned to any predictor value greater than specified input value 2 and less than or equal to specified input value 3 (columns IV 46-51).
- IV 46-51 R Specified input value 3.
- IV 52-54 R Specified recode category 4, the value of which is assigned to any predictor value greater than specified input value 3 and less than or equal to specified input value 4 (columns IV 55-60).
- IV 55-60 R Specified input value 4.
- IV 61-63 R Specified recode category 5, the value of which is assigned to any predictor value greater than specified input value 3 and less than or equal to specified input value 5 (columns IV 64-69).
- IV 64-69 R Specified input value 5.
- IV 70-80 Not used.

If more than three categories are required, additional predictor cards can be used by duplicating columns IV 1-24 and continuing in columns IV 25-69 respectively. On second and following continuation cards, start next specified recode category in columns IV 25-27. As many cards per predictor as necessary can be used as long as the maximum of 40 recode categories is not exceeded; i.e., 00-39. The last specified input value must always equal 999999 and the previous input value acts as the upper bound by which any predictor value greater than the upper bound is excluded. The last specified recode category (before the 999999's) must be left blank.

And the second states and the second states and

## AID-4 CRITERION CARD

i. V

н мет., р

**9**#

÷ •

, **1** 

j t

,DA

Card Column	3	Description
1	R	The card type, punched equal to 5, which identifies the criterion card. One criterion card is required for each run.
2-19	R	The name of the criterion which is printed at the beginning of the output under "Control Card Parameters " Eighteen alphanumeric characters are provided for the name.
20-22	R	The field number of the criterion as specified by the format card or cards.
23-24	0	The field number of the weight as specified by the format card or cards.
		If this field is left blank, the weight is set to one.
25-30	R	YMAX, the maximum allowable value of the criterion variable. If the value of the criterion is greater than YMAX, the case is excluded. The maximum value that can be specified for YMAX is 999999.
31-36	R	MD1, the first deletion value, by which any criterion value equal to MD1 will be excluded. <u>MD1 can be deactivated</u> by setting MD1 greater than YMAX.
37-42	R	MD2, the second deletion value, by which any criterion value equal to MD2 will be excluded. <u>MD2 can be deacti-</u> <u>vated by setting MD2 greater than YMAX.</u>
49-80		Not used.

i

ł.

### AID-4 END OF JOB CARD

Card	
Columns	Description

1 R The card type, punched equal to 9, which identified the end-of-job card and terminates the job.

The purpose of the end-cf-job card is to indicate the end of a run executed within a given job. The card should be thought of strictly as a control card such as the title card, format card, etc., and not as part of the original input file.

2-80 Not used.

1997 - 1997 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 - 1995 -

, * , ***

્રક્રો

÷

٢

**AP** 

 $\sim$ 

بة ب

, *** 

di.

A

## AID-4 LIMITATIONS

For the current version of AID-4, the following list of limitations apply:

1. The maximum number of input variables to be read in is 83 (80 fields for predictors, one field for the criterion, one field for identification, and one field for weights).

2. The maximum number of predictor variables is 80.

3. The maximum number of generated categories for all predictor variables taken together must not exceed 700. The total number of categories for all predictors,  $N_{C}$ , can be computed by the following formula:

 $N_{C} = 4N_{E} + 2N_{SR} + 2N_{SC}$ 

- Li.

**37**2

where  $N_E$  is the number of predictor variables with equal intervals,  $N_{SR}$  is the number of specified recode categories for predictors with equal intervals, and  $N_{SC}$  is the number of specified recode categories for predictors not with equal categories.

4. The maximum number of criterion variables for any one analysis is one.

5. The maximum number of candidate groups at any given iteration is 90.

6. The maximum number of cards for the variable input format is four.

7. The maximum number of final groups, including the candidate groups and the groups that cannot be split, is 99999, the maximum value of MAXGP. Note that the number of final groups is also dependent upon P1, P2, and NMIN.

8. The range of the recode categories is 0 to 39.

9. The range of the predictor values is -99999 to 999999.

10. The range of the criterion values is -99999 to 999999.

11. The range of the weight values is determined by the variable format.

12. The minimum value for NMIN is 2.

**#**;

342

. ni

and the

а. , **А**. , ( А. , ( А. , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А.) , ( А

1

. 2**11**4. . . . . . . . . . . . .

5 . A

.....

.....

## AID-4 LIST OF TERMINAL ERRORS

,**#** ,

ă.

,

đ,

. .

್ಧ≵

. t j

n

3

\$

The second second second second second second second second second second second second second second second s

Error Number	Description
1	AID-4 title card (card type 1) expected.
2	IRUN out-of-range (greater than 4).
3	IRUN equal to 1, 2, or 3 and NCPERM'equal to zero.
4	IRUN equal to 1, 2, or 3 and NSAMA equal to zero.
5	IRUN equal to 2 or 3 and NSAMB equal to zero.
6	IRUN equal to 1, 2, or 3 and NUM equal to zero.
7	IRUN equal to 1, 2, or 3 and NUM is a multiple of 17.
8	IRUN equal to 1, 2, or 3 and NUM is a multiple of 5882353.
9	IFMT out-of-range (equal to zero or greater than 4).
10	AID-4 description card (card type 3) expected.
11	Pl out-of-range (greater than 1).
12	P2 out-of-range (greater than 1).
13	MAXGP out-of-range (less than 1).
14	KRW out-of-range (greater than 1).
15	IOPT out-of-range (greater than 1).
16	KREJ out-of range (greater than 1).
17	IRESID out-of-range (greater than 3).
18	IRUN greater than 1 and IRESID greater than 1.
19	The field number of a predictor card is greater than NV.
20	The field number of a predictor card is less than 1.
21-24	Either KBL1 or KBL2 out-of-range (not equal to zero or 1).
25	Interval length out-of-range (less than 0).
26	Number of categories over all predictors greater than 700.
27	Number of categories over all predictors greater than 699.
28	AID-4 predictor card (card type 4) expected.
29	Continuation card incorrect.
30	Number of categories greater than 40.
31	Total number of categories greater than 700.
32	AID-4 criterion card (card type 5) expected.
33	The field number of the criterion card is greater than NV.
34	The field number of the criterion card is less than 1.
35	The weight field in the criterion card is greater than NV.
36	The weight field in the criterion card is negative.

and a substantial for the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the second state of the

- 37 YMAX is less than or equal to YMIN.
- 38 NV is greater than 83.
- 39 Not used.

 ${}^{4}t$ 

े ह

3

8

Æ

'n

ι «

الم معدد و المع المع

26.00

- 40 Interval length is less than zero.
- 41 ICARD greater than 1.
- 42 IRUN greater than 2 and ICARD equal to 1.

COMPANY IN COMPANY

## An Example Problem:

Figure C-1 shows the input deck for a problem with 12 data points. The first four cards are the system control cards required to execute AID4UT/AIDTRE. There are three predictor variables,  $(X_1, X_2, X_3)$ . Figure C-2 shows the standard output set produced by all AID4UT runs. Figure C-3 shows the detailed output for the first iteration of the splitting process. This detailed output is optional and controled by the input parameter, KSTOP.

Figure C-4 shows the residuals table for groups 8 and 9 and the summary information available at the end of the splitting process. It also shows the BSS/TSST and BSS/TSS(i) profiles and the corresponding ranking of the predictor variables based on the PEP coefficient (RECSUM). This latter coefficient was proposed by Finifter and represents the average explanatory power of a predictor over all groups in the AID-Tree. This output is only available with IRUN  $\leq 1$ .

Figure C-5 presents the skeleton version of the AID-Tree that is produced when ITREE = 2. Figure C-6 presents the detailed AID-Tree that is produced when ITREE  $\leq 2$ . If an AID-Tree is not desired for a run, removal of the EXECPF(4409, AIDTRE) card will eliminate the tree but will not affect the output shown in Figures C-2 through C-4.

#### Setting of Parameters in AID4UT/AIDTRE:

In initial exploratory analysis runs on a data set, the following parameter settings are recommended:

> IRUN = 0, P1 = .0001, P2 = .0001, MAXGP  $\simeq$  100, NMIN = 2, KSTOP = 5, IRESID = 0, ITREE = 2.

These settings allow maximum splitting to occur and detection of any coding errors and refinement of predictor categories (if these categories were arbitrarily defined originally). These parameters do allow considerable overfitting to occur if the data set is small and further runs with restricted parameter values will generally be required.



r.

4

*

•

. .

. . .

۱ ۴

3° ₆,

•.. • •*≁‡

**#** 

l inens



Ś

Nor of charge

UNIVENSIT OF TEARS VENSION OF ALUA AS ACAPTED FROM APMAL,LACKLEND AFM Mouiffed for cucobod at capt LL wunch. DCT 72

L YB

i e j

CNTHOL CARD PAHANETFRS

TITLE CARD

1**. 1** 1**. 1** 0 ALIN d 45 ×8 IRIIM NCP TRM NREELS NSANA A 17 0 0 AIDAUT LENDNSTHATION RUN FORMAT CARD OR CARDS CARD 17PE

[\$[*]]]

DESCRIPTION CARD

ctise: 570, DFV. 144,70535 MH KRY 1001 KREJ RESIN ICARO ITREE NOGO IESS 18501 1 0 0 1 1 1 2 0 7 MEAN 142.50000 2 2.5127500 **WEIGHT IN FILLO NO.** 1 5 5 251275.00 NANT BHINE THAN MONDIONIC MOND TONIC DIND [ CHIC TYPE 504 7-5004RE 566150,00 501. CCRRESFONUS TO ~ N N V LESS THAN 2 TO 3 TO 4 OD UVER YFAX. IS 500...YMIN. IS -50...FXCLUDE 541 LESS THAN Z OR UVEN LESS THAN 2 OR UVEN DEPENDENT VARIABLE-V IN FIELD NO. 5 15 HESP (Y) DD NOT SPLIT UNLESS ISS MORE THAN 2.5127500 12 Valit Cases. Group J Values Areş v 504 0F Y 1954,0000 NHIN KSTUP FIELD NO. RECOCE 2 4 3 -P2 MAKGP .00001 12 TOTAL WEIGHT [2,000000 m ~ 2 14 .0000. VANJABLE 2 ۲ Ş GROUP L CARD TYPE

----

~

m

FIGURE C-2

330

1       6       κ.0000000       C00.0000       C420.000       170404.33       49,464165       170404.33       49,464165       170404.33       49,464165       170404.33       49,464165       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,664       104,6		1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ICTON 1 AL MEIGHT	SUH OF Y	SUR T-SQUAKE	MEAN	- A30 - DEV -	5 5 10	5
<ul> <li>A5. 455 IT04CR.33 455/155 =1417 LETVER CODES 1</li> <li>TYY ON PERFICION 2 K2</li> <li>COUE N TOTAL WEIGHT 655,00000 JU075,00 214,1667</li> <li>COUE N TOTAL WEIGHT 655,00000 JU075,00 214,16667</li> <li>A5. 3203.333 45.5</li> <li>A5. 4000.000 45.5</li> <li>A5. 4000.000 45.5</li> <li>A5. 4000.000 45.5</li> <li>A5. 4000.000 45.5</li> <li>A5.55000 15.5</li> </ul>				0000""""""""""""""""""""""""""""""""""	201°10247478	2H1.66667 Heefseffefefe 13.33333	164.52790 ++***********************************	170408, 33	H9460, h61
COLE       NU       YSSUARE       LU       YSSUARE       LZT.65322       DEV.       R S S         L       6       0000000       L255.0000       L/0575.003       L/0575.003       L27.66332       DEV.       R S S         L       6       0000000       L255.0000       L/0575.003       L27.66333       L27.66332       L33.333       L27.66335       L33.333       L27.444       L24.444       L24.1456.00       L24.1475.0       L24.1475.0       L25.0000       L24.1475.0       L24.1475.0       L24.1475.0       L25.0000       L24.1475.0       L24.1475.0       L25.0000       L24.1475.0       L24.1475.0       L24.1475.0       L24.1475.0       L24.1475.0       L04.1675.0       L24.1475.0       L24.1475.0       L24.1475.0       L24.1475.0       L24.1475.0       L41.475.0       L44.1475.0       L44.1475.0       L44.1475.0       L44.1475.0       L44.1475.0       L44.1675.0	AA. B	issa A Barar	170408.33	B\$\$/T\$\$ .	ATULT VETVEEN CODE! And codes	0			
I & & 9000000 1265.0000 JY515.00 ZI, 16667 1.2.0135 J2014.314 ZIV2014 INY ON FREDICTOR 1 k3 COE N TOTAL EIGHT SUM TSQUARE DODES 1 TOTAL EIGHT SUM PSQUARE 112,50000 6.3,234.45 1500.000 236275.0 2 4 0.0000000 699.00000 cu155.000 143.54.45 15000.000 236275.0 2 4 0.0000000 699.00000 cu155.000 143.54.45 96.00.0000 241.75.0 3 4 0.0000000 697.0 00000 262.50000 143.54.47 96.00.0000 241.75.0 3 4 0.0000000 85775 .05970 8672.50000 143.54.45 96.00.0000 241.75.0 3 4 0.0000000 85775 .05970 8672.50000 143.54.47 96.00.0000 241.75.0			TOTAL WEIGHT	5UH QF Y 665,00000	504 Y-SGUARE 1/0575.00	HEAR 110,01333	127.66352 DEV.	5	
<pre>AX. 455= 32031.333 855/TSS = .12746 EETWEEN CDDES 1 INY ON FREDICTOR 3 X3 CODE N TOTAL BETGHT SUM OF Y SULME THEAA 51.23445 UEV. B 5 5 CODE N TOTAL BETGHT SUM OF Y SULME 112.50000 b12.33445 UEV. B 5 5 CODE N 4.0000000 450.00000 culm50.000 1172.50000 143.23445 UEV. B 5 5 Code N 4.0000000 450.00000 culm50.00 275.0 C 4 4.0000000 610.00000 culm50.00 262.50000 143.54475 UEV. B 5 5 C 4 4.0000000 610.00000 culm50.00 262.50000 143.5445 UEV. B 5 5 C 4 4.0000000 610.00000 culm50.00 262.50000 143.54475 UEV. B 5 5 C 4 4.0000000 610.00000 culm50.00 262.50000 143.54475 UEV. B 5 5 C 4 4.0000000 610.00000 culm50.00 262.50000 143.54475 UEV. B 5 5 C 4 4.0000000 810.00000 culm50.00 262.50000 144.445 C 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5</pre>			5.000000	1265.0000	397575°AD	214.14467	eseecesseeeeee 142.81155	EEE*F542F	19.102412
CODE       w       TOTAL METOR       SUM DF Y       SUM DF	(AX. 8	55 <b>-</b>	52033,333	955/155 a	.12746 BETWEEN CODE! And Codes	<b>6</b>			
Z 4 4.0000000 699.00000 236275.0 236275.0 236275.0 236275.0 236275.0 236275.0 236275.0 236275.0 24.000000 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.00000 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.075.0 24.07			CIUM 3 A3 TOTAL WEIGHT 4.0000000	50,0000 PF Y	2UM Y+SGUAWE 66550.000	ME 43 112_50006	510. DEV. 63.234445	л 2	10 10
3 4 4.0000000 610.0000 2410.0000 241.75.0 4.1. 455 15000.000 455/155 <b>6597U 8574EEK COUES 0</b> 4.0 CODE <b>S 2</b> 3	• •			**************************************		174,50000		15000.000	236275 <b>.0</b> 0
4ÅÅ. 855# 15000.000 855/155 <b>0597U 8ETVEEN CODES 0</b> Ånd Cod <b>es 2</b> 3	8 17) 9 9			10+00000		252+50000	######################################	9610.0000	241675.00
	4 <b>7</b> . 5	ESS:	15000,000	ess/15s	05970 8574ER COURT				
PELIT GROUP & ON PREVICTOR & AI ALL INTO GROUP & WIYE COES 0	96698 PL11	40049	1 ON PREDICTOR	,		24400440040000 2 MITH CODES 0			
435 # 170408433 855/755 # 487437 7-4415 4459		859	= 170408.33	855/155 =	.eltil T-VALIJE	4.59			

....

0 DF2 υ£ i ---F-ANOVA 21,0728 10 UF 2 uf l 21.072⁸ 1 F-RSO **.**82351 æ 47417464 R-SQUARED TOTAL #SS 80866.667 TOTAL ESS 170404.33 TOTAL ISS 251275.00 N MCF

CANUJUATE GROUPS ARE AS FULLONS.

5TD. DFV. 104.62790 49.888765 MEAN 201.56667 43.33333 1 5 5 65933,333 14933,333 SUM Y-SQUARE 3+1450_03 6+240,000 SUM OF Y 1690.0835 265.05839 TOTAL WEIGHT 6.0090000 6.0000000 2 ** 6ROUP 2 3

FIGURE C-3

¥ ¢

6	4 63 •599679 •619312 •223799 • *5597	45854 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44625 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44655 44555 44555 44555 44555 44555 44555 44555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 45555 455555 455555 455555 455555 455555 455555 4555555			
15 uțtk <b>0</b> 55/155	2 42 -1276\$3 -1276\$3 -1716\$\$	2 ( 4 4 5 . 2 4 4 4 5 . 6 4 4 1 6 . 5 2 5 2 6 4 . 5 2 5 2 6 4 . 7 3 2 6 4 8 . 7 3 2 6 4 8 . 7 3 2 6 4 8 . 7 3 2 6 4 8 . 7 3 5 6 4 8 . 7 3 5 6 6 8 . 7 3 5 6 6 8 . 7 4 6 6 8 . 7 5 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	1.251275.6 8.1 8.2 8.2 8.2 8.2 8.2	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	
्रित २ <del>०</del> २	In ki + \$780-75 8 - 3880-75 8 - 880-85 8 - 880-85		ue∩ CR1f, e .010155 10 ¥Eax <b>h 8</b> mcu95 1.02201 <b>1</b> 1.02213 4	УЧЮ СМІГ, а "№10751 0 АААМ № 6-20065 1.643955 1	0
	TKLAL/AKP 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	9.0450 1214.5 1214.5 1214.5 1214.5 1214.5 1214. 1214. 1214. 1214. 1214.	1.093711 AC 1.093711 AC 8.00005 24 9.00005 24	1.912547 Rg 1.912547 Rg 1.9124055 Pc1 94 1.98005 151	IC 48968.
	64946 65649 61102915 241116 91116	20205- 0 201010 201010 201010 20102- 20102- 20102-	• 64140 MEAN • • 66140 MEAN • • 10028 • 96850 • 101146 • 101146	- 25160 - 26611 - 212 - 20057 - 21200 - 21200 - 21200 - 21200 - 21200	FIGURE C-4
125/1521 M11	K2 127463 (876476 (82757 (82757	5412E 8 8 11985 11985 11985 11985	<pre>\$ \$usgacues 1\$ \$ \$ \$usgacues 1\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$</pre>	5 946680495 15 1 946680495 15 1 846657846716 1 2 4465571 1 2 46639	
N SİSATENY	K1 105425 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105455 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 105555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 1055555 10555555 1055555 1055555 1055555 10555555 10555555 10555555 10555555 10555555 10555555 10555555 10555555 10555555 10555555 10555555 105555555 105555555 1055555555	L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L 4 L 10 L	N Ca (231457514) 1976 1976 1976 1986 1986 1986 1986 1987 1987 1987 1987 1987 1987 1987 1987	N 04 63136 940 44 205 14 205 205 14 205 2040 2042 2040 2040	
	_ ****			TABLE 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	7
	18.84/14 20 20 20 20 20 20 20 20 20 20 20 20 20	REALE CF CAR CF CAR CF CAR CF CAR CF CAR		20 20 20 20 20 20 20 20 20 20 20 20 20 2	•

RESIDUAL ANCION NE JECT SUMMARY

5 64009 IU

F-ANDVA DF1 •

ÿ

-----٠ 41E8" i + 41910. 8-3843420 ,959837e5 TGTAL **855** 19866,667 TGTAL 655 241288.33 TGTAL TSS 251275.00 J.

* 642

CMD OF AIDAUT - DATA FOR TREE ON TAPES IN ~ 14

. . . . . . .

. .V

j,

.

đ.

' . .

ć. ..

2,3

.

. آھر بھت

. .,

332

TREE SIMUCTURE OF FINST EIGHT LEVELS, GHOUP NUMBEN IS GIVEN IN CENTER OF THE BOX, THE PAIR MEMBEN WITH THE MIGHEN MEAN IS ON TOP

,

y

••

1

••

÷

**4**, 1

••••••••••••••••••••••••••••••••••••••				
4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4			• • • • • • • • • • • • • • • • • • •	•••••• •••0010 •••••••
	2007 2007 2007	• • • • • • • • • • • • • • • • • • •		

FIGURE C-5

						2000 60 600 800 800 800 800 800 800 800 8		JROUP (FINA Na 2 Pheulctum 2 Coues 1	г мгали 5.0.8 Х2	H 1037 cc.c.	
				COULS 2 3			• • •			77.00 NSO	Ŭ D N U U U
680UP 2 NB 6 NB 6	KEANS S.D.S.	281.01 #56 = 281.01 #56 =	- 001 - 001					COUES 0	<b>.</b>		
CODES D							•		•	****	
					жет. Кирле Карна Карна						
	• • •			COUES C			• • •				
••••••••••••••••••••••••••••••••••••••		162.50	-			•	* * 8 *				
			•								
				EGRIUP 9-FINAL	85 ANK	90.00 FKDHa	.020.				
	• • • •		•	PREULCIUM 2 COUES 1	2		• • • •				
	* * * * 2 * * 4 0 * br>* * * * * * * * * * * * * * * * * *	•••••• •••••• •••••• •••••• •••••• •••••					•				
							***				

ų

3

FIGURE C-6

In detailed analysis curs, the critical parameters become P1, P2, MAXGP and NMIN. These values are dependent upon the size of the data set. In cases where there are less than 500 data units, the following values are suggested.

> P1  $\geq$  .01 P2  $\geq$  .005 MAXGP  $\leq$  90 NMIN  $\geq$  5 percent of data set, KSTOP  $\leq$  10

Other parameter settings are dependent only on the output desires of the analyst. For example, the residuals list may only be desired on final runs.

Use of AID4UT/AIDTRE with other Computer Programs as a Policy Capturing System:

In the process of Policy Capturing, various ancillary computer programs were used with AID4UT/AIDTRE to analyze and develop regression models. These programs were:

Program	Purpose	Documentation
GCORR	Contingency Table Generation for Data Analysis	Anderberg (1971)
REG REJ (EDSTAT-J)	Multiple regression model determination and vross validation	Jennings (1971)
MCA (SOCLIB)	Multiple Classification Analysis	UT-Computation Center
STEP-01	Stepwise Regression	CFHR, ESB, UT Austin
NCRIT	Prediction of Binary Vector	(Listing Included)

ر معرف

ь **ў** 

.....

(stal)

Figure C-7 is a Process Diagram for using the various programs in the Policy Capturing process. Alternative coefficient determination schemes allow the analyst considerable flexibility. The

In had a day the g



FIGURE C-7

use of the EDSTAT-J system represents a particularly important option since the analyst can write a driver program which uses the standard statistical subroutines to perform a statistical analysis that is tailored to his specific needs. As an example, the listing for a driver program named CRV is given later in this appendix. This is the routine that was written to perform the interative determination/cross-validation of regression coefficients discussed in Chapter IV. A complete description of the EDSTAT-J package is given in Jennings (1971).

In the cross-validation phase of the Policy Capturing process, either the PREDICT routine of the EDSTAT-J package or the NCRIT routine can be used. The NCRIT routine was adapted for use in this research from a program obtained from the AFHRL. It is very similar to the PREDICT routine. It does allow the specification of an arbitrary objective function and operates as a "canned program". A complete listing of NCPIT, including comment cards describing the input, is included.

#### Program Listings:

. . . . . .

 $\rho$ 

чb

JL.

.

÷.,

э.,

g.

 In order that a master listing of the programs used in this research could be referenced, source listings for those programs that are not formally described elsewhere are provided. The programs included are:

- AID4UT (AIDIV, ERROR, HEADER, LINK3, AIDFIN, RANKSIT)
- AIDTRE (TREEC, BLOCK, PLEVEL, PLOTH, PLOTH, PLOTV, PLOTV1, STOA1, PLOTHI, ITOA1, PLOTHF, RTOA1, AIDPLT)
- CRV (The various subroutines referenced are described in Jennings, 1971)
- NCRIT (DRIVER, DA TRAM, TMT, REGWTS, TABLES, OBJFUN) (This program also uses some EDSTAT-J routines.)

ar ar in

MSAMA-1ABS(NSAMA) MUMS-1ABS(NSAMA) MUMS-1ABS(NSAM3) MUMS-1ABS(NLMM) If (17.60.9) GO TO B40 If (17.60.9) GO TO B40 If (17.80.9) CALL EROR (1) If (180%-651.80%-180% (2) If (180%-651.80%-80% (2) If (180%-651.80%-80% (2) If (180%-551.80%-60%) CALL EROR (1) If (180%-551.80%-16) If (180%-562)551.64% (2) If (180%-582)551.64% (2) If (180%-582)551.64% (24L EROR (1) If (180%-582)551.64% (24L EROR (1) If (161%-6%-582)551.64% (24L EROR (1) If (161%-6%-562)551.64% (24L EROR (1) If (161%-6%-562)551.64% (24L EROR (1)) k CAFS TYFE=3 k F1-F4E F30P0410h OF THE T014L SUM OF SQUARES TAAT 495F BE CONTAINED IN A GROUP IF THAT GFOUP IS TO BECC4E A CANDIDATE GROUP FOR SFLITTNU K P2-THE FEST SPLIT UN THE TTH CANDIDATE GROUP MUST REQUEE THE UNEXPLAINED SUM OF SOUARES MUST REQUEE THE UNEXPLAINED SUM OF SOUARES BUT P2 PEODETION OF THE TOTAL SUM OF SOUARES UN THAT GROUP WILL NOT BECOME ACANDIDATE GROUP ADDIT FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL ADDIT FILL NOT BECOME A CANDIDATE GROUP ADDIT FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDATE FILL NOT BECOME A CANDIDAT PAINT 1050 PAINT 8-50. II LABEL I RUN. NCPERM. NKFELS. NSAMA. NSAMA. NMM. IFMT WEAD FORMAT CAND OF CANDS DEPENDING ON IFWI WEAD FOUNNS DESCRIPTION CARD COLUMNS DESCRIPTION I 1-78 R IFM IS DIMENSIONED 5.2 TO PROCESS A MAXAMM OF FOUR FORMAT CARDS) 79-80 NOI USED HEUDIRLENT. HEUDIRLENT. IF FORCED SALITING IS CALLED FOR THIS IF ORCED SALITING IS CALLED FOR THIS NUMEEM MAI NOT EXCEED 90 HNIN-THE MINIMUM NUMEEM OF CASES IN BOTH HESULTANT GROUPS HEOUTHED FOR A SPLIT x IFXT=JFMTe13 12-16 11-2 4-2 4 ຂຶ້ນບູບ **...**. .............

11-8 đ <u>_</u>_____

9 2

qiả lợ **1** 1

<u>,</u> ;

. •

۰,

ł

**4**522

2.2 

. . . . .

R.

NOIL FART CAND TYPE-9 DENDIES END OF NUM 1 TILE 1 RUN-THF RUN OPTION 0-RUN 101AL 0-RUN 101AL 1 RUN-THF RUN BFFORCE B 2-RUN A AFORCE B 1 -RUN A AFORCE B 2-RUN A AFORCE B 1 -RUN A AFORCE B 2-RUN A AFORCE B 1 -RUN A AFORCE B 2 -RUN AFORCE B 1 -RUN A AFORCE B 2 -RUN A AFORCE B 1 -RUN A AFORCE B 2 -RUN A AFORCE B 1 -RUN A AFORCE B 2 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 2 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AFORCE B 1 -RUN A AI D4UT PHOGRAM AIDIV(INPUT-OUTPUT-FUMCH-TAPES=INPUT-TAPE1.TAPE2.TAPE1.TAP IE4.TAPEZPUNCH-TAPE6.TAPE10.TAPE14.TAPE5.TAPE10. IE4.TAPEZPUNCH-TAPE6.TAPE10.TAPE14.TAPE5.TAPE10. IE4.TICP101.11CP101.11ALTAPE10.TAPE25.SS02.ACRT00.12(4) IE4.TICP101.11CP101.11ALTAPE10.1204.11APE2.TAPE2.TAPE10.12(4) IE4.TICP101.11CP10.11ALTAPE2.TAPE10.11APE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TAPE2.TA COMMON XPTXAPHED.NAMI(1301,1TR1A(183) COMMON XPTXAPHED.NAMI(1301,1TR1A(183) DETA RELAMXAM DATA RELAMXAM T CHITISHED AM CRIT .6H PHED / DATA ALIISISHED  .8) .6H [FORC.6HE B] .6H (NORW.6HAL B] .6M (FORC. 12) AM (NORW.6HAL A) .6H [FORC.6HE B] .6H (NORW.6HAL B] .6M (FORC. 26FE A) / REL NAMI . NITIALIZE FARAMETERS JCE = 0 + KEAD AID TITLE CARD + CARD COLUMNS DESCRIFTION - CARD COLUMNS DESCRIFTION - 1 R CARD TYPE=1 - 1 NOTE THAT CARD TYPE=9 DEMOTES END OF RUN E017 TITLE CARD
 NCPERM=TABSINCPERM) 65-70 51~56 57-58 11-74 26-80 54-40 25 9 **= =** [

 $\cup \cup \cup$ 

NREELS= [ABS (NREELS)

U

338

ж

12-21

92-22	NA NATORATINE NUMBER OF TTERATIONS TO BE PRIMIED A		
• •			
•	UNE SUPPORT IN UNATION ABOUT STUDIES IS		
•	PRINTED. IJF KSTOP IS LEFT BLANK, XSJOP .	د	
•	WILL GE SET TO D)	U	
•	IF KSTOP EQUALS -1.0MLY A SPLIT SUMMARY -		READ 1660. IT.P1.P2.MAXGP.KHIN.KSIOP.KY.NR.KKW.IOFI.KRFJ.IKESID.IC
•	WILL BE PRINTED		14RD+17REE+NOGO+16S5+1RS01+1RS02
02-10 e	P NV-THE NUMBER OF VARIARIES TO BE READ IN	L	
	INCLUDENCE THE COLLECTION MEDIACE AND THE C		• Fort Descendence, CARD
	A THE PART AND THE AND AND AND AND AND AND AND AND AND AND	, <b>.</b>	
	NAUTROLATION VANADAL AND TOTAL AND TOTAL	)	tstastop fit. At KSIOP = A
	TARLENCERT FUSI DE LESS INAN UN EVUAL		
26-06 .	M MK-IHE LIEFD KOMBER OF THE IDEKILICATION		
•	VARSABLE		
• 33	O KRETHE REMIND OFIION		
•	D-ND REWIND OF INPUT TAPE		KES 0=[AFS1]RESID]
•	• I-REWIND INPUT INPUT		[CAMD=[ABS [[CAMD]]
	0 TOPT-THE EXCLUDE OPTION FOR PREDICTOR		I M E = 1 A H S { I T R E E ]
•	LARTARLES		10902=1 ¥ + X + X + X + X + X + X + X + X + X +
•	5-20 NOT FICTIONE CISES WITH CHI-OF-PANGE 0		[f(lf2, .,f, 1) 1f55 = 0
•			[[1]\$54] .r. [] [853] = 0
			IFURSu2 At 1 IRSu2 a d
	ALL ALL ACCOUNTS AND ALL ALL ALL ALL ALL ALL ALL ALL ALL AL		IF III.NF.3] CALL ERROW (10)
, ,			
•			
۰ JS	C RKEJ-IKE KEJECI SUMMANT PAINI UPIJUN		
•	0-NO FEJECT SUMMARY TO BE PRINTED .		[] [RAAGP_[]]+]] CALL EXAGN 1:3]
•	L-HEJECI SUMMARY TO ZE PRINTED.		IF INV.GI.631 CALL ERROR (33)
• 36	O _ MESID-THE RESIDUAL SUMMARY OPTION		JF (*F#.GI.I) CALL EMROR (14)
•	0-NO RESIDUALS		JF (1007.61.1) CALL ERMOR (15)
•	I-PWINI RESIDUALS		JF (KAEJ.Cî.)) CALL ERROR (16)
•			IF (INESID.GT.3) CALL CAROP (17)
•			15 (1/480-61-1) CALL ERROR (41)
•	ALLOT TE DESTRUMENTE ON TAPE		IS ITERA. G. P. AND. ICARD. FO. J. CALL FRROR (42)
•			1
	D FREDRATING FLOW AND AND AND AND AND AND AND AND AND AND		THE ALL OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CON
			The state of successform states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and states and sta
•	A TANK TAOUT ADUAL MART AT LEFT TAKE TO TA	ſ	a a a a a a a a a a a a a a a a a a a
	LINFUL UALA UN KAFLUDI		A TALLAR UNIONALIANSA AYAA AYAA AYAA AYAA AYAA AYAA AYAA
		J	20101 010 01 00 00 00 00 00 00 00 00 00 0
•			
	3+STRUCTURE PLUT UNLY		「 たんに ニー・パー オンビー インション ビーション ション ション ション ション ション ション ション ション ション
	2-STRUCTURE FLOT AND DETAIL FLUT	ر	
			11 11 11 11 11 11 11 11 11 11 11 11 11
6 <b>2</b>	C MUGU-IME NUGU UMILUN		
•			2.2.1.2.1.4.2.2.2.2.2.2.2.2.2.2.2.2.2.2.
	- I-MUGG		
			1
• •	FURCEU SPLIIS(CRUSS+YALIUA), CURPUTEU - Foor 40,555 of abriants 6,500 5		
			C = 20 - 20 - 20 - 20 - 20 - 20 - 20 - 20
•			에 다 하는 것은 1.4
	- INSUL-OFICE TO THINK		
•	IN PURCED SPETISFEURPOIED PAUN REARS OF T		
•			
•			
	O POSUD-DPTION TO POINT RETAILORDOSED SEDENITITYS	u	
	IN FORCED SPLITS.WITH BETA BEJORTSFROM	1	[f INCPERM.E0.6) NCPERM±1006300

.

, Alexandre

..

.

k

.

f mar.finitester masses manuality
 f mar.finitester
 f mar.finitester masses
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester
 f mar.finitester mar.finitester mar.finitester
 f mar.finitester mar.finitester mar.finitester mar.finitester
 f mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester mar.finitester

11-36 PARTENE WILE TO BE CAVEN RECODE 22-35 PARTENE WILE TO BE CAVEN RECODE 22-35 PARTENE TO BE CAVEN RECODE 22-35 PARTENE TO BE CAVEN RECODE 22-35 PARTENE PARTENE 22-35 PARTENE PARTENE 22-35 PARTENE PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE 22-35 PARTENE

. *. .*.

. 7

, j

**. 1** 

۰. ا

و يەر مىر كە

ت پېلې د سوري

340

-

į

ŧ.

ł

THE INOUTIUN ARRAY STORES THE FOLLOWING INFORMATION FOR THE VART FOR THE PRECICTORS FOR THE CALTORS INOUTINE WINA-WALUE INOUTINE WINA-WALUE INOUTINE MIRERVAL INOUTINE MIRERVAL INOUTINE MIRERVAL INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE INOUTINE SPECIFIED JECODE FIND MAXIMUM AND MINIMUM RANGES AND INTERVAL LENGTH FOR EACH PRE . If THE PREDICIOR HAS SPECIFIED CATEGORIES.GO TO STATEMENT 350 KEAU &00° []•AA•88•CC•J8L•K8L•f[WOU]{J]•J≠[•]5} K8E±1825.K8L |{F4{}0}=31=K2 00 T0 42 PRIAT 1161 11641(2):EASU-1001CR(1,3) 11 f1(28(1,3):L10) CALL ERACR (21) 11 f1(28(1,3):L10) CALL ERACR (22) 15 f1(24(1,2):L1:0) CALL ERACR (22) 15 f1(24(1,2):L1:0) CALL ERACR (23) 15 f1(24:L2) GAL(23:AA X=2*) NoUT:J=1000*1NOUT(K-1)+1NGUT(K) NOUT:J=1000*1NOUT(K-1)+1NGUT(K) 16 (1NOUT12)-LE-01 CALL ERROR (46) 17 1 NOUT12)-LHOUT(1) 18 1 NOUT12)-LHOUT(1) 18 1 NOUT12)-LE-1 18 1 NOUT12) 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT12] 18 1 NOUT1 INOUT (7) - INUUT (10) REPEATS STEPS FOR INOUT (5) - INUUT (6) ICR(1,2)=XZ 1F (1)C9(1+2)+60+1) 60 T0 160 M±3 ACP (1,2)=88 ACK (1,3) +CC 5.1=L 02

DC 130 JF71343 FF (190UT(J).L1.D) GO 10 140 KF2*/J 1801[K-1]*13000*1MOUT(J.1)*1MOUT(J+2) 1801[K+2]*1 1801[K+2]*1 1001[K+2]*1 CKL MEADF (1) PFEAT [D30* [MOUT(K1.[MOUT(K+1) IF (L.LT.0) CALL ERROR (25) P×INT 1000+ J+INDUT(1)+LA LINER±LINEM+1 CALL MEAUER (1) 60 10 (118, 95), ROPT 17(18 .60. 0) 60 70 100 1 = 1.1 LC = LU •!W - 1 PAINT 1600-1-144-LC LIMER = LIMER • 1 PRIVI 962• INOUTI2) UC 10 120 CALL MEADER 11) PRIMI 1000: J.Le.LC L=L+1 PRIAT 1620. L.LH • LATTNOUT (3) LC=L8+LA-1 1-37=87

> 2 ¥,

3

28

79

..

.

t 200.00

CONTINUE 1-E [COMP AWAY STORES THE FOLLOWING LLE T. JICHIJJ3 FOB FOUL INTERVALS FLORPERT = MIXIMUM MALUE FCOMPERTS = INTERVAL LENGTH FCOMPERTS = INTERVAL LENGTH FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCOMPERTS = INPUT VALUE FCO LE=INGUT(1)+LA GG TO T5+661+ KOPT IF INOUT(1)+E9+6) HAUUT(1)+9 FF:NT 946+ 1+(ACR(1+N)+N+1+3)+IICR(1+1)+**INOUT(1**) PF:NT 946+ 1+(ACR(1+N)+N+1+3)+IICR(1+1)+**INOUT(1**)

100 10 120 341

ŝ
U	° FOR IOPT ≈ IIOISCARK CASSS WITH ONI-OF-RANKS POEDICIOOC)	1 CD&P (K + 1 ) = 1 & 1
• •		
J	ICOMPIX+1) = LOWER BOUNDIDISCARD CASES WITH PREDICTOR I LESS	LC=1C00P(K+1)
u	TAL OR EQUAL TO LONGR BOUND	IF (K.EQ.K3) GO TO 209
U ·	ICOMP(K+2) = RECOUT VALUE (SUIPUI) FOR ALL VALUES LESS THAN OR	CALL MEADER (1)
	E ECUSE TO SEAT ENDUT VALUE ([COMP(K+3)]	PAINT 1000 (COMPIX) LEGLC
		200 LB#LU-1
5	• ICOMP(K+4) = RECOVE VALUE / OR ALL VALUES LESS THAN OR EQUAL	
γı	TO NEXT INFUT VALUE BUT GREATER THAN PREVIOUS	210 EUNI 190E C anderzenaerreereereereereereereereereereereereer
، ر		C + MEAN CONTINUATION CARAC FOR PRENICTORS WITH S
, .		C UNTILL THE LAST INPUT RECORDE VALUET MANDATORY
	e trokpik il a brook valle	
10	ICOMPTIANT INTERVIEW ROLADING CASES WITH PREDICTOR 1	
u	GREATER THAN UPPER BOUND	HEAD 890. [[.AA.68.CC.JBL.KHL.([NOUT(J).J=].[5]
u	C FOR JOPT = 0100 NOT DISCARD CASES with GUT-OF-RANGE PREDICTORS)	IF (11.46.4) CALL EPROP (28)
U	* ICUMP(4) = RECODE VALUE (OUTPUT) FOR ALL VALUES LESS THAN OR	IF (JBL.WE.IICH(1.1)) CALL ERROR (29)
u	EQUAL TO MEXT INPUT VALUE (ICOMP(K+1))	CC 10 160
<b>U</b> 1	100H2 (M+1) = [NPUT VALUE	220 PMIN1 950, LH
0	* ICOMP(K.2) * RECODE VALUE FOR ALL VALUES LESS THAN OR EQUAL	CALL MEADEW (])
	CONCELLINGUE AUTOR AUTORATER THAN PREVIOUS	00 10 640 234 57040763-1700-1771
ب	CONVEXED FIXED RALEE	
		PRIMI 1620. SCONDEX1.IN
, c	• TCAMO(K.) - DECTOR VALUE FOD ALL VALUES COSATED TVAN DDEVICUE	
	PICONTINED & ARTONG FOR ALL ARTORS GAMAIRA TAK TARE FOR	ZED CONTRACT (2)
<b>،</b> ر		7-2x1-2-1
L		IF (1.67.39) CALL ERROR (30)
140	2 × X + 1	IF (K2-67.700) CALL ERHOK (31)
	00 150 J×1 **	ICR(1.1) #52
		R JIR J-S
	IF (K.GT.TGD) CALL EKROK 126)	K2≖x2+1
	[COMP(K)=INOUT(J)	250 COMIJNUE
150	CONTINUE	CALL E-ROR 1321
	60 TO 240	260 IF (IT.AE.5) CALL ERROR (32)
U I	÷*************************************	IF (JEL.GT. WW) CALL ERROH (33)
	• 111 EDITION (111 EAST CONSTANT)	IF (JUL, [1.]) CALL EXHUR 1347 15 (4.2) (T AV) CALL EXHUCK 1347
ى ر		IF (#BL.JT.C) CALL FARDE ING
υ	***************************************	UC 270 J=1.4
160	L*O	C ∎2 ± N
		IN041(1) = 1 600+ 1004 (K-1)+ 1004 (K)
		ZTO CONTINUE
	LB=1000*1M0UT(2)*1M0UT(3)	
	11 (14)+50,01/840 / 1 12 / 140 / 140 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 121 / 1	THE ATENCOTICS TO THE THE THE THE THE THE THE THE THE THE
170	94 14 1994 1414 14 1414 141 141 141 141	If [NoU](2)
		IF (1'OUT(3), EQ.0) [NOUF(3)=0
180	J= 1 NOUT (1)	IF (INDUT(4).40.0) [NOUT(4)=0
	PHIAT 960. I. (ACREIAN	
0.51		IZ+12001(4) If itmutititif lumnitidit fan faand inti
	I & N * # 1000 * 1404 I + 1 + 1 + 1 + 2 + 2 + 2 + 2 + 2 + 2 + 2	AM# [485 FKBL]
	IF (1AMT.FG.999993 60 TO (230-2201, KOPT	PGINT 940, JUL.AA.83.CC
	I COMP (K) = I WOUT (J)	IMOUT (84) * ]

******************

Ξ.

י א א

. . .

mathematics and a

۰.

SPECIFIED CATEGORIES ( 999999) IS READ

***************

.

342

₩EWIND KE HEWIND KO Keinnd Ke Finnse: Finnse0-2.08-18um.EQ.3.08.(18um.EQ.4.4MD.KRM.EQ.1))REMIND KF Iml(T=0 LN = 1 IF NO REJECTS PRIMIED.NO RANDOM SAMPLING 2 IF REJECTS PRIMIED 3 IF NO REJECTS PRIMIED ANDOM SAMPLING NL # FIELD NUMBER OF THE VARIABLE OF COTTERION N = VALUE OF THE VARIABLE (CRITENIOH) N = VALUE OF THE TOUMP(J) ARAV PRINT 1010. (FNOUT4J).J=1.41.KM IF (NOGG.E2.1) CALL EXIT IF (NOGG.E2.1) CALL EXIT NPENPIL NPENPIL IF (IN-BOUNTANN) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 IF (IN-BUL) NNENN-1 NEL-21-1 IF (IN-1) NNENN-1 NEL-21-1 NNEL-21-1 NNELxM = FIELD NO. GF WEIGHT # = value of weight « СМ1 =0 L M = NBL L M = NBL + 155 L x8=156 2002 310 250 130

60 10 479 2001 20 479 6 (0 [N MAI]. (360.450) 6 (1 0 31 FM)(135.3F1.NV) 6 0 10 31 FM)(135.3F1.NV) 6 1 (300.60 JL) 8 MNSHAND(23.7KNM).05005001) 6 1 (300.7(30.5KNM).05005001) 8 MNSHAND(23.7KNM).9.9099995-9 8 MNSHL03T(KNMM).9.9999995-9 8 MNSHL03T(KNMM).9.9999995-9 8 MNSHL03T(KNMM).9.9999995-9 8 MNSHL03T(KNMM).9.999995-9 8 MNSHL03T(KNMM).9.999995-9 8 MNSHL03T(KNMM).9.999995-9 8 MNSHL03T(KNMM).9.99995-9 8 MNSHL03T(KNMM).9.9995-9 8 MNSHL03T(KNMM).9.9995-9 8 MNSHL03T(KNMM).9.9995-9 8 MNSHL03T(KNMM).9.9995-9 8 MNSHL03T(KNMM).9.9995-9 8 MNSHL03T(KNMM).9.9995-9 8 MNSH PRUDUCE REJECTS FOR MISSING DATA CASES LATLA Meas (Ka. Ifm) (Inoutiu) (Jal.Mu) If (Nuutikm) (Lf.g) [noutikm)=1 m=Inuitiem) 1510H (LM) ** 60 10 490 ReelPHES |× = ] / 3L L M = L M 4 . 380

υ

្ត្រ 9 **\$**2

360

965 004

ĩ 350

3 ,

4

8

ω,

- 12 ', **4** 

, , ,

• r. ۰ ۷<u>۲</u>

: 2

14=1NUUT (KL) K2=1CR11 M6E=J0E+1

**8**94 ¢14

J2(106-1)=K2 J2(106-2)=H0UT(#4) If (HC 2)=H0UT(#4) J2(106-2)=H0UT(#4) J2(106-2)=H0UT(#4) J2(106-2)=H0UT(#4) J2(106-1)=K4) IF (106-11-200).60.3) PRIMT 3090 PH MT 1090. (1J2(1].4K1).4K4[.44).1[F=1.4) U0000 (1J2(1].4K1).4K4[.44).1[F=1.4) U0000 (1J2(1].4K1).4K4[.44).1[F=1.4) U0000 (1J2(1).4K1).4K4[.44).1[F=1.4)

IF (IRÉSID.€44.0) 60 TC 506 |SIGR(LM)=[NOUT(NR] LM=LM+1

YOAS 200

ųυ

00 590 K#1.1# 1f (1.61.4P) 60 TG 520 L*11CH(1.1]

Lts![Cd(1.]) 1 (![Cd(1.2].50.1) GO 10 536 11 01128(1.2].50.1) GO 10 536

......

PPEDICTORS WITH EQUAL INTERVAL

[f (w.67.5COMP(LK*)) M=1COMP(LK*)]
[f (w.67.5COMP(LK*) M=1COMP(LK*)]
[f (w.51.5COMP(LK*) M=1COMP(LK)]
[f (w.51.6CMP(LK*))/TCOMP(LK*2)]
[f (1COMP(LK*))/TCOMP(LK*2)]
[f (1COMP(LK*))/TCOMP(LK*2)]
[f (1COMP(LK*))/TCOMP(LK*2)]
[f (1COMP(LK*))/TCOMP(LK*2)]
[f (1COMP(LK*2))/TCOMP(LK*2)]

510

**Li** = 0 60 10 

PREDICTORS WITH SPECIFIED CATEGORIES

LL- VALUE TO BE GIVEN TO THE CATEGORY

KJ=IICP(1,3)+10P2 K4=11C8(1+1+3)-3 00 540 L=K3+K+2 17 (*.LE.IC0MP(L-1))

60 10 1576+580, JUMP RE(R1:20+LT(M+1)+1-0 Set 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 (M+2)+4 SPL 11 (M+2) +5PL 11 ( CCATINUE [STOKILM] = 1.088 LEICONPILS 555 566 570 540

**JUNI 1 NOD** 

8 30 -

۴

580

590

909

60 [0 [6]0+620]+ JUMP KCM1±KCMT+1

610

ちょうしょう しょう 629

([ × • ] | = Y T

2.26.1=36.1

LWALLY -LWALLY -40 TU 330 4811E (KF) KJ,f[5TOF(J),J=]56+L8L} LMB-N9L+]55

049

Gũ TO 330 IF (LM-6F.#×L) Gũ TU LAa=LAA+2

660

650

LHALLM

114.64.KHL) 60 TO AND 30 10 690 I=9×.

6 7 O

LA=LE+2 40 10 330 LM=MAL

660

0 6 Q

##JTE (KC) K1+([5T0P(J)-J=]+KBL) 15 TU J3D CCh([huf

100

110

ăkucă=ikuct+1 1 fouut.lta.amd.Jue.ke.giprint 1686 1 fouc.dta.amd.Jue.ke.giprint 1686 1 fouc.dta.gi print 1800, f(J2([[:KK].Kke].e]+1]]=[+J0E) 1 four-25[eu.2560 to 71]

ITRUN.ED.Z.CH.IRUN.EQ.J)NENHNSEL #Ifloatimgutijfloatimgut+nenij#100. 149-LT.=00001)65 TO 71

(IHUN.LT.2.GH. JHUN.EQ.4) 60 TU 730 11

	345	
	· •	
	. 16.2	
· · ·	13.61	
	e e e e e e e e e e e e e e e e e e e	
	90 91 91 91 91 91 91 91 91	
	0 10 1	
	<b>A</b>	
	s s	
	52 ° 74 6 37 * 5 2 ° 7	
	· · · · · · · · · · · · · · · · · · ·	
	بن الله من الله الله الله الله الله الله الله الل	
	156+L1 156+L1 1-KBL 1-KBL 1-KBL 1-1160 1-1160 1-1160 1-1160	
	1 1 1 2 1 0 2 0 0 1 0 1 1 2 1 0 2 0 0 1 0 1	
	1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1     1 <td></td>	
	22 46 74 74 74 74 74 74 74 74 74 74	

•

i

:

i

- 070 FORMAT YOM GRUUP+XXIEMETTAL WEIGHTOKL WEIGHTOKLOKYNI OF YNGHLIF 150W Y-50UARE 10A:5HT 5 \$12X++MEAWIGX*9H51D. OEV.) 050 FORMAT (14112040113)12:10134 101 FORMAT (1411264111) 102 FORMAT (1411264114) 101 FORMAT (1411264114) 101 FORMAT (1411264114) 102 FORMAT (141125'1115X*115:2X+15:2X+13:3X+11:5X+11:5X+11:5X+11: 10211F9.5:2X+F8.5:2X+15:2X+15:2X+13:2X+13:3X+11:5X+11:5X+11: 10211F9.5:2X+F8.5:2X+15:2X+15:2X+13:2X+13:3X+11:5X+11:5X+11: 10211F9.5:2X+F8.5:2X+15:2X+15:2X+13:2X+13:3X+11:5X+11:5X+11: 10211F9.5:2X+F8.5:2X+15:2X+15:2X+13:2X+13:3X+11:5X+11:5X+11: 10211F9.5:2X+58.5:2X+15:2X+15:2X+13:2X+13:2X+13:5X+11:5X+11:5X+11: 10211F9.5:2X+58.5:2X+15:2X+15:2X+13:2X+13:2X+13:5X+11:5X+11:5X+11: 10211F1.5X+11:5X+11:5X+11:5(5X+11) 102011F0.5X+11:5X+11:5(5X+11:2X+13:2X+13:2X+13:2X+13:5X+11:5X+11:5X+11:5X+11:5X+11:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+13:5X+12:5X+13:5X+12:5X+12:5X+13:5X+12:5X+13:5X+13:5X+12:5X+13:5X+12:5X+13:5X+12:5X+13:5X+12:5X+13:5X+12:5X+13:5X+12:5X+12:5X+13:5X+12:5X+13:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+12:5X+1

- 6 A MARGE AUM ITATIO 1660 FGRANT (11.46.5:13):1011 1020 FGRANT (11.46.5:13):21.1011 1020 FGRANT (11.46.5:13):21.011 1000 FGRANT (11.41.45049).1411.2049.01) 1000 FGRANT (11.41.45041014.113.2049.01) 1000 FGRANT (11.4.46.11044113.2040.01) 1100 FGRANT (11.4.46.11044113.2040.01) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 1124) 1110 FGRANT (11.4.46.00 FALASE 11460 FALASE 11460 1110 FGRANT (11.4.46.00 FALASE 11460 FALASE 11460 1110 FGRANT (11.4.46.00 FELASE 11460 FALASE - 0 % 3
- SUHHOUTTNE ERROR (IX) Print 10, 1X Stop
- FOXMAT (//* COMTROL CARD ERROR *.13.*.TERMIMATE*) END ວ **ຂ**

LUBOUTINE MEADER (1) CONDU YMHY LINER LINERALINERAL IF (LINEN-LT-52) 60 TO 1 F (LINEN-LT-52) 60 TO 1 PRIMT 20 LINERAL SIBFTC HEADER

1

, ...

<u>,</u># •

<u>े</u>

\$

Â.

j. J **)** 

ų.

- 2ບ ຄ
- FOMMAT (1M1.4X.+VAR1ASLE*+8X.*FIELD NG.+.2X.*RECODE*.6X.*C**MARESPON** 105 10*.51.*17PE*//1 End

SUBROUTIME LINK3 SUBROUTIME LINK3 SUMMARY OF SAILTERS AND ARRAYS IFS - 1 IF FORCED SPLITING - 2 IF REGULAR SPLITING IFS - 1 IF FORCED SPLITING - 1 IF FORCLAR SPLITING IFS - 2 IF REGULAR SPLITING - 2 IF REGULAR SPLITING - 2 IFR 2 (CULAR SPLITING - 1 IFFE 2 (CULAR SPLITING - 2 IFR 2 (CULAR SPLITING - 3 IFRE ALA - 3 IFRE ALA - 3 IFRE ALA - 3 IFRE ALA - 4 IFRE ALA - 3 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IFRE ALA - 4 IF [V=?-]⊎ N!=2 IF (IRESID+EQ_0) N]=] L2=1 KENDS#=1

Ų

Ų

 

 IDIUTES
 IDIUTES

 DG 265 [±];;# If (L]=61.4520-) 60 TC 100 Print [450: ].(ACR[[.]).J=[.]) Continue GƏ TO 910 TISS=60ATA(1.6) WHIYE (19.9886) TISS Iclut=3 LJ=] K1=] 60 TG (10.40), IFSM MT=60ATA(1.2) B=60ATA(1.3) T=66ATA(1.4) AT=77B IAN2 KCRAF [=8]

R

.... #1 = 1.....Cl FGF THE FIRST VARIABLE = (Ci=1)....C2 FGR THE SECOND VARIABLE = (C2=1).....C3 FGR THE SECOND VARIABLE #TEWE C1=ND = MUNER OF CLASSES IN THE FIRST VARIABLE C3=ND = MUNER OF CLASSES IN THE SECOND VARIABLE C3=ND = MUNER OF CLASSES IN THE FIRST VARIABLE CUMPUTE FEST STATISTICS ONE VARIABLE AT A TIME

4 1

. . .

4

.

ń

ू . त . . . . .

***** *****

. Andre ą *.*#;

9 . 4 . . . . . . . . . . . . . . .

i **dis**it. Lestigit

**,** 1

347

106

2 Q N

IS COMPARED WITH PREVIOUS MAX.BSS AND THE LARGER-DW MAX.855 855/755 efteen codes and codes TOTAL MEIGHT Sum of Y Sum Y-Souare 510.DEV. HEAN 005 855 Ĩ 72 85 .... 5 IF PREDICTOR IS MORDIONIC(ICR(1.4)=1).60 IO 1880
 IF PREDIC.OR IS FREE FLOATING.SORT CLASS MEANS INTO ASCENDING A = CLASS MŁAN Averikie 40°05 for all k îmitially = Larest Class Mean for current k = 40°05 if k is incremented in do loop 1060 COMPUTE CLASS STATISTICS ONE CLASS AT A TIME WHITE HEADING FOR CLASS STATISTICS IAVERIK).E0.AVER(40)) 60 TC 160 AVEMINE = A SEIKI=SPLITiJ++J/SPLIT(J+2)-A®A SEIKI=SGKT(SEIKI) CONTIAUE IF (N.EG.0) 60 TO 170 A=SPLIT(J,3)/SPLIT(J,2) If (ICR(1,3).EQ.1) 63 TO 156 DG 120 K=1.39 15 (A.LT.AVERIK)) GO TO 330 L±+0-K L±+0-K HK=41-M AKE414M)=KKEK[HH-1] AKE414M)=KKEK[HH-1] SD[HM)=KCCE[HM]=[COCE[HM]] 1C0DE (K) = 1 485 ( 7-M1) D0 170 J=#1,#2 N=5PL17(J+1) If (M+EQ+0) 60 1 00 118 Ja1+40 AVEH(J)za6**14 M;#[CR([+2] M2#[CR[[+]) PRINT 1430 CCNTINUE 10 160 CONT INUE CONTINUE I-ara ŝ 5 120 140 150 170 180 110 uuuuu (, **() () () ()** () () ()

Y = TGTAL SUM OF WEIGHTED CHITERION SCOKES IN ALL CLASSES(TDTAL AN = NUMBER OF OBSERVATIONS SUMMED OVER CLASSES OF THIS VARIABLE = TOTAL SUM OF WEIGHTS FOW ALL CLASSES OF THIS VARIABLE * SUM OF WEIGHTED CHITERION SCORES FOR THE (TOTAL - K) CLASSE E SUM OF WEIGHTS OF (ALL-K) CLASSES OF THIS VARIABLE. . . . . . IF AT LEAST ONE OF THE CLASSES QUALIFY. HE FOLLOWING ARE PRINTE Phint internal symbol S = BETwEEN SUM OF SQUARES(BSS) = [Y]==2]/#] - [Y2==2]/[W-W])-(Y= IF THE NUMBER OF CREENATIONS IN THE RESULTANT GROUPS AFTER THE ATTERPTED SOLIT IS LESS THAN THE MININUM SPECFIED ON CARD MG.J-CCL.J-JOT THE POINTER MA WILL BE ZERO AND A MISSAGE IS PRINED. MOJCATING THE IMABILITY TO SPELT PARENT GROUP MI DAINED. MOJCATING THE IMABILITY TO SPELT PARENT GROUP MI ENDING WITH STATEMENT NO. 260 SEARCH FOR THE LARGEST BSS AMONG THE CLASSES OF THIS VARIABLE. B ≈ CURPENT LARGEST 855. M4 ≈ Pointer set to The class(K) mith The largest current 855 AFTER EACH CYCLE OF THIS LOOP,THE FOLLUNING ARE PRIMIED PRINT INTERNAL SYMBOL 5 = 8/60474(1+6) 1CODE (M) +K=1+M4 1CODE (M) +X=L+L3 YI # SUM OF WEIGHTED CHITERION SCORES FOR THIS CLASS(K) N = KUUBER OF OBSERVATIONS IN THIS CLASS(K) WI * SUM OF WEIGHTS FON THIS CLASS(A) N SPL [1(J+2) SPL [1(J+3) SPL [1(J+4) 1 CODE (K) AVER (K) MJ & NUMBER OF CLASSES OF THIS VARJABLE MJ = SAME AS DEFINED IN DO LOOP 1100 ABOVE J = 1+2+---*NV Ś

Ę

1. - 1. B. - - - -

. 5. 1.

. . .

÷,  348

1

Same share a

* INOUT(43) # CODE NUMBER WHERE SPLIT GCCURS * INOUT(42) # CODE NUMBER OF THAT CLASS WITH THE LARGEST 855 RENSY#2 F=GGATA(1.4)/GDATA(1.3) F=GGATA(1.4)/GDATA(1.3) F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F=CAT F CAT F=CAT F CAT F=CAT F CAT F=CAT F CAT F CAT F CAT F CAT F CAT F CAT F CA **.**RC ITRATIONT=K1 IF N456.66.RC) 60 1350.480). IF5W2 60 10 (270.280). IF5W2 60 10 (270.280). IF5W2 61 16.60.91 60 10 290 IF (16.60.91 60 10 390 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILSTUP-L(.0160 10 300 FILST PRINT 1420 Print 1400, kl·n,(GDATA(1.K.,.**n=3.6),f**•S K1-604TA11.) F1K120P-L1-8160 TC 300 F1A1 1500. K1 F1 (12.00.0) 60 TO 340 F 402ATA11.9)/GDATA(1.3) A460ATA(1.2) A460ATA(1.2) A460ATA(1.1) F160ATA(1.1) 4-60ATA(1.1) 65 TO 895 15 K4510-11-6066 TO 339 PAINT 1510-13412 15 (148510-60401 60 TO 348 0 315 J≠1+6 6581411-J1+60ATA11+1+J 66NTNUE DC 32C 1=1.LZ INOUT (45) #1 INOUT (45) #M INOUT (43] #L INOUT (42) #M4 CONTINUE IDI01-10101-1 145 I VOVI 2=12-3 HS#×1 1 260 280 290 300 204 320 330 919 140 ....... [f (M3.LT.1) WRITE(19.9841) 60060L5.60060L5 [f (M3.LT.1) 60 10 210 0 200 xe1.m3 ]=[COE(KY.M1 M*SPL[1(J.1) M*SPL[1(J.1) M*SPL[1(J.2)+W1 M*SPL[1(J.2)+Y1 W155PL[1(J.2)+Y1 W155PL[1] W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 W1 1500 15 STORED. NON TRY THE NEAT PREDICTOR AND ITS CLASSES. J=[CODE[W3]•M1 h=SPL[T[J_J] h=SPL[T[J_J] h=SPL[T][SOD[C[M3]•K+(SPL[F[J+L]+L=2+4)+AVER[M3]+SD[M3] S=8/GDA[A[]+6] S=8/GDA[A[]+6] [F[L]_SOT_SSIPP] GD T2 20 [F[L]_SOT_SSIPP] GD T2 20 FRINT [AAG(-M1)+1(ACR[1+J)+J=1-3) FRINT [AAG(-M1)+1(ACR[1+J)+J=1-3) Inout(45) * PREDICTOR NUMBER WITH THE LARGEST BSS
 Inout(44) * Number of Classes in THIS Predictor
 Nout(44) * Number of Classes in THIS Predictor IF (L3.61.x51GP) GO TO 246 PRINT 1460. 8.5.(1005E(M).M=1.M6) Print 1470. (100DE(M).M=1.M3) (13.61.4510P1 G0 T0 220 CONTINUE RAXEDTTSS Vales/GDATA11.6) #811E (19,9001) KXA.YAZ IF (J.LT.855) GJ TO 260 855=8 00 250 x41.43 10 260 CONTINUE 1.64=64 1.**** 5.55 3 240 200 210 220 230 190

3

8 . . . . . .

**P** 

· ňa 2.4

4

1 i

, ice

. .

.......

349

Fert Four Life(1) C 1) = 40° [FORCE(1]FC.[]. + 1WOUT(K) Ff (1 WF. = 60.0) G0 T0 390 = Auf + 1 = Auf + 1 = Auf + 1 = 0 0 300 1 = 1 + 1 WF MARTI 160466116.48)*81°1604661176,01.400414% 161466116.48) 16145769411.0150 TO 300 16145769411.0150 TO 300 IFORCE (IFC. I) = 40 * IFORCE (IFC. I) + INOUT (K) IF (MAMA(#1.2).60.1) 60 10 410 IfC=IfC-1 IIEP-=FCHCE(FC+8)/81 IN=FCHCE(FC-8)-IIEMP*1 IN=FCHCE(FC-8)-IIEMP*1 M=FFCHCE(FC-2) M=FCHCE(FC-2) M=FCHCE(FC-3) M=FCH(N-2) M=H+4 H==H+4 H==H+4 H==H+4 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H==H+1 H=H+1 FCERFC-1 FFCERFC-1 FFCERFC-3).E0.00 G0 T0 359 FFCERFC-3).E0.00 G0 T0 359 FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCERFC-2).FINDUTIKS FFCEFC-2).FINDUTIKS FFCEFC-2).FINDUTIKS FFCEFCERFC-2) D0 421 K=M1.44 If (SPL2T1K+1).60.00.10 421 IFUNCE (IFC+81=0 MI.1=L 035 DQ 359 360 390 +10 *****20

UU 430 J=1+4 Leve:forge(ffc.m)/0] Nout(n)=1aus(fforce(ffc.e)-17€MP+0]) Ifonce(ffc.e)=11€MP 00 460 J≠1.4m 18E#P=160xCE(FC.4.1/40 1WUUTKF=160xCE(FC.4.1/40 1WUUTKF=1605(FFORE(FFC.4.1-4]TEMPa40) 1K04CE(1FC.4.1=1TEMP K=×-1 HI = [C4(1K.2) H2 = 1NO(1(42) H2 = 1NO(1(42) H2 = 1NO(1(42) L = 1 = 1NO(1(42) DC 47) = 1.44 DC 47) = 1.44 F 15/L1(1.41) .65, 0.1 60 T0 472 H2 = 1.41 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 471 - 2 L.43 DC 472 - 2 L.43 DC 472 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 473 - 2 L.43 DC 474 - 2 L.43 DC 474 - 2 L.43 DC 474 - 2 L.43 DC 474 - 2 L.43 D DG 422 K±M3.L If (SPLIT(K+1).EQ.0.) 60 TO 422 CUNTINUE CUNTINUE INOUTINATIA INOUTINATIA INOUTINATIA INOUTINATIA INOUTINATIA INOUTINATIA CUTINATIA INOUTINATIA REALI 16 (NEC.LT.11 60 70 470 20 460 1=1.745 Langf-1+1 ISS1 - 1 IF(L -€0. 1) ISS1 = 2 INUU1(42) = L=1 D0 +76 + 1.4M [=]+1 |\00T(|)=|AB5(K - M]) |#1+1 |NOUTill=1ABSfK ~ MJ) COMTINUE IF (1.60.0) 1551=2 [ \$71 LAONI = 3 CONTINUE 1552=1 542 F 25

440 450

400

**3** 1

Ħ

jant.

<u>.</u> ð

2**61** 275

421

422

627

504

A CONTRACTOR OF THE OWNER OF

350

K.l 610 620 وې نو پ 590 629 630 550 253 570 585 585 45 *2.5 41 *1.50 *201411*10*5014J 41 *1.50 *601411*10*196041411*175041411*31 15715+5041411*51-6547411*4.*6047411*31 N(L)*K 15541:00.2) 60 T0 560 11541:0-661141:+-0-66011413+4)/66011415-2) 210 * 2:5 210 * 2:5 20 1:1.4 20 1:1.4 ŝ z = i t (44) = M3 t (44) = M3 t (44) = M3 t (45) = M3 t (45) = M3 t (45) = M3 moutus moutus s (40,11,41) moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus moutus N 475 ្ឋ ||+[-| -23 **2+5 |1*55(X|+5PLIT(LL+*-1) |NUE 3 MI • IMOUTIL) SPLIT[J+1] •61. 0.5 6 •3. 1 *1. 5. 4500 10 476 *1. 5. 41. LM3 *1. 1. 1. NGUT(J+1) TW-55.2) TW-55.2) TW-55.43 TW-55.43 TW-55.43 Feo14-6014 Feo15-15715 Feo15-15715 10 476 **JNCE** . . . .

**i**,

• • •

5. 2. ₹. ₹.

 $\left| e^{i \theta} \right|$ 

(¥**)** įšū.

алат 1971 А. 1971 352

йСТ:[м-1[+] М.С.21.4.*1 1.11.П.ЕЕ. Б.А.МО.1F522.мЕ.].АМО.F].(E.,1791MR]E6101 ISTOR(5).5D13 1.5.5D17.4C1 (1185.6E..AMO.F12.1M).ISTOR(6):5D14).5D(0).4MC2.(1MOUTLJ 2: 1.4.1.).1M.1R2.6F5.1.4C1 2: 1.4.1.).1M.1R2.6F5.1.4C1 2: 1.4.1.1.1M.1R2.6F5.01 (0 0 0.0 1F 1.5.11.1.1.1 K=E.5.11.1.1.3 (0 10 0.0 1F 1.4.4.5.3 (0 10 0.70 LINLIST K2=K5-2 E[K5100-L[.0]PRIMT ]700+K2+[5]0R(5]+5D(])+K3+[5]0R(6]+5D(4)+1+ Ink2EK Ink2EK 4=60414(J++)/60414(J+3) B=244166414(J+6)/60414(J+3)) B121 1400, 1.**(60414(J+K)+K=3+6;+A+8 D0 930 J=1,700 ₩₹!ΤΕ (KD) L+(!STOR(J)+J=1+KBL) ₩Ε«!h9 KD GO TO (892,940), KEMDSW IF (12,60,41) GO TO 915 IF (12,600, b) GO TO 915 Frint 1410 Frint 1410 Frint 1410 744.Eu.#21 60 TO 890 DG 900 J#1+L2 K3_K1-1 [*=[NOUT(44) [L#[NOUT(43) 11.034 666665=7 S-1117S 1=60A1 A=F 2 F 2 = F ] 113112 113112 113112 . - K.O K2=13 €¥=¥ 1 5 = P J= H y [ # # ]

DU 1270 L#41-J61.2 DU 1270 L#41-J61.2 16 (L.Mr.411 GU TO 990 16 (L.Mr.411 GU TO 990 16 (L.Mr.411 GU TO 990 16 (1120-1280.960). 14 17 (L.M.10 10 980 J#1.ML 10 980 J#1.ML 10 081 J#1.500R(LM) 17 (LODE [J*1350R(LM) 17 (LODE [J*1350R(LM) 17 (LODE [J*1350R(LM) 17 (LODE [J*1350R(LM) 60 10 1270 If 11-1444-11-46.144 60 10 11160-1140-1270). W2 UP 10 1680 If 114418-141-66-115 60 10 11160-12701. W2 жак-40°L: |f ilverns+1).me.9999 60 f0 (1008-1076)+ 1LL |f kiyye .lt. 01 60 10 1270 |D=[f*0E(1) Fish(1,4); GG TO (10001520:1209). 1A If (1.61.400) GG TO 1120 UC TO (1000-1010). 1C UC TO (1000-1010). 1C UC TO (10 MD) GG TO MD If (14ES1/..E0.0) GO TO 130C АЧЕЙН.-155 Анайн.-318 Меай (Rai La([510R(J)-J=],884) IF (196510.EU.0) 10=0 **188510 IF (182.LF.50) 60 TO 1950 =Gi'ATA1J++1/GDATA(J+3) R*6041413.13 16 18 16 0.10 10 10 20 COMINUE FRICT SAD 60 10 973 A=11 05 (12)/10 LL=1/40 D0 920 x=1.4 5PL [1 (Jek)=5 C0%] [NUE I = AVE HILA) 60 10 340 1.2-1-2-1 CONTINUE L V = 3 1 1 L = = 3 1 1 12=241 JOE 2=0 RANK ¥C≢t 1~2=0 1..... 1-258 1059 1010 1020 1030 10-01 1950 1960 1070 876 5 940 470 980 007

**386** 4 ⁴

**4**00

ø

. इ

ŵ

(5) 1=0
(15 20 J#1+ML
(15 200 J#1+ML
(15 200 J#1+ML
(15 200 J#1+ML
(15 200 J#1+ML
(15 200 J#1+M+)
(15 200 JE0 K J#1+M+)
(15 11 20 J*10
(15 11 20 J*10
(15 11 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 20 J*10
(10 12260 24101 1610 25101 1620• ((J22([[+KK]+KK=[+4]+[[=[+4] 192=192+1 Leggggg del[ fsc| L+([5]0R[J]+J=[+KBL) Mel[ fsc| L+([5]0R[J]+J=[+KBL) 60 T0 (1]50.1325.1340)+ 15 62.0 Tc0 (1510R[J]+J=[+KAL) 9 F10 Tc0 (150 0 130 10 112 (40) L+([5]0R[J]+J=[+KAL) 11 [ (40) L+([5]0R[J]+J=[+KAL) 12 112 (40) L+([5]0R[J]+J=[+KAL) 13 112 (40) L+([5]0R[J]+J=[+KAL) 14 112 (40) L+([5]0R[J]+J=[+KAL) 14 112 (40) L+([5]0R[J]+J=[+KAL) 14 112 (40) L+([5]0R[J]+J=[+KAL) 15 112 (40) L+([5]0R[J]+J=[+KAL) 15 112 (40) L+([5]0R[J]+J=[+KAL) 15 112 (40) L+([5]0R[J]+J=[+KAL) 15 112 (40) L+([5]0R[J]+J=[+KAL) 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) L+([5]0R[J]+J=[+KAL)] 15 112 (40) JOE2=0 60 10 1270 60 10 1270 60 10 1270 10 10 1270 IS (AEMINWLEQLZ) 60 TO 1350 UC=1 de3 14=2 60 TO 950 16+2 14+3 18+3 110=+ 1249 1305 0501 13,0 1250 1310 1230 1260 1240 1350

3 اھ

" A

· · ·

**्रें** -

**فلا**ر, ۵

н 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 1997 - 1997 - 1997 1997 - 1997 - 1997 1997 - 1997 - 1997 1997 - 1997 - 1997 - 1997 1997 - 1997 - 1997 - 1997 1997 - 1997 - 1997 - 1997 - 1997 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1

60 TO (1099.1150.1270). M3 AFF(LENTY AFF(LENTY SFCR(LENTY SFCR(LENTY SFCR(LENTY D 1100 J=1.00 T 100 J=1.00 T 100 J=1.00 T 100 J=1.00 T 100 J=1.00 F 100 L=1.00 
CFB_I []HI.]3[[]F.].*064'0UP*.]3.* CANNOT BE SPLIT FGR FAILURE TO FGFB_I []HI.]3][]F.].*064'0UP*.]3.* CANNOT BE SPLIT FGR FAILURE TO [!EC.CE THE OFIAL SW OF SQUARES FF P2**FF.6.55X.']H **PRODO 3106. 7-11:4'H. *BAILBES.FOR THIS GROUP ON PREDIETOR*[13.2X3346.64 3.**E3.8.']H. *60X**ISS =*E20.8.']H *57X*8H?2*ISS **E20.8'']H *1 SJICHTIV) FCAMI (11-11-11(14-1/140)*ALL PREDICIORS FOR GAOUP*13+54×FAILFT 1/2 MCET 15: Min. Number of Observation requirements*/140+131(14+,1 FSHE FSHE SECCES-1-1 (2)196 F2-5 (2)196 F2-5 (2)190 F2-K (2)190 F2-K (2)190 F2-K (2)14(1,2)=662TA(1,2).59LT(1,2) (2)14(1,2)=662TA(1,2).59LT(1,2) (2)14(1,5)=660TA(1,5).59LT(1,2) (2)14(1,4)=550ATA(1,5).59LT(1,4) (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20 (2)14(1,4)=20(2)14(1,4)=20 (2)14(1,4)=20(2)14(1,4)=20(2)14(1,4)=20 (2 AC=L REWID KA REWID KA REWID KC REWID KC REWID KC REWID KC REWIN KC ACCI = XW(1)•6DATA(1.4) ACCI = XW(1)•6DATA(1.3) ACCI = XW(1)•62)•6DATA(1.3) ACCI = X ACCI = XW(1)•62)•6DATA(1.3) ACCI = C IF S # *1•0 (0 10 950 (0 10 (1370+53)• 1 FS#=2 3=4.2

JOPAL 140413111401//584.0CURRENT SUMMAR.2/1 FOMMAT 1414-13111401//584.0FUMAL SUMMARY8/1 FOMMAT 1/10 MCF 101AL 155 TOLAL 655 1 - 47-544EC & 101AL 155 TOLAL 655 2 - 7207/141333617.0413.0410.5426K *112 101AL 755 101AL 155 œ FCRMAT (//ª NCF 1651 FURMAT 111. NCF H-SUUARED 31-411 FORMATI 11/11 11. 1690 FCP 1570 1049 1660 1670 1680 1530 9522 1560 1580 1650 1535 1590 1660 1410 1620 1640

FORMAT (*ELIMER NUMBER OF FIMAL GROUPS EQUALS MAXGE OK NUMBER OF 12#20104E GROUPS EQUALS 94-71841147E NUMBER OF FIMAL FPO:P FORMAT 1554.9MAND GROUP-154.11M WITH CODES.113.7794.17137 FORMAT 14 1554.9MAND GROUP-154.11M WITH CODES.113.7794.17137 FORMAT 14.1111.11M FORMAT 14.0049-14.1 WITH CODES.1713.7794.1713.7794.1713 FORMAT 14.0131114.1 FORMAT 14.0131114.1 FORMAT 14.04131114.1 FORMAT 14.04131114.1

1510

· 1997年,1997年,1997年,1997年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,1998年,199

* . . . . .

4 (c) **2** 

.

8 1 : .

, У

FORMAT (//* NEF TOTAL TSS TOTAL 855 TOTAL 455 R-SQUARED R F-MSO DFT DFZ F-AMOVA YT (f2*//1x+11+361F.8+F13+6+F10+5+12+4+15+17+F12,4+15+17+/*0*+131(

TOTAL ISS TOTAL ESS

F-ANOVA DF1 *F12+4+15+17+//*0*,131 LF2*//j4+13+3617+8+F13+8+F19+5+2+X

TOTAL WSS F-ANOVA

2

+F12.4+15+17+//*2*+1314 SPLIT GROUP=.14.* [NIO CROUP*.14.

2" AND GRUPP-141 108"AT TODESS IS UNDEFINED SINCE EITHER GROUP-14." ON GROUP-14." 108"AT TODESS IS UNDEFINED SINCE EITHER GROUPS IS NOT IN

1700 FCRMAITA.0ROUP 0.13.0 Ne0.16.0 MEANED.FG.2081.0GROUP 0.13. 10 NE0.16.0 MEANED.FG.2044.0120011.000 KSUED.FP.5. 20 FRSUED.611.0111.111.101.011.011 1710 FURMIT.PL.500.55PL.17 SUMMART.) 1711 FCRMIT.0 GROUP 1 010 NOT 5PL.51.1ERNINATED) 1700 5

SubRoufine aftern SubRoufine aftern DIRENSION MAILION: ANNION: OLIPSGRTIVEBIANC(1) (2 CONTINUE 65%Ex6GFREAMIN 66%Ex6GFREAMIN 61 J01 1=1.W FLIPTTTTER (1) CONTINEE (1) CONTINEE (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTTER (1) FLIPTTTER (

**, e** 

....**!** 

÷

356

00 102 1=1.wv W.angusetti-Sumeij/Groupsetj 3854ran -Gusmran .mrangsettj 44fiangetj-10-mrangsettj**·nrangssettj·groupsett**j/groupsettj

BU-11400

({5x+ 2+fi2.5+fi0+fi0+ ) [ = =+46+fi5.5+5(fi5.5)	[[0 0,46,F]5,4%;5!F]5,4Å]] [[[6,1%,410,2%,F]5,5;F]5,F]5,4]}	((2615.8)		
CRMATISA.	-CHMAT (* *	CRMAT ( 261	361084	043

56840011xE KANKSIT15ET KRAWF.MF.KOUNDER) DIKENSION SET(20) KRAWF.MF.KOUNDER) THEGE ORD-OFUSUD1.0RDSUBJ DE 1 191.N Af11×5E1(1) Af11×5E1(1) Af11×5E1(1) Af11×5E1(1) Af11×5E1(1) Af11×5E1(1) Af11×5(1) ٠ CCNTIMLE (T. X1)-ROUNDEA.61.X1(J.1)) 69 70 J--11 1620-Y #ICOUMT-1 62 70 J 62 70 J J=1-1 D=1-1 D=2 × 41-ICOUNT D=2 × 41-ICOUNT D=2 × 41-ICOUNT AAMAIL0ASUB11×1 (ICOUMT-1)/2 C=11NLE D=2 T0 7 D=2 T0 7 C=11RN ENT 111-1 111-1 1111-1 2055-912-00011 2484-104050411 1000% ±1 Ξ

AIDTRE

Ų

. л 8.8 . г. *.* 

.

,

. și.

**, 1**961

a A S P S

3

. ,869

, t

÷

ALUIKE Puggam REECTAPUT-OUTUTTIEFT-TAPET9) C COMMAN HITES DEGRAM FIELS DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION HITES DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNSION DEFNS 60 1021 141.4 14048201940 60 10 1050 17705642 ). [r]] ]=]-64 [=[w-]]]]=0 0 = 2 3001 1001 1010 1021 1045 1030 1001 1640

00 60 114 16 = 1 = 0.06 70 0 16 = 1 = 0.06 70 0 CALL ELENK(1.9.16) NUMLARDS(169.1) NUMLARDS(169.1) 17 (NUML-Q.160.10 10 NUMEARDS(169.1) 17 (NUML-Q.160.10 10 17 (NUML-Q.160.10 10 17 (NUML-2.10.1) .60.10 10 18 (NTAL2.2) 0 10 11 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 = NUML 18 (NTAL2.2) 1 ŝ 52 .44)60 TO 2008 (#06K3 (K+1) #%UM2 (#06K3 (K) #NUM1 D0 1051 1=1.4 1=0241(1)=3 28 Ş -

SCIIIICAT OF FREE FOR SAMPLE C. FORMITIEISTART OF FREE FOR SAMPLE C. UO TO 100C EAL CAIT EAD NAL 1 14-1-0050 10 90 14-16-405161 10 90 Call Block (1+7+16P) Call Block (1+7+16P) Call Plot 1030-1940-1940) 194456 10 11030-1940-1940 10 2010 17 1104-40-2060 10 2010 17 1104-40-2060 10 2010 00 00 14.6 16 18 45.5 4.2 30 70 81 16 18 456 42113 16 4 8 8 42113 Ef(1PHASE_E0_2)GD T0 76
T#0%X2(2*1)=NUMI
T#0%X2(2*1)=NUM2
T#0%Z(2*1)=NUM2
GD T0 BD
GD T0 BD K=2*1*L | MORK3*K+1) =NUM1 | MORK3*K+1) =NUM1 | MORK3*K+1) =NUM2 CONTINUE =XM62 2002 3965 2010 è 5 2 2 3 8

1

(marini)

P 4 8 ;; 1 1 1 1 1 1 1 1 ... о. - Б ۰. ., ¥ . ۰,5 1

CALL PLOTV [[HTP05.43.4.[hE.8.8H******] CALL PLOTV [[HTP05.43.4.[hE.8.8H******] CALL PLOTH [[HTP05.14.[hE.1.6.6H9690P ] [GR0UP.3] CALL PLOTH [[HTP05.11.[NE.1.6.6H96719.4] [F(x:1051[GR0UP.1].EC.01CALL PLOTM:]NFP05.10.L[NE.1.6.6H97199.2] CALL PLOTH [[HTP05.2.1.[NE.1.55:5H820 ~ 8501[GR0UP].6.3] CALL PLOTH [[HTP05.2.1.[NE.1.55:5H820 ~ 8501[GR0UP].6.3] CALL PLOTH [[HTP05.2.1.[NE.2.55:5H820 ~ 8501[GR0UP].6.3] CALL PLOTH [[HTP05.2.1.[NE.2.55:5H820 ~ 8501[GR0UP].6.3] CALL PLOTH [[HTP05.2.1.[NE.2.55:5H820 ~ 8501[GR0UP].6.3] CALL PLOTH [[HTP05.2.1.] F(CC.2.0.1000.)DF221000. CALL PLOTE([RTP05.17.1] F(CF.6.1.1000.)DF221000. CALL PLEVEL11.00F2.FF8001GR0UP].PR009 CALL PLEVEL11.00F2.FF8001GR0UP].PR009 CALL PLEVEL1.00F2.FF8001GR0UP].PR009 current HulwailREE44G1.AMG1.SDG1.MG2.KMG2.SDG2.ACR(80.3).MOG3 CUMMON/ELT/MAREANSD.ELCOUE.TREPEOFSEG.FESGT.MCGDE.KID5 DIFFLSION NILL1.NKEAN[18].SD118[1].ECODE(18].SO).1PRED1180). 18S0[160].FFSQ1180].T[180].K[18].SD138[1].FCODE(18].S9).1PRED1180). 28C0E[17].4.(15).11[16].K[15]. 28C0E[17].4.(15).11[16].K[15].4.(15).2.(179.2). 28C0E[17].4.(15).11[16].K[15].4.(15).2.(179.2). 12.4.5.7.(16).2).4.7.1.8.[5.22.29.36.43.50]. CALL PLCTHF (IRTPOS-32.4LINE-2.5.5%PROB.4PROB.6.3) CALL PLCTHF(IRTPOS-32.4LINE-3.10+1049RED1CTOR .1PRED116ROU9).2) iPPatrocs-12 Litatrecs-12 00 1912 L#1.0448 Call PLOTHI (IPP.LN.0.0.1CODE (IGROUP.IS2).3) 00 1002 [4...] 1021 PLOTH (1PP.LINE-3.0.4CR(1AL+L)) 10214 PLOTH (1PTP0S.2.LINE ++.5.5HCODES) CALL PLOTH (1RTP0S.2.LINE ++.5.5HCODES) 152×152+1 60 fu (**1003+1003+1015)+KK** LAT_INE ... 60 TO 1010 IF (NC.LE.0160 TO 1915 IF INC.LT.IZ) NVARENC IF INC.LT. 14) HVAPENC [PP#1HTPOS+1 MC#"+CODE (]GROUP} |PP-JRTP^S+7 RC=AC-AVAR [.dd[=jd] 21=-444 **A**R = ] **4** 152=1 8 e -Ņ 33 5 512

1015 HETURN End

SudaduTimE PLEVELIOF1.0F2.4.9) DIRENSIGN Y(6).486(1).642.4.9) DIRENSIGN Y(6).486(1).644.4.3) BASHWIT2.7.39(5.5).4106(1).6.4.913)333333.5-1-2*(.2778 T786-2-2*(.7985.7).45-2*(.5952.306955-3-2*6.83)335333.5-1-2*(.2778 TFU6F1.F1..08.0F2.4F1.-00.F5.160 T0 7 TFU6F1.F1..08.0F2.4F1.-00.F5.160 T0 7 A66(1).60.0F1.4MD.A86(2).60.0F2.160 T0 7 A66(1).60.11.60.11.4MD.A86(2).50.0F2.160 T0 7 A66(1).60.11.60.11.4MD.A86(2).50.0F2.160 T0 7 A66(1).60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.11.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60.10.60. × * 0. ⊎0 13 [=1.10 GC TO(11.11.11.11.11.11.11.10).1 Y(6) × Y(2) Z = 1./(U*U) GAMA(I) = GAMA(I)-BASMLM(Z*U) Z = 1./(T*T) GAMA(I) = GAMA(I)+BASMLM(Z*T) ) If (CN-ME.0.) 60 TO 95 XX = Y(2)-C F(X1.1.0.)6C TO 95 P=P=E=P(XX-692) P=P=E=P(XX-692) X = 0. 64MA(1) = \$572364943 CONTINUE 2

11 D0 13 J=1.6
 x = x+M
 X = S1(x)
 X = S1(x)
 X = S1(x)
 X = (1)-CC(MALDG(XS)+CH+ALDG(1,-XS*LS)
 F(2)12,12,02
 F(2)12,12,02
 F(2) = 1,02
 F(2 >>>-SLEMEUTIVE PLOIC (IPMPOS.LIME.ICHAR) COMPON PLIATY LARVI (201132-66) IF (IPMPOS .LT. 2) RETURN IF (IPMPOS .GT. 132) RETURN IF (ILME .GT. 1) STUDN IF (LINE .GT. 4) RETURN ARTIPAPOS.LIME1 = (CAAR EAD 00 10 J=1.60 Print 100. (Eart(1.J). 1=1.132) Return Return Ead SUBROUTINE PLOTB COMMON /PLARY/ [ARY (132.66) UDIA (LIAN /14 / (ARY (132.66) DS 10 11.1.12 DS 10 11.1.12 DS 10 11.1.13 LAP(14.1) = (BLMA AFTLAN) = (BLMA 568×00Т1м6 РL0ТР Ссмном /РL1АRY/ 14R¥1132•601 Data 11 /141/ 14k*(1+11 #11 END 100 2 2

ş

đ 5

8

, es 1. 1

· 🎽

:

, , ,

* ∳ *

÷.

ð

1

e '. . e K

" U

**A** 1

÷ Ĵ,

·,.

į. ñ. 

ř. ε

i

i

ŀ

÷

ķ

ŗ

SUBROUTINE PLOTH (TPRPOS, LINE,NCHAR, ISTRNG) Subron Judals/Stangill) Call Sidailostrug, Istangi Call Plothit[Prpos, Line,nchar, Jstrng) Rétor Edi

SUBROUTINE PLOTV(1PRPOS.LINE.MCMAR.ISTRNG) COMON.MORAZ.STRNG[13]) CALL STORIUSTRNG, ISTNAG) CALL FLOTVI(1PRPOS.LINE.MCMAR.JSTRNG) RELUAM EPD

SUBROUTINE PLOTHILIPHODS.LINE.NCHAR,JSTRNG) ILMA A NCHAM ICHAR ACTAN IFLICHAR GT, 131 ICHAR = 131 IFLICHAR GT, 131 ICHAR = 131 J 1 1 1FLICHAR J 1 1 1FLICHAR J 2 1 1 FLICHAR CELL PLOTCIJ.LINE,JSTANG(I)) ELUR

SUBRCUTINE PLOTVIIIPAPOS.LINE.WCNAR.JSTANG) Dimension Jstng (131) Dimesion Jstng (131) ICHAR - 67. 60) ICHAR = 60 DG 11 Jijchar J = 1 & Line - 1 Call Plot(IPAPOS,JJJTMG(I)) Elon Elon

100

SUBROUTINE STORI CUSTANG, ISTRAG) Subroutine Stori Custang(131) Subragion Istrag(14). Justang(131) Secos (131),100.(Strag) Justang Retuan Retuan Edrat (131) 2

SUBHOUTINE PLETMI (TPRPDS,LINE,MCHAF,ISTRNG,IYALUE,IW) CONNON / VOKAZ/ JSTRNG13) Call Stoat (JSTRNG,ISTRNG) Call Stoat (JSTRNG,NCHAR,ISTRNG) Call Stoat (JSTRG,NCHAR,ISTRNG) Call Pletmi (1PRPOS,LINE,ICMAR,JSTRNG) EEU RETURN

. 00 14 J#1,10 FM.13) = HEAK ENCODE (19:104/FMT1) FFM ENCODE (19:1471-14AFT1) FVALUE DECUDE (100-100-1AFT1) FART2 DEECUDE (100-100-1AFT1) FART2 DEECUDE (100-100-1AFT1) STAV6(1) = FAHT2(1) | = 4 |fm(]) = (=(2) | = [ = 1 |fm(]) = ]M) | = [ = 1 RETUKN FCRMAT (12) FCRMAT(241) FCRMAT(201) FCRMAT(1001) 5.D

2

1

SUBHOUTINE FLOTHF (IPPPOS, LINE, MCMAR, ISTRNG, FVALUE, JU, ID) COMMUN / WORKAZ/ USTPNG(13) Call Ftoai (USTMAG, ISTRNG) Call Ftoai (USTMAG, ISTRNG) Call Ftoai (USTMAG, NCMAR, I, FVALUE, IL, ID, JW) (CHM & NCMAR & JW Call FLOTH(IPPPOS, LINE, ICHAN, JSTRNG) CALL PLOTH(IPPPOS, LINE, ICHAN, JSTRNG) ELUR

SUBROUTINE RTGAI (ISTRMS.ISC.FYALUE.IK.10. JW) COUMON YWOKKAJ/ JARY1(10). IAR72(108). IFW(1). LW(2). LD(2) DATA JELME /JM DATA JELME /JM DATA JELME /JM DATA JELME /JM C = WOD(19:100) IF 1JW G(1 = 1)2-15C FCCOE (2:100.KG) JD FCCOE (2:100.KG) JD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE (2:100.KG) LD FCCOE ( FUTURE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE STATE ГСКМАТ 'ZA!) FORMAT(1041) FORMAT(1041) I = I • 1 00 |4 J=1, 10 Ĥ • [ * [ •

14

SUUROUTINE ALDPLT(H1.JD.NOGP) COMMN.APT1/ 0(641-22):ICORMILD3).ICOKVIJB3).LEVEL(103) COMMN.APT1/ 0(641-22):ICORMILD3).ICOKVIJB3).LEVEL(103) LOGICAL UN INTEGEN H1.M1 INTEGEN H1.M1 INTEGEN H1.M1 INTEGEN H1.M1 INTEGEN H1.M1 INTEGEN H1.M1 INTEGEN H1.M1 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEGEN H1.M2 INTEG D0 50 L * ICOPH". JMID ALL ICOPVI) * 100TL ALLGCOPP 2 ICOPVP) * D0TF ALIMICA ICOPVI) * 100TF ALIMICA I ALIAL DG 30 L = 1410, 100849 K(L, 100441) = 1001L 1640 - 640 UG 19 1 - 1. 123 LEVELTI - 8 UG 25 1 - 1. 641 DO 27 J - 1. 22 * [C.JAHP+JC K(1, J) = 14L [CURH(1) = 32] [COHV(1) = 2 K(322. 1) - 1451 K(322. 1) - 1451 50 10 67 IN LD

> 9 20

. . . . . .

8

0

å

. .

ð v

Æ

362

P 9 99

 WIUM1 = INTD-1

 CC0R1 = IC00011 = IASF

 KIUMIDP1 - IC00011 = IASF

 KIUMIDP1 - IC00011 = IASF

 XII = M0011/100-11011

 JJ = M0011/100-1011

 JJ = M0011/100-1011

 JJ = M0011/100-1011

 JJ = M0011/100-1011

 KI = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = AASS.0MUMI(11)

 K = COR1) = LOND

 K = COR1) = LONP

 K = COR1) = LONP

 K = COR1) = AND. K(1. J) . . J = L. 22)

 K = COR1NE

 K = COR1NE

 K = COR1NE

 K = COR1

 K = COR1NE

 K = COR1NE

 K = AAD. K(1. J) . . J = L. 22)

 K = CUAN

 K = COR1NE

 K = CUAN

 

:

... 1. 1. :

. .

. .

• **A** • ) •

ĸ

л 3.

е., 7

- 8 80
- FORMAT(*)TREE STRUCTURE OF FIRST FIGHT LEVELS, GROUP*, 1ª Mumrep IS Given in center of the rola, the pair member*, 2ª Mith the migher mean is Gm top*///) End 200 700
  - 10

1

Ĵ

KCKTT SUBROUTINE YMT(N=TO-FROM) U MERCION TO(N) -FROM(N) UC 5 Tel.N UC 5 Tel.N UC 5 Tel.N READPF(2350,205747J) REL.7777 Roreduce. Load(140:605747J) Erecute. 0 1 (al.NV 1 al.NV 4 (al.NV  4 (al.NV 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al.NV) 4 (al. JOJ • 1 M=25 • PK=25 • RUM ( 5 ) CP • -EN15 POL IST ۰ 2 ..... • ah. PROGRAM CRY (APUTOUTAUT) DIEGNON FMT1623.0KFMT1223, KFMT2(22), KFMT2(22) DIEGNON FMT10573.4 CUMMCN FMT105MT3.4 CUMMCNFMT105MT3.4 EQUIVALENCE (FMT1.6FMT3), (FMT2.4FMT2), (AoKA) CUMMCNFMT3.4 EQUIVALENCE (FMT1.6FMT1), (FMT2.4FMT2), (AoKA) S COMMT (55.6 NV: MT1.4NV018), 2 241 - 500 READ 5. MSR (AF NV: MT1.4NV018), 2 241 - 055 (LF 1.4NM.NV) MV2 MV2 5. MS2 (FMT2.4539), 2 241 - 055 (LF 1.4NM.NV) MV2 MV2 5. MV2 (MT1.955 ) MV2 MV2 5. S201 5 CALL 055 (LT 1.4N1 % CALL 055 (LF 1.4NM.NV) CALL 057 (LT 2.1NM.NV)55 (L 055 (LT 1.4N1 % V2 35) 5 240 % V2 365 (LF 1.4NM.NV) CALL 057 (LT 2.1NM.NV)55 (L 055 (LT 1.4N1 % V2 35) 5 240 % V2 365 (LF 1.4NM.NV) CALL 057 (LT 2.1NM.NV)55 (L 055 (LT 1.4N1 % V2 35) 5 240 % V2 365 (LF 1.4NM.NV) CALL 057 (LT 2.1NM.NV)55 (L 055 (LT 1.4N1 % V2 35) 5 240 % V2 365 (LF 1.4NM.NV) CALL 057 (LT 1.4N1 % V1 4.4N1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1.4N1 % V1 1.81) CALL 057 (LT 1 SJBROUTHE DATMAN TIMEGINNY) & COMBON BEAGGETERS Call 2003 (July 14 NV) Kaecia CRV Y [13)=C(5) Return 5 End ŝ

WARS --WURER DF WARTABLES IN CURRELATION MATWIA FROM REGRESSIGN PMASE WARS --WURER DF PAREDICTOR WAIABLES IN REWEISSION PHASE Incomally Area of use is une less than Naway UNLESS Some PAREDICTOR waitables dar Not use is the Name Name of Paredictor waitables of the Name of USE Referencession Phase refore confession in Metaression Phase refore the Last become and to the Cariferion warable all BE 2000 refore the Last Elevent is the regression version in BE 2000 refore the Last Elevent is the regression ways and under a Failoura Poole and to the Cariferion warable all BE 2000 Inverte Last Elevent is the regression were and the last refore the Last Elevent is the regression waser, pad weight conder a score regression weight and the second refore the last found the formation of the second refore the last formation of the regression of the second refore regression waser, pade weight of refore regression phase phase number of refore regression phase phase number of refore regression phase regression refore regression phase regression regression refore regression phase regression regression were and refore regression phase regression regression regression refore regression phase regression regression were and refore regression phase regression regression regression were and refore regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regression regr METUR. SEND Sucroutine Begentanderusilaguts) Werkers Regession Versens Punched frum Juilon into an Array Starting in Location Ailaguts)

ъ. . .

,

ļ

ŝ 0

> ... ۰.

. .

.

•

.ن تو . بىر

÷

. . ,tł E s · 6· ÷

2.

۰. Ϋ.

5

• •

LABLTPIN WCRIT-L 1-2415415+14C-1 STORAGE .3063 SUMY280.0 SUMXY80.0 107AL980.0 107AL980.0 MRV8NRV1.830 NRV8NRV1.330 NORECE0 SUMIZED. 210UAL +1 C SPECIAL 5 1021 596 222 • 220 4 S S ŝ v . TABLFS Subboutine Tablesidtapf.file.ncases.nvars.ncmit.lmbuts.ledrk.n) Generates Prediction System Tables 1 and 2 UIMEneton FMT1 (22) «KFMT1 (22) «FMT2 (22) «KFMT2 (22) «A(1) «KA(1) COMON FMT1 «FMT2» EQUIVALENCE (FATI)«FATI)«IFATZ«KFATZ)«(A«KA) Common /Auc/sthum«Phhum«Numpe.sumsam«Numpru»Hsq«MHVI A(LRG475) a0 0 CALL TH(MV854, (LRGW15.1), A(LRGW15)) D 1 1 1 1, MPREDS D 1 1 1 1, MPREDS FRAD 500.1DPRED.WT EVBLACHT15.1CPRED-1 ISUBLACHT5.1CPRED-1 I 502 FORMAT (20X FT5-8) LSUB-LRGETS-NVARS LNCORRELPG+101 LOFELMCORM+101 LENDELOF LP3=LP94].101 ARGUMENTS IDT IPF-TAF RF "URN

\$18FTC

0 000 0 0 000 00 00 0000

280

ωu

2

RESTALTUAL-SURLC SF11000+E4219REMT 805 P0HAAT15544RES10UAL TABLE0/J843CASE01844ACTUAL+74098E05C1512006E09RES C SPECIAL MILE) Legic Logic For Data COVED as 1 ard 2 fom Einary 0 and 1 Leinumpr.ge.[0] 40 to 1732 UČ 865 1L00001.0KCASES Call Dirtpicks(0.111.1.1-LTPIW.LTPENU) Call Dirtpick(0.3.45.9.3.2000.10.100000.NTV) Call Dirtpick(0.3.12.0000) UCCODE(80.KFM72.A12000) UCCODE(80.KFM72.A12000) (A1J).JIC30000.NTV UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UCCODE(80.KFM72.410.VA35) UC LF(1L)005-62, 2)60 TO 594 HEIT 555-(4(14)-1144LTPIN4LTPENU) ,5UMLC PORMAT(213+12F10,3) CONTINUE CONTINUE 51 CONTINUE 52 FOHANT 523 A123 52 FOHANT 723 A123 52 FOHANT 720 A123 54 FOHANT 7200 MCRT 56 FOHAT 700 CRT WARP,16/01Maw DATAP, FRIN 61 KFNT 2 CRT WARP,16/01Maw DATAP, FRIN 61 KFNT 2 CRT WARP,16/01Maw DATAP, RAW MEIGHTAL FAINT 964 ILOOP ACTUAL SUMLCARES FORMATIJIS, 51, 314 % F7, 4, 41) CONTINUE SUMY2=SUMY2=ACTUAL992 SUMXY=SUMXY+SUMLC0ACTUAL SUMXY=SUMXY+SUMLC0ACTUAL UETE04TNE INTERVAL FOR SUMSC If (SUMLC=6E_1,000160 TC 901 TO ITON PESIDUAL TABLE 1027 (LUOPS - 1 1 LEMU) UKLCESUMLC+4(]³.44{**]**²) (3+LRGMTS+NVARS 14(C-1-10) Sumcc=sumcc-a(13) 1f(1100P-61.116C T0 350 SUMERCLARSUMLC SUME2eSUME2+SUMLC+R SUMESUME+ACTUAL FRINT 552 | W.A([]) FORMATTIX: 10, F15, B) FORMAT(•] VARIAPLE UO 551 [#41,44435 11=[464T5•[4-]

÷

.

Ą

Æ

\$

, đ

Ğ

8⁴⁸

ş

365

o

TINU In Idtar Umutte Pial Doum and Pare UP Umutte Pial Strijoo-K Gilatkujoo-K 00) =4 (LINT+100)+1.0 TO 964 AND POAL LP1 AND LP9 AL 1-A([12] TAL 0-A(110) 44(K5)+A(K5-1) A (LINT) +1.0 *A (K2) KE=1,100 101.1.101 6.1.1.1. стро5•1546 Ггро5•1546 Пад (18)•1•0 (1-2× ž 5,0 823 CO 57 606 903 C 828 201 Net *****06 196 U υ υ

C.L.[80,81160 T0 902

INUS IN

- . A (KS) •0
- AND LOF

# Fair assimilation in the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second secon

1. . . . . .

i

. .

. .

ş J

T D

367

Server of the server of the server of the server of the server of the server of the server of the server of the

## References

1

¢.

ž

. .

æ

:

×

÷ 🖇

r: - 5

- Anderberg, M. R., Cluster Analysis for Applications, Ph.D. Dissertation, The University of Texas at Austin, December, 1971.
- Jennings, E., EDSTAT-J: A Subroutine System for Data Processing in the Behavioral Sciences, Department of Educational Psychology, University of Texas at Austin, 1971.
- Koylyay, J. B., C. D. Gott, and J. H. Elton, Automatic Interaction Detection AID-4, Personnel Research Div., AFHR L, Lackland AFB, Texas, July, 1971, (Personal Communication).
- SOCLIB Program Manual (Program MCA), The University of Texas at Austin Computational Center.
- Step-01; Statistical Computer Program for Stepwise Multiple Regression, Center for Highway Research, The University of Texas at Austin, 1970.

## **EIBLIOGRAPHY**

. . .

.5

2

A

10

3.8

- Allais, D. C., <u>The Selection of Measurements for Predictions</u>, Tech. Report No. -6103-9, Stanford Electronics Laboratories, Stanford, California, November 1964.
- Anderberg, M. R., <u>Cluster Analysis for Applications</u>, Unpublished Ph. D. Dissertation, The University of Texas at Austin, December 1971.
- 3. Anderson, R. L. and T. A. Bancroft, <u>Statistical Theory in</u> Research, McGraw-Hill, 1952.
- Andrews, F., Morgan, J., and J. Sonquist, <u>Multiple Classifi-</u> cation Analysis, ISR, University of Michigan, Ann Arbor, 1967.
- 5. Beach, Lee Roy, "Multiple Regression as a Model for Human Information Utilization", Organizational Behavior and Human Performance, Vol. 2, p. 276-289, 1967.
- Bettman, James R., <u>Applying a New Methodological Approach</u> to a Problem of Information Processing Model Validation, Working Paper No. 152, Western Management Science Institute, UCLA, 1969.
- 7. Bettman, James R., <u>Behavior Simulation Models in Marketing</u> <u>Systems</u>, Unpublished Dissertation, Yale University, Boston, Massachusetts, 1969.
- Biborosch, R. A., "Numerical Credit Scoring", <u>Credit World</u>, Vol. 53: no. 9 & 10, June-July 1965.
- 9. Blaylock, H. M. Jr., <u>Theory Construction</u>, Prentice-Hall, 1969.
- Bottenberg, R. A. and R. E. Christal, <u>An Iterative Technique</u> for Clustering Criteria which Retains Optimum Predictive <u>Efficiency</u>, WADD-TN-61-30, Personnel Research Laboratory, Lackland, Air Force Base, Texas, March 1961.

- 11. Bottenberg, R. A. and Joe H. Ward, Jr., <u>Applied Multiple</u> <u>Linear Regression</u>, PRL. TDR-63-6, 6470th Personnel Research Laboratory, Lackland Air Force Base, Texas, March 1963.
- Bowman, E. H., "Consistency and Optimality in Managerial Decision Making", <u>Management Science</u>, No. 9, p. 310-321, 1963.

C

. . . . .

į

.. ð

हैः स्वर्ग ह

> . بنج:

> > 38

- Brehmer, B., "The Structure of Policy Conflicts", <u>UMEA</u> <u>Psychological Reports</u>, No. 29, Department of Psychology, University of UMEA, Sweden, 1970.
- 14. Bruner, J. S.; Goodnow, J. J. and G. A. Austin, <u>A Study in</u> Thinking, Wiley and Sons, New York, 1956.
- Brunswik, Egon, "Thing Constancy as Measured by Correlation Coefficients", <u>Psychological Review</u>, Vol. 47, p. 69-78, 1940.
- 16. Brunswik, E., <u>Perception and the Representative Design of</u> <u>Experiments</u>, University of California Press, Berkeley, 1956.
- Christai, R. E., "JAN: A Technique for Analyzing Group Judgment", <u>The Journal of Experimental Education</u>, Vol. 36, NR 4, p. 24-27, AD 675295, Summer 1968 (issued earlier under the same title as report No. PRL-TDR-63-3, 6570th Personnel Research Laboratory, Lackland AFE, Texas, AD 403813, February 1963).
- Christal, R. E., "Selecting a Harem and Other Applications of the Policy Capturing Model", PRL-TR-67-1, AFHRL, Lackland AFB, Texas, March 1, 1967.
- Churchman, C. W., <u>Prediction and Optimal Decisions:</u> <u>Philosophical Issues in a Science of Values</u>, Prentice-Hall, 1961.
- 20. Cohen, Kalman J., and Frederick S. Hammers, Eds., Analytical Methods in Banking, Irwin, 1966.

- 21. Credit Research Foundation, <u>Credit Limits Established by</u> Formula and Computer, September 1970.
- 22. Dalkey, Norman C., and Daniel L. Rourke, <u>Experimental</u> <u>Assessment of Delphi Procedures with Group Value Judgments</u>, R-612-ARPA, RAND Corp., Santa Monica, California, February, 1971.
- 23. Dawes, Robyn M., and Richard D. Diller, "The Prediction of Bootstrapping and a Method of Amalgamation", <u>Oregon Research</u> Institute Research Bulletin, Vol. 10, No. 6, October 1970.
- Dawes, Robyn M., "A Case Study of Graduate Admission: Application of Three Principles of Human Decision Making", <u>American Psychologist</u>, Vol. 26, No. 2, February 1971.

ć

ł

ъ

- Draper, N. R., and H. Smith, <u>Applied Regression Analysis</u>, Wiley, 1966.
- Dudycha, D. S., and Naylor, J.C., The Effect of Variations in the Cue R Matrix Upon the Obtained Policy Equation of Judges", <u>Educational and Psychological Measurement</u>, Vol. 26, p. 583-603.
- 27. Durand, David, <u>Risk Elements in Consumer Installment</u> <u>Financing</u>, Study #8, National Bureau of Economic Research, New York, 1941.
- 28. Earle, T. C., "Task Learning, Interpers al Learning and Cognitive Complexity", <u>Program on Cognitive Processes</u> <u>Report</u>, No. 129, Institute of Behavorial Science, University of Colorado, Boulder, May 1970.
- 29. Einhorn, Hillel J., "The Use of Non-Linear, Non-Compensatory Models in Decision Making", <u>Psychological Bulletin</u>, Vol. 3, No. 3, p. 221-230, 1970.
- 30. Einhorn, Hillel J., "Use of Non-Linear, Non-Compensatory Models as a Function of Task and Amount of Information", <u>Organizational Behavior and Human Performance</u>, Vol. 6, p. 1-27, January 1971.

- 31. Einhorn, Hillel J., Expert Measurement and Mechanical Combination, II: An Analysis of the Validity, Reliability and Agreement of Expert Judges, Working Paper 21-71-2, Carnegie Mellon University, September 13, 1971.
- Einhorn, Hillel J., "Expert Measurement and Mechanical Combination", Organizational Behavior and Human Performance, March 1972.
- 33. FARADA Handbooks, Vol. 1 through 6, Naval Ordinance System Command, FMSAEG, Corona, California, April 1969.
- 34. Finifter, B. M., "ERIV: A Computer Program for Evaluating Relative Importance of Variables in the Analysis of Interaction Effects", <u>Behavioral Science</u>, Vol. 16, No. 5, pp. 511-512, September 1971.
- Finifter, Bernard M., Program ERIV and PEP Statistic for Measuring Relative Importance, personal correspondence, May 1972.
- Fishburn, P. C., "Utility Theory", <u>Management Science</u>, No. 14, p. 335-378, 1968.
- Fishburn, P. C., <u>Utility Theory for Decision Making</u>, Wiley & Sons, New York, 1970.

Rui

- Garner, W. R., "The Stimulus in Information Processing", American Psychologists, Vol. 25, p. 350-358, 1970.
- Glasser, Gerald J., "Statistical and Mathematical Application in Consumer Credit Management", <u>Credit World</u>, Vol. 55, No. 1 and 2, p. 18-21, October - November 1966.
- Goldberg, L. R., "Simple Models or Simple Processes? Some Research on Clinical Judgments", <u>American Psychologist</u>, Vol. 23, p. 483-496, 1968.
- 41. Goldberg, L. R., "Man Versus Model of Man: A Rationale plus Some Evidence for a Method of Improving on Clinical Inferences", <u>Psychological Bulletin</u>, Vol. 73, p. 422-432, 1970.

A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF

- 42. Goldberg, Lewis R., "Five Models of Clinical Judgment: An Empirical Comparison between Linear and Non-Linear Representation of the Human Inference Frocess", Organizational Behavior and Human Performance, Vol. 6, p. 458-479, 1971.
- Goodman, Leo A., "The Multivariate Analysis of Qualitative Data: interactions Among Multiple Classifications", <u>Journal</u> of <u>American Statistical Association</u>, Vol. 65, p. 226-256, March 1970.

, ien

÷...:

æ

. Q

> с Слада

, *н* 

- 44. Graybill, J. A., <u>An Introduction to Linear Statistical Models</u>, McGraw-Hill, New York, 1961.
- 45. Green, Bert F., "Description and Explanations: A Comment on Papers by Hoffman and Edwards", in <u>Formal Representa-</u> tion of Human Judgment, B. Kleinmuntz (ed.), Wiley, 1968.
- 46. Grinnell, Merilyn; Keeley, Stuart: and Michael Doherty, "Bayesian Predictions of Faculty Judgments of Graduate School Success", Organizational Behavior and Human Performance, Vol. 6, p. 379-387, May 1971.
- 47. Hammond, K. R., Hursch, C. J., and F. J. Todd, "Analyzing the Components of Clinical Inference", <u>Psychological Review</u>, Vol. 71, p. 438-456, 1964.
- 48. Hayes, John R., "Strategies in Judgmental Research", in Formal Representation of Human Judgment, B. Kleinmuntz (ed.), Wiley, 1968.
- 49. Heiner, Olaí, <u>Anclysis of the Future: The Delphi Method</u>. RAND Corp., Santa Monica, Calif., March 1967.
- 50. Herzberg, Paul A., <u>The Parameters of Cross Validation</u>, Ph. D. Dissertation, University of Illinois, NTIS AD #667467, Dec., 1967.
- 51. Hicks, Charles, R., <u>Fundamental Concepts in the Design of</u> <u>Experiments</u>, Holt, Rinehart, and Winston, New York, 1964.

 Hoffman, Paul J., "The Paramorphic Representation of Clinical Judgment", <u>Psychological Bulletin</u>, Vol. 47, p. 116-131, 1960.

ł

þ

ĉ

2

4

, : 1

- **8** 

e.

- 53. Hoffman, Paul J., "Assessment of the Independent Contributions of Predictors", <u>Psychological Bulletin</u>, Vol. 59, No. 1, pp. 77-80, 1962.
- 54. Hoffman, Paul J., "Cue Consistency in Human Judgment", in Formal Representation of Human Judgment, B. Kleinmuntz (ed.), Wiley, 1968.
- 55. Hoffman, Paul J., P. Slovic, and L. G. Rorer, "An Analysisof-Variance Model for the Assessment of Configural Cue Utilization in Clinical Judgment", <u>Psychological Bulletin</u>, Vol. 69, p. 338-349, 1968.
- 56. Holt, Robert R., "Yet Another Look at Clinical and Statistical Prediction: Or, Is Clinical Psychology Worthwhile?", <u>American Psychologis</u>t, Vol. 25, p. 337-349, 1970.
- 57. Huber, G. P.; Daneshgor, R., and D. L. Ford, "An Empirical Comparison of Five Utility Models for Predicting Job Preferences", <u>Organizational Behavior and Human Performance</u>, Vol. 6, p. 267-282, May 1971.
- 58. Jennings, E., "Fixed Effects Analysis of Variance by Regression Analysis", <u>Multivariate Behavorial Research</u>, Vol. 2, p. 95-108, 1967.
- 59. Jennings, E., EDSTAT-J: A Subroutine System for Data <u>Processing in the Behavorial Sciences</u>, Department of Educational Psychology, University of Texas at Austin, Feb., 1971.
- 60. Kempthorne, O., The Design and Analysis of Experiments, New York, Wiley & Co., 1952.
- 61. Kennedy, Henry A., Judged Importance of the Dimensions of <u>A Loan</u>, University of Washington, Department of Psychology, May 1971.

A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL PROPERTY AND A REAL

374

- 62. Klahr, David, "Deci ion Making in a Complex Environment: The Use of Similarit Judgments to Predict Preferences", <u>Management Science</u>, Vol. 15, p. 595-609, July 1969.
- 63. Kleinmuntz, B. (ed.), Formal Representation of Human Judgment, Carnegie-Mellon University, Wiley, 1968.
- 64. Koplyay, J. B., Field Test of the Weighted Airman Promotion System: Phase I. Analysis of the Promotion Board Component in the Weighted Factors System, Report No. AFHRL-TR-69-101, AF Human Resources Laboratory, 20 pp., AD 689751, April 1969.
- 65. Koplyay, J. B., Field Test of the Weighted Airman Promotion System: Phase II. Validation of the System for Grades E-4 Through E-7., Report No. AFHRL-TR-69-102, 18 pp., AD 697798, May 1969.
- 66. Koplyay, James B., etal., "Automatic Interaction Detection AID-4, Personnel Division, AFHRL, Lackland Air Force Base, Texas, (personal communication, 1971).
- 67. Krieger, Sam, "Appraising Credit Applications by Point Count Method", <u>Credit World</u>, Vol. 53, No. 6, p. 18-20, March 1965.
- Kunreuther, Howard, "Extensions of Bowman's Theory on Managerial Decision-Making", <u>Management Science</u>, Vol. 15, p. 415, April 1969.
- 69. La Motte, L. R., et al., <u>Computational Efficiency in the</u> <u>Selection of Regression Variables</u>, Themis TR#14, Texas A&M College Station, Texas, July 1969.
- Madden, J. M., <u>An Application to Job Evaluation of a Policy-Capturing Model for Analyzing Individual and Group Judgments</u>, 6570th Personnel Research Laboratory, Aerospace Medical Division, Air Force Systems Command, PRL-TDR-63-15, May 1963.

375

dia Sila da

- 71. Majone, Grandemenico, <u>Classification and Discrimination in</u> <u>the Analysis of Credit Risks: I,</u> Carnegie-Mellon University, Pittsburgh, February 1968.
- 72. Mead. D. F., <u>Development of an Equation for Evaluating Job</u> <u>Difficulty</u>, Report No. AFHRL-TR-70-42, AF Human Resources Laboratory, Lackland AFB, Iexas, 19 pp., AD 720253, November 1970.
- Mead, D. F., <u>Continuation Study on Development of a Method</u> for Evaluating Job Difficulty, Report No. AFHRL-TR-70-43, AF Human Resources Laboratory, Lackland AFB, Texas, 20 pp., AD 720254, November 1970.
- 74. Mead, D. F. and R. E. Christal, <u>Development of a Constant</u> <u>Standard Weight Equation for Evaluating Job Difficulty</u>, Report No. AFHPL-TR-70-44, AF Human Resources Laboratory, Lackland AFB, Texas 20 pp., AD 720255, November 1970.

。 。

- 75. Meehl, P. E., <u>Clinical Versus Statistical Prediction</u>: <u>A</u> <u>Theoretical Analysis and Review of the Literature</u>, University of Minnesota Press, Minneapolis, 1954.
- 76. Mendenhall, W., Introduction to Linear Models and the Design and Analysis of Experiment, Wadsworth Publishing Co., 1968.
- 77. Milburn, T. W., et. al., <u>The Echo Method and the Study of</u> <u>Values</u>, Tech. Memo 951, General Research Corp., Santa Barbara, Calif., November, 1968.
- Miller, George A., "The Magical Number Seven, Plus or Minus Two: Some Limits on Our Capacity for Processing Information", Psychological Review, Vol. 63, p. 81-98, 1956.
- 79. Mood, A. M. and F. A. Graybill, Introduction to the Theory of Statistics, McGraw-Hill, 1963.
- 80. Moore, Geoffrey, H. and Philip A. Klein, <u>The Quality of</u> <u>Consumer Installment Credit</u>, National Bureau of Economic Research, Columbia University Press, New York, 1967.

- 81. Morrison, D. F., <u>Multivariate Statistical Methods</u>, McGraw-Hill, New York, 1966.
- Mullins, Cecil J. and Usdin, Eugene, <u>Estimation of Validity</u> in the Absence of a Criterion, AFHRL-TR-70-36, Personnel Division, AFHRL, Texas, 1970.
- 83. Myers, James H., "Numerical Scoring Systems for Retail Credit Evaluation", <u>Credit World</u>, Vol. 50, No. 7, p. 5-7, April, 1962.

1

:

- Myers, James H. and Edward Forgy, "The Development of Numerical Credit Evaluation Systems", <u>American Statistical</u> <u>Association Journal</u>, Vol. 58, p. 799, 1963.
- 85. Naylor, J. C. and R. J. Wherry, Sr., <u>Feasibility of Distinguishing Supervisors' Policies in Evaluation of Subordinates by Using Rating of Simulated Job Incumbents</u>, PRL-TR-64-25, Air Force Human Resources Laboratory, Lackland AFB, Texas, October 1964.
- 86. Naylor, J. C., and Schenck, E. A., "ρ_m As An Error Free Index of Rater Agreement", <u>Educational and Psychological</u> <u>Measurement</u>, Vol. 26, p. 815-824, 1966.
- 87. Norman, James E., David, H. A., "Restricted Ranking", Psychometrics, Vol. 34, No. 1, p. 85-110, March 1969.
- Phillips, P. J., <u>An Evaluation of the Discriminant Assignment</u> <u>Model</u>, Master's Thesis, Air Force Institute of Technology, AD 806484, 59 pp., August 1966.
- Roberts, Fred S., What if Utility Functions Do Not Exist, R-528-ARPA, RAND Corp., Santa Monica, Calif., September 1970.
- 90. Robinson, William O., <u>Multivariate Probability Assessment</u>, Technical Report No. 60, Operations Research Center, MIT, Cambridge, Massachusetts, 1971.
- 91. Rosenthal, R., <u>Experimental Effects in Behavioral Research</u>, Appleton-Century-Crofts, New York, 1966.
- 92. Sawyer, Jack, "Measurement and Prediction, Clinical and Statistical", <u>Psychological Bulletin</u>, Vol. 66, p. 178-200, 1966.
- 93. Searle, S. R., Linear Models, Wiley, New York, 1971.

| { -****T**.

- 94. Slovic, P., "Cue Consistency and Cue Utilization in Judgment", <u>American Journal of Psychology</u>, Vol. 79, p. 42/-434, 1966.
- 95. Slovic, Paul, "Analyzing the Expert Judge: A Description Study of A Stockbroker's Decision Processes", Journal of Applied Psychology, Vol. 53, p. 255, 1969.
- 96. Slovic, Paul, <u>Structural Determinants of Cue Utilization in</u> Judgment, Oregon Research Institute, May 1969.
- 97. Slovic, P., and S. Lictenstein, "Comparison of Bayesian and Regression Approaches to the Study of Information Processing in Judgment", ORI Monograph, Vol. 10, No. 1, July 1970.
- 98. Smith, R. L., et al., <u>Acquisition of Restore Time Data by</u> <u>Subjective Techniques</u>, AMRL-TR- 0-76, Aerospace Medical Research Laboratory, Wright Patterson AFB, Ohio, NTIS AD 727015, March 1971.
- 99. Smith, T. H., C. D. Gott, and R. A. Bottenberg, <u>Predicting</u> the Potential for Active Duty Success of Rehabilitated Air <u>rorce Prisoners</u>, PRL-TR-67-16, Air Force Human Resources Laboratory, Lackland AFB, Texas, NTIS AD 672958, October 1967.
- 100. Sonquist, J. A. and J. N. Morgan, <u>The Detection of Interaction</u> <u>Effects</u>, Survey Research Center, University of Michigan, Ann Arbor, 1964.

- Sonquist, J. A. "Finding Variables that Work", <u>Public Opinion</u> <u>Ouarterly</u>, Vol. 33, No. 1, pp. 83-95, Spring 1969.
- 102. Sonquist, John A., <u>Multivariate Model Building</u>, Survey Research Center, University of Michigan, 1970.
- 103. Sonquist, J. A., E. L. Baker and J. N. Morgan, <u>Searching</u> for Structure, ISR, Ann Arbor, Michigan, 1971.
- 104. Stephenson, R. and Joe H. Ward, Jr., "Computer Assisted Decisions", <u>Proceedings of the 78th Annual Convention</u>, American Psychological Association, 1970.
- Stimson, David H., "Utility Measurement in Public Health Decision", <u>Management Science</u>, Vol. 16, p. 17-31, October 1969.
- 106. Tatsuaka, M. M., "Joint Probability of Membership and Success in a Group: An Index Which Combines the Information from Discriminant and Regression Analysis as Applied to the Guidance Problem", <u>Harvard Studies in Career Development</u>, No. 6, Harvard University, Boxton, 1957.
- 107. Taylor, Bernard H., <u>Groups of Homogeneous Sriterion Vari-</u> ables in Multivariate Prediction, University of Washington, Seattle, January 1962.
- 108. Thames, James P., Integrated Personnel Management Systems, <u>Ouantified for Greater Efficiency</u>, Master's Thesis, T-1694, Graduate School of Business Administration, The University of Texas at Austin, 1964.

, . .

**R** 

- 109. Valenzi, Enzo R., <u>Configural Cue Utilization in Human Decision</u> <u>Making</u>, Ph. D. Dissertation, Bowling Green State University, August, 1970.
- 110. Vandell, R. F., "Management Evaluation in the Quantitative World", <u>Harvard Business Review</u>, Vol. 48, p. 83, January-February 1970.

- 111. Wagner, G. R. and L. L. Gooch, "Quantifying Human Judgments for more Meaningful Computer Decision Models", in process, 1972.
- 112. Waligore, Robert S., "Presentation to American Industrial Bankers Association, November 12, 1971," The Morris Planning Company of California, San Francisco.
- 113. Ward, J. H., Jr., <u>An Application of Linear and Curvilinear</u> <u>Joint Functional Regression in Psychological Prediction</u>, <u>AFPTRC-TR-54-86</u>, Air Force Personnel and Training <u>Research Center</u>, Lackland AFB, Texas, December 1954.
- 114. Ward, J. H., Jr., <u>Use of a Decision Index in Assigning Air</u> <u>Force Personnel</u>, WADC-TN-59-38, Personnel Research <u>Iaboratory</u>, Lackland Air Force Base, Texas April 1959.

.

ii.

-31

**4**5

y,

- 115. Ward, Joe H., Jr., and Marion E. Hook, <u>A Hierarchical</u> <u>Grouping Procedure Applied to a Problem of Grouping</u> <u>Profiles</u>, ASD-TN-61-55, Aeronautical Systems Division, <u>Personnel Laboratory</u>, Lackland Air Force Base, Texas, October 1961.
- 116. Ward, Joe H., Jr., "Comments on the Paramorphic Represen tation of Clinical Judgment", <u>Psychological Bulletin</u>, Vol. 59, p. 74-76, 1962.
- Ward, J. H., "Multiple Linear Regression Models", <u>Computer Applications in the Behavioral Sciences</u>, H. Borko (ed.), Prentice-Hall, Englewood Cliff, p. 205-237, 1962.
- 118. Ward, Joe H., Jr., and Kathleen Davis, <u>Teaching a Digital</u> <u>Computer to Assist in Making Decisions</u>, PRL-TDR-63-16, 6570th Personnel Research Laboratory, Lackland Air Force Base, Texas 1963.
- Ward, J. H., Comment on "The Partitioning of Variance in School Effects Studies", <u>American Educational Research Journal</u>, Vol. 6, No. 3, p. 467-474, 1969.

- 120. Ward, Joe H., Jr., "Analyzing Judgment and Decision Processes", Working Paper, Southwest Educational Development Laboratory Octiber 1970.
- 121. Ward, J. H. Jr., and E. Jennings, <u>Introduction to Linear Models</u>, Prentice-Hall, Englewood Cliffs, New Jersey, (in press), 1972.
- 122. Weber, Donald C., "The Analysis of Experimental Designs Using Time-Sharing Computer Terminals", Paper presented at Joint Statistical Meeting, Colorado State University, August 23, 1971.
- Wiggins, N. S., and Hoffman, P. J., "Three Models of Clinical Judgment", Journal of Abnormal Psychology, Vol. 73, p. 70-78, 1968.
- 124. Wilt, G. A., Jr. and J. M. Tierney Jr., "Progressive Risk Analysis Through Credit Scoring", <u>Credit World</u>, Vol. 56, No. 6, p. 10, March 1968.
- 125. Yntema, D. B., and Torgerson, W. S., "Man-computer Cooperation in Decisions Requiring Common Sense", <u>IRE</u> <u>Transactions on the Profession-1 Group on Human Factors in</u> <u>Electronics</u>, HFE 2, 1961.
- 126. Yntema, D. B., and L. Klem, "Telling a Computer How to Evaluate Multi-dimensional Situation", <u>IEEE Transactions</u> on Human Factors in Electronics, HFE-6, p. 1, 1965.
- 127. Zaegel, Richard J., "A Point Rating System for Evaluating Customers", <u>Credit World</u>, Vol. 52, No. 1, p. 9-13, October 1963.
- 128. Zaegel, Richard J., "After 10 Years of Credit Scoring", Credit World, Vol. 59, No. 10, p. 14, July 1971.
- 129. Zyskind, G., et al., Linear Models, Statistical Information and Statistical Inference, Aerospace Research Laboratories ARL 71-0076, Wright Patterson Air Force Base, Ohio, May 1971.